# Nonlinear Differential Equations in the Framework of Regular Variation

Pavel Řehák Institute of Mathematics Academy of Sciences of the Czech Republic

A-Math-Net

Brno 2014

## Nonlinear Differential Equations in the Framework of Regular Variation

Pavel Řehák Institute of Mathematics Academy of Sciences of the Czech Republic rehak@math.cas.cz

Brno 2014



Preparation of this text was supported by the project A-Math-Net (Applied Mathematics Knowledge Transfer Network) No. CZ.1.07/2.4.00/17.0100.

## Preface

This text deals particularly with nonlinear differential equations which are examined with the help of regular variation. It was created within the A-Math-Net project, and its primary purpose is to serve as a textbook for (almost) graduate students. But it can be useful also for experts or for anybody who is interested in asymptotic theory of differential equations.

The theory of regularly varying functions has been shown very useful in some fields of qualitative theory of differential equations, see, in particular, the important monograph by Marić [105] summarizing themes in the research up to the year 2000. This book served as an excellent source for our text, but of course, much material is taken also from other sources. Actually, our treatise includes many of the results that appeared after Marić's book. Moreover, it reveals new relations among various results, and revise some of them.

In view of the main aim of this text, we cannot give a complete treatment with all details. We rather focus on presenting a wide variety of methods which shows how powerful tool regular variation is. For reference purposes we present brief comprehensive surveys of sub-themes.

The first chapter summarizes useful information about regularly varying functions and related concepts. Our style of treatment with this topic is affected by requirements on applications in differential equations.

Although this text is focused on nonlinear equations, there is a big part (Chapter 2) which deals with linear differential equations. We offer a comprehensive survey of the results in which linear differential equations are studied in the framework of regular variation. Proofs are given only exceptionally, but we present many comments which sometimes include a description of the main ideas. The objective of this chapter is multiple. Some of the linear results are used in the nonlinear theory, thus we can easily refer them. Some of the results are extended to a nonlinear case, thus we can easily make a comparison; at the same time, some of the statements in Chapter 2 may serve as a motivation for an extension (which has not been made yet) to a nonlinear case. Moreover, our survey includes also the results which are not contained in the above mentioned Marić's book (especially the recent ones), and we point out relations among various results. Chapter 3 deals with second-order half-linear differential equations in the framework of regular variation. The most of the results can be understood as a half-linear extension of some of the statements from Chapter 2. But this part offers more, especially as far as the methods are concerned. Indeed, many steps in the proofs require a quite new approach or at least a highly nontrivial modification comparing with the existing linear case.

Emden-Fowler type equations are examined in Chapter 4. We consider various types of such objects, second order equations, higher order equations, but also systems. Older as well as recent results are presented. Concerning recent results, because of their big amount, we give only a brief (but quite comprehensive) survey which is followed by a detailed description of several selected results. We try to make make a selection in such a way which shows a variety of typical approaches.

Chapter 5 is devoted to investigation of some other nonlinear differential equations where the theory of regular variation has been shown to be helpful. We deal with equations which involve a generalized Laplacian, partial differential equations, a class of third order nonlinear equations, perturbed first order equations, and second order nearly linear equations.

The last chapter briefly discusses utilization of regular variation in some other differential (or integral) equations; among others, equations with deviating arguments are mentioned. Further, it offers a short survey of the literature devoted to examination of difference equations, *q*-difference equations, and dynamic equations on time scales in the framework of regular variation.

Preparation of this text was supported by the project A-Math-Net (Applied Mathematics Knowledge Transfer Network), No. CZ.1.07/2.4.00/17.0100.

I would like to express my thanks to Robert Mařík and Marek Sas for reading the manuscript and useful comments.

Pdf version of this text can be found on http://www.amathnet.cz/ or on users.math.cas.cz/~rehak/ndefrv.

Pavel Řehák Brno, February 2014

# Contents

## Notation

1	Reg	egular variation						
	1.1	Karamata theory						
		1.1.1	Basic concepts	14				
		1.1.2	Uniform convergence and representation	16				
		1.1.3	Karamata theorem	17				
		1.1.4	Monotonicity	18				
		1.1.5	Further selected properties of <i>RV</i> functions	19				
		1.1.6	Regularly bounded functions	22				
		1.1.7	Rapid variation	23				
	1.2	aan theory	24					
		1.2.1	$Class \Pi$	24				
		1.2.2	Class Γ	28				
	1.3	Some	other related classes	32				
		1.3.1	Generalized regularly varying functions	32				
		1.3.2	Subexponential functions	34				
		1.3.3	Regular variation on various time scales	35				
		1.3.4	Hardy field	37				
2	Line	ear seco	and order differential equations	39				
-	2.1	Karar	nata solutions: coefficient with sign condition	40				
	2.2	Karar	nata solutions: coefficient with no sign condition	41				
	2.3	Gener	ralization and self-adjoint equation	43				
	2.0	2.5 Concratization and sen adjoint equation						
	2.1	regula	poularly varying solutions					
	2.5	Regul	arly varying solutions of Friedmann	17				
	2.0	equat	ions	47				
	2.6	More	precise information about asymptotic behavior	48				
	2.0	2.6.1	De Haan type solutions	48				
		0.1		10				

11

		2.6.2	A different approach	53			
	2.7	A not	e about higher order equations; an alternative				
		appro	ach to second order equations	58			
	2.8	On ze	ros of oscillatory solutions, asymptotic behavior of maxima				
		and of	f eigenvalues	66			
3	Hal	f-linear	second order differential equations	71			
	3.1	Introd	luction	71			
	3.2	$\mathcal{RV}$ as	nd $\mathcal{RB}$ solutions of half-linear equations	72			
		3.2.1	Auxiliary statement	72			
		3.2.2	$\mathcal{RV}$ solutions with different indices $\ldots \ldots \ldots \ldots$	74			
		3.2.3	$\mathcal{RV}$ solutions in the border case $\ldots \ldots \ldots \ldots \ldots \ldots$	79			
		3.2.4	$\mathcal{RB}$ solutions	82			
	3.3	More	precise information about $SV$ solutions	83			
	3.4	4 More general case					
	3.5	Asymptotic formulas for nonoscillatory solutions of					
		condi	tionally oscillatory half-linear equations	89			
	3.6	De Ha	an type solutions	99			
		3.6.1	Solutions in the class $\Gamma$	99			
		3.6.2	Solutions in the class $\Pi$	106			
	3.7	Half-l	inear differential equations having regularly				
		varyir	ng solutions	107			
4	Emo	den-Fo	wler type equations and systems	109			
	4.1	Introd	luctory and historical remarks	109			
	4.2	Former results					
		4.2.1	Superlinear second order equations in the work by Avaku-				
			mović and related results	111			
		4.2.2	Superlinear second order equations in the former works				
			by Marić and Tomić	113			
		4.2.3	Rapidly varying coefficient and nonlinearity	116			
	4.3	Select	ion of recent results	117			
		4.3.1	Introductory remarks	117			
		4.3.2	Asymptotic behavior of $SV$ and $\mathcal{RV}(1)$ solutions of sublin-				
			ear				
			second order equations	121			
		4.3.3	$SV$ solutions of second order super-linear equations $\ldots$	125			
		4.3.4	Intermediate generalized $\mathcal{RV}$ solutions of second order				
			sub-half-linear equations	129			
		4.3.5	Strongly monotone solutions of coupled systems	131			
		4.3.6	Overall structure of $\mathcal{RV}$ solutions to odd-order equations .	138			
		4.3.7	$\mathcal{P}_{\omega}$ solutions of <i>n</i> -th order equations	140			
		4.3.8	Strongly monotone solutions of <i>n</i> -th order nonlinear systems	149			

Some other nonlinear differential equations							
5.1	Equations with a general $\Phi$ -Laplacian						
	5.1.1 A boundary value problem on a half-line	166					
	5.1.2 Regularly varying solutions	168					
5.2	Partial differential equations	169					
	5.2.1 Radial $\mathcal{RV}$ solutions of partial differential systems	169					
	5.2.2 Almost radial symmetry	170					
5.3	$\mathcal{RV}$ and $\mathcal{RPV}$ solutions for a class of third order						
	nonlinear differential equations	172					
	5.3.1 The self-adjoint equation	172					
	5.3.2 $\mathcal{RV}$ and $\mathcal{RPV}$ solutions of the perturbed equation	174					
5.4	.4 Classification of convergence rates of solutions to						
	perturbed first order ODE's	175					
5.5	Nearly linear differential equations	179					
Con	cluding remarks	191					
6.1	More on differential equations in the framework						
	of regular variation	191					
6.2	Regularly varying sequences and difference						
	equations	192					
6.3 Regular variation on time scales and dynamic							
	equations	192					
6.4	<i>q</i> -regular variation and <i>q</i> -difference equations	192					
bliog	raphy	195					
	Som 5.1 5.2 5.3 5.4 5.5 Con 6.1 6.2 6.3 6.4 bliog	Some other nonlinear differential equations5.1Equations with a general $\Phi$ -Laplacian5.1.1A boundary value problem on a half-line5.1.2Regularly varying solutions5.2Partial differential equations5.2Partial differential equations of partial differential systems5.2.1Radial $\mathcal{RV}$ solutions of partial differential systems5.2.2Almost radial symmetry5.3 $\mathcal{RV}$ and $\mathcal{RPV}$ solutions for a class of third ordernonlinear differential equations5.3.15.3.1The self-adjoint equation5.3.2 $\mathcal{RV}$ and $\mathcal{RPV}$ solutions of the perturbed equation5.3.3 $\mathcal{RV}$ and $\mathcal{RPV}$ solutions of the perturbed equation5.4Classification of convergence rates of solutions to perturbed first order ODE's5.5Nearly linear differential equations in the framework of regular variation6.1More on differential equations in the framework of regular variation6.2Regularly varying sequences and difference equationsequations					

## Notation and convention

- $\mathcal{RV}(\vartheta)$ : regularly varying functions (at infinity) of index  $\vartheta$  [Section 1.1]
- *SV*: slowly varying functions (at infinity) [Section 1.1]
- *RB*: regularly bounded functions (at infinity) [Section 1.1]
- $\mathcal{RPV}(\pm\infty)$ : rapidly varying functions (at infinity) of index  $\pm\infty$  [Section 1.1]
- NRV(θ): normalized regularly varying functions (at infinity) of index θ [Section 1.1]
- NSV(θ): normalized slowly varying functions (at infinity) of index θ [Section 1.1]
- tr- $\mathcal{RV}(\vartheta)$ : trivial regularly varying functions (at infinity) of index  $\vartheta$  [Section 1.1]
- $SV_0$ : slowly varying functions at zero, similarly for  $RV_0(\vartheta)$ ,  $RPV_0(\pm)$ ,  $RB_0$ , tr- $RV_0(\vartheta)$  [Section 1.1]
- *RV*: the class of all regularly varying functions (at infinity), similarly for *RV*<sub>0</sub>, *RPV*, *RPV*<sub>0</sub>, tr-*RV* tr-*RV*<sub>0</sub>; *"RV"* sometimes may mean the abbreviation of "regularly varying", similarly for *SV*, *RPV*, *RB* etc. [Section 1.1]
- Karamata functions:  $\mathcal{RV} \cup \mathcal{RPV}$  [Section 1.1]
- $f^{\leftarrow}$ : generalized inverse of f [Section 1.1]
- $\Pi(w)$ : class  $\Pi$  in the de Haan sense with auxiliary function w [Section 1.2]
- $\Gamma(w)$ : class  $\Gamma$  in the de Haan sense with auxiliary function w [Section 1.2]
- $\Pi R_2(w, z)$ : Omey-Willekens type functions [Section 1.2]
- *BSV*: Beurling slowly varying functions [Section 1.2]
- *SN*: self-neglecting functions [Section 1.2]
- $\mathcal{RV}_{\omega}(\vartheta)$ : (generalized) regularly varying functions (at infinity) with respect to  $\omega$  of index  $\vartheta$ , similarly for  $\mathcal{SV}_{\omega}$ ,  $\mathcal{RPV}_{\omega}(\pm\infty)$ ,  $\mathcal{RB}_{\omega}$ ,  $\mathcal{NRV}_{\omega}$  [Subsection 1.3.1]
- $L_f$ : slowly varying component of  $f \in \mathcal{RV}$ , i.e.,  $L_f(t) = f(t)/t^{\vartheta}$  for  $f \in \mathcal{RV}(\vartheta)$ .
- $\sim, \approx, o, O$ : For eventually positive functions *f*, *g*, we denote:

 $f(t) \sim g(t) \text{ if } \lim_{t \to \infty} f(t)/g(t) = 1,$   $f(t) \approx g(t) \text{ if } \exists c_1, c_2 \in (0, \infty) \text{ s. t. } c_1g(t) \leq f(t) \leq c_2g(t) \text{ for } \text{ large } t,$  $f(t) = o(g(t)) \text{ if } \lim_{t \to \infty} f(t)/g(t) = 0,$  f(t) = O(g(t)) if  $\exists c \in (0, \infty)$  s. t.  $f(t) \le cg(t)$  for large t.

• We adopt the usual conventions:  $\prod_{j=k}^{k-1} u_j = 1$  and  $\sum_{j=k}^{k-1} u_j = 0$ .

•  $\Phi, \Phi_{\lambda}$ : The notation  $\Phi(u)$  is typically used in connection with half-linear equations and means  $\Phi(u) = |u|^{\alpha-1} \operatorname{sgn} u$  with  $\alpha > 1$ . The notation  $\Phi_{\lambda}(u)$  is typically used in connection with Emden-Fowler type equations and means  $\Phi_{\lambda}(u) = |u|^{\lambda} \operatorname{sgn} u$  with  $\lambda > 0$ .

Notice that the powers are shifted by 1, but since  $\alpha$  starts from 1 while  $\lambda$  starts from 0, the functions are practically the same. We decide to use such convention in accordance with some literature.

Chapter

# Regular variation

The subject of regular variation as we use the term today was initiated by Jovan Karamata in a famous paper of 1930 [74], see also [73, 75], though preliminary or partial treatments may be found in earlier work of Landau in 1911, Valiron in 1913, Pólya in 1917, Schmidt in 1925, and others.

In its basic form, regular variation may be viewed as the study of relations such as

$$\frac{f(\lambda t)}{f(t)} \to g(\lambda) \in (0,\infty) \quad (t \to \infty) \quad \forall \lambda > 0,$$

together with its numerous ramifications. This study is referred to as Karamata theory. More general than the relation above is

$$\frac{f(\lambda t) - f(t)}{g(t)} \to h(\lambda) \in \mathbb{R} \quad (t \to \infty) \quad \forall \lambda > 0.$$

The study of relations of this kind is referred to as de Haan theory.

Mathematically, regular variation is essentially a field in classical real variable theory, together with its applications in integral transforms – Tauberian theorems, probability theory, analytic number theory, complex analysis, differential equations, and elsewhere.

Our style of dealing with regular variation in this chapter is designed for the purpose to study asymptotic behavior of differential equations.

A quite comprehensive treatment of regular variation can be found in the book by Bingham, Goldie, and Teugels [14]; much of material presented in this chapter can be found (and is proved) in that book. Other main sources for this chapter are the book by Seneta [156], the book by Geluk and de Haan [47], and the thesis by de Haan [50]. Several useful concepts and statements are taken from some papers on differential equations.

## 1.1 Karamata theory

#### 1.1.1 Basic concepts

We start with two essential definitions.

**Definition 1.1.** A measurable function  $f : [a, \infty) \to (0, \infty)$  is called *regularly varying* (*at infinity*) of index  $\vartheta$  if

$$\lim_{t \to \infty} \frac{f(\lambda t)}{f(t)} = \lambda^{\vartheta}$$
(1.1)

for every  $\lambda > 0$ ; we write  $f \in \mathcal{RV}(\vartheta)$ . The class of all regularly varying functions is denoted as

$$\mathcal{RV} = \bigcup_{\vartheta \in \mathbb{R}} \mathcal{RV}(\vartheta).$$

**Definition 1.2.** A measurable function  $L : [a, \infty) \to (0, \infty)$  is called *slowly varying* (*at infinity*) if

$$\lim_{t \to \infty} \frac{L(\lambda t)}{L(t)} = 1; \tag{1.2}$$

we write  $L \in SV$ .

A slowly varying function is customarily denoted by *L* because of the first letter of the French "lentement" which means "slowly"; note that the foundation papers by Karamata were written in French.

It is clear that SV = RV(0). Consequently, the set of slowly varying functions forms a subset of the set of regularly varying ones. This however might be somehow misleading statement, since the class of slowly varying functions is the one which presents itself, due to wealth of interesting properties, as a major novelty in the classical analysis and applications. In the sequel the term "regularly varying functions" sometimes will include the slowly varying ones and sometimes not. The context, however, will prevent any ambiguity.

It is known that the conditions in the definition of  $\mathcal{RV}$  functions can be weakened. Indeed, the limit relation in (1.1) is sufficient to hold only for  $\lambda$  in a set of positive measure and then the regular variation follows. Moreover, if the limit

$$\lim_{t \to \infty} \frac{f(\lambda t)}{f(t)} = g(\lambda) \in (0, \infty)$$

exists for  $\lambda$  in a set of positive measure, then *g* necessarily takes the form  $g(\lambda) = \lambda^{\vartheta}$ , where  $\vartheta$  is some real number.

It is easy to show that  $f \in \mathcal{RV}(\vartheta)$ ,  $\vartheta \in \mathbb{R}$ , if and only if it can be written in the form

$$f(t) = t^{\vartheta}L(t), \quad \text{where } L \in \mathcal{SV}.$$
 (1.3)

Thus, for most purposes, to study regular variation, it suffices to study the properties of slowly varying functions. Here are some examples of slowly varying functions:

$$L(t) = \prod_{i=1}^{n} (\ln_{i} t)^{\mu_{i}}, \text{ where } \ln_{i} t = \ln \ln_{i-1} t \text{ and } \mu_{i} \in \mathbb{R},$$
  

$$L(t) = \exp\left\{\prod_{i=1}^{n} (\ln_{i} t)^{\nu_{i}}\right\}, \text{ where } 0 < \nu_{i} < 1,$$
  

$$L(t) = 2 + \sin(\ln_{2} t),$$
  

$$L(t) = (\ln \Gamma(t))/t,$$
  

$$L(t) = \frac{1}{t} \int_{a}^{t} \frac{1}{\ln s} ds,$$
  

$$L(t) = \exp\left\{(\ln t)^{\frac{1}{3}} \cos(\ln t)^{\frac{1}{3}}\right\}.$$

The last example shows the SV function which exhibits "infinite oscillation", i.e.,  $\lim \inf_{t\to\infty} L(t) = 0$ ,  $\limsup_{t\to\infty} L(t) = \infty$ . This phenomenon can be somewhat in a contrast with the intuitive notion of a "slowly varying" behavior and it reveals that the class RV includes a wide variety of functions. In particular, slowly varying functions do not need to be monotone eventually. It is clear that the power function  $t^{\vartheta}$  is a trivial example of regularly varying function (of index  $\vartheta$ ), but for  $\vartheta \neq 0$ , we have  $t^{\vartheta} \notin SV$ . The exponential functions  $\exp t$ ,  $\exp(-t)$  are not regularly varying at all. But, for example,  $1 + \exp(-t)$  is slowly varying. Undamped oscillatory functions such as  $2 + \sin t$  are not slowly or regularly varying. It is interesting to observe that while  $2 + \sin(\ln t)$  is not slowly varying, the function  $2 + \sin(\ln_2 t)$  is slowly varying.

Some authors speak about the class of *trivial regularly varying function of index*  $\vartheta$ , we write tr- $\mathcal{RV}(\vartheta)$ , where  $f \in \text{tr-}\mathcal{RV}(\vartheta)$  if

$$f(t) \sim Ct^{\vartheta}$$
 as  $t \to \infty$ ,

*C* being some positive constant; we consider a positive measurable *f*. We denote

$$\operatorname{tr}-\mathcal{RV} = \bigcup_{\vartheta \in \mathbb{R}} \operatorname{tr}-\mathcal{RV}(\vartheta).$$

We have,  $f \in \text{tr-}\mathcal{RV}(\vartheta)$ , if and only if

$$f(t) = t^{\vartheta}h(t)$$
 where  $h(t) \sim C \in (0, \infty)$  as  $t \to \infty$ ,

cf. (1.3). It is clear that tr- $\mathcal{RV} \subset \mathcal{RV}$ . Thus, regular variation of a function can be understood as a (one-sided, local) asymptotic property which arises out of trying to extend in a logical and useful manner the class of functions whose asymptotic behavior is that of a power function, to functions where asymptotic behavior is that of a power function multiplied by a factor which varies "more slowly" than a power function. As examples of slowly varying functions show, such an extension is far from being trivial. We have defined regular variation at infinity. Of course, this is not the only possibility. A measurable function  $f : (0, a] \rightarrow (0, \infty)$  is said to be *regularly varying* at zero of index  $\vartheta$  if  $\lim_{t\to 0+} \frac{f(\lambda t)}{f(t)} = \lambda^{\vartheta}$  for every  $\lambda > 0$ ; we write  $f \in \mathcal{RV}_0(\vartheta)$ . Since regular variation of  $f(\cdot)$  at zero of index  $\vartheta$  means in fact regular variation of f(1/t) at infinity of index  $-\vartheta$ , properties of  $\mathcal{RV}_0$  functions can be easily deduced from theory of  $\mathcal{RV}$  functions. Regular variation can now be defined at any finite point by shifting the origin of the function to this point. It is thus apparent that it suffices to develop the theory of regular variation at infinity, which we shall do, frequently omitting the words "at infinity" in the sequel. For example,  $-\ln t$  is slowly varying at t = 0+,  $\ln t$  is slowly varying at t = 1+,  $-\ln t$  is slowly varying at t = 1-, and  $-\ln(1-t)$  is slowly varying at t = 1.

In connection with investigation of solutions to some differential equations, the concept of nearly regularly varying functions was introduced: If a positive continuous function f satisfies  $f(t) \approx g(t)$  as  $t \to \infty$  for some  $g \in \mathcal{RV}(\vartheta)$ , then f is called a *nearly regularly varying function of index*  $\vartheta$ .

#### 1.1.2 Uniform convergence and representation

The following result (the so-called Uniform Convergence Theorem) is one of the most fundamental theorems in the theory. Many important properties of  $\mathcal{RV}$  functions follow from it.

**Theorem 1.1.** If  $f \in \mathcal{RV}(\vartheta)$ , then the relation (1.1) (and so (1.2)) holds uniformly on each compact  $\lambda$ -set in  $(0, \infty)$ .

The second fundamental result is the following Representation Theorem. It follows from the previous result and vice versa.

**Theorem 1.2.** A function L is slowly varying if and only if it has the form

$$L(t) = \varphi(t) \exp\left\{\int_{a}^{t} \frac{\psi(s)}{s} \, ds\right\},\tag{1.4}$$

 $t \ge a$ , for some a > 0, where  $\varphi, \psi$  are measurable with  $\lim_{t\to\infty} \varphi(t) = C \in (0,\infty)$  and  $\lim_{t\to\infty} \psi(t) = 0$ .

Since  $L, \varphi, \psi$  may be altered at will on finite intervals, the value of *a* is unimportant; if a = 0 one can take  $\psi \equiv 0$  on a neighborhood of 0 to avoid divergence of the integral at the origin. In view of (1.3), a function  $f \in \mathcal{RV}(\vartheta)$  may be written as

$$f(t) = t^{\vartheta}\varphi(t)\exp\left\{\int_{a}^{t}\frac{\psi(s)}{s}\,ds\right\},\tag{1.5}$$

where  $\varphi$  and  $\psi$  are as in the theorem. Alternatively,  $f \in \mathcal{RV}(\vartheta)$  if and only if it has the representation

$$f(t) = \varphi(t) \exp\left\{\int_{a}^{t} \frac{\delta(s)}{s} ds\right\},$$
(1.6)

where  $\delta$  is measurable with  $\lim_{t\to\infty} \delta(t) = \vartheta$ .

The Karamata representation (1.4) is essentially non-unique: within limits, one may always adjust one of  $\varphi(\cdot)$ ,  $\psi(\cdot)$ , making a compensating adjustment to the other. It turns out that the function  $\psi$  may be arbitrarily smooth, but the smoothness properties attainable for  $\varphi$  ale limited by those present in *L*. However, replacing  $\varphi(t)$  by its limit  $C \in (0, \infty)$ , we obtain a *SV* function which is asymptotic to the original one, but with much enhanced properties; this topic will be discussed also later.

As already indicated above, from some points of view (for instance, the measuring of scales of growth like in studying asymptotic behavior of relevant functions), slowly varying functions are of interest only to within an asymptotic equivalence. We then lose nothing by restricting attention to the case  $\varphi(t) \equiv C$  in (1.4) or (1.5) or (1.6). The following definition is pertinent to this situation.

**Definition 1.3.** The regularly varying function of index  $\vartheta$ 

$$f(t) = t^{\vartheta} C \exp\left\{\int_{a}^{t} \frac{\psi(s)}{s} \,\mathrm{d}s\right\},\tag{1.7}$$

 $\lim_{t\to\infty} \psi(t) = 0, C \in (0, \infty)$ , is called *normalized*; we write  $f \in NRV(\vartheta)$ . The set of normalized slowly varying functions, i.e., NRV(0), is denoted as NSV.

For  $L \in NSV$ ,  $\psi(t) = tL'(t)/L(t)$  almost everywhere. Conversely, given a function *L* with  $\psi(t) := tL'(t)/L(t)$  continuous and o(1) at infinity, we may integrate to obtain (1.7), showing *L* to be normalized slowly varying.

The class *NSV* coincides with the so-called Zygmund class which is defined as follows: A positive measurable function *f* belongs to the *Zygmund class* if, for every  $\vartheta > 0$ ,

 $t^{\vartheta}f(t)$  is ultimately increasing and  $t^{-\vartheta}f(t)$  is ultimately decreasing.

### 1.1.3 Karamata theorem

As we will see later, the results of this subsection are extremely useful in applications to the theory of differential equations.

**Theorem 1.3** (Karamata's theorem; direct half). *If*  $L \in SV$ , *then* 

$$\int_{t}^{\infty} s^{\zeta} L(s) \, \mathrm{d}s \sim \frac{1}{-\zeta - 1} t^{\zeta + 1} L(t) \tag{1.8}$$

*provided*  $\zeta < -1$ *, and* 

$$\int_{a}^{t} s^{\zeta} L(s) \,\mathrm{d}s \sim \frac{1}{\zeta + 1} t^{\zeta + 1} L(t) \tag{1.9}$$

provided  $\zeta > -1$ . The integral  $\int_a^{\infty} L(s)/s \, ds$  may or may not converge. The function

$$\tilde{L}(t) = \int_{t}^{\infty} \frac{L(s)}{s} \, \mathrm{d}s \ resp. \ \tilde{L}(t) = \int_{a}^{t} \frac{L(s)}{s} \, \mathrm{d}s \tag{1.10}$$

is a new SV function and  $L(t)/\tilde{L}(t) \rightarrow 0$  as  $t \rightarrow \infty$ .

We shall see in Section 1.2 that actually the integral in (1.10) is, more precisely, a de Haan function, to be defined in that section. For later use, it can be convenient to write the above theorem in a slightly different form.

**Theorem 1.4.** Let  $f \in \mathcal{RV}(\zeta)$  and be locally bounded in  $[a, \infty)$ . Then (*i*) For any  $\sigma < -(\zeta + 1)$  (and for  $\sigma = -(\zeta + 1)$  if  $\int_a^{\infty} s^{-(\zeta+1)} f(s) ds < \infty$ ),

$$t^{\sigma+1}f(t)\Big/\int_t^\infty s^\sigma f(s)\,\mathrm{d} s\to -(\sigma+\zeta+1)\,as\,t\to\infty;$$

(*ii*) for any  $\sigma \ge -(\zeta + 1)$ ,

$$t^{\sigma+1}f(t)\Big/\int_a^t s^{\sigma}f(s)\,\mathrm{d}s\to\sigma+\zeta+1 \ as\ t\to\infty.$$

The above theorems tell us in detail how SV functions behave when multiplied by powers and integrated. It is a remarkable fact that such behavior can only arise in the case of regular variation.

**Theorem 1.5** (Karamata's theorem; converse half). Let f be positive and locally *integrable in*  $[a, \infty)$ .

(i) If for some  $\sigma < -(\zeta + 1)$ ,

$$t^{\sigma+1}f(t)\Big/\int_t^\infty s^\sigma f(s)\,\mathrm{d}s \to -(\sigma+\zeta+1)\,\mathrm{as}\,t\to\infty$$

then  $f \in \mathcal{RV}(\zeta)$ . (ii) If for some  $\sigma > -(\zeta + 1)$ ,

$$t^{\sigma+1}f(t)\Big/\int_a^t s^{\sigma}f(s)\,\mathrm{d}s \to \sigma + \zeta + 1\,\mathrm{as}\,t \to \infty,$$

then  $f \in \mathcal{RV}(\zeta)$ .

#### 1.1.4 Monotonicity

Various aspects of the theory of regular variation are simplified if the functions in questions are assumed monotone. For instance:

• If *L* is eventually positive and monotone and there exists  $\lambda_0 \in (0, \infty) \setminus \{1\}$  with

$$\lim_{t\to\infty}\frac{L(\lambda_0 t)}{L(t)}=1,$$

then  $L \in SV$ .

• If *f* is eventually positive and monotone and

$$\lim_{t \to \infty} \frac{f(\lambda t)}{f(t)} = g(\lambda) \in (0, \infty)$$

for all  $\lambda$  in some dense subset of  $(0, \infty)$  or just for  $\lambda = \lambda_1, \lambda_2$  with  $(\ln \lambda_1) / \ln \lambda_2$  finite and irrational, then  $f \in \mathcal{RV}$ .

• A monotone positive function f is  $\mathcal{RV}$  if and only if there exist two sequences  $\{c_n\}$  and  $\{a_n\}$  of positive numbers with

$$\lim_{n \to \infty} \frac{c_{n+1}}{c_n} = 1, \qquad \lim_{n \to \infty} a_n = \infty$$

such that for all positive  $\lambda$ 

$$\lim_{n\to\infty}c_nf(\lambda a_n)$$

exists, is positive and finite.

## 1.1.5 Further selected properties of $\mathcal{RV}$ functions

- If  $f \in \mathcal{RV}(\vartheta)$ , then  $\ln f(t) / \ln t \to \vartheta$  as  $t \to \infty$ . It then clearly implies that  $\lim_{t\to\infty} f(t) = 0$  provided  $\vartheta < 0$ , and  $\lim_{t\to\infty} f(t) = \infty$  provided  $\vartheta > 0$ .
- If  $f \in \mathcal{RV}(\vartheta)$ , then  $f^{\alpha} \in \mathcal{RV}(\alpha \vartheta)$  for every  $\alpha \in \mathbb{R}$ .
- If  $f_i \in \mathcal{RV}(\vartheta_i)$ ,  $i = 1, 2, f_2(t) \to \infty$  as  $t \to \infty$ , then  $f_1 \circ f_2 \in \mathcal{RV}(\vartheta_1 \vartheta_2)$ .
- If  $f_i \in \mathcal{RV}(\vartheta_i), i = 1, 2$ , then  $f_1 + f_2 \in \mathcal{RV}(\max\{\vartheta_1, \vartheta_2\})$ .
- If  $f_i \in \mathcal{RV}(\vartheta_i)$ , i = 1, 2, then  $f_1 f_2 \in \mathcal{RV}(\vartheta_1 + \vartheta_2)$ .
- If  $f_1, \ldots, f_n \in \mathcal{RV}$ ,  $n \in \mathbb{N}$ , and  $R(x_1, \ldots, x_n)$  is a rational function with positive coefficients, then  $R(f_1, \ldots, f_n) \in \mathcal{RV}$ .
- If  $L \in SV$  and  $\vartheta > 0$ , then  $t^{\vartheta}L(t) \to \infty$ ,  $t^{-\vartheta}L(t) \to 0$  as  $t \to \infty$ .
- Let *f* be eventually positive and differentiable, and let

$$\lim_{t \to \infty} \frac{tf'(t)}{f(t)} = \vartheta$$

Then  $f \in \mathcal{NRV}(\vartheta)$ .

• If  $f \in \mathcal{RV}(\vartheta)$  with  $\vartheta \leq 0$  and  $f(t) = \int_{t}^{\infty} g(s) \, ds$  with *g* nonincreasing, then

$$\frac{-tf'(t)}{f(t)} = \frac{tg(t)}{f(t)} \to -\vartheta \text{ as } t \to \infty.$$

• If  $f \in \mathcal{RV}(\vartheta)$  with  $\vartheta \ge 0$  and  $f(t) = f(t_0) + \int_{t_0}^t g(s) \, ds$  with *g* monotone, then

$$\frac{tf'(t)}{f(t)} = \frac{tg(t)}{f(t)} \to \vartheta \quad \text{as } t \to \infty.$$

• (Similar to the above ones but much earlier results.) If the derivative of  $L \in SV$  is monotone, then  $tL'(t)/L(t) \to 0$  as  $t \to \infty$ .

Suppose that  $F \in \mathcal{RV}(\vartheta)$  with  $\vartheta \in \mathbb{R}$  and that there exists a monotone function f such that for all positive t,  $F(t) = \int_0^t f(s) \, ds$ . Then

$$\lim_{t \to \infty} \frac{tf(t)}{F(t)} = \vartheta$$

Hence, for  $\vartheta \neq 0$ , it holds  $f \operatorname{sgn} \vartheta \in \mathcal{RV}(\vartheta - 1)$ .

• (*Almost monotonicity*) For a positive measurable function *L* it holds:  $L \in SV$  if and only if, for every  $\vartheta > 0$ , there exist a (regularly varying) nondecreasing function *F* and a (regularly varying) nonincreasing function *G* with

$$t^{\vartheta}L(t) \sim F(t)$$
  $t^{-\vartheta}L(t) \sim G(t)$  as  $t \to \infty$ .

In particular, if  $L \in SV$  and  $\vartheta > 0$ ,  $t^{\vartheta}L(t)$  is asymptotic to a nondecreasing function,  $t^{-\vartheta}L(t)$  to a nonincreasing one. Or, a regularly varying function of index  $\vartheta \neq 0$  is almost monotone. Recall that a positive function f is called almost increasing on  $[a, \infty)$  if for some constant M > 0,  $f(s) \ge f(Mt)$ ,  $s \ge t \ge a$ . We may write  $f(t) = O(\inf_{s \ge t} f(t))$  or even  $f(t) \asymp \inf_{s \ge t} f(s)$ . Similarly we define an almost decreasing function f; then we have  $f(t) \asymp \sup_{s > t} f(t)$ .

• (*Asymptotic inversion*) If  $g \in \mathcal{RV}(\vartheta)$  with  $\vartheta > 0$ , then there exists  $g \in \mathcal{RV}(1/\vartheta)$  with

$$f(g(t)) \sim g(f(t)) \sim t \text{ as } t \to \infty.$$

Here g (an "asymptotic inverse" of f) is determined uniquely up to asymptotic equivalence. One version of g is the *generalized inverse* 

$$f^{\leftarrow}(t) := \inf\{s \in [a, \infty) : f(s) > t\}.$$

• (*de Bruijn conjugacy*) If  $L \in SV$ , there exists  $L^{\#} \in SV$ , unique up to asymptotic equivalence, with

$$L(t)L^{\#}(tL(t)) \to 1, \ L^{\#}(t)L(tL^{\#}(t)) \to 1$$

as  $t \to \infty$ . Then  $L^{\#} \sim L$ . The function  $L^{\#}$  is the *Bruijn conjugate* of *L*; (*L*,  $L^{\#}$ ) is a *conjugate pair*.

• (Smooth variation) First we give definition of the class SRV: A positive function varies smoothly with index  $\vartheta \in \mathbb{R}$ , we write  $f \in SRV(\vartheta)$ , if  $h(t) := \ln f(e^t)$  is  $C^{\infty}$ , and

$$h'(t) \to \vartheta, \ h^{(n)}(t) \to 0 \ (n = 2, 3, ...) \text{ as } t \to \infty.$$
 (1.11)

If  $f \in \mathcal{RV}(\vartheta)$ , then there exist  $f_1, f_2 \in \mathcal{SRV}(\vartheta)$  with  $f_1 \sim f_2$  and  $f_1 \leq f \leq f_2$ on some neighborhood of infinity. In particular, if  $f \in \mathcal{RV}(\vartheta)$ , there exists  $g \in \mathcal{SRV}(\vartheta)$  with  $g \sim f$ . Thus for many purposes it suffices to restrict attention to the smoothly varying case.

Note that if  $f \in SRV(\vartheta)$ , then  $tf'(t)/f(t) = h'(\ln t) \rightarrow \vartheta$  and is continuous, whence  $f \in NRV(\vartheta)$ . Condition (1.11) is equivalent to

$$\frac{t^n f^{(n)}(t)}{f(t)} \to \vartheta(\vartheta - 1) \cdots (\vartheta - n + 1),$$

 $n = 1, 2, ..., \text{ as } t \to \infty$ . Smooth variation is well adapted to the processes of integration and differentiation. If  $f \in SRV(\vartheta)$ ,  $\vartheta \neq 0$ , then  $|f'| \in SRV(\vartheta - 1)$ . If  $f \in SRV(\vartheta)$ , then for  $\vartheta > -1$ ,  $\int_a^t f(s) \, ds \in SRV(\vartheta + 1)$ , and for  $\vartheta < -1$ ,  $\int_t^\infty f(s) \, ds \in SRV(\vartheta + 1)$ .

Note that if  $L \in SV$ , then there exists another, infinitely differentiable, SV function  $L_1$  such that  $L_1(t) \sim L(t)$  as  $t \to \infty$  and  $L_1(n) = L(n)$  for all large  $n \in \mathbb{N}$ .

If  $f \in SRV(\vartheta)$  and  $\vartheta \notin \{0, 1, 2, ...\}$ , each derivative will ultimately have constant sign, so  $|f^{(n)}| \in SRV(\vartheta - n)$ . In particular, for  $\vartheta > 1$  the first  $[\vartheta] - 1$  derivatives will be ultimately convex and the  $[\vartheta]$ -th derivative ultimately concave. A related result is the following: If  $f \in RV(\vartheta)$ ,  $\vartheta \notin \{0, 1, 2, ...\}$ , then there exists a  $C^{\infty}$ function g, all of whose derivatives are monotone, with  $f(t) \sim g(t)$ .

• If  $f \in \mathcal{RV}(\vartheta)$ ,  $\vartheta \in \mathbb{R}$ , then for all sequences  $\{a_n\}, \{b_n\}$  of positive numbers with  $\lim_{n\to\infty} a_n = \lim_{n\to\infty} b_n = \infty$  and  $\lim_{n\to\infty} a_n/b_n = \lambda \in (0, \infty)$ , we have

$$\lim_{n \to \infty} \frac{f(a_n)}{f(b_n)} = \lambda^{\vartheta}.$$
(1.12)

If  $\vartheta \in \mathbb{R} \setminus \{0\}$ , the conclusion is also true for  $\lambda = 0$  and  $\lambda = \infty$ .

• (See [2].) Let  $L \in SV$  and assume that  $\int_{a}^{\infty} t^{\eta} |f(t)| dt$  converge for some  $\eta > 0$ . Then  $\int_{a}^{\infty} f(t)L(xt) dt$  exists and

$$\int_{a}^{\infty} f(t)L(xt) \, \mathrm{d}t \sim L(x) \int_{a}^{\infty} f(t) \, \mathrm{d}t$$

as  $x \to \infty$ .

### 1.1.6 Regularly bounded functions

The main limitation of the theory so far developed is the need to assume the existence of the limit  $\lim_{t\to\infty} \frac{f(\lambda t)}{f(t)}$ . A natural and useful generalization of regular variation is the concept of regularly bounded functions (or O-regularly varying functions) introduced by Avakumović in 1935.

**Definition 1.4.** A measurable function  $f : [a, \infty) \to (0, \infty)$  is called *regularly bounded* if

$$0 < \liminf_{t \to \infty} \frac{f(\lambda t)}{f(t)} \le \limsup_{t \to \infty} \frac{f(\lambda t)}{f(t)} < \infty \quad \text{for every } \lambda \ge 1;$$
(1.13)

we write  $f \in \mathcal{RB}$ .

Equivalently,  $f \in \mathcal{RB}$  can be defined via  $\limsup_{t\to\infty} \frac{f(\lambda t)}{f(t)} < \infty$  for all  $\lambda > 0$ . The set of regularly bounded functions at zero (i.e., we take the limit as  $t \to 0$ ) is denoted as  $\mathcal{RB}_0$ .

Clearly any  $\mathcal{RV}$  function is  $\mathcal{RB}$ . Any positive and measurable function which is bounded away from both 0 and  $\infty$  satisfies this definition; thus various simple oscillating functions noted hitherto as not being regularly varying, such as  $2 + \sin t$ and  $t^{\gamma}(1 + \alpha \sin(\ln t))$  with  $\alpha$  small, are regularly bounded, though  $e^t$  still not. It is evident that if measurability is strengthened to monotonicity one of the bounds in  $m \le f(\lambda t)/f(t) \le M$  (such inequalities can alternatively define  $\mathcal{RB}$ ) is automatically satisfied. Further, for instance, if f is nondecreasing, instead of  $\limsup_{t\to\infty} \frac{f(\lambda t)}{f(t)} < \infty$  for some  $\lambda_0 > 1$ .

Here are selected properties of  $\mathcal{RB}$  functions:

- (Uniform convergence theorem for  $\mathcal{RB}$ ) If  $f \in \mathcal{RB}$ , then, for every  $\Lambda > 1$ , (1.13) holds uniformly in  $\lambda \in [1, \Lambda]$ .
- (*Representation theorem for*  $\mathcal{RB}$ ) A function f is regularly bounded if and only if it has the representation

$$f(t) = \exp\left\{\xi(t) + \int_a^t \frac{\eta(s)}{s} \,\mathrm{d}s\right\},\,$$

 $t \ge a$ , where  $\xi$  and  $\eta$  are bounded and measurable on  $[a, \infty)$ . If  $\xi(t) \equiv \text{const}$  in the representation, then *f* is referred to as a *normalized regularly bounded* function.

- A positive continuous function f is regularly bounded if and only if there exist  $\gamma, \delta \in \mathbb{R}, \gamma > \delta$ , such that  $t^{\gamma} f(t)$  is eventually almost increasing and  $t^{\delta} f(t)$  is eventually almost decreasing.
- It holds,  $f \in \mathcal{RB}$  if and only if there exists  $\delta \in \mathbb{R}$  such that

$$\int_{a}^{t} s^{\delta-1} f(s) \, \mathrm{d}s \asymp t^{\delta} f(t) \quad \text{as } t \to \infty.$$

• It holds,  $f \in \mathcal{RB}$  if and only if there exists  $\gamma \in \mathbb{R}$  such that

$$\int_{t}^{\infty} s^{\gamma-1} f(s) \, \mathrm{d}s \asymp t^{\gamma} f(t) \quad \text{as } t \to \infty$$

#### 1.1.7 Rapid variation

We proceed with another problem which naturally arises out: We examine functions for which the limit in (1.1) attains the extreme values.

**Definition 1.5.** A measurable function  $f : [a, \infty) \to (0, \infty)$  is called *rapidly varying of index*  $\infty$ , we write  $f \in \mathcal{RPV}(\infty)$ , if

$$\lim_{t \to \infty} \frac{f(\lambda t)}{f(t)} = \begin{cases} 0 & \text{for } 0 < \lambda < 1, \\ \infty & \text{for } \lambda > 1, \end{cases}$$
(1.14)

and is called *rapidly varying of index*  $-\infty$ , we write  $f \in \mathcal{RPV}(-\infty)$ , if

$$\lim_{t \to \infty} \frac{f(\lambda t)}{f(t)} = \begin{cases} \infty & \text{for } 0 < \lambda < 1, \\ 0 & \text{for } \lambda > 1. \end{cases}$$
(1.15)

The class of all rapidly varying solutions is denoted as  $\mathcal{RPV}$ .

While  $\mathcal{RV}$  functions behaved like power functions (up to a factor which varies "more slowly"),  $\mathcal{RPV}$  functions have a behavior close to that of exponential functions. In particular,  $e^t \in \mathcal{RPV}(\infty)$  and  $e^{-t} \in \mathcal{RPV}(-\infty)$ .

If, for  $\lambda > 0$ , we adopt the convention

	0	for $\lambda < 1$		$\infty$	for $\lambda < 1$
$\lambda^{\infty} = \left\{ \right.$	1	for $\lambda = 1$	$\lambda^{-\infty} = \langle$	1	for $\lambda = 1$
	$\infty$	for $\lambda > 1$		0	for $\lambda > 1$

then regular and rapid variation of *f* can be expressed in a unique formula  $\lim_{t\to\infty} \frac{f(\lambda t)}{f(t)} = \lambda^{\vartheta}$ , where  $\vartheta \in \mathbb{R} \cup \{\pm\infty\}$ .

We now present selected properties of  $\mathcal{RPV}$  functions. Notice that some of the results for regularly varying functions have partial analogues for rapid variation.

- (Uniform convergence theorem for  $\mathcal{RPV}$ ) If  $f \in \mathcal{RPV}(\infty)$ , then (1.14) holds uniformly in  $\lambda$  over all intervals  $(0, M^{-1})$  and  $(M, \infty)$  for every M > 1.
- To establish (1.14), only  $f(\lambda t)/f(t) \to \infty$  for  $\lambda > 1$  has to be proved. Similarly for (1.15).
- Let *f* be positive and differentiable, and let there exist

$$\lim_{t \to \infty} \frac{tf'(t)}{f(t)} = \pm \infty$$

Then  $f \in \mathcal{RPV}$ .

• If  $f \in \mathcal{RPV}$  with f' increasing and  $\lim_{t\to\infty} f(t) = 0$ , then

$$\lim_{t \to \infty} \frac{-tf'(t)}{f(t)} = \infty$$

- There is a connection between a slowly and rapidly varying function. Let *f* be positive, locally bounded, and (globally) unbounded on [0,∞). If *f* ∈ SV, then *f*<sup>←</sup> ∈ RPV(∞) (in fact, RPV(∞) can be replaced here by a certain subclass, see [14, Theorem 2.4.7]). If *f* ∈ RPV(∞), then *f*<sup>←</sup> ∈ SV.
- If  $\vartheta = \pm \infty$ , the conclusion (1.12) is true for monotone function f and  $c \in ([0, \infty) \cup \{\infty\}) \setminus \{1\}$ .
- Suppose the function  $f : (0, \infty) \to (0, \infty)$  is nonincreasing. If  $f \in \mathcal{RPV}(-\infty)$ , then for all  $\vartheta \in \mathbb{R}$ ,  $\int_{1}^{\infty} t^{\vartheta} f(t) dt < \infty$  and

$$\lim_{t \to \infty} \frac{t^{\vartheta + 1} f(t)}{\int_t^\infty s^\vartheta f(s) \, \mathrm{d}s} = \infty.$$
(1.16)

If for some  $\vartheta \in \mathbb{R}$  the integral  $\int_{a}^{\infty} t^{\vartheta} f(t) dt$  converges and (1.16) holds, then  $f \in \mathcal{RPV}(-\infty)$ .

Suppose the function  $f : (0, \infty) \to (0, \infty)$  is nondecreasing. If  $f \in \mathcal{RPV}(\infty)$ , then for all  $\vartheta \in \mathbb{R}$  for which the integral  $\int_0^1 t^\vartheta f(t) dt$  converges, we have

$$\lim_{t \to \infty} \frac{t^{\vartheta + 1} f(t)}{\int_0^t s^\vartheta f(s) \, \mathrm{d}s} = \infty.$$
(1.17)

If for some  $\vartheta \in \mathbb{R}$  the integral  $\int_0^1 t^\vartheta f(t) dt$  converges and (1.17) holds, then  $f \in \mathcal{RPV}(\infty)$ .

## **1.2 De Haan theory**

In this section we work with somehow more general relations than the one defining regular variation, namely such as  $(h(\lambda t) - h(t))/w(t) \rightarrow k(\lambda)$  as  $t \rightarrow \infty$ . Note that functions satisfying this relation (which leads to the class  $\Pi$ ) were introduced by Bojanić and Karamata in 1963. They were rediscovered, and definitely studied by de Haan in his thesis of 1970 [50], who introduced and studied related class  $\Gamma$  which is also discussed below.

### **1.2.1** Class ∏

The Karamata theory considered so far concerns asymptotic relations such as  $f(\lambda t)/f(t) \rightarrow g(\lambda)$  as  $t \rightarrow \infty$ . Writing  $h = \ln f$  and  $k = \ln g$ , this becomes

$$h(\lambda t) - h(t) \rightarrow k(\lambda)$$
 as  $t \rightarrow \infty$ .

One can work instead with the more general relation

$$\frac{h(\lambda t) - h(t)}{w(t)} \to k(\lambda) \quad \text{as } t \to \infty, \text{ for all } \lambda > 0, \tag{1.18}$$

where  $w : (0, \infty) \rightarrow (0, \infty)$  is called *auxiliary function* of *f*. By that a new class of functions — *de Haan class* — is introduced, see de Haan [50] and Geluk, de Haan [47]. The case  $w \equiv 1$  leads to the Karamata case.

First note one context in which such relations naturally arise, namely the "limiting case" in the Karamata theorem. Here the converse half yields no assertion, while the direct half tells us that if *L* varies slowly and  $h(t) := \int_a^t \frac{L(s)}{s} ds$ , then  $\frac{h(t)}{L(t)} \rightarrow \infty$ . Much more precise links between *L* and *h* exist, however, involving differencing, as the Uniform Convergence Theorem yields

$$\frac{h(\lambda t) - h(t)}{L(t)} = \int_{1}^{\lambda} \frac{L(tu)}{L(t)u} \, \mathrm{d}u \to \int_{1}^{\lambda} \frac{\mathrm{d}u}{u} = \ln \lambda$$

as  $t \to \infty$ .

The denominator w in (1.18) needs in general to be taken regularly varying. If  $w \in \mathcal{RV}(\vartheta)$ , then  $k(\lambda)$ , if it exists finite for all  $\lambda > 0$ , has to be of the form

$$k(\lambda) = \begin{cases} \ln \lambda & \text{for } \vartheta = 0, \\ \frac{\lambda^{\vartheta} - 1}{\vartheta} & \text{for } \vartheta \neq 0. \end{cases}$$

If  $\vartheta \neq 0$ , nothing new is obtained (essentially we get regular variation). But  $\vartheta = 0$  leads to a new useful class. This class is, after taking absolute values, a proper subclass of the Karamata class SV.

**Definition 1.6.** A measurable function  $f \in [a, \infty) \to \mathbb{R}$  is said to belong to the class  $\Pi$  if there exists a function  $w : (0, \infty) \to (0, \infty)$  such that for  $\lambda > 0$ 

$$\lim_{t \to \infty} \frac{f(\lambda t) - f(t)}{w(t)} = \ln \lambda; \tag{1.19}$$

we write  $f \in \Pi$  or  $f \in \Pi(w)$ . The function *w* is called an *auxiliary function* for *f*.

Note that de Haan [50] studied the class  $\Pi_C(w)$  of SV functions for which a positive and measurable function w exists such that for all  $\lambda > 0$ ,

$$\lim_{t \to \infty} \frac{f(\lambda t) - f(t)}{w(t)} = C \ln \lambda.$$
(1.20)

If  $C \neq 0$ , then w must be SV. If C = 0 the slow variation of w is assumed. Bingham et al. [14] studied classes of functions satisfying general asymptotic relations related to (1.20). For instance, for  $w \in RV(\vartheta)$ , they consider the class of measurable f satisfying for all  $\lambda \geq 1$ 

$$\lim_{t\to\infty}\frac{f(\lambda t)-f(t)}{w(t)}=Ck_{\vartheta}(\lambda), \quad \text{where } k_{\vartheta}(\lambda)=\int_{1}^{\lambda}u^{\vartheta-1}\mathrm{d}u.$$

The resulting theory (de Haan theory) is both a direct generalization of the Karamata theory above and what is needed to fill certain gaps, or boundary cases, in Karamata's main theorem. The term "second-order theory" is sometimes used for this study. The original motivation was probabilistic. De Haan class presents itself as very fruitful in various applications; as we will see later they include applications in differential equations.

Let us give several examples. The functions f defined by

$$\begin{split} f(t) &= \ln t + o(1), \\ f(t) &= (\ln t)^{\alpha} (\ln_2 t)^{\beta} + o(\ln t)^{\alpha - 1}, \ \alpha > 0, \beta \in \mathbb{R}, \\ f(t) &= \exp\{(\ln t)^{\gamma}\} + o(\ln t)^{\gamma - 1} \exp\{(\ln t)^{\gamma}\}, \ 0 < \gamma < 1, \\ f(t) &= t^{-1} \ln \Gamma(t) + o(1) \end{split}$$

are in  $\Pi$ . The function

$$f(t) = 2\ln t + \sin\ln t$$

is in SV, but not in  $\Pi$ .

Now we present selected properties of functions in the class  $\Pi$ .

- If  $f \in \Pi$ , then for  $0 < c < d < \infty$  relation (1.19) holds uniformly for  $\lambda \in [c, d]$ .
- Auxiliary function is unique up to asymptotic equivalence.
- The statements  $f \in \Pi$  and

$$\lim_{t \to \infty} \frac{f(\lambda t) - f(t)}{f(t) - \frac{1}{t} \int_{a}^{t} f(s) \, \mathrm{d}s} = \ln \lambda$$

for  $\lambda > 0$  are equivalent.

• The statements  $f \in \Pi$  and there exists  $L \in SV$  such that

$$f(t) = L(t) + \int_{a}^{t} \frac{L(s)}{s} ds$$
 (1.21)

are equivalent.

• If *f* satisfies (1.21), then  $f \in \Pi(L)$ . Hence

$$L(t) \sim f(t) - \frac{1}{t} \int_{a}^{t} f(s) \,\mathrm{d}s$$
 (1.22)

as  $t \to \infty$ . If  $f \in \Pi(L)$  is integrable on finite intervals of  $(0, \infty)$ , then (1.22) holds.

• If  $f \in \Pi$ , then  $\lim_{t\to\infty} f(t) =: f(\infty) \le \infty$  exists. If the limit is infinite, then  $f \in SV$ . If the limit is finite, then  $f(\infty) - f(t) \in SV$ .

- If  $-f_i \in \Pi(w_i)$ , where  $f_i$  is eventually positive, i = 1, 2, then  $-f_1 f_2 \in \Pi(f_1 w_2 + f_2 w_1)$ .
- If  $f \in \Pi(w)$ , *g* is measurable and  $\{f(t) g(t)\}/w(t) \to c \in \mathbb{R}$  as  $t \to \infty$ , then  $g \in \Pi(w)$ .
- If  $f \in \Pi(w)$ , then for any  $\varepsilon > 0$  there exist  $s_0, M \in (0, \infty)$  such that for  $s \ge s_0, t \ge 1$ ,

$$\left|\frac{f(st) - f(s)}{w(s)}\right| \le Mt^{\varepsilon}$$

For further properties see [14, 47, 50]. There are another classes, closely related to  $\Pi$ , which can also be important for our purposes. Geluk in [44] introduces  $\Pi$ -regular variation:

$$f \in \Pi \mathcal{RV}(\vartheta)$$
 if an only if  $\frac{f(t)}{t^{\vartheta}} \in \Pi$ .

Further, in Omey, Willekens [134], the following class of functions was introduced. As we will see, it opens further possibilities in obtaining more precise information about certain solutions of certain differential equations.

**Definition 1.7.** Let  $f : (0, \infty) \to \mathbb{R}$  be measurable. If there exist measurable functions w, z such that  $z \in SV$  and

$$\frac{f(\lambda t) - f(t) - w(t) \ln \lambda}{z(t)} \to H(\lambda) \quad \text{as } t \to \infty \text{ for } \lambda > 0,$$

for some function  $H(\lambda)$ , then we write  $f \in \prod R_2(w, z)$ .

Next we present selected information about the class  $\Pi R_2(w, z)$ .

• If z(t) = o(w(t)), then  $f \in \prod R_2(w, z)$  implies that  $f \in \prod$ .

• (Auxiliary concepts) Consider functions h satisfying

$$h(\lambda t) - h(t) \sim k(\lambda)g(t)$$

as  $t \to \infty$ . If  $g \in SV$ , the limit function  $k(\lambda)$  can be characterized as follows: for each  $\mu, \lambda > 0$ , we have

$$k(\lambda \mu) = k(\lambda) + k(\mu).$$

Hence  $k(\lambda) = c \ln \lambda$  for some real *c*. The corresponding class of functions *h* will be denoted by  $\Pi V(c, g)$ . Note that  $h \in \mathcal{RV}(\vartheta)$  if and only if  $\ln h \in \Pi V(\vartheta, 1)$ . If  $c \neq 1$ , the class  $\Pi V(c, g)$  is the class  $\Pi$ .

• (*Representation theorem*) Suppose that  $z \in SV$  and f is locally bounded and define

$$G(t) = f(t) - \frac{1}{t} \int_0^t f(s) \,\mathrm{d}s,$$

t > 0. Then  $f \in \prod R_2(w, z)$  if and only if there exists  $c \in \mathbb{R}$  such that  $G \in \prod V(c, z)$ . Moreover, if  $f \in \prod R_2(w, z)$ , then

$$\lim_{t \to \infty} \frac{w(t) - G(t)}{z(t)} = c_0$$

exists and  $f \in \prod R_2(G, z)$  with limit function  $H(\lambda) = c \ln \lambda + \frac{1}{2}c(\ln \lambda)^2$ .

### 1.2.2 Class Γ

For  $\mathcal{RV}$  functions (generalized) inversion gives again an  $\mathcal{RV}$  function. For nondecreasing unbounded functions in the class  $\Pi$  (which forms a proper subset of  $\mathcal{SV}$ ) we obtain the following class by inversion; it is a useful subclass of rapidly varying functions.

Another view of understanding is an extension of the notion of  $\mathcal{RV}$  functions defined by  $\lim_{t\to\infty} f(\lambda t)/f(t) = \lambda^{\vartheta}$  in the sense that we consider the class of functions satisfying the following property: There exists a function  $g : (0, \infty) \to (0, \infty)$  and  $\vartheta \in \mathbb{R}$  such that

$$\lim_{t \to \infty} \frac{f\left(t\lambda^{g(t)}\right)}{f(t)} = \lambda^{g(t)}$$

for all positive  $\lambda$ . First we confine our considerations to nondecreasing functions f and ask for a characterization of the class of functions for which this relation holds with  $\vartheta > 0$ . Without loss of generality we may take  $\vartheta = 1$  (this only involves a trivial change in g).

It turns out to be more convenient to start with the following definition which is a mere transformation of the one just given.

**Definition 1.8.** A nondecreasing function  $f : \mathbb{R} \to (0, \infty)$  is said to belong to the class  $\Gamma$  if there exists a function  $v : \mathbb{R} \to (0, \infty)$  such that for all  $\lambda \in \mathbb{R}$ 

$$\lim_{t \to \infty} \frac{f(t + \lambda v(t))}{f(t)} = e^{\lambda};$$
(1.23)

we write  $f \in \Gamma$  or  $f \in \Gamma(v)$ . The function v is called an *auxiliary function* for f.

The following functions satisfy (1.23) with the given auxiliary functions *v*:

$$f(t) = \exp(t^{\alpha}) \text{ for fixed } \alpha > 0 \quad \text{with } v(t) = \begin{cases} 1 & \text{for } t \le 0 \\ t^{1-\alpha}/\alpha & \text{for } t > 0 \end{cases}$$
$$f(t) = \exp(t \ln t) \quad \text{with } v(t) = \begin{cases} 1 & \text{for } t \le 1 \\ 1/\ln t & \text{for } t > 1, \end{cases}$$
$$f(t) = \exp(e^{t}) \quad \text{with } v(t) = e^{-t}.$$

We now give selected properties of functions in the class  $\Gamma$ .

- If  $f \in \Gamma$ , then  $f \in \mathcal{RPV}(\infty)$ .
- Relation (1.23) holds uniformly on each bounded interval.
- Any positive function *z* is an auxiliary function for *f* if and only if  $z(t) \sim v(t)$  as  $t \to \infty$ .
- An auxiliary function v in (1.23) cannot grow so fast:  $v(t)/t \rightarrow 0$  as  $t \rightarrow \infty$ .
- If  $f \in \Pi(w)$  and  $f(\infty) = \infty$ , then  $f^{\leftarrow} \in \Gamma(v)$  with  $v(t) \sim w(f^{\leftarrow}(t))$  as  $t \to \infty$ .
- If  $g \in \Gamma(v)$  and  $g(\infty) = \infty$ , then  $g^{\leftarrow} \in \Pi(w)$  with  $w(t) \sim v(g^{\leftarrow}(t))$  as  $t \to \infty$ .
- The statements  $f \in \Gamma$  and

$$\lim_{t \to \infty} \frac{f(t) \int_0^t \int_0^s f(\tau) \, \mathrm{d}\tau \, \mathrm{d}s}{\left(\int_0^t f(s) \, \mathrm{d}s\right)^2} = 1$$

are equivalent.

• If  $f \in \Gamma$ , then for all positive  $\alpha$ 

$$\lim_{t\to\infty}\frac{\int_0^t f^{\alpha}(s)\,\mathrm{d}s}{f^{\alpha-1}(t)\,\int_0^t f(s)\,\mathrm{d}s}=\frac{1}{\alpha}.$$

Conversely, if a positive nondecreasing function satisfies this relation for some positive  $\alpha \neq 1$ , then  $f \in \Gamma$ .

• (*Representation*) The statements  $f \in \Gamma$  and

$$f(t) = \exp\left\{\eta(t) + \int_0^t \frac{\psi(s)}{\varphi(s)} \,\mathrm{d}s\right\},\,$$

where  $\eta(t) \to c \in \mathbb{R}$ ,  $\psi(t) \to 1$  as  $t \to \infty$ ,  $\varphi$  is positive, absolutely continuous with  $\varphi'(t) \to 0$  as  $t \to \infty$ , are equivalent. The auxiliary function of *f* may be taken as  $\varphi$ .

• (*Representation*) The statements  $f \in \Gamma$  and

$$f(t) = \exp\left\{\eta(t) + \int_0^t \frac{1}{\omega(s)} \,\mathrm{d}s\right\},\,$$

where  $\eta(t) \to c \in \mathbb{R}$  as  $t \to \infty$  and  $\omega \in SN$  (*SN* being defined below), are equivalent. The auxiliary function of *f* may be taken as  $\omega$ .

• If  $g \in SN$  (*SN* being defined below), then *f* defined by

$$f(t) = \exp\left\{\int_0^t \frac{\mathrm{d}s}{g(s)}\right\}$$

satisfies  $f \in \Gamma(g)$ .

- If  $f \in \Gamma$  has a nondecreasing positive derivative f', then  $f' \in \Gamma$ .
- If  $f \in \Gamma(v)$ , then

$$v(t) \sim \frac{\int_0^t f(s) \,\mathrm{d}s}{f(t)}$$
 as  $t \to \infty$ .

Hence v can always be taken measurable.

- If  $f \in \Gamma(v)$ , then  $v(t + \lambda v(t)) \sim v(t)$  as  $t \to \infty$  uniformly on finite intervals of  $\mathbb{R}$  (that is v is in the below defined class SN).
- If  $f \in \Gamma(v)$ , then  $\int_0^t f(s) \, ds \in \Gamma(v)$ .
- If  $f_1 \in \mathcal{RV}(\vartheta)$  with  $\vartheta > 0$  and  $f_2 \in \Gamma$ , then  $f_1 \circ f_2 \in \Gamma$ .
- If  $f_1 \in \Gamma$  and  $f'_2 \in \mathcal{RV}(\vartheta)$  with  $-1 < \vartheta < \infty$ , then  $f_1 \circ f_2 \in \Gamma$ .
- If  $f_1 \in \Gamma$  and  $f'_2 \in \Gamma$ , then  $f_1 \circ f_2 \in \Gamma$ .
- For a nondecreasing function  $f, f \in \Gamma \cup \mathcal{RV}$  if and only if

$$\lim_{t \to \infty} \frac{f(t) \int_0^t \int_0^s f(\tau) \, \mathrm{d}\tau \, \mathrm{d}s}{\left(\int_0^t f(s) \, \mathrm{d}s\right)^2} = C.$$
 (1.24)

Under these conditions necessarily  $C \in [1/2, 1]$ . If C = 1, then  $f \in \Gamma$ . If C < 1, then  $f \in \mathcal{RV}(1/(1 - C) - 2)$ .

We already know that  $\Gamma \subseteq \mathcal{RPV}(\infty)$ . For an example showing that the inclusion is strict see the proof of [14, Proposition 2.4.4].

In applications to differential equations we will also use the class  $\Gamma_-$  defined as follows. A function  $f \in \Gamma_-(v)$  if  $1/f \in \Gamma(v)$ . Alternatively we can define: A function  $f : (0, \infty) \to (0, \infty)$  is in the class  $\Gamma(\sigma; v)$ ,  $\sigma$  being a real number and  $v \in SN$ (see below), if

$$\lim_{t \to \infty} \frac{f(t + \lambda v(t))}{f(t)} = \exp(\sigma \lambda)$$

for all real  $\lambda$ . Clearly,  $\Gamma(1; v) = \Gamma(v)$  and  $\Gamma(-1, v) = \Gamma_{-}(v)$ . It holds:

If 
$$f' \in \Gamma(v)$$
, then  $\lim_{t \to \infty} f(t) = A$  exists,  $A - f(t) \sim v(t)f(t)$ ,

and  $A - f(t) \in \Gamma_{-}(v)$ . (1.25)

The class of auxiliary functions for functions in the class  $\Gamma$  is an interesting class in its own right since it can be used in either context as well.

**Definition 1.9.** A measurable function  $f : \mathbb{R} \to (0, \infty)$  is *Beurling slowly varying* if

$$\lim_{t \to \infty} \frac{f(t + \lambda f(t))}{f(t)} = 1 \quad \text{for all } \lambda \in \mathbb{R};$$
(1.26)

we write  $f \in BSV$ . If (1.26) holds locally uniformly in  $\lambda$ , then f is called *self-neglecting*; we write  $f \in SN$ .

Here are selected properties concerning the classes  $\mathcal{BSV}$  and  $\mathcal{SN}$ .

- If  $f \in \mathcal{BSV}$  is continuous, then  $f \in \mathcal{SN}$ .
- (*Representation*) It holds  $f \in SN$  if and only if it has the representation

$$f(t) = \varphi(t) \int_0^t \psi(s) \, \mathrm{d}s,$$

where  $\lim_{t\to\infty} \varphi(t) = 1$  and  $\psi$  is continuous with  $\lim_{t\to\infty} \psi(t) = 0$ .

- If  $f \in SN$ , then  $\lim_{t\to\infty} f(t)/t = 0$ .
- If  $f \in \mathcal{BSV}$  is continuous, then there exists  $g \in C^1$  such that  $f(t) \sim g(t)$  and  $g'(t) \to 0$  as  $t \to \infty$ .

Consider  $f \in \mathcal{RV}(\vartheta), \vartheta > -1$ . Then

$$\frac{\int_0^t f(s) \, \mathrm{d}s}{tf(t)} \to \frac{1}{\vartheta + 1} \quad \text{and} \quad \frac{\int_0^t \int_0^s f(\tau) \, \mathrm{d}\tau \, \mathrm{d}s}{t \int_0^t f(s) \, \mathrm{d}s} \to \frac{1}{\vartheta + 2}$$

as  $t \to \infty$ , see the Karamata theorem and (1.24). By setting

$$h(t) = \int_0^t \int_0^s f(\tau) \,\mathrm{d}\tau \,\mathrm{d}s$$

and combining (1.1) with the above two relations we find that h satisfies the differential equation

$$h(t)h''(t) = \varphi(t)h'^{2}(t), \qquad (1.27)$$

where  $\varphi(t) \rightarrow (\vartheta+1)/(\vartheta+2)$ . This leads to the idea to start from (1.27) and study the asymptotic behavior of nonnegative solutions of (1.27) under various conditions on  $\varphi$ , see Omey [132]. For instance, the following holds, cf. (1.24). Suppose that *h* is a nonnegative solution of (1.27) and  $\varphi(t) \rightarrow C \in \mathbb{R} \cup \{\pm\infty\}$  as  $t \rightarrow \infty$ . If C < 1 or C > 1, then  $h \in \mathcal{RV}(\beta)$  where  $\beta = 1/(1 - C)$ ; here  $\beta = 0$  if  $C = \pm\infty$ . If C = 1, then  $h \in \Gamma(d, v)$  where  $v = |h/h'| \in SN$  and d = 1 or d = -1 depending on the sign of *h'*. Second order behavior is also studied in Omey [132].

The behavior of a function  $f \in BSV$  is controlled by the function f itself. A similar class of functions controlled by another function z is defined as follows

(see e.g. [131, 133]). Suppose that  $z \in SN$ . A positive and measurable function f is controlled by z if for all real  $\lambda$ ,

$$\frac{f(t+\lambda z(t))}{f(t)} \to 1$$

as  $t \to \infty$ ; we write  $f \in SC(z)$ . It holds: If  $f' \in SC(z)$ , then for all real  $\lambda$ ,

$$f(t + \lambda z(t)) - f(t) = (1 + o(1))\lambda z(t)f'(t)$$
(1.28)

locally uniformly in  $\lambda$ . For other classes related to  $\Gamma$  see [133].

## **1.3** Some other related classes

#### 1.3.1 Generalized regularly varying functions

The following type of extension of  $\mathcal{RV}$  functions was introduced in [60]. Motivation was primarily for purposes of studying differential equations. Consider a continuously differentiable function  $\omega$  which is positive and satisfies  $\omega'(t) > 0$  for  $t \in [b, \infty)$  and  $\lim_{t\to\infty} \omega(t) = \infty$ .

**Definition 1.10.** A measurable function  $f : [a, \infty) \to (0, \infty)$  is called *regularly* varying of index  $\vartheta$  with respect to  $\omega$  if  $f \circ \omega^{-1} \in \mathcal{RV}(\vartheta)$ ; we write  $f \in \mathcal{RV}_{\omega}(\vartheta)$ . If  $\vartheta = 0$ , then f is called *slowly varying with respect to*  $\omega$ ; we write  $f \in \mathcal{SV}_{\omega}$ .

The following selected properties of generalized  $\mathcal{RV}$  functions are mostly immediate consequences of the properties of  $\mathcal{RV}$  functions.

•  $f \in \mathcal{RV}_{\omega}(\vartheta)$  if and only if  $f(t) = \omega^{\vartheta}(t)L_{\omega}(t)$ , where  $L_{\omega} \in \mathcal{SV}_{\omega}$ .

- If  $L_{\omega} \in SV_{\omega}$  and  $\vartheta > 0$ , then  $\omega^{\vartheta}(t)L_{\omega}(t) \to \infty$ ,  $\omega^{-\vartheta}(t)L_{\omega}(t) \to 0$  as  $t \to \infty$ .
- (*The representation theorem*):  $L_{\omega} \in SV_{\omega}$  if and only if

$$L_{\omega}(t) = c(t) \exp\left\{\int_{a}^{t} \frac{\omega'(s)}{\omega(s)} h(s) \,\mathrm{d}s\right\},\tag{1.29}$$

 $t \ge a$ , for some a > 0, where c, h are measurable and  $c(t) \to c \in (0, \infty)$ ,  $h(t) \to 0$ as  $t \to \infty$ . If  $c(t) \equiv c$  in (1.29), then  $L_{\omega}$  is called *normalized slowly varying with respect to*  $\omega$ ; we write  $L_{\omega} \in NSV_{\omega}$ . A generalized regularly varying function  $f(t) = \omega^{\vartheta}(t)L_{\omega}(t)$  with  $L_{\omega} \in NSV_{\omega}$  is called *normalized regularly varying* of index  $\vartheta$  with respect to  $\omega$ ; we write  $f \in NRV_{\omega}(\vartheta)$ .

• Representation of a generalized  $\mathcal{RV}$ -function can alternatively be written as follows:  $f \in \mathcal{RV}_{\omega}(\vartheta)$  if and only if

$$f(t) = c(t) \exp\left\{\int_{a}^{t} \frac{\omega'(s)}{\omega(s)} \delta(s) \,\mathrm{d}s\right\},\tag{1.30}$$

 $t \ge a$ , for some a > 0, where  $c, \delta$  are measurable and  $c(t) \rightarrow c \in (0, \infty), \delta(t) \rightarrow \vartheta$  as  $t \rightarrow \infty$ .

• If  $\omega(t) \sim K\varphi(t)$  as  $t \to \infty$  for some constant K > 0, then

$$\mathcal{RV}_{\omega}(\vartheta) = \mathcal{RV}_{\omega}(\vartheta)$$

for any  $\vartheta \in \mathbb{R}$ .

• It holds

$$\mathcal{RV}_{\omega^{\gamma}}(\vartheta) = \mathcal{RV}_{\omega}(\vartheta\gamma)$$

for any  $\vartheta \in \mathbb{R}$  and  $\gamma \in (0, \infty)$ .

• There hold  $\mathcal{RV}_{id}(\vartheta) = \mathcal{RV}(\vartheta)$  and  $\mathcal{NRV}_{id}(\vartheta) = \mathcal{NRV}(\vartheta)$ .

It would be of interest to observe that there exists a function which is slowly varying in the generalized sense but is not slowly varying in the sense of Karamata, so that, roughly speaking, the class of generalized Karamata functions is larger than that of classical Karamata functions. In fact, using the notation

$$\exp_0 t = t, \ \exp_n t = \exp(\exp_{n-1} t), \quad n = 1, 2, \dots,$$
$$\ln_0 t = t, \ \ln_n t = \ln(\ln_{n-1} t), \quad n = 1, 2, \dots,$$

we define the functions  $\phi_n(t)$  and  $f_n(t)$  for  $n \in \mathbb{Z}$  by

$$\phi_n(t) = \exp_n t, \ \phi_{-n}(t) = \log_n t, \ n = 0, 1, 2, \dots,$$

and

$$f_n(t) = 2 + \sin \phi_n(t), \quad n = 0, \pm 1, \pm 2, \dots$$

Since  $\phi_n^{-1}(t) = \phi_{-n}(t)$  and  $\phi_m \circ \phi_n(t) = \phi_{m+n}(t)$  for any  $m, n \in \mathbb{Z}$ , we have

$$f_n \circ \phi_m^{-1}(t) = f_{n-m}(t)$$

for any  $m, n \in \mathbb{Z}$ , from which, by taking into account the fact that

$$f_n(t) \in SV$$
 for  $n \leq -2$  and  $f_n(t) \notin SV$  for  $n \geq -1$ ,

we conclude that

$$f_n(t) \notin SV$$
 and  $f_n(t) \in SV_{\phi_m}$  if  $n \ge -1$  and  $m \ge n+2$ .

Regular boundedness is generalized as follows.

**Definition 1.11.** A measurable function  $f : [a, \infty) \to (0, \infty)$  is called *regularly bounded with respect to*  $\omega$  if  $f \circ \omega^{-1} \in \mathcal{RB}$ ; we write  $f \in \mathcal{RB}_{\omega}$ .

A function *f* belongs to  $\mathcal{RB}_{\omega}$  if and only if it has the representation

$$f(t) = \exp\left\{\eta(t) + \int_a^t \frac{\omega'(s)}{\omega(s)}\xi(s)\,\mathrm{d}s\right\},\,$$

where  $\eta$  and  $\xi$  are bounded measurable functions on  $[a, \infty)$ .

#### 1.3.2 Subexponential functions

Now we recal the concept of subexponential functions which have been shown useful in the study of asymptotic properties of differential equations, and are also somehow related to  $\mathcal{RV}$  functions, see e.g. [5, 7].

The convolution of two appropriate functions f, g defined on  $[0, \infty)$  is denoted, as usual, by

$$(f * g)(t) = \int_0^t f(t-s)g(s) \,\mathrm{d}s, \quad t \ge 0.$$

**Definition 1.12.** Let  $f : [0, \infty) \to (0, \infty)$  be a continuous function. Then we say that *f* is (*positive*) *subexponential* if

$$\lim_{t \to \infty} \frac{(f * f)(t)}{f(t)} = 2 \int_0^\infty f(s) \, \mathrm{d}s \tag{1.31}$$

(where we assume the convergence of the integral) and

$$\lim_{t \to \infty} \sup_{0 \le s \le T} \left| \frac{f(t-s)}{f(t)} - 1 \right| = 0 \quad \text{for all } T > 0 \tag{1.32}$$

(i.e.,  $\lim_{t\to\infty} f(t-s)/f(t)$  uniformly for  $0 \le s \le T$  for all T > 0).

The nomenclature subexponential is suggested by the fact that (1.32) implies that, for every  $\varepsilon > 0$ ,  $f(t)e^{\varepsilon t} \to \infty$  as  $t \to \infty$ , see e.g. [8]. It is also true that  $\lim_{t\to\infty} f(t) = 0$ . In the definition above, condition (1.31) can be replaced by

$$\lim_{T \to \infty} \limsup_{t \to \infty} \frac{1}{f(t)} \int_{T}^{t-T} f(t-s)f(s) \, \mathrm{d}s = 0$$

and this latter condition often proves to be useful in proofs.

The properties of subexponential functions have been extensively studied, for example, in [7, 8, 21]. Simple examples of subexponential functions are

$$f(t) = (1 + t)^{-\alpha} \text{ for } \alpha > 1,$$
  

$$f(t) = e^{-(1+t)^{\alpha}} \text{ for } 0 < \alpha < 1,$$
  

$$f(t) = e^{-t/\ln(t+2)}.$$

The class of subexponential functions therefore includes a wide variety of functions exhibiting polynomial and slower-than-exponential decay: nor is the slower-than-exponential decay limited to a class of polynomially decaying functions. It is noted in [7] that the class of (positive) subexponential functions includes all continuous, integrable functions which are regularly varying at infinity. If  $g \in \mathcal{RV}(\vartheta)$  with  $\vartheta < -1$ , g is subexponential.

### **1.3.3 Regular variation on various time scales**

The concept of regular variation can be extended in such a way that it allows us to study asymptotic behavior of difference equations or q-difference equations or dynamic equations on time scales, see Section 6.2, Section 6.3, Section 6.4, respectively. Since this text is focused on differential equations, we will be very brief.

The concept of regularly sequences was introduced already by Karamata in 1930. In fact, two main approaches are known in the basic theory of regularly varying sequences: the approach due to Karamata [73], based on a definition that can be understood as a direct discrete counterpart of simple and elegant continuous definition, and the approach due to Galambos and Seneta [43], based on purely sequential definition.

**Definition 1.13** (Karamata [73]). A positive sequence  $\{y_k\}$ ,  $k \in \mathbb{N}$ , is said to be *regularly varying of index*  $\vartheta$ ,  $\vartheta \in \mathbb{R}$ , if

$$\lim_{k \to \infty} \frac{y_{[\lambda k]}}{y_k} = \lambda^{\vartheta} \text{ for all } \lambda > 0, \qquad (1.33)$$

where [*u*] denotes the integer part of *u*.

**Definition 1.14** (Galambos and Seneta [43]). A positive sequence  $\{y_k\}$ ,  $k \in \mathbb{N}$ , is said to be *regularly varying of index*  $\vartheta$ ,  $\vartheta \in \mathbb{R}$ , if there is a positive sequence  $\{\alpha_k\}$  satisfying

$$\lim_{k \to \infty} \frac{y_k}{\alpha_k} = C, \quad \lim_{k \to \infty} k \left( 1 - \frac{\alpha_{k-1}}{\alpha_k} \right) = \vartheta, \tag{1.34}$$

*C* being a positive constant.

If  $\rho = 0$  in Definition 1.13 or 1.14, then  $\{y_k\}$  is said to be *slowly varying*.

In Bojanić, Seneta [17], it was shown that Definition 1.13 is equivalent to Definition 1.14.

In Matucci, Řehák [119] the authors are interested in applying regularly varying sequences to certain second order difference equations. For this purpose a slight modification (in the equivalent sense) of Definition 1.14 is proposed there; the latter condition in (1.34) is replaced by

$$\lim_{k \to \infty} \frac{k \Delta \alpha_k}{\alpha_k} = \vartheta$$

In Bojanić, Seneta [17] and Galambos, Seneta [43], the so-called embedding theorem was established (and the converse result holds as well):

**Theorem 1.6.** If  $\{y_k\}$  is a regularly varying sequence, then the function R (of a real variable), defined by  $R(t) = y_{[t]}$ , is regularly varying.

Such a result makes it then possible to apply the continuous theory to the theory of regularly varying sequences. However, the development of a discrete theory, analogous to the continuous one, is not generally close, and sometimes far from a simple imitation of arguments for regularly varying functions, as noticed and demonstrated in Bojanić, Seneta [17]. Simply, the embedding theorem is just one of powerful tools, but sometimes it is not immediate that from a continuous results its discrete counterpart is easily obtained thanks to the embedding; sometimes it is even not possible to use this tool and the discrete theory requires a specific approach, different from the continuous one.

For further properties of regularly varying sequences and some other related useful concepts (such as rapidly varying sequences) see e.g. [17, 25, 26, 43].

References concerning applications of regular variation in difference equations are given in Section 6.2.

In Řehák [141], the concept of regularly varying functions on time scales (or measure chains) was introduced; the primary purpose was to investigate dynamic equations on time scales.

Recall that the calculus on time scales (or, more generally, on measure chains) deals essentially with functions defined on nonempty closed subsets of  $\mathbb{R}$ , see Hilger's initiating work [55] and the monograph [16] by Bohner, Peterson. Hence, it unifies and extends usual calculus and quantum (*q*- or *h*-) calculi.

A theory of regular variation on time scales offers something more than the embedding result, and has the following advantages: Once there is proved a result on a general time scale, it automatically holds for the continuous and the discrete case, without any other effort. Moreover, at the same time, the theory works also on other time scales which may be different from the "classical" ones.

A time scale  $\mathbb{T}$  is assumed to be unbounded above. The following definition is motivated by a modification of the purely sequential criterion mentioned above.

**Definition 1.15.** A measurable function  $f : \mathbb{T} \to (0, \infty)$  is said to be *regularly varying* of index  $\vartheta$ ,  $\vartheta \in \mathbb{R}$ , if there exists a positive rd-continuously delta differentiable function  $\alpha$  satisfying

$$f(t) \sim C\alpha(t)$$
 and  $\lim_{t \to \infty} \frac{t\alpha^{\Delta}(t)}{\alpha(t)} = \vartheta$ ,

*C* being a positive constant. If  $\vartheta = 0$ , then *f* is said to be *slowly varying*.

In Řehák, Vítovec [153], a Karamata type definition for  $\mathcal{RV}$  functions on  $\mathbb{T}$  is introduced and an embedding theorem is proved. Note that conditions posed on the behavior of the graininess  $\mu(t)$  of a time scale  $\mathbb{T}$  plays a crucial role in the theory. In particular we suggest to distinguish three cases: (a) The graininess satisfies the condition  $\mu(t) = o(t)$  as  $t \to \infty$ . Then we obtain a continuous like theory which unifies the above discrete and continuous theories. (b) The case where  $\mu(t) = Ct$ with  $C \in (0, \infty)$  leads to the *q*-case, which is discussed below. (c) Other cases — in
particular when the graininess is "too large" or a "combination of large and small" — give no reasonable theory of regular variation in a certain sense.

Applications of this theory (especially when  $\mu(t) = o(t)$ ) to dynamic equations on time scales can be found in the works listed in Section 6.3.

The concept of the so-called *q*-regularly varying functions was introduced in Řehák, Vítovec [153]. Let  $q^{\mathbb{N}_0}$  denote the *q*-uniform lattice  $q^{\mathbb{N}_0} = \{q^k : k \in \mathbb{N}_0\}, q > 1$ . Let  $D_q f$  denote the Jackson derivative of *f*. Further we set  $[a]_q = (q^a - 1)/(q - 1)$  for  $a \in \mathbb{R}$ . Quantum version of regular variation of  $f : q^{\mathbb{N}_0} \to (0, \infty)$  can be defined in accordance with Definition 1.15 as follows.

**Definition 1.16.** A function  $f : q^{\mathbb{N}_0} \to (0, \infty)$  is said to be *q*-regularly varying of index  $\vartheta, \vartheta \in \mathbb{R}$ , if there exists a function  $\beta : q^{\mathbb{N}_0} \to (0, \infty)$  satisfying

$$f(t) \sim C\beta(t)$$
, and  $\lim_{t \to \infty} \frac{tD_q\beta(t)}{\beta(t)} = [\vartheta]_q$ .

*C* being a positive constant. If  $\vartheta = 0$ , then *f* is said to be *q*-slowly varying.

However, as was shown in the quoted paper, thanks to the structure of  $q^{\mathbb{N}_0}$ , we are able to find much simpler (and still equivalent) characterization which has no analogue in the classical continuous or the discrete case. Such a simplification is possible since *q*-regular variation can be characterized in terms of relations between f(t) and f(qt), which is natural for discrete *q*-calculus, in contrast to other settings. In particular, for  $f : q^{\mathbb{N}_0} \to (0, \infty)$  we have

*f* is *q*-regularly varying of index 
$$\vartheta \iff \lim_{t \to \infty} \frac{f(qt)}{f(t)} = q^{\vartheta}$$
.

For further properties of *q*-regularly varying functions (and other related functions such as *q*-rapidly varying ones) see [142] and [154]. Note that in [145], certain generalization of *q*-regular variation was introduced which somehow involves also *q*-rapid variation or *q*-hypergeometric functions.

Applications of this theory to *q*-difference equations can be found in the works listed in Section 6.4.

#### 1.3.4 Hardy field

The so-called *logarithmico-exponential function* is defined as a real-valued function defined on  $[a, \infty)$  by

a finite combination of the ordinary symbols  $(+, -, \cdot, /, \sqrt[n]{})$ and the functional symbols  $\ln(\cdot)$ ,  $\exp(\cdot)$ , operating on the variable *x* and on real constants,

see Hardy [52, III.2].

#### More generally, see Bourbaki [18], a Hardy field is

a set of germs of real-valued functions on  $[a, \infty)$  that is closed under differentiation and that form a field under the usual addition and multiplication of germs,

see also Rosenlicht [155]. Loosely speaking, Hardy fields are the natural domain of asymptotic analysis, where all rules hold without qualifying conditions.

In Hardy's own words [52, V.6]: "No function has yet presented itself in analysis the laws of whose increase, in so far as they can be stated at all, cannot be stated, so to stay, in logarithmico-exponential terms." This statement of Hardy was basically influenced by the fact that the arithmetic functions occurring in the number theory having often very complicated structure and for which he expected "would give rise to genuinely new modes of increase," so far obey the log-exp laws of increase.

That indicates a possible significance of the results in this treatise. For any logarithmico-exponential function f (or any element of Hardy fields) together with the derivatives, is ultimately continuous and monotonic, of constant sign and  $\lim_{x\to\infty} f(x)$  exists as a finite or infinite one. On the other hand, as we could see above, a slowly varying function may somehow oscillate, even infinitely. As some of the results in the next chapter show, solutions of differential solutions (for instance, of a second-order (half-)linear one) may behave as SV functions. Therefore, the solutions of such a (simple) equation may exhibit a "genuinely new mode of increase." To support our point we emphasize here that no hypothesis of some of the theorems which lead to the above statement about solutions is related to regular variation.

For completeness, note that Rosenlicht in [155] looks for asymptotic formulas for solutions to the linear differential equation y'' = p(t)y with the help of the theory of Hardy fields. It is assumed that  $p(t) \rightarrow \infty$  as  $t \rightarrow \infty$  and p belongs to a Hardy field. In fact, the problem is transferred to the investigation of the associated Riccati equation.

## Chapter 2

# Linear second order differential equations

Although this text is focused primarily on applications of the theory of regular variation to nonlinear differential equations, we start with linear equations. The objective of this chapter is multiple: comparison purposes, reference purposes, but also other ones. Indeed, many of the results which we present here will be later generalized to a nonlinear case. Second, some of the results for linear equations will be utilized in the study of nonlinear equations in various ways. Third, some of the statements may serve as a motivation for an attempt to extend them (in particular, to the half-linear case or to the nearly linear case). Moreover, our survey includes also the results that appeared after Marić's book [105] and we point out new relations among various results, and revise some of them.

Since we want primarily to deal with nonlinear equations, we present the results in this chapter without proofs and give only comments; they include the main ideas in some instances.

Perhaps the first paper where linear differential equations were studied in the framework of regular variation is [130] (Omey, 1981). Note that a connection of regular variation with nonlinear equations was shown much earlier by Avakumović in 1947, see [9] and Section 4.2.

We consider linear second order equation

$$y'' + p(t)y = 0, (2.1)$$

where *p* is a continuous function on  $[a, \infty)$ . Many of the results in this chapter were obtained by Marić, Tomić, and their collaborators. Another important authors in this direction are Howard, Geluk, Grimm, Hall, Omey, and Radašin – this concerns the research up to the year 2000. For more information see the monograph [105] by Marić. Recent results for linear equations presented here are mainly due to Jaroš, Kusano, Marić, and Řehák.

#### 2.1 Karamata solutions; coefficient with sign condition

Instead of (2.1) it is more convenient to consider now

$$y'' - p(t)y = 0. (2.2)$$

In the following theorem, the sign condition on *p* is assumed. Note that under this condition the equation is nonoscillatory (by the Sturm comparison theorem). Without loss of generality, we analyze only positive solutions.

Note that although the statements refer to regular or rapid variation, none of the hypotheses of the theorem requires this notion.

#### **Theorem 2.1** (Marić, Tomić [105, 114]). *Let p be positive.*

(*i*) Equation (2.2) has a fundamental set of solutions  $y_1(t) = L_1(t) \in SV$ ,  $y_2(t) = tL_2(t) \in RV(1)$  if and only if

$$\lim_{t\to\infty}t\int_t^\infty p(s)\,\mathrm{d}s=0.$$

*Moreover,*  $L_1, L_2 \in NSV$  with  $L_2(t) \sim 1/L_1(t)$ . All positive decreasing solutions of (2.1) are in NSV and all positive increasing solutions are in NRV(1).

(ii) Equation (2.2) has a fundamental set of solutions  $y_1(t) = t^{\vartheta_1}L_1(t) \in \mathcal{RV}(\vartheta_1)$ ,  $y_2(t) = t^{\vartheta_2}L_2(t) \in \mathcal{RV}(\vartheta_2)$  if and only if

$$\lim_{t\to\infty}t\int_t^\infty p(s)\,\mathrm{d}s=C,$$

where  $\vartheta_1 < \vartheta_2$  are the roots of the equation  $\vartheta^2 - \vartheta - C = 0$ . Moreover,  $L_1, L_2 \in NSV$  with  $L_2(t) \sim 1/((1 - \vartheta_1)L_1(t))$ . All positive decreasing solutions of (2.1) are in  $NRV(\vartheta_1)$  and all positive increasing solutions are in  $NRV(\vartheta_2)$ .

(iii) Equation (2.2) has a fundamental set of solutions  $y_1 \in \mathcal{RPV}(-\infty)$ ,  $y_2 \in \mathcal{RPV}(\infty)$  if and only if for each  $\lambda > 1$ 

$$\lim_{t \to \infty} t \int_t^{\lambda t} p(s) \, \mathrm{d}s = \infty$$

Moreover, all positive decreasing solutions of (2.1) are in  $RPV(-\infty)$  and all positive increasing solutions are in  $RPV(\infty)$ .

It would be of interest whether the integrals in conditions in all three parts of the theorem were the same. To make a conclusion, the following observations are important (see Grimm, Hall [49]):  $t \int_t^{\infty} p(s) ds \to 0$  as  $t \to \infty$  if and only if, for each  $\lambda > 1$ ,  $t \int_t^{\lambda t} p(s) ds \to 0$  as  $t \to \infty$ . Also,  $t \int_t^{\infty} p(s) ds \to C > 0$  as  $t \to \infty$  if and only if, for each  $\lambda > 1$ ,  $t \int_t^{\lambda t} p(s) ds \to 0$  as  $t \to \infty$ . Also,  $t \int_t^{\infty} p(s) ds \to C > 0$  as  $t \to \infty$  if and only if, for each  $\lambda > 1$ ,  $t \int_t^{\lambda t} p(s) ds \to C(\lambda - 1)/\lambda$  as  $t \to \infty$ . Thus, the interval of integration in all three conditions may be taken to be  $(t, \lambda t)$ . On the other hand, these are easier to verify for the interval  $(t, \infty)$  as it appears in the first of them. However in the third condition the interval  $(t, \lambda t)$  cannot be replaced by  $(t, \infty)$  even when

the integral  $\int_{t}^{\infty} p(s) ds$  converges. For, the condition  $t \int_{t}^{\infty} p(s) ds \to \infty$  as  $t \to \infty$  does not necessarily imply even that  $\lim_{t\to\infty} t \int_{t}^{\lambda t} p(s) ds$  exists for all  $\lambda > 1$ . A counter-example is presented in Grimm, Hall [49].

From the previous theorem we get the following statement. Recall that by Karamata functions we mean SV or RV or RPV functions.

**Corollary 2.1.** All (eventually positive) solutions are Karamata functions if and only if there exists for each  $\lambda > 1$ ,

$$\lim_{t\to\infty} t \int_t^{\lambda t} p(s) \,\mathrm{d}s$$

as a finite or infinite one.

Marić and Tomić in [113] proved a similar result under a more restrictive assumption on p, namely that the limit  $\lim_{t\to\infty} t^2 p(t)$  exists. This is a sufficient condition for (2.1) to be in the Karamata class. More precisely, we have the following

**Corollary 2.2** (Marić, Tomić [113]). If  $\lim_{t\to\infty} t^2 p(t) = A$ , then all decreasing solutions of (2.2) are slowly or rapidly or regularly varying functions with the index  $\vartheta = (1 - \sqrt{1+4A})/2$  in the latter case, according to as A = 0,  $A = \infty$ ,  $A \in (0, \infty)$ , respectively.

The second linearly independent solution can be treated as in the general case. That is, with the use of the reduction of order formula in the SV and RV cases, and directly in the RPV case.

Such a result was in essence first discovered by Omey in [130, Theorem 2.1] (with an additional assumption) and formulated for solutions tending to infinity. This excludes the case A = 0 or  $\vartheta = 0$ , since SV solutions cannot cannot increase. Thus the "trichotomy" character of the result is lost in such formulation.

An important feature of Corollary 2.2 consists in opening various possibilities of a subtle use of classes  $\Pi$ ,  $\mathcal{BSV}$  and  $\Gamma$  in further analysis of solutions under consideration. Thus, by specifying the way in which  $t^2 f(t)$  tends to the finite limit A, Geluk proved in [45] some refinements of Corollary 2.2, see Subsection 2.6.1. In that he makes use of the class  $\Pi$ . Analogous results for the case  $A = \infty$  are obtained in [130, 131] by Omey using classes  $\mathcal{BSV}$  and  $\Gamma$ , see Subsection 2.6.1. Also some of these results were generalized to the half-linear case and improved even in the linear case, see Section 3.6. In Section 5.5 we discuss this type of results for nearly linear equations.

#### 2.2 Karamata solutions; coefficient with no sign condition

Now consider equation (2.1) with no sign condition on p. Such an equation may have oscillatory solutions. However, since we are interested in solutions belonging to Karamata class whose elements are positive, only nonoscillatory solutions have to be considered. Observe that expressions in conditions guaranteeing regular

variation or regular boundedness resemble expressions which appear in the well known Hille-Nehari type (non)oscillation criteria.

In contrast to the case when p(t) < 0 in (2.1) and eventually positive solutions always exist, here we have to establish the existence of nonoscillatory solutions first. This can be achieved either by proving it ab ovo, in fact simultaneously with the regularity like in subsequent Theorems 2.3 and 2.4, or by applying the following auxiliary result. In addition to the methods used by Howard, Marić, Radašin [57, 56], there exists also a different approach, based on the Banach contraction mapping principle, see Jaroš, Kusano [58] — both, principal and nonprincipal solutions can be directly constructed. Results in Section 3.2 can be understood as a half-linear extension of this approach.

The following result very well suits our needs. The proof uses the method of successive approximation and a variant of the Riccati technique can be revealed in it.

Proposition 2.1 (Howard, Marić, Radašin [57, 105]). Put

$$P(t) = \int_{t}^{\infty} p(s) \,\mathrm{d}s. \tag{2.3}$$

If there exists a positive continuous function h with  $h(t) \to 0$  as  $t \to \infty$ , and such that for  $t \ge t_0$ ,  $|P(t)| \le h(t)$ ,  $\int_t^{\infty} h^2(t) dt \le Ah(t)$  with 0 < A < 1/4, then (2.1) is nonoscillatory and there exists a solution of the form

$$y(t) = \exp\left\{\int_{a}^{t} (P(s) - Z(s)) \,\mathrm{d}s\right\}.$$
 (2.4)

*Here Z is a solution of the integral equation* 

$$Z(t) = -\int_{t}^{\infty} (Z(s) - P(s))^{2} ds$$
(2.5)

satisfying Z(t) = O(h(t)) as  $t \to \infty$ .

The previous proposition plays a key role in the proof of the following generalization of Theorem 2.1-(i).

**Theorem 2.2** (Howard, Marić [56, 105]). Equation (2.1) has a fundamental set of solutions  $y_1(t) = L_1(t) \in SV$ ,  $y_2(t) = tL_2(t) \in RV(1)$  if and only if

$$\lim_{t \to \infty} t \int_{t}^{\infty} p(s) \, \mathrm{d}s = 0.$$
(2.6)

*Moreover,*  $L_1, L_2 \in NSV$  *with*  $L_2(t) \sim 1/L_1(t)$  *as*  $t \to \infty$ .

In the next result, Proposition 2.1 cannot be applied, and one has to prove regularity and thus nonoscillation directly.

**Theorem 2.3** (Howard, Marić [56, 105]). Let  $A \in (-\infty, 1/4)$ ,  $A \neq 0$ , and let  $\vartheta_1 < \vartheta_2$  be the roots of the equation  $\vartheta^2 - \vartheta + A = 0$ . Equation (2.1) has a fundamental set of solutions  $y_1(t) = t^{\vartheta_1}L_1(t) \in \mathcal{RV}(\vartheta_1)$ ,  $y_2(t) = t^{\vartheta_2}L_2(t) \in \mathcal{RV}(\vartheta_2)$  if and only if

$$\lim_{t \to \infty} t \int_{t}^{\infty} p(s) \, \mathrm{d}s = A. \tag{2.7}$$

*Moreover,*  $L_1, L_2 \in NSV$  with  $L_2(t) \sim 1/((1 - \vartheta_1)L_1(t))$  as  $t \to \infty$ .

The next theorem deals with the border case when A = 1/4 in (2.7). Note that in general no conclusion concerning oscillation or nonoscillation of (2.1) can be drawn. Also in this result, Proposition 2.1 cannot be applied.

**Theorem 2.4** (Howard, Marić [56, 105]). Let A = 1/4 and suppose that the integral  $\int_{s}^{\infty} \frac{\phi(s)}{s} ds$  converges. Put

$$\phi(t) := t \int_t^\infty p(s) \, \mathrm{d}s - A \text{ and } \psi(t) := \int_t^\infty \frac{\phi(s)}{s} \, \mathrm{d}s$$

Assume

$$\int^{\infty} \frac{\psi(s)}{s} \, \mathrm{d}s < \infty.$$

Then equation (2.1) has a fundamental set of solutions  $y_1(t) = \sqrt{t}L_1(t) \in \mathcal{RV}(1/2)$ ,  $y_2(t) = \sqrt{t} \ln tL_2(t)$  if and only if (2.7) holds. Moreover,  $L_1, L_2 \in \mathcal{NSV}$ , tend to constants and  $L_2(t) \sim 1/L_1(t)$  as  $t \to \infty$ .

In the above theorems the existence of the limit of  $t \int_{t}^{\infty} p(s) ds$  is required. If one relaxes that request to a condition of Hille-Nehari type, then we get regular boundedness. Proposition 2.1 finds application in the proof.

**Theorem 2.5** (Howard, Marić [56, 105]). If, for large t,

$$\left| t \int_t^\infty p(s) \, \mathrm{d}s \right| \le A < \frac{1}{4},$$

then all (eventually positive) solutions of (2.1) are in  $\mathcal{RB}$ .

#### 2.3 Generalization and self-adjoint equation

It is natural to ask whether the results of the previous section can be extended to the more general equation

$$y'' + g(t)y' + h(t)y = 0$$
(2.8)

or to the equation in the self-adjoint form

$$(r(t)y')' + p(t)y = 0.$$
 (2.9)

The next result is an easy consequence of Theorem 2.3, it is based on a suitable transformation.

**Theorem 2.6** (Marić, Tomić [105, 116]). Let  $A, B \in \mathbb{R}$  be such that

$$\frac{B}{2} - \frac{B^2}{4} + A < \frac{1}{4} \tag{2.10}$$

and let  $\alpha_1 < \alpha_2$  be the roots of the equation  $\alpha^2 - \alpha + \gamma = 0$  with  $\gamma = B/2 - B^2/4 + A$ . Further, let *h* be continuous and *g* continuously differentiable on  $[t_0, \infty)$  and such that  $tg(t) \rightarrow B$ as  $t \rightarrow \infty$ . Then there exist two linearly independent regularly varying solutions  $y_1, y_2$  of (2.8) of the form  $y_i(t) = t^{\alpha_i - B/2}L_i(t)$ , i = 1, 2, if and only if

$$\lim_{t\to\infty}t\int_t^\infty h(s)\,\mathrm{d}s=A.$$

*Here*  $L_1, L_2 \in NSV$  *are such that* 

$$L_2(t) \sim \frac{1}{(1 - 2\alpha_1)L_1(t)} \exp\left\{\int_a^t \frac{\varepsilon(s)}{s} \, \mathrm{d}s\right\}$$

with some  $\varepsilon(t) \to 0$  as  $t \to \infty$ .

Condition (2.10) excludes the case  $\alpha_1 = 0$ . Observe that A = B = 0 implies  $\alpha_1 = 0, \alpha_2 = 1$  so that  $y_1 \in NSV$  and  $y_2 \in NRV(1)$ .

From the previous theorem, taking g(t) = r'(t)/r(t), h(t) = p(t)/r(t), we get the result for self-adjoint equation (2.9).

**Corollary 2.3.** Let A, B,  $\alpha_1$ ,  $\alpha_2$  be as in Theorem 2.6. Let p be continuous, r positive and twice continuously differentiable on  $[t_0, \infty)$ , and such that

$$\frac{tr'(t)}{r(t)} \to B \quad as \ t \to \infty. \tag{2.11}$$

Then (2.9) has two  $\mathcal{RV}$  solutions  $y_i$  having the same form as in Theorem 2.6 if and only if

$$\lim_{t \to \infty} t \int_t^\infty \frac{p(s)}{r(s)} \, \mathrm{d}s = A.$$

Using the method of Theorem 2.1, Grimm and Hall [49] obtained similar result but for p(t) < 0 only, requiring of p to increase and tend to a (finite) limit as  $t \rightarrow \infty$ . This is more restrictive than (2.11) but, as a compensation, r may be only once continuously differentiable. They also generalized Theorem 2.6-(iii) using the same method:

**Theorem 2.7** (Grimm and Hall [49]). Let y be a decreasing solution of (2.9), p(t) < 0, and  $r \in NSV$  be nondecreasing with  $r(t) \to 1$  as  $t \to \infty$ . Then  $y \in RPV$  if and only if for each  $\lambda > 1$ ,  $t \int_{t}^{\lambda t} p(s) ds \to \infty$  as  $t \to \infty$ .

Jaroš and Kusano in [60] used a different approach to investigation of  $\mathcal{RV}$ type solutions to (2.9). It utilizes the concept of generalized regular variation, see Subsection 1.3.1, which was introduced for this purpose in [60]. The contraction mapping theorem plays also a role. These results were generalized later to the half-linear case, see Section 3.4. For comparison purposes we give the results for linear equation here. Note however that some differences may occur. For instance, in the linear case, once we know the behavior of one solution, then — in many cases — it is easy to examine behavior of a linearly independent solutions. Further, once we have a fundamental set of  $\mathcal{RB}$  solutions, it is almost immediate (because of linear combinations) to show that all positive solutions are  $\mathcal{RB}$ . On the other hand, the solution space for half-linear equations is not linear and the reduction of order formula cannot be used.

No differentiability condition on r and no sign condition on p is assumed. Essentially, we require p, r to be continuous with r > 0. The two cases are distinguished; this is quite usual when dealing with equations of the form (2.9):

$$\int_{a}^{\infty} \frac{\mathrm{d}s}{r(s)} \,\mathrm{d}s = \infty,\tag{2.12}$$

and

$$\int_{a}^{\infty} \frac{\mathrm{d}s}{r(s)} \,\mathrm{d}s < \infty. \tag{2.13}$$

We denote  $R(t) = \int_{a}^{t} \frac{ds}{r(s)} ds$  if (2.12) holds and  $\tilde{R}(t) = \int_{t}^{\infty} \frac{ds}{r(s)} ds$  if (2.13) holds. In the next theorem we assume that *p* is integrable on  $[a, \infty)$ .

#### Theorem 2.8 (Jaroš, Kusano [60]). Let (2.12) hold.

(*i*) Let  $A \in (-\infty, 1/4)$  and denote by  $\vartheta_1, \vartheta_2, \vartheta_1 < \vartheta_2$ , the real roots of the quadratic equation  $\vartheta^2 - \vartheta + A = 0$ . Equation (2.9) has a fundamental set of solutions  $\{y_1, y_2\}$  such that  $y_i \in NRV_R(\vartheta_i)$ , i = 1, 2, if and only if

$$\lim_{t\to\infty} R(t) \int_t^\infty p(s) \,\mathrm{d}s = A.$$

(ii) Assume

$$\lim_{t\to\infty} R(t) \int_t^\infty p(s) \,\mathrm{d}s = 1/4.$$

Put

$$\phi(t) = R(t) \int_t^\infty p(s) \,\mathrm{d}s - \frac{1}{4}$$

and suppose that

$$\int_{-\infty}^{\infty} \frac{|\phi(s)|}{r(s)R(s)} \, \mathrm{d}s < \infty \quad and \quad \int_{-\infty}^{\infty} \frac{\psi(s)}{r(s)R(s)} \, \mathrm{d}s < \infty,$$

where  $\psi(t) = \int_t^\infty \frac{|\phi(s)|}{r(s)R(s)} ds$ . Then equation (2.9) possesses a fundamental set of solutions  $\{y_1, y_2\}$  such that  $y_i \in NRV_R(1/2)$ , i = 1, 2, and  $y_1(t) = \sqrt{R(t)}L_1(t)$ ,  $y_2(t) = \sqrt{R(t)}L_1(t)$ .

 $\sqrt{R(t)}L_2(t) \ln R(t)$ , where  $L_1, L_2 \in NSV_R$  and  $\lim_{t\to\infty} L_i(t) = C_i \in (0, \infty)$ , i = 1, 2, with  $C_1C_2 = 1$ . (iii) If

$$-\frac{1}{4} < \liminf_{t \to \infty} R(t) \int_t^\infty p(s) \, \mathrm{d}s \le \limsup_{t \to \infty} R(t) \int_t^\infty p(s) \, \mathrm{d}s < \frac{1}{4},$$

then (2.9) is nonoscillatory and all its eventually positive solutions are in  $\mathcal{RB}_R$ .

The next theorem is a complement of the previous one, in the sense of condition (2.13). We assume that  $\tilde{R}^2(t)p(t)$  is integrable on  $[a, \infty)$ .

Theorem 2.9 (Jaroš, Kusano [60]). Let (2.13) hold.

(*i*) Let  $B \in (-\infty, 1/4)$  and denote by  $\mu_1, \mu_2, \mu_1 < \mu_2$ , the real roots of the quadratic equation  $\mu^2 - \mu + B = 0$ . Equation (2.9) has a fundamental set of solutions  $\{y_1, y_2\}$  such that  $y_i \in NRV_{1/\tilde{R}}(\mu_i)$ , i = 1, 2, if and only if

$$\lim_{t\to\infty}\frac{1}{\tilde{R}(t)}\int_t^\infty \tilde{R}^2(s)p(s)\,\mathrm{d}s=B.$$

(ii) Assume

$$\lim_{t \to \infty} \frac{1}{\tilde{R}(t)} \int_t^\infty \tilde{R}^2(s) p(s) \, \mathrm{d}s = 1/4.$$

Put

$$\phi(t) = \frac{1}{\tilde{R}(t)} \int_t^\infty \tilde{R}^2(s) p(s) \, \mathrm{d}s - \frac{1}{4}$$

and suppose that

$$\int^{\infty} \frac{|\phi(s)|}{r(s)\tilde{R}(s)} \, \mathrm{d}s < \infty \ and \ \int^{\infty} \frac{\psi(s)}{r(s)\tilde{R}(s)} \, \mathrm{d}s < \infty,$$

where  $\psi(t) = \int_t^\infty \frac{|\phi(s)|}{r(s)\tilde{R}(s)} ds$ . Then equation (2.9) possesses a fundamental set of solutions  $\{y_1, y_2\}$  such that  $y_i \in \mathcal{NRV}_{1/\tilde{R}}(-1/2), i = 1, 2, and$ 

$$y_1(t) = \sqrt{\tilde{R}(t)}L_1(t), \ y_2(t) = \sqrt{\tilde{R}(t)}L_2(t)\ln\frac{1}{\tilde{R}(t)},$$

*where*  $L_1, L_2 \in NSV_{1/\tilde{R}}$  *and*  $\lim_{t\to\infty} L_i(t) = C_i \in (0, \infty)$ , i = 1, 2, with  $C_1C_2 = 1$ . (*iii*) If

$$-\frac{1}{4} < \liminf_{t \to \infty} \frac{1}{\tilde{R}(t)} \int_{t}^{\infty} \tilde{R}^{2}(s) p(s) \, \mathrm{d}s \leq \limsup_{t \to \infty} \frac{1}{\tilde{R}(t)} \int_{t}^{\infty} \tilde{R}^{2}(s) p(s) \, \mathrm{d}s < \frac{1}{4},$$

*then* (2.9) *is nonoscillatory and all its eventually positive solutions are in*  $\mathcal{RB}_{1/\tilde{R}}$ *.* 

### 2.4 Linear differential equations having regularly varying solutions

Kusano and Marić in [94] deals with the question whether for any distinct real constants  $\vartheta_1$  and  $\vartheta_2$  there exists a differential equation of the form (2.9) which possesses a pair of solutions  $y_i \in \mathcal{RV}(\vartheta_i)$ , i = 1, 2. The problem was solved for more general, half-linear, equation. Here we formulate the linear version. The general statement along with the proof is presented in Section 3.7. The function  $\omega$  which appears in the theorem is assumed to satisfy conditions from the definition of regularly varying functions with respect to  $\omega$ , see Definition 1.10.

**Theorem 2.10** (Kusano and Marić [94]). Let  $\vartheta_1$  and  $\vartheta_2$  be any given real constants such that  $|\vartheta_1| \neq |\vartheta_2|$ .

(i) Suppose that r satisfies  $1/r(t) \sim K\omega^{\vartheta_1+\vartheta_2-1}(t)\omega'(t)$  as  $t \to \infty$  for some positive constant K. Let  $\vartheta_1 + \vartheta_2 > 0$  and p be conditionally integrable on  $[a, \infty)$ . Then equation (2.9) possesses a fundamental set of solutions  $y_i \in NRV_{\omega}(\vartheta_i)$ , i = 1, 2, if and only if

$$\lim_{t \to \infty} K \omega^{\vartheta_1 + \vartheta_2}(t) \int_t^\infty p(s) \, \mathrm{d}s = \frac{\vartheta_1 \vartheta_2}{\vartheta_1 + \vartheta_2}$$

(*ii*) Suppose that  $1/r(t) = K\omega^{\vartheta_1+\vartheta_2-1}(t)\omega'(t)$  for some positive constant K. Let  $\vartheta_1 + \vartheta_2 < 0$  and  $\omega^{2(\vartheta_1+\vartheta_2)}p$  be conditionally integrable on  $[a, \infty)$ . Then equation (2.9) possesses a fundamental set of solutions  $y_i \in NRV_{\omega}(\vartheta_i)$ , i = 1, 2, if and only if

$$\lim_{t \to \infty} K \omega^{-(\vartheta_1 + \vartheta_2)}(t) \int_t^\infty \omega^{2(\vartheta_1 + \vartheta_2)}(s) p(s) \, \mathrm{d}s = -\frac{\vartheta_1 \vartheta_2}{\vartheta_1 + \vartheta_2}.$$

### 2.5 Regularly varying solutions of Friedmann equations

Mijajlović, Pejović, Šegan, and Damljanović in [125] applied the theory of  $\mathcal{RV}$  functions to the asymptotical analysis at infinity of solutions of Friedmann cosmological equations. Their analysis is strongly based on Theorem 2.2 and Theorem 2.3.

Let us consider Friedmann equations

$$\left(\frac{u'}{u}\right)^2 = \frac{8\pi G}{3}\rho - \frac{kc^2}{u^2} \quad \text{(Friedmann equation)} \tag{2.14}$$

$$\frac{u''}{u} = -\frac{4\pi G}{3} \left( \rho + \frac{3p}{c^2} \right) \quad \text{(acceleration equation)} \tag{2.15}$$

that describe the evolution of the expansion scale factor u(t) of the universe. Here, p = p(t) is the energy pressure in the universe,  $\rho = \rho(t)$  is the density of matter in the universe, k is the space curvature, G is the gravitational constant and c is the speed of light. The variable t represents the cosmic time.

In [125], it is established a necessary and sufficient condition for solutions that satisfy the generalized power law expressed as  $u(t) = t^{\vartheta}L(t)$ , where  $L \in SV$ . The analysis is strongly based on Theorem 2.2 and Theorem 2.3. For this reason it is introduced a new parameter  $\mu(t) = q(t)(H(t)t)^2$  where q(t) is the deceleration parameter and H(t) is the Hubble parameter. It is proved that the Friedmann equations (2.14) and (2.15) have an asymptotical solution u(t) that satisfies the generalized power law if and only if the integral limit

$$\Upsilon = \lim_{t \to \infty} t \int_{t}^{\infty} \frac{\mu(s)}{s^2} \,\mathrm{d}s \tag{2.16}$$

exists and  $\Upsilon < 1/4$ . It is proved that the values of the constant  $\Upsilon$  completely determines the asymptotical behavior of all cosmological parameters u(t), H(t), q(t), p(t), and  $\rho(t)$ . It appears that this approach covers all results on cosmological parameters for the Standard model of the universe, as presented for example in [103] or in [139]. The crucial role in this analysis is played by the linear functional related to (2.16)

$$\mathbf{M}(f) = \lim_{t \to \infty} t \int_{t}^{\infty} \frac{f(s)}{s^2} \, \mathrm{d}s$$

**M** is defined on the class of real functions that satisfy (2.16) for some  $\Upsilon$ . It is proved that  $f \in \ker \mathbf{M}$  if and only if there are real functions  $\varphi$  and  $\eta$  such that

$$f(t) = t\varphi'(t) + \eta(t), \quad \varphi(t), \eta(t) \to 0 \text{ as } t \to \infty.$$

This representation of  $f \in \ker \mathbf{M}$  yields the asymptotical representations of the mentioned cosmological parameters, even assuming that the Einstein's cosmological constant  $\Lambda$  is non-zero. Detailed proofs and physical interpretations of these results can be found in [125].

#### 2.6 More precise information about asymptotic behavior

#### 2.6.1 De Haan type solutions

To some extent, the results here can be understood as ramifications and refinements of observations related to Corollary 2.2.

We start with the statements established for (2.2) by Geluk in [45], see also Marić [105], which essentially concern a description of behavior of SV solutions. Under the conditions posed on p (in addition to  $t^2p(t) \rightarrow 0$  as  $t \rightarrow \infty$ , we have a second order condition), the Hartman result [53, Chapter XI, Ex. 9.9b] can be refined. Note that — for other comparison purposes — the Hartman result is recalled below in (2.33).

**Theorem 2.11** (Geluk [45]). Assume  $t^2p(t) \to 0$  as  $t \to \infty$ . Let y be an eventually positive decreasing solution of (2.2). If  $p \in \mathcal{RV}(-2)$ , then  $-y \in \Pi(-ty'(t))$ . Moreover, if

 $\int_{a}^{\infty} sp(s) \, \mathrm{d}s = \infty, \, then$ 

$$y(t) = \exp\left\{-\int_{a}^{t} sp(s)(1+\varepsilon(s)) \,\mathrm{d}s\right\},\tag{2.17}$$

and if  $\int_{a}^{\infty} sp(s) \, ds < \infty$ , then

$$y(t) = y(\infty) \exp\left\{\int_{t}^{\infty} sp(s)(1+\varepsilon(s)) \,\mathrm{d}s\right\},\tag{2.18}$$

where  $\varepsilon(t) \to 0$  as  $t \to \infty$ .

Note that if  $\int_{a}^{\infty} sp(s) ds$  converges in Theorem 2.11, then solutions under consideration tend to a positive constant, whereas if it diverges the representation does not imply in general that

$$y(t) \sim y(a) \exp\left\{-\int_{a}^{t} sp(s) \,\mathrm{d}s\right\}$$

as  $t \to \infty$ , see e.g. (2.19). A half-linear extension of the above result and related observations are presented in Section 3.6, while in Section 2.7 we offer a "4-th order extension." In Section 5.5 we discuss this type of formulas in connection with the so called nearly linear equations.

The following result is an extension of the previous theorem to the case of  $\mathcal{RV}$  solutions.

**Theorem 2.12** (Geluk [45]). Assume  $L(t) := t^2 p(t) - A \to 0$  as  $t \to \infty$ , where A > 0 is a constant. Let y be an eventually positive decreasing solution of (2.2). If  $L \in SV$ , then  $-t^{\vartheta}y(t) \in \Pi(-t(t^{\vartheta}y(t))')$  and

$$y(t) = t^{-\vartheta} \exp\left\{-\frac{1}{2\vartheta + 1} \int_a^t \frac{L(s)}{s} (1 + \varepsilon(s)) \,\mathrm{d}s\right\},\,$$

where  $\vartheta > 0$  and A are related by  $\vartheta(\vartheta + 1) = A$ , and  $\varepsilon(t) \to 0$  as  $t \to \infty$ .

A remark analogous to that after Theorem 2.11 holds also here. Further, see the remark after Theorem 2.17 which is concerned with certain generalization of the above theorem.

The class  $\Pi R_2$  opens further possibilities in obtaining more precise information about considered solutions *y* of (2.2). The following statement is a refinement of Theorem 2.11.

**Theorem 2.13** (Geluk [45]). Assume  $t^2p(t) \to 0$  as  $t \to \infty$ . Let y be an eventually positive decreasing solution of (2.2). If  $-t^2p(t) \in \Pi(h)$ , then  $-y \in \Pi \mathbb{R}_2(v, w)$  with v(t) = -ty'(t) and

$$w(t) \sim ty'(t) + t^2y''(t) \sim (h(t) + t^4p^2(t))y(t)$$

as  $t \to \infty$ . The following three cases are possible: (i) If  $\int_{a}^{\infty} s^{3}p^{2}(s) ds = \infty$ , then

$$y(t) = \exp\left\{-\int_{a}^{t} sp(s) \,\mathrm{d}s + \int_{a}^{t} (s^{3}p^{2}(s) + h(s)/s)(1 + \varepsilon(s)) \,\mathrm{d}s\right\},\tag{2.19}$$

where  $\varepsilon(t) \to 0$  as  $t \to \infty$ . (ii) If  $\int_a^{\infty} s^3 p^2(s) ds < \infty$  and  $\int_a^{\infty} sp(s) ds = \infty$ , then

$$y(t) = \exp\left\{-\int_{a}^{t} sp(s) \,\mathrm{d}s + C - \int_{t}^{\infty} (s^{3}p^{2}(s) + h(s)/s)(1 + \varepsilon(s)) \,\mathrm{d}s\right\},\tag{2.20}$$

where  $C \in \mathbb{R}$  is a constant and  $\varepsilon(t) \to 0$  as  $t \to \infty$ . (*iii*) If  $\int_{a}^{\infty} sp(s) \, ds < \infty$ , then

$$y(t) = y(\infty) \exp\left\{\int_t^\infty sp(s) \,\mathrm{d}s - \int_t^\infty (s^3 p^2(s) + h(s)/s)(1+\varepsilon(s)) \,\mathrm{d}s\right\}$$
(2.21)

where  $\varepsilon(t) \to 0$  as  $t \to \infty$ .

Also in this setting we have an extension to the case of regularly varying solutions.

**Theorem 2.14** (Geluk [45]). Assume  $L(t) := t^2 p(t) - A \rightarrow 0$  as  $t \rightarrow \infty$ , where A > 0 is a constant. Let y be an eventually positive decreasing solution of (2.2). If  $L \in \Pi(h)$ , then  $-t^{\vartheta}y(t) \in \prod R_2$  and

$$y(t) = t^{-\vartheta} \exp\left\{-\frac{1}{2\vartheta + 1} \int_a^t \frac{L(s)}{s} ds + \frac{1}{(2\vartheta + 1)^2} \int_a^t \frac{h(s) + (2\vartheta + 1)^{-1}L^2(s)}{s} (1 + \varepsilon(s)) ds\right\},$$

where  $\vartheta > 0$  and A are related by  $\vartheta(\vartheta + 1) = A$  and  $\varepsilon(t) \to 0$  as  $t \to \infty$ .

Under the conditions of Theorem 2.11, the linearly independent solution x(t) = $y(t) \int_{a}^{t} \frac{1}{y^{2}(s)} ds$  satisfies  $x(t)/t \in \Pi$ . A representation can also be given. For example, under the additional condition  $\int_{a}^{\infty} sp(s) \, ds = \infty$ ,

$$x(t) = t(1 - 2t^2 p(t)(1 + o(1))) \exp\left\{\int_a^t (1 + o(1)) sp(s) \, \mathrm{d}s\right\}.$$

A similar representation can be given in other cases.

Assuming  $p \in \mathcal{RV}(-2)$  and  $t^2 p(t) \to 0$  as  $t \to \infty$  it follows from one of (2.17) and (2.18) that  $-y \in \Pi(-ty'(t))$ . Similarly  $-t^2p(t) \in \Pi(h)$ ,  $t^2p(t) \to 0$  as  $t \to \infty$ , and one of (2.19), (2.20), (2.21) imply  $-y \in \prod R_2(v, w)$ . In the first case of Theorem 2.11 relation (2.17) implies

$$y(t) = y(a) \exp\left\{-(1+o(1))\int_{a}^{t} sp(s) \,\mathrm{d}s\right\}$$

but the last representation does not imply  $-y \in \Pi$  any more. A similar remark applies to the other expansions. In Theorem 2.13 it follows from [11, Theorem 3] that  $t^2p(t) \sim \int_t^{\infty} h(s)/s \, ds$  as  $t \to \infty$ , hence in (2.20) and in (2.21)

$$\int_t^\infty \frac{h(s)(1+\varepsilon(s))}{s} \, \mathrm{d}s = (1+o(1))t^2 p(t)$$

as  $t \to \infty$ .

We finish this section with refinements of the results for  $\mathcal{RPV}(\infty)$  solutions of (2.2) using the classes  $\mathcal{BSV}$  and  $\Gamma$ . The core of the following result is due by Omey and appeared in [47, 130, 131, 133]. Later it was extended to the half-linear case and additional observations were made (some of the being new also in the linear case), see Section 3.6. Denote

$$\mathbb{M}^+ = \{y : y \text{ is a solution of } (2.2), y(t) > 0, y'(t) > 0 \text{ for large } t\}$$

and

$$\mathbb{M}^+_{\infty} = \{ y \in \mathbb{M}^+ : \lim_{t \to \infty} y(t) = \infty \}.$$

**Theorem 2.15** (Omey [47, 130, 131, 133]). *If*  $1/\sqrt{p} \in \mathcal{BSV}$ , then  $\emptyset \neq \mathbb{M}^+ = \mathbb{M}_{\infty}^+ \subseteq \Gamma(1/\sqrt{p})$ .

A half-linear extension of the above theorem is presented in Subsection 3.6.1

Note that, under the assumptions of the theorem, for a solution  $y \in \mathbb{M}^+$  we have

$$\lim_{t \to \infty} \frac{y'(t)}{\sqrt{p(t)}y(t)} = 1.$$

If  $p \in C^1$ , then the assumption from the previous theorem — in view of the properties of  $\mathcal{BSV}$  functions — yields  $(1/\sqrt{p(t)})' = -p'(t)p^{-3/2}(t)/2 \rightarrow A$ , where A = 0, as  $t \rightarrow \infty$ , cf. Hartman, Wintner [54]. If A > 0, then we obtain regular variation of the solution; see Omey [130]. See also the discussion around equation (2.39).

Let *y* be a solution as in the previous theorem and consider the linearly independent solution

$$x(t) = C \int_t^\infty \frac{1}{y^2(s)} \,\mathrm{d}s,$$

C > 0. Note that  $\lim_{t\to\infty} x'(t) = 0$  and x is nonprincipal solution while y is principal solution. From (1.25) we get  $x \in \Gamma_-(1/\sqrt{p})$ . Moreover,  $x'(t)/(\sqrt{p(t)}x(t)) \to -1$  as  $t \to \infty$ .

In the following considerations (Omey [130, 133]) we get a second-order result by taking a closer look at H(t) := y(t)/h(t) - y'(t), where  $h \ge 0$  is defined by  $h^2(t)p(t) = 1$ . In order to obtain a rate of convergence result we assume that h'(t) < 0 for large *t* and that

$$-h' \in SC(h).$$

Introduce the functions *u* and *G* by

$$u(t) = \exp\left\{\int_{a}^{t} \frac{1}{h(s)} ds\right\}$$
 and  $G(t) = u(t)H(t)$ .

Since  $h'(t) \rightarrow 0$ , we have  $h \in SN$  and  $u \in \Gamma(h)$ . Also we have

$$G'(t) = -\frac{u(t)h'(t)}{h^2(t)}y(t).$$

Note that G'(t) > 0 for large *t* and  $G'(t) = u(t)(\sqrt{p})'y(t)$ . [If  $(\sqrt{p})'' > 0$ , we have that G''(t) > 0 and G' is nondecreasing.] It is not difficult to see that

$$\frac{G'(t+\lambda h(t))}{G'(t)} \to \exp(2\lambda)$$

so that  $G' \in \Gamma(h/2)$ . Now this implies that  $G(t) \sim G'(t)h(t)/2$ . We conclude that

$$G(t) \sim -\frac{u(t)h'(t)y(t)}{2h(t)}$$

and then also that  $H(t) \sim -h'(t)y(t)/(2h(t))$ . It follows that

$$\frac{h(t)y'(t)}{y(t)}-1\sim\frac{h'(t)}{2},$$

or

$$\frac{y'(t)}{y(t)} = \frac{1}{h(t)} + (1 + \varepsilon(t))\frac{h'(t)}{2h(t)},$$

where  $\varepsilon(t) \rightarrow 0$ . Integration between *t* and  $\lambda h(t)$  gives

$$\ln \frac{y(t+\lambda h(t))}{y(t)} - \lambda = I_1(t) + I_2(t),$$

where

$$I_1(t) = \int_0^{\Lambda} \left( \frac{h(t)}{h(t+sh(t))} - 1 \right) ds,$$
  

$$I_2(t) = \int_0^t (1 + \varepsilon(t+sh(t))) \frac{h'(t+sh(t))h(t)}{h(t+sh(t))} ds.$$

We get (see (1.28))

$$I_1(t) \sim -\frac{h'(t)\lambda^2}{2}, \quad I_2(t) \sim h'(t)\lambda.$$

It follows that

$$\ln\left(\frac{y(t+\lambda h(t))}{y(t)}\exp(-\lambda)\right) \sim h'(t)\left(\lambda-\frac{\lambda^2}{2}\right).$$

Since  $h'(t) \rightarrow 0$ , we obtain

$$\frac{y(t+\lambda h(t))}{y(t)}\exp(-\lambda) - 1 \sim h'(t)\left(\lambda - \frac{\lambda^2}{2}\right)$$

#### 2.6.2 A different approach

Similarly as in the first part of the previous section also here we obtain more precise information about (slowly and regularly varying) solutions of (2.1). While previously the considerations were based on  $\Pi$ -variation here we offer an alterative procedure. Besides one method of successive approximations used in the proof of an auxiliary proposition, the properties of regularly varying solutions are the only essential tool.

To determine asymptotic behavior of solutions under consideration of equation (2.1), due to representations in Theorem 2.2 and Theorem 2.3, since  $L_2(t) \sim 1/((1 - \vartheta_1)L_1(t))$ , one has to do it for  $L_1$  only. In principle the same procedure applies for both slowly and regularly varying solutions. However, we consider separately two cases: A = 0 and  $A \neq 0$ ,  $-\infty < A < 1/4$  (see (2.7)). The reason is that in the former case one obtains a more general result than in the latter one.

We begin with an important auxiliary result which utilizes some of the ideas of Proposition 2.1 on p. 42.

**Proposition 2.2** (Howard, Marić, Radaśin, [57, 105]). Let the functions *P* and *h* be defined as Proposition 2.1. If there exists a continuous decreasing function q(t) such that

$$\int_{t}^{\infty} h^{2}(s) \,\mathrm{d}s \le q(t)h(t) \tag{2.22}$$

and

$$0 < q(t) \le C < 1/4, \tag{2.23}$$

and if for some  $n \in \mathbb{N}$ 

$$\int_{0}^{\infty} q^{n}(t)h(t) \,\mathrm{d}t < \infty, \tag{2.24}$$

then the solution y of (2.1) given by (2.4) satisfies

$$y(t) \sim B \exp\left\{\int_a^t (P(s) - Z_{n-1}(s)) \,\mathrm{d}s\right\}$$

as  $t \to \infty$ , where  $Z_0(t) := 0$ ,  $Z_n(t) := -\int_t^\infty (P(s) - Z_{n-1}(s))^2 ds$ ,  $n \in \mathbb{N}$ ,  $P(t) := \int_t^\infty p(s) ds$ , B is some positive constant, but one may take it to be 1.

Note that in general solutions y in the previous proposition need not to be slowly varying. But as established in Theorem 2.2, these are such under the condition (2.6). This, in addition, implies the existence of a function q(t) satisfying (2.22) and (2.23) by taking h(t) = q(t)/t. Consequently, we get the following statement.

**Theorem 2.16** (Howard, Marić, Radaśin, [57, 105]). Let (2.6) be fulfilled. If condition for some  $n \in \mathbb{N}$  (2.24), being here of the form  $\int_{-\infty}^{\infty} c^{n+1}(t)/t \, dt < \infty$  holds, then two linearly independent solutions of (2.1)  $y_1(t) = L_1(t)$ ,  $y_2(t) = tL_2(t)$ , where  $L_1, L_2 \in NSV$  with  $L_2(t) \sim 1/L_1(t)$ , possess the following asymptotic representation for  $t \to \infty$ 

$$y_{1}(t) \sim \exp\left\{\int_{a}^{t} (P(s) - Z_{n-1}(s)) \, \mathrm{d}s\right\},$$
  

$$y_{2}(t) \sim t \exp\left\{-\int_{a}^{t} (P(s) - Z_{n-1}(s)) \, \mathrm{d}s\right\},$$
(2.25)

and  $ty'_1(t)/y_1(t) \to 0$ ,  $ty'_2(t)/y_2(t) \to 1$ .

By Theorem 2.3, condition (2.7) which can be written in the form

$$\phi(t) := t \int_t^\infty p(s) \, \mathrm{d}s - A \to 0 \quad \text{as } t \to \infty, \ A \neq 0,$$

implies the existence of two linearly independent NRV solutions  $y_i(t) = t^{\vartheta_i}L_i(t)$ ,  $i = 1, 2, \vartheta_1 < \vartheta_2$  being the roots of  $\vartheta^2 - \vartheta + A = 0$ . Next we describe behavior of these solutions in a more precise way. The result was firstly proved in Geluk, Marić, Tomić [48] for the case p(t) < 0. The general case was treated in Marić [105]. Denote

$$\varrho(t) = \exp\left\{\int_1^t \frac{2(\vartheta_1 + \phi(s))}{s} \,\mathrm{d}s\right\}.$$

**Theorem 2.17** (Marić [105]). Let  $A \in (-\infty, 1/4)$ ,  $A \neq 0$ . If

$$\int^{\infty} \frac{\phi^2(s)}{s} \, \mathrm{d}s < \infty, \tag{2.26}$$

then (2.1) possesses linearly independent solutions  $y_1, y_2$  satisfying

$$y_1(t) \sim t^{\vartheta_1} \exp\left\{\int_a^t \left(\frac{\phi(s)}{s} + 2\vartheta_1 \int_s^\infty \frac{\varrho(\tau)}{\varrho(s)} \cdot \frac{\phi(\tau)}{\tau^2} \,\mathrm{d}\tau\right) \,\mathrm{d}s\right\},\tag{2.27}$$

$$y_2(t) \sim \frac{1}{(1-2\vartheta_1)L_1(t)}$$

and  $y'_i(t) \sim \vartheta_i y_i(t)/t$ , i = 1, 2, as  $t \to \infty$ .

In Geluk, Marić, Tomić [48, Theorem 1.2] the condition of integrability (2.26) is replaced  $\phi \in SV$ . That result can also be seen a generalization of Theorem 2.11 and Theorem 2.12.

Note that in the asymptotic representation formula (2.27) two summands occurring in the exponential function need not in general to be of the same order or growth and consequently none of these can be disposed of. This is exemplified by the result which follows where also by strengthening the conditions, the behavior of solutions becomes much more legible.

**Theorem 2.18** (Marić [105]). If in addition to (2.26) one has

$$\int_{a}^{\infty} \frac{\phi(t)}{\varrho(t)} \int_{t}^{\infty} \frac{\varrho(\tau)|\phi(\tau)|}{\tau^{2}} \, \mathrm{d}\tau \, \mathrm{d}t < \infty, \tag{2.28}$$

then

$$y_1(t) \sim t^{\vartheta} \exp\left\{\frac{1}{1-2\vartheta_1} \int_a^t \frac{\phi(s)}{s} \,\mathrm{d}s\right\}$$
(2.29)

as  $t \to \infty$ .

If  $\phi(t)$  tends to zero sufficiently fast, e.g. like some power  $t^{-\delta}$ ,  $\delta > 0$  (or faster), then  $\int^t \phi(s)/s \, ds$  converges and (2.27) becomes  $y_1(t) \sim Bt^{\vartheta_1}$ . From that point of view, the following simple (and easy to apply) observation which uses properties of  $\mathcal{RV}$  functions is of interest. If  $\phi(t) = O(\Phi_0(t))$  as  $t \to \infty$ , where  $\phi_0 \in \mathcal{SV}$  satisfies (2.26), then the behavior determined by (2.29) holds.

In the special case of p(t) < 0, the root  $\vartheta_1$  is always negative. Therefore, by the use of inequality  $\varrho(t)/\varrho(s) \le (t/s)^{2\vartheta_1+2\varepsilon}$ , which holds for any  $\varepsilon > 0$  and  $t > s, s \ge s_0(\varepsilon)$ , condition (2.28) can be replaced by the simpler one

$$\int_{a}^{\infty} |\phi(t)| \int_{t}^{\infty} \frac{|\phi(s)|}{s^{2}} \, \mathrm{d}s \, \mathrm{d}t < \infty$$

Theorem 2.18 (simplified by this remark) then extends the following result of Mařík and Ráb [117]. If for some c > 0

$$\int_a^\infty t|p(t)-c/t^2|\,\mathrm{d}t<\infty,$$

then (2.2) has a pair of solutions  $y_1$ ,  $y_2$  such that

$$y_i(t) \sim t^{-\vartheta_i} \text{ and } y'(t) \sim \vartheta_i t^{\vartheta_i - 1},$$
 (2.30)

where  $\vartheta_i$  are the roots of  $\vartheta^2 - \vartheta - c = 0$ . Observe that if we put  $\psi(t) = t(p(t) - ct^{-2})$ , then the preceding condition implies

$$\int_{a}^{\infty} \int_{s}^{\infty} \frac{|\psi(\tau)|}{\tau} \, \mathrm{d}\tau \, \mathrm{d}s < \infty \text{ and } \int_{t}^{\infty} |\psi(s)| \, \mathrm{d}s \to 0$$

as  $t \to \infty$ . Consequently for  $t \to \infty$ ,

$$|\phi(t)| = \left| t \int_t^\infty p(s) \, \mathrm{d}s - c \right| \le t \int_t^\infty \frac{|\psi(s)|}{s} \, \mathrm{d}s \le \int_t^\infty |\psi(s)| \, \mathrm{d}s \to 0$$

and also  $\phi(t)/t = \int_t^{\infty} \phi(s)/s \, ds$  is absolutely integrable over  $(a, \infty)$ . Therefore all conditions of Theorem 2.18 are fulfilled and the result follows. This illustrates again kind of generalization we have: The slowly varying function  $L(t) = \exp\left\{\frac{1}{1-2\vartheta_1}\int_a^t \phi(s)/s \, ds\right\}$  in Theorem 2.18 multiplying by  $t^{\vartheta_1}$  is in the result by Mařík and Ráb the special one which tends to a constant as  $t \to \infty$ .

In the rest of this section we assume p(t) < 0 and again we study slowly varying solutions. Of course, in this special case asymptotic representation of solutions remains the same as before (i.e., (2.25)). However here if  $y_1$  is a SV solution we know according to the results of Section 2.1 that it is normalized and decreasing. Further, by the Representation Theorem, it is of the form  $y_1(t) = \exp\left\{-\int_a^t \eta(s)/s \, ds\right\}$ , where  $\eta(t)$  is positive and tends to zero as  $t \to \infty$ . It also satisfies the integral (Riccati type) equation

$$\eta(t) = t \int_t^\infty p(s) \, \mathrm{d}s - t \int_t^\infty \left(\frac{\eta(s)}{s}\right)^2 \mathrm{d}s.$$

Furthermore by Theorem 2.1, condition (2.6) is fulfilled. Therefore by putting  $Z(t) - P(t) = \eta(t)/t$  and  $Z_n(t) - P(t) = \eta_n(t)/t$ ,  $n \in \mathbb{N}$ , where *Z* is a solution of (2.5) and  $P(t) = \int_t^{\infty} p(s) \, ds$ , solution  $y_1$  has the form (2.4) and if a condition of the type (2.24) holds, the asymptotic representation for  $y_1$  will follow. Since condition (2.6) is necessary and sufficient for the existence of SV solutions, one can reverse the above argument: Assume first (2.6) holds and then continue as above. It is worthwhile mentioning that not only for the existence but also for the asymptotic representation of SV solutions do the rest. Condition (2.6) is the sole one needed. Properties of SV solutions do the rest. Condition (2.24) is only a technical one. We emphasize here also that in contrast to the general case (no sign condition on p) where all conditions that appear, (2.24) in particular, refer to functions h and q which have to be constructed, whereas here those conditions can be expressed in terms of the known function P(t) as defined by (2.3).

Observe that condition (2.6) and the negativity of *p* imply that for any  $\varepsilon > 0$  there exists  $t_0$  such that for  $t \ge t_0$ , one has

$$\int_{t}^{\infty} P^{2}(s) \, \mathrm{d}s \le \varepsilon |P(t)|. \tag{2.31}$$

For, by a partial integration

$$\int_{t}^{\infty} P^{2}(s) \, \mathrm{d}s = -tP^{2}(t) + 2 \int_{t}^{\infty} s(-P(s))(-p(s)) \, \mathrm{d}s$$

and (2.31) follows.

Put

$$U_1(t) := \int_t^\infty P^2(s) \, \mathrm{d}s, \quad U_{n+1}(t) := 2 \int_t^\infty -P(s) U_n(s) \, \mathrm{d}s, \ n \in \mathbb{N}.$$

**Theorem 2.19** (Marić, Tomić [105, 115]). *If for some*  $n \in \mathbb{N}$ 

$$\int^{\infty} U_n(s) \,\mathrm{d}s < \infty, \tag{2.32}$$

then for any SV solution  $y_1$  of (2.1) with p(t) < 0 and for the linearly independent solution the asymptotic representation (2.25) holds.

Condition (2.32) might be cumbersome to verify. It can be replaced by a simpler in general a cruder one, as it is done in corollary below.

Estimate (2.31) implies the existence of a positive continuous function q(t) decreasing to zero and such that  $U_1(t) = \int_t^{\infty} P^2(s) ds \le q(t)|P(t)|/2$ . Then inequality  $U_{n+1}(t) \le \varepsilon U_n(t)$  with q(t) replacing  $\varepsilon$ , leads to  $U_n(t) \le q^n(t)|P(t)|$ . Hence the previous theorem implies the following statement.

**Corollary 2.4.** If for some  $n \in \mathbb{N}$ ,  $\int_{0}^{\infty} q^{n}(t) |P(t)| dt < \infty$ , the asymptotic representation (2.25) holds.

For comparison purposes we now recall the result by Hartman and Wintner ([53, Chapter XI, Ex. 9.9-b]); note that we utilized it already earlier, see the text before Theorem 2.11. Let p(t) be a continuous complex function defined for  $t \ge a$ . If for some  $\alpha \in [1, 2] \int_{0}^{\infty} t^{2\alpha-1} |p^{\alpha}(t)| dt < \infty$ , then equation (2.1) has a pair of solutions such that

$$y_1(t) \sim \exp\left\{\int_a^t sp(s) \,\mathrm{d}s\right\} \text{ and } ty'_1(t)/y_1(t) \to 0,$$
  
 $y_2(t) \sim t \exp\left\{-\int_a^t sp(s) \,\mathrm{d}s\right\} \text{ and } ty'_2(t)/y_2(t) \to 1$  (2.33)

as  $t \to \infty$ .

The results for the derivatives show that when  $y_1$ ,  $y_2$  are real, they respectively are NSV and NRV(1).

E.g. for p(t) < 0 Theorem 2.19 gives the above behavior for  $y_1$  and  $y_2$  with n = 1. For,  $y_1(t) \sim \exp\left\{\int_a^t p(s) ds\right\}$ . By integrating partially and then using (2.6) one obtain the above behavior for  $y_1$ , similarly for  $y_2$  and for derivatives.

Instead of Hartman-Wintner conditions we have by (2.32) for n = 1, again after a partial integration,  $\int_a^t sP^2(s) ds < \infty$ . These two conditions are not comparable in general. However, for the rather general example  $p(t) = \varepsilon(t)/t^2$  where  $\varepsilon(t)$  is almost decreasing, their condition is reduced to  $\int_{\infty}^{\infty} \varepsilon^{\alpha}(s)/s ds$  and ours to  $\int_{\infty}^{\infty} \varepsilon^2(s)/s ds < \infty$ . They coincide for  $\alpha = 2$  whereas for the remaining values of  $\alpha$  it might happen that the latter is fulfilled but the former is not. The asymptotics of solutions for  $t \to \infty$  of the type we consider here is obtained for the more general system y' = [A + B(t)]y, where A is a constant  $n \times n$  matrix whose characteristic roots are all simple and the continuous matrix B(t) is such that  $\int_{\infty}^{\infty} |B(s)| ds < \infty$  and hence for the *n*-th order equation

$$y^{(n)} + b_1(t)y^{(n-1)} + \dots + b_n(t)y = 0.$$

Compare with Coppel [23, Chapter IV, Th. 2, Th. 3]. We formulate explicitly the result for the *n*-th order equation with n = 2,  $b_1(t) = 0$ ,  $b_2(t) = p(t)$ , in order to compare it with the present ones, e.g. that of Theorem 2.19 when p(t) < 0: If the function p(t) is continuous for  $t \ge t_0$  and  $-\int_{\infty}^{\infty} sp(s) ds < \infty$ , then equation (2.1) has two linearly independent solutions  $y_1, y_2$  such that

$$y_1(t) \to 1, y_2(t) \sim t, ty'_1(t) \to 0, y'_2(t) \to 1$$

as  $t \to \infty$ . On the other hand, Theorem 2.19, with n = 1, gives, as mentioned above: If  $\int_{-\infty}^{\infty} sP^2(s) \, ds < \infty$ , then for the above mentioned solutions  $y_1, y_2$  one obtains (2.33). But in (the special case of) the Coppel theorem the integral  $\int_{-\infty}^{\infty} sp(s) \, ds$  converges, this is to say that the SV functions  $\exp\left\{\pm \int_{a}^{t} sp(s) \, ds\right\}$  become the special ones which tend to a constant at variance with the present result where these can be completely general ones.

### 2.7 A note about higher order equations; an alternative approach to second order equations

At the end of the previous section we mentioned the concept of a higher order equation. Here we consider a general *n*-th order linear differential equation again, and briefly describe the approach by which it was investigated in Řehák [148]. The main ideas are simple: To apply the classical Poincaré-Perron type result [138, 140], to find a fundamental set of real solutions, and to use a suitable transformation. We will see that consequences of the main statement can yield known results (in particular for second order equations); the method however is different.

For a given function  $\tau$ , we define the operator  $\mathfrak{D}_{\tau}$  as  $\mathfrak{D}_{\tau}u(s) = \tau(s)\frac{du}{ds}$ . Further, we set  $\mathfrak{D}_{\tau}^{n}u(s) = \tau(s)\frac{d}{ds}\mathfrak{D}_{\tau}^{n-1}u(s)$ ,  $n \in \mathbb{N}$ , with  $\mathfrak{D}_{\tau}^{0} = \text{id}$ . Consider the linear equation

$$\mathfrak{D}_{\tau}^{n} u + \tilde{a}_{n-1}(s)\mathfrak{D}_{\tau}^{n-1} u + \dots + \tilde{a}_{1}(s)\mathfrak{D}_{\tau} u + \tilde{a}_{0}(s)u = 0, \qquad (2.34)$$

where  $\tilde{a}_0, \ldots, \tilde{a}_{n-1}$  are continuous functions and  $\tau$  is a positive continuous function on  $[a, \infty)$ .

**Theorem 2.20** (Řehák [148]). Let  $\lim_{t\to\infty} \tilde{a}_i(s) = A_i$ , i = 0, ..., n-1, and  $\varrho_i$  be the roots of  $\varrho^n + A_{n-1}\varrho^{n-1} + \cdots + A_1\varrho + A_0 = 0$  all assumed to be real and of distinct moduli. Let

$$\int_{a}^{\infty} \frac{1}{\tau(z)} \, \mathrm{d}z = \infty$$

Then equation (2.34) possesses a fundamental set of (real) solutions  $\{u_1, \ldots, u_n\}$  such that

$$u_i \in \mathcal{NRV}_{\omega}(\varrho_i), \quad i = 1, \ldots, n,$$

where  $\omega(s) = \exp\left\{\int_a^s \frac{1}{\tau(z)} dz\right\}$ . Every nontrivial (real) solution u of (2.34) satisfies  $|u| \in \mathcal{NRV}_{\omega}(\varrho_m)$  for some  $m \in \{1, ..., n\}$ .

Note that one important step of the proof consists of transformation of equation (2.34) into

$$\frac{d^n y}{dt^n} + a_{n-1}(t)\frac{d^{n-1}y}{dt^{n-1}} + \dots + a_1(t)\frac{dy}{dt} + a_0(s)y = 0$$
(2.35)

by means of the relations  $s = \xi(t)$  and  $u(s) = y(\xi^{-1}(s)) = y(t)$ , where  $\xi$  is defined as the inverse of the function  $s \mapsto \int_a^s \frac{1}{\tau(z)} dz$ .

Theorem 2.20 can be applied in various ways. We mention several corollaries which are related to the results presented in this text.

Consider first the second order equation (such an equation is considered e.g. in (2.8))

$$u'' + b_1(s)u' + b_0(s)u = 0, (2.36)$$

where  $b_0$ ,  $b_1$  are continuous on  $[a, \infty)$ .

**Corollary 2.5.** Let there exist a function  $\tau$  with the properties

$$\tau \in C^1([a,\infty)), \ \tau(s) > 0 \ for \ s \ge a, \ and \ \int_a^\infty \frac{1}{\tau(z)} \ dz = \infty$$

such that

$$\lim_{s \to \infty} (\tau(s)b_1(s) - \tau'(s)) = A_1 \neq 0,$$
$$\lim_{s \to \infty} \tau^2(s)b_0(s) = A_0,$$

where  $A_1^2 > 4A_0$ . Then equation (2.36) possesses a fundamental set of solutions  $\{u_1, u_2\}$  such that

$$u_i \in \mathcal{NRV}_{\omega}(\varrho_i), \quad i = 1, 2$$

where  $\varrho_{1,2} = \frac{1}{2} \left( -A_1 \pm \sqrt{A_1^2 - 4A_0} \right)$  and  $\omega(s) = \exp \left\{ \int_a^s \frac{1}{\tau(z)} dz \right\}$ . If *u* is a nontrivial (real) solution of (2.36), then |*u*| is normalized regularly varying with respect to  $\omega$  of index  $\varrho_1$  or  $\varrho_2$ .

If we choose  $\tau(s) = s$ , then  $\omega(s) = s$  (up to a negligible multiplicative constant) and  $\mathcal{NRV}_{\omega} = \mathcal{NRV}$ . Hence we get the following corollary. Note that with this choice, the function  $s = \xi(t)$  defining the new variable in the proof of Theorem 2.20 becomes  $\xi(t) = e^t$ . Recall also the two useful relations which hold generally:  $\tau = \xi' \circ \xi^{-1}$  and  $\omega = \exp \circ \xi^{-1}$ .

#### Corollary 2.6. If

$$\lim_{s \to \infty} sb_1(s) = A_1 + 1 \neq 1, \quad \lim_{s \to \infty} s^2 b_0(s) = A_0 \tag{2.37}$$

with  $A_1^2 > 4A_0$ , then equation (2.36) possesses a fundamental set of solutions  $\{u_1, u_2\}$  such that  $u_i \in NRV(\varrho_i)$ , i = 1, 2, where  $\varrho_1, \varrho_2$  are as in the previous corollary. If u is a nontrivial (real) solution of (2.36), then |u| is normalized regularly varying of index  $\varrho_1$  or  $\varrho_2$ .

A more special choice in the previous corollary, namely  $b_1(s) \equiv 0$ , yields the existence of normalized  $\mathcal{RV}$  solutions of the indices  $(1 \pm \sqrt{1 - 4A_0})/2$  to the equation  $\frac{d^2u}{ds^2} + b_0(s)u = 0$  provided  $\lim_{s\to\infty} s^2b_0(s) = A_0 \in (-\infty, 1/4)$ . Recall that using a different technique the same result was obtained by Marić and Tomić [113] under the condition  $b_0 < 0$ , see also Corollary 2.2. This result was in essence first discovered by Omey in [130] (with additional conditions) and formulated for solutions tending to infinity. Later, the sufficient condition from [113] was improved and expressed in terms of  $\lim_{s\to\infty} s \int_s^{\infty} b_0(z) dz = A_0$ , and this integral condition was shown to be also necessary for the existence of  $\mathcal{RV}$  solutions, see Theorem 2.1.

Next we observe how Corollary 2.5 yields the result for a second order equation in the self-adjoint form

$$(r(s)u')' + p(s)u = 0, (2.38)$$

where  $p \in C, r \in C^1, r(s) > 0$  on  $[a, \infty)$ , when some special (and somehow optimal) setting is made. Recall that this equation was considered also in Section 2.3. Assume first  $\int_a^{\infty} \frac{1}{r(z)} dz = \infty$ . Set  $\omega(s) = \int_a^s \frac{1}{r(z)} dz$ . Because of the relation between  $\omega$  and  $\tau$ , which reads as  $\tau = \frac{\omega}{\omega'}$ , it means that  $\tau(s) = r(s) \int_a^s \frac{1}{r(z)} dz$ . Equation (2.38) can be written as (2.36), where  $b_1 = \frac{r'}{r}$  and  $b_0 = \frac{p}{r}$ . Since  $\tau(s)b_1(s) = r'(s) \int_a^s \frac{1}{r(z)} dz$ ,  $\tau'(z) = 1 - r'(z) \int_a^s \frac{1}{r(z)} dz$ , and

$$\tau^{2}(s)b_{0}(s) = p(s)r(s)\left(\int_{a}^{s} \frac{1}{r(z)} dz\right)^{2},$$

we have that (4.49) reduces to  $-1 = A_1$  and (4.50) reads as

$$\lim_{s\to\infty} p(s)r(s) \left(\int_a^s \frac{1}{r(z)} \,\mathrm{d}z\right)^2 = A_0.$$

Similarly, if  $\int_{a}^{\infty} \frac{1}{r(z)} dz < \infty$ , then we set  $\omega(s) = 1/\int_{s}^{\infty} \frac{1}{r(z)} dz$ , and (4.49) reduces to  $1 = A_1$ , while (4.50) becomes

$$\lim_{s\to\infty} p(s)r(s) \left(\int_s^\infty \frac{1}{r(z)} \,\mathrm{d}z\right)^2 = A_0.$$

Thus, applying Corollary 2.5, we get the following statement.

**Corollary 2.7.** (i) Let  $\int_{a}^{\infty} \frac{1}{r(z)} dz = \infty$ . Denote  $R(s) = \int_{a}^{s} \frac{1}{r(z)} dz$ . If

$$\lim_{s\to\infty}p(s)r(s)R^2(s)=A\in\left(-\infty,\frac{1}{4}\right),$$

then equation (2.38) possesses a fundamental set of solutions  $\{u_1, u_2\}$  such that  $u_i \in$  $\mathcal{NRV}_R(\varrho_i)$ , i = 1, 2, where  $\varrho_{1,2} = (1 \pm \sqrt{1-4A})$ . If u is a nontrivial (real) solution of (2.38), then |u| is normalized regularly varying with respect to R of index  $\varrho_1$  or  $\varrho_2$ . (ii) Let  $\int_a^{\infty} \frac{1}{r(z)} dz < \infty$ . Denote  $\tilde{R}(s) = \int_s^{\infty} \frac{1}{r(z)} dz$ . If

$$\lim_{s \to \infty} p(s)r(s)\tilde{R}^2(s) = \tilde{A} \in \left(-\infty, \frac{1}{4}\right),$$

then equation (2.38) possesses a fundamental set of solutions  $\{u_1, u_2\}$  such that  $u_i \in$  $\mathcal{NRV}_{1/\tilde{R}}(\tilde{\varrho}_i), i = 1, 2, where \tilde{\varrho}_{1,2} = (-1 \pm \sqrt{1 - 4\tilde{A}})$ . If u is a nontrivial (real) solution of (2.38), then |u| is normalized regularly varying with respect to  $1/\tilde{R}$  of index  $\tilde{\varrho}_1$  or  $\tilde{\varrho}_2$ .

As we could see in Theorem 2.8, using a different approach (more precisely, a combination of the Riccati technique with the contraction mapping theorem), Jaroš and Kusano in [60], showed that the condition

$$\lim_{s \to \infty} R(s) \int_s^\infty p(z) \, \mathrm{d}z = A < \frac{1}{4}$$

is sufficient and necessary for the existence of a fundamental set of solutions with the properties from the last corollary, provided  $\int_a^{\infty} \frac{1}{r(z)} dz = \infty$ . Since — provided the latter limit exists and  $\int_{a}^{\infty} p(z) dz$  converges — the L'Hospital's rule yields

$$\lim_{s \to \infty} R(s) \int_s^\infty p(s) \, \mathrm{d}z = \lim_{s \to \infty} p(s) r(s) R^2(s);$$

we see that we offer an alternative (and quite simple) approach to the result closely related to Theorem 2.8. Similarly, if we assume  $\int_a^{\infty} \frac{1}{r(z)} dz < \infty$ , then the necessary and sufficient condition for the existence of generalized  $\mathcal{RV}$  solutions from Theorem 2.8 reads as

$$\lim_{s\to\infty} (\tilde{R}(s))^{-1} \int_s^\infty \tilde{R}^2(z) p(z) \, \mathrm{d}z = \tilde{A} < \frac{1}{4}.$$

To see a relation with Corollary 2.7-(ii), note that

$$\lim_{s\to\infty} (\tilde{R}(s))^{-1} \int_s^\infty \tilde{R}^2(z) p(z) \, \mathrm{d}z = \lim_{s\to\infty} p(s) r(s) \tilde{R}^2(s),$$

provided the assumptions for the use of the L'Hospital rule are satisfied.

Corollary 2.5 can be utilized to study self-adjoint form (2.38) also in the following, slightly different, way. We set  $\tau(s) = s$ , which yields  $\omega(s) = s$ . Recall that equation (2.38) can be written as (2.36), where  $b_1 = \frac{r'}{r}$  and  $b_0 = \frac{p}{r}$ . Since then  $\tau(s)b_1(s) - \tau'(s) = \frac{sr'(s)}{r(s)} - 1$  holds, condition (4.49) can be understood as  $r \in NRV(A_1 + 1)$ . Condition (4.50) reads as  $p(s) \sim A_0 s^{-2} r(s)$  which among others means that also p is required to be regularly varying, namely of index  $(A_1 + 1) - 2 = A_1 - 1$ . Thus we get the following statement.

Corollary 2.8. If

$$r \in \mathcal{NRV}(A_1 + 1)$$
 and  $s^2 p(s) \sim A_0 r(s)$  as  $s \to \infty$ ,

with  $A_1^2 > 4A_0$ , then equation (2.38) possesses a fundamental set of solutions  $\{u_1, u_2\}$  such that  $u_i \in NRV(\varrho_i)$ , i = 1, 2, where  $\varrho_1, \varrho_2$  are as in Corollary 2.5.

We conclude this part devoted to second order equations with showing another way how Corollary 2.5 can be utilized. In addition, we present related observations and indicate relations with known results. We consider the equation

$$u^{\prime\prime} = p(s)u, \tag{2.39}$$

where p > 0,  $p \in C^1$ . If  $b_1 = 0$  and  $b_0 = -p$ , then (2.36) reduces (2.39). Clearly, equation (2.39) is nonoscillatory. Set  $\tau = 1/\sqrt{p}$ . Then condition (4.49) reads as

$$\lim_{s \to \infty} \left( \frac{-1}{\sqrt{p(s)}} \right)' = A_1 \tag{2.40}$$

and condition (4.50) reduces to  $A_0 = -1$ . Assume first  $A_1 \neq 0$ . Then  $\int_a^{\infty} \frac{dz}{\tau(z)} dz = \infty$  and (2.39) possesses solutions  $u_i \in NRV_{\omega}(\vartheta_i)$ , i = 1, 2, where

$$\vartheta_{1,2} = \frac{1}{2} \left( -A_1 \pm \sqrt{A_1^2 + 4} \right)$$

(with  $\vartheta_{1,2} \neq -1, 0, 1$ ) and  $\omega(s) = \exp \left\{ \int_a^s \sqrt{p(z)} dz \right\}$ , by Corollary 2.5. Hence,  $\frac{u'_i(s)}{u_i(s)} \sim \vartheta_i \sqrt{p(s)}$  as  $s \to \infty$ , i = 1, 2. Since  $u_i$  is positive,  $u'_i$  is eventually positive or negative. We have

$$\frac{u_i''(s)u_i(s)}{(u_i'(s))^2} = \frac{p(s)u_i^2(s)}{(u_i'(s))^2} \sim \frac{1}{\vartheta_i^2} \neq 1,$$
(2.41)

i = 1, 2. We then get  $u_i \in \mathcal{RV}\left(\frac{\vartheta_i^2}{\vartheta_i^2 - 1}\right)$ , i = 1, 2, cf. Omey [130] where increasing solutions are considered. Because of monotonicity of u', we have

$$u_i \in \mathcal{NRV}\left(\frac{\vartheta_i^2}{\vartheta_i^2 - 1}\right),\tag{2.42}$$

i = 1, 2. Observe that condition (2.40) with  $A_1 \neq 0$  implies the existence of the limit  $\lim_{s\to\infty} s^2 p(s)$ ; compare with the latter condition in (2.37). The expected correspondence between the indices of regular variation in (2.42) and in Corollary 2.6 with  $b_1 = 0$  and  $b_0 < 0$  can be now easily revealed; the details are left to the reader.

Assume now  $A_1 = 0$  in (2.40) and  $\int_a^{\infty} \sqrt{p(z)} dz = \infty$ . Then by Hartman-Wintner [54, Paragraph 24], (2.39) has a pair of solutions satisfying

$$u_i'(s) \sim \pm \sqrt{p(s)} u_i(s) \tag{2.43}$$

as  $s \to \infty$ , i = 1, 2. Notice that the roots of the associated characteristic equation  $\rho^2 - 1 = 0$  are  $\pm 1$ , and so Corollary 2.5 can not be used. In terms of generalized regular variation, we can write  $u_i \in N\mathcal{RV}_{\omega}(\pm 1)$ , where  $\omega(s) = \exp\left\{\int_a^s \sqrt{p(z)} dz\right\}$ . For increasing solutions, this statement was rediscovered in [130] by Omey. In addition, by the same author, relations with the class  $\Gamma$  were shown and condition (2.40) with  $A_1 = 0$  was relaxed to  $1/\sqrt{p} \in \mathcal{BSV}$  (without requiring  $p \in C^1$ ) in [131], see Theorem 2.15. Other type of relaxation of (2.40) with  $A_1 = 0$  can be found in Hartman-Wintner [54], namely to the condition

$$\sup_{0 \le k < \infty} \frac{|\ln p(s+k)/p(s)|}{1 + \int_s^{s+k} \sqrt{p(z)} \, \mathrm{d}z} \to 0$$

as  $s \to 0$ , which is sufficient and also necessary for equation (2.39) to posses a pair of solutions satisfying (2.43). Let us show how the classes  $\Gamma(\pm 1; v)$  are involved. In the set  $\{u_1, u_2\}$  satisfying (2.43), clearly one solution must be increasing (say  $u_1$ ) while other is decreasing. Similarly as in (2.41), we obtain  $\frac{u''_i(s)u_i(s)}{(u'_i(s))^2} \sim 1$  as  $s \to \infty$ , i = 1, 2. Hence we get  $u_i \in \Gamma(\sigma_i; v)$ , i = 1, 2, where  $\sigma_1 = 1, \sigma_2 = -1$ , and v = |u/u'|. Thus

$$u_i = \Gamma(\sigma_i; p^{-\frac{1}{2}}),$$

i = 1, 2, by (2.43). From the transformation relations between u(s) and y(t) (see the text after (2.35)) we obtain a pair of Poincaré-Perron solutions { $y_1, y_2$ } of

$$\frac{d^2y}{dt^2} + a_1(t)\frac{dy}{dt} - y = 0$$
(2.44)

satisfying  $\frac{dy(t)}{dt} \cdot \frac{1}{y(t)} \to \pm 1$  as  $t \to \infty$ , where  $a_1(t) = \tilde{a}_1(\xi(t)) = -\tau'(\xi(t)) \to 0$  as  $t \to \infty$ ;  $\xi$  being the inverse of  $s \mapsto \int_a^s \frac{1}{\tau(z)} dz$ . In this connection we recall the result [23, Chapter IV, Theorem 2] according to which (2.44) possesses a pair of solutions satisfying  $y(t) \sim e^t$ ,  $y_2(t) \sim e^{-t}$ ,  $\frac{dy_1(t)}{dt} \sim e^t$ ,  $\frac{dy_2(t)}{dt} \sim e^{-t}$  as  $t \to \infty$  provided the (stronger) condition  $\int_a^\infty |a_1(z)| dz < \infty$  is fulfilled. Clearly then  $y_1 \in \Gamma(1; 1)$ ,  $y_2 \in \Gamma(-1; 1)$ . Another result of Coppel ([23, Chapter IV, Theorem 14]) says that if p > 0 and  $\int_a^\infty |p^{-3/2}(t)p''(t)| dt < \infty$ , then (2.39) has a fundamental set of solutions satisfying

$$u_i(s) \sim p^{-1/4}(s) \exp\left\{\pm \int_a^s p^{1/2}(t) \, \mathrm{d}t\right\},\$$
$$u_i'(s) \sim p^{1/4}(s) \exp\left\{\pm \int_a^s p^{1/2}(t) \, \mathrm{d}t\right\}$$

as  $s \to \infty$ , i = 1, 2.

For completeness, we mention another two classical results which are somehow related to the topic of this paragraph. Hartman's result [53, Ex. 9.9-b] is recalled in (2.33). Mařík's and Ráb's statement reads as (2.30) The conditions in these statements (including Coppel's one) differ among each other but lead to the same type of conclusions. As we could see above, such a behavior is typical for regularly (or rapidly) varying solutions. Concerning the requirement imposed on *p*, the behavior of  $s^2p(s)$  (or, possibly, of  $s \int_s^{\infty} p(t) dt$  or of  $s \int_s^{\lambda s} p(t) dt$ ) is crucial in that respect. Recall that the existence of the finite limit  $\lim_{s\to\infty} s^2p(s)$  leads to a regularly varying solutions while  $\lim_{s\to\infty} s^2p(s) = \infty$  yields rapidly varying solutions of (2.39), see Section 2.1. The fact that a solution is a  $\mathcal{RV}$  or  $\mathcal{RPV}$  function is less precise than the formulas in the quoted results, but requires milder hypotheses.

Next we examine the third order equation in the self-adjoint form

$$u''' + 2b(s)u' + b'(s)u = 0, (2.45)$$

where  $b \in C^1([a, \infty))$ . First note that this equation is not so special in comparison with the general equation

$$u''' + b_2(s)u'' + b_1(s)u' + b_0(s)u = 0,$$
(2.46)

as it might seem. Indeed, the second derivative term in (2.46) can always be removed by the transformation

$$u(s) = v(s) \exp\left\{-\frac{1}{3} \int_{a}^{s} b_{2}(z) \,\mathrm{d}z\right\}$$

to obtain the equation v''' + 2b(s)v' + (b'(s) + c(s))v = 0. This form has frequently occurred in the literature ([65, 157]), thus we can easily make a comparison. Our method (which leads to the existence of generalized regularly varying solutions) and is based on Theorem 2.20 applies also to general equation (2.46).

#### Corollary 2.9. If

$$\lim_{s \to \infty} (-s^3 b'(s)) = B < 1, \tag{2.47}$$

then (2.45) possesses a fundamental system of solutions  $\{u_1, u_2, u_3\}$  such that  $u_i \in NRV(\varrho_i)$ , i = 1, 2, 3, where  $\varrho_1 = 1$ ,  $\varrho_2 = 1 + \sqrt{1-B}$ ,  $\varrho_3 = 1 - \sqrt{1-B}$ . If u is a nontrivial (real) solution of (2.45), then |u| is normalized regularly varying of index 1 or  $1 + \sqrt{1-B}$  or  $1 - \sqrt{1-B}$ .

In Jaroš, Kusano, Marić [65], see Section 5.3 — using a different approach which is based on the correspondence of (2.45) with certain second order equation — it is proved that the statement of Corollary 2.9 holds under the assumption  $\lim_{s\to\infty} \int_s^{\infty} b(z) dz = \frac{B}{2}$ . A simple use of the L'Hospital rule shows how (2.47) implies this assumption. Note however that (2.47) does not require the convergence of  $\int_a^{\infty} b(z) dz$ .

We conclude this section with observations related to the two-term fourth order linear differential equation

$$u^{(4)} + b(s)u = 0, (2.48)$$

where  $b \in C([a, \infty))$ . Denote

$$G(x) = x^4 - 6x^3 + 11x^2 - 6x.$$

Applying Theorem 2.20 we get the following statement.

Corollary 2.10. If

$$\lim_{s\to\infty}s^4b(s)=A\in\left(-\frac{9}{16},1\right),$$

then equation (2.48) possesses a fundamental set of solutions  $\{u_1, u_2, u_3, u_4\}$  such that  $u_i \in NRV(\varrho_i)$ , i = 1, 2, 3, 4, where  $\varrho_i$  are the (real) roots of  $G(\varrho) + A = 0$ . If u is a nontrivial (real) solution of (2.48), then |u| is normalized regularly varying of index  $\varrho_1$  or  $\varrho_2$  or  $\varrho_3$  or  $\varrho_4$ .

Note that as a by-product of Corollary 2.10 we get a nonoscillation criterion for equation (2.48). For related criteria see [157, Chapter 3].

Let us now examine slowly varying solutions of (2.48) where b < 0 and A = 0. Let us write the equation in the form

$$u^{(4)} = p(s)u, (2.49)$$

i.e., p = -b > 0. Since A = 0, the roots of  $G(\varrho) + A = 0$  are 0, 1, 2, 3, and so SV solutions of (2.49) indeed exist. Eventually positive solutions of (2.49) are eventually monotone and therefore a solution  $u \in SV$  is eventually monotone. It can be shown that SV solutions cannot increase. Hence, if we deal with SV solutions, in fact, we deal with all positive decreasing solutions of (2.49), since non-SV solutions are in  $RV(1) \cup RV(2) \cup RV(3)$ . As for the existence of SV solutions, the condition  $\lim_{s\to\infty} s^4 p(s) = 0$  can be relaxed to

$$\lim_{s \to \infty} \int_s^\infty z^2 p(z) \, \mathrm{d}z = 0. \tag{2.50}$$

More precisely, it then holds that any eventually positive decreasing solution u of (2.49) is in NSV.

Next we give asymptotic formula for a solution  $u \in SV$  under the assumption  $p \in RV(-4)$ . Denote  $H(s) = s^3 p(s)/6$ . It holds  $H \in RV(-1)$  and any of the possibilities, convergence or divergence of  $\int_a^{\infty} H(z) dz$ , can in general occur.

**Theorem 2.21** (Řehák [148]). Let  $p \in \mathcal{RV}(-4)$  with  $L_p(s) \to 0$  as  $s \to \infty$ ;  $L_p$  being the SV component of p. Then the set of eventually positive decreasing solutions of (2.49) (which is nonempty) is a subset of SV. For each solution  $u \in SV$  (which is necessarily decreasing) one has  $-u \in \Pi(-su'(s))$  and one of the following formula holds:

(i) If  $\int_{a}^{\infty} H(z) dz = \infty$ , then u(s) tends to zero as  $s \to \infty$  and satisfies the formula

$$u(s) = \exp\left\{-\int_a^s (1+o(1))H(z)\,\mathrm{d}z\right\}.$$

(ii) If  $\int_{a}^{\infty} H(z) dz < \infty$ , then u(s) tends to a positive constant  $u(\infty)$  as  $s \to \infty$  and satisfies the formula

$$u(s) = u(\infty) \exp\left\{\int_{s}^{\infty} (1 + o(1))H(z) \,\mathrm{d}z\right\}.$$

The above theorem can be understood as a "fourth-order extension" of Theorem 2.11.

### 2.8 On zeros of oscillatory solutions, asymptotic behavior of maxima and of eigenvalues

The problem of determining the asymptotic behavior of the number of zeros is an old one and studied by many authors, see e.g. Hartman [53, Chapter XI]. Here we show how the class of Beurling slowly varying functions can be utilized to obtain some information about this behavior. The results were achieved by Hačik and Omey [51], see also Marić [105], Omey [130].

In this section we assume that p in (2.1) is a positive and continuous function. In the sequel,  $\{t_n\}, t_n \to \infty$ , denotes the sequence of zeros of a solution y(t) of (2.1) and N(I) the number of these in the interval I.

As an example, consider equation (2.1) with  $p(t) = k^2$ ,  $k \in (0, \infty)$ . Every solution of (2.1) has the form  $y(t) = c_1 \sin(kt) + c_2 \cos(kt)$ , where  $c_c, c_2 \in \mathbb{R}$ . We see that this solution has an infinite number of zeros  $t_n$ ,  $n \in \mathbb{N}$ , which satisfy  $t_{n+1} - t_n = \pi/k$  and  $kt_n \sim n$  as  $n \to \infty$ . This last statement says that the number of zeros less than  $t_n$  asymptotically equals  $\sqrt{p(t_n)}t_n$ .

In the general case Titchmarsh [165], p. 146, proved that, whenever p(t) is continuous and of bounded variation, there holds that

$$\left| N((a,t)) - \frac{1}{\pi} \int_{a}^{t} \sqrt{p(s)} \, \mathrm{d}s \right| \le 1 + \frac{1}{2\pi} \int_{a}^{t} 1/\sqrt{p(s)} \left| \mathrm{d}(\sqrt{p(s)}) \right|.$$

We will discuss conditions under which we can relate the asymptotic behavior of p(t) to that of N((0, t]), to that of the sequence of zeros  $\{t_n\}$  and to that of the sequence  $\{t_{n+1} - t_n\}$ .

First note that if  $1/\sqrt{p} \in \mathcal{BSV}$ , then  $t^2p(t) \to \infty$  as  $t \to \infty$  so that equation (2.1) is oscillatory.

**Theorem 2.22** (Hačik and Omey [51]). Suppose  $1/\sqrt{p} \in \mathcal{BSV}$ . Then

(*i*)  $N((0, t]) \sim \frac{1}{\pi} \int_0^t \sqrt{p(s)} \, ds \text{ as } t \to \infty,$ (*ii*)  $N((t, t + s/\sqrt{p(s)}]) = \frac{t}{\pi} + O(1)$  for each s > 0. From the properties of  $\mathcal{BSV}$  functions we have that if  $1/\sqrt{p} \in \mathcal{BSV}$ , then p may be regularly varying and likewise belong to the class  $\Gamma$ . This justifies the following two corollaries.

**Corollary 2.11.** (i) If  $p \in \mathcal{RV}(\alpha)$ ,  $\alpha > -2$ , then  $N((0, t]) \in \mathcal{RV}((\alpha + 2)/2)$  and

$$N((0,t]) \sim \frac{2}{\pi(\alpha+2)} t \sqrt{p(t)}$$

as  $t \to \infty$ 

(*ii*) If  $p \in \Gamma(h(t))$ , then  $N((0, t]) \in \Gamma(2h(t))$  and

$$N((0,t]) \sim \frac{2}{\pi} h(t) \sqrt{p(t)}$$

as  $t \to \infty$ .

The proof of this corollary follows from the previous theorem and elementary properties of the classes  $\mathcal{RV}$  and  $\Gamma$ .

If we replace *t* by  $t_n$  and use  $\lim_{n\to\infty} \sqrt{p(t_n)}(t_{n+1} - t_n) = \pi$ , we get the following statement for the sequence of zeros.

**Corollary 2.12.** (i) If  $p \in \mathcal{RV}(\alpha)$ ,  $\alpha > -2$ , then  $t_n \sqrt{p(t_n)} \sim n\pi(\alpha + 2)/2$  and

$$n\frac{t_{n+1}-t_n}{t_n} \to \frac{2}{\alpha+2}$$

as  $n \to \infty$ .

(*ii*) If  $p \in \Gamma(h(t))$ , then  $h(t_n) \sqrt{p(t_n)} \sim n\pi/2$  and  $t \to -t$ 

$$n\frac{t_{n+1}-t_n}{h(t_n)} \to 2$$

as  $n \to \infty$ .

Up to now we assumed  $1/\sqrt{p(t)} \in \mathcal{BSV}$  which implies  $t^2p(t) \to \infty$  as  $t \to \infty$ . When  $t^2p(t) \to A \ge 1/4$  however (2.1) may remain oscillatory. To deal with this kind of equations we transform (2.1) into a more suitable form. Generally, consider the differential equation (2.9), r(t) > 0,  $t \ge a \ge 0$ . For a suitable function  $\Psi$  we can define

$$\xi(t) = \int_{a}^{t} \frac{1}{r(s)\Psi^{2}(s)} \,\mathrm{d}s, \ \eta(\xi) = \frac{y(t)}{\Psi(t)}.$$

With this transformation, (2.9) becomes

$$\frac{\mathrm{d}^2\eta}{\mathrm{d}\xi^2} + \tilde{p}(\xi)\eta(\xi) = 0, \quad 0 \le \xi \le \xi(\infty),$$

where  $\tilde{p}(\xi) = [(r\Psi')' + p\Psi]\Psi^3 r$ , see Willett [166, p. 597]. Now  $t_n$  is a zero of y if and only if  $\xi_n \equiv \xi(t_n)$  is a zero of  $\eta$ . Hence the number of zeros of y less than or equal T is the same as the number of zeros of  $\eta$  less than  $\xi(T)$ . Now Theorem 2.22 yields

**Theorem 2.23** (Hačik and Omey [51]). *Suppose that*  $\tilde{p}$  *satisfies the conditions of Theorem 2.22. Then for equation* (2.9) *we have* 

$$N((0,t]) \sim \frac{1}{\pi} \int_{a}^{t} \sqrt{\tilde{p}(\xi(s))} \xi'(s) \,\mathrm{d}s$$

as  $t \to \infty$ .

In the case of (2.1) we have  $r \equiv 1$ . If we choose  $\Psi(t) = \sqrt{t}$ , a = 1, then  $\xi = \ln t$  and  $\tilde{p}(\xi) = t^2 p(t) - 1/4$ . Hence we get

**Corollary 2.13.** (*i*) If  $\tilde{p}(t) = e^{2t}p(e^t) - 1/4 \in \mathcal{RV}(\alpha)$ ,  $\alpha > -2$ , then

$$N((0,t]) \sim \frac{1}{\pi} \ln t \sqrt{t^2 p(t) - 1/4} \frac{2}{\alpha + 2}$$

as  $t \to \infty$ .

(*ii*) If  $\tilde{p} \in \Gamma(h)$ , then

$$N((0,t]) \sim \frac{1}{\pi} 2h(\ln t) \sqrt{t^2 p(t) - 1/4}$$

as  $t \to \infty$ .

This corollary of course remains valid when  $t^2 p(t) \rightarrow \infty$ .

Omey in [130] studies also asymptotic behavior of the sequence  $\{y^2(s_n)\}$  of successive maxima of  $y^2$ , y being a solution of (2.1),  $s_n$  denotes the value at which |y(t)| reaches a maximum. For illustration, we present here one selected result. Note that its proof uses some of the previous results in this section, the Sturm comparison theorem and Wiman's method [167].

**Theorem 2.24** (Omey [130]). (*i*) Let *p* be a monotone function such that

$$\lim_{t\to\infty}\frac{tp'(t)}{p(t)}=\alpha\in(-2,\infty).$$

Then the sequences  $\{y'^2(t_n)\}, \{y^{-2}(s_n)\}\$  are in  $\mathcal{RV}(\alpha/(\alpha+2))$ .

(*ii*) Let  $p' \in \Gamma(h(t))$  and

$$\lim_{t \to \infty} \frac{\ln(p(t))}{h(t)\sqrt{p(t)}} = 0.$$

Then the sequences  $\{y'^2(t_n)\}, \{y^{-2}(s_n)\}\$  are in  $\mathcal{RV}(1)$ .

Omey [130] further studies the asymptotic behavior of the eigenvalues of the linear operator

$$\mathcal{L} = q(t) - \frac{\mathrm{d}^2}{\mathrm{d}^2 t}$$

in  $(0, \infty)$ . Consider the differential equation  $\mathcal{L}[y] = \lambda y$ , where  $\lambda \in \mathbb{R}$ . A function which satisfies this equation and also some boundary conditions (e.g. y(0) = 0,

y'(0) = 0 is called an eigenfunction. The corresponding value of  $\lambda$  is called an eigenvalue. We will assume that q(t) diverges to infinity as  $t \to \infty$ , since the operator  $\mathcal{L}$  then has an infinite number of eigenvalues  $\{\lambda_n\}$ ; also the eigenfunction associated with  $\lambda_n$  has exactly *n* zeros [165, p. 110].

In the theorem below it is examined how the distribution of the eigenvalues  $\{\lambda_n\}$  is determined by the function *q*. We present the result without a proof; note just that it utilizes the Sturm comparison theorem, an Abelian type theorem involving  $\mathcal{RV}$  functions, and some the previous results in this section. Let M(x) denote the number of eigenvalues of the operator  $\mathcal{L}$  not exceeding *x*.

**Theorem 2.25** (Omey [130]). Let q be a continuous, increasing function with q(0) = 0. If  $q \in \mathcal{RV}(\alpha)$ ,  $\alpha \in (0, \infty)$ , then  $M \in \mathcal{RV}((\alpha + 2)/(2\alpha))$ . Furthermore,

$$q(M(t)/\sqrt{t}) \sim t \left(\frac{B(3/2, 1/\alpha)}{\alpha\pi}\right)^{\alpha}$$

as  $t \to \infty$ , where  $B(\cdot, \cdot)$  denotes the beta function.

Under the conditions of this theorem, the sequence  $\{\lambda_n\}$  is regularly varying with the index  $2\alpha/(\alpha + 2)$  and

$$q(n/\sqrt{\lambda_n}) \sim k_n \left(\frac{B(3/2, 1/\alpha)}{\alpha\pi}\right)^{\alpha}$$

as  $n \to \infty$ .

Solving this "asymptotic functional equation", we can derive an explicit asymptotic formula for  $\{\lambda_n\}$ ; we give a sufficient condition to do this (the proof uses the concept of conjugate *SV* solutions). Suppose  $q(t) = (tL(t))^{\alpha}$ ,  $\alpha \in (0, \infty)$ , where  $L \in SV$ . If for all real  $\beta$ ,

$$\lim_{t\to\infty}\frac{L(tL^{\beta}(t))}{L(t)}=1,$$

then

$$\lambda_n \sim \left(\frac{n\alpha\pi}{B(3/2,1/\alpha)}\right)^{2\left(\frac{\alpha}{\alpha+2}\right)^2} \left(q\left(\left(\frac{n\alpha\pi}{B(3/2,1/\alpha)}\right)^{\frac{2}{\alpha+2}}\right)\right)^{\frac{\alpha}{\alpha+2}}$$

as  $n \to \infty$ .



## Half-linear second order differential equations

#### 3.1 Introduction

In this chapter we consider the half-linear second order differential equation

$$(r(t)\Phi(y'))' + p(t)\Phi(y) = 0, \quad \Phi(u) := |u|^{\alpha - 1} \operatorname{sgn} u, \ \alpha > 1, \tag{3.1}$$

Recall that the terminology *half-linear* differential equation (systematically used by Bihari and Elbert for the first time) reflects the fact that the solution space of (3.1) is homogeneous, but not additive. The works of Mirzov [126] and Elbert [30] are usually regarded as pioneering ones. A quite comprehensive treatment on half-linear differential equations can be found in the book [28] by Došlý, Řehák.

We suppose that the functions r, p are continuous and r(t) > 0 in the interval under consideration. Many of the results for (3.1) can be formulated under weaker assumptions that the functions 1/r, c are locally integrable. However, since we are interested in solutions of (3.1) in the classical sense (i.e., a solution y of (3.1) is a  $C^1$ function such that  $r\Phi(y') \in C^1$  and satisfies (3.1) in an interval under consideration), the continuity assumption is appropriate for this setting.

Half-linear equations are closely related to the partial differential equations with *p*-Laplacian. In fact, (3.1) is sometimes called the *differential equation with the one-dimensional p-Laplacian*. Recall that the  $\alpha$ -Laplacian is a partial differential operator of the form

$$\Delta_{\alpha} u := \operatorname{div} \left( \|\nabla u\|^{\alpha-2} \nabla u \right),$$

where, for  $u = u(x) = u(x_1, ..., x_N)$ ,  $\nabla u = \left(\frac{\partial}{\partial x_1}, ..., \frac{\partial}{\partial x_N}\right)$  is the Hamilton nabla operator and, for  $v(x) = (v_1(x), ..., v_N(x), \text{ div } v(x) = \sum_{i=1}^N \frac{\partial v}{\partial x_i}(x)$  is the usual divergence operator. If u is a radially symmetric function, i.e., u(x) = y(t), t = ||x||,  $||\cdot||$  being the Euclidean norm in  $\mathbb{R}^N$ , the (partial) differential operator  $\Delta_\alpha$  can be reduced to the ordinary differential operator

$$\Delta_{\alpha}u(x)=t^{1-N}\left(t^{N-1}\Phi(y'(t))\right)',\quad '=\frac{\mathrm{d}}{\mathrm{d}t}.$$

There are at least another two ways how half-linear equations can be understood. First, as a natural generalization of second order linear differential equations. Indeed, if  $\alpha = 2$ , then (3.1) reduces to (2.9). Second, as a special case of certain Emden-Fowler type equations, which are studied in the next chapter.

Some of the results will be formulated for the half-linear equation in a more special form

$$(\Phi(y'))' + p(t)\Phi(y) = 0. \tag{3.2}$$

An important role in some of the proofs will be played (not surprisingly) by the generalized Riccati differential equation

$$w' + p(t) + (\alpha - 1)|w|^{\beta} = 0, \qquad (3.3)$$

which is associated to (3.2) by the substitution  $w = \Phi(y'/y)$ . Note that the Riccati technique is extremely useful also in many other parts of qualitative theory of half-linear equations, see Došlý, Řehák [28].

Basic classification and asymptotics of nonoscillatory solutions to half-linear equations are discussed e.g. in Cecchi, Došlá, Marini [19] and Chanturiya [20], see also [28, Chapter 4].

The results in this chapter can be understood as a half-linear extension of some of the results from the previous chapter, but there are also some original formulas. However we should emphasize that due to the lack of the additivity of the solutions space of equation (3.1), many steps in the proofs (if not the proof entire) require a quite new approach or at least a highly nontrivial modification comparing with the linear case. As an example of the problematic point we can mention that there is no reduction of order formula for (3.1), and so we cannot so simply construct a linearly independent solution provided one solution (with some known properties) is at disposal.

#### 3.2 $\mathcal{RV}$ and $\mathcal{RB}$ solutions of half-linear equations

The results of this section are based on the paper [59] by Jaroš, Kusano, and Tanigawa.

#### 3.2.1 Auxiliary statement

**Proposition 3.1.** Put  $P(t) = \int_t^{\infty} p(s) ds$  and suppose that there exists a continuous function  $h : [t_0, \infty) \to (0, \infty), t_0 \ge 0$ , such that  $\lim_{t\to\infty} h(t) = 0, |P(t)| \le h(t), t \ge t_0$ , and

$$\int_{t}^{\infty} h^{\beta}(s) \, \mathrm{d}s \le \frac{1}{\alpha - 1} a^{\beta - 1} h(t), \quad t \ge t_{0}, \tag{3.4}$$
for some positive constant

$$a < \frac{1}{\alpha} \left(\frac{\alpha - 1}{\alpha}\right)^{\alpha - 1}.$$
(3.5)

*Then equation* (3.2) *is nonoscillatory and has a solution of the form* 

$$y(t) = \exp\left\{\int_{t_0}^t \Phi^{-1}[v(s) + P(s)] \,\mathrm{d}s\right\}, \ t \ge t_0,$$
(3.6)

where v is a solution of the integral equation

$$v(t) = (\alpha - 1) \int_{t}^{\infty} |v(s) + P(s)|^{\beta} \mathrm{d}s, \ t \ge t_{0},$$
(3.7)

satisfying

$$v(t) = O(P(t)) \text{ as } t \to \infty.$$
(3.8)

*Proof.* Consider the function *y* defined by (3.6). Recall that if *w* is a solution of generalized Riccati equation (3.3), then  $\exp\left\{\int_{t_0}^t \Phi^{-1}(w(s))\,ds\right\}$  is a (nonoscillatory) solution of (3.2). Hence, *y* is a solution of (3.2) if *v* is chosen in such a way that w = v + P satisfies (3.3) on  $[t_0, \infty)$ . The differential equation for *v* then reads  $v' + (\alpha - 1)|v + P(t)|^{\beta} = 0$ , which upon integration under the additional requirement that  $\lim_{t\to\infty} v(t) = 0$ , yields (3.7). We denote by  $C_h[t_0, \infty)$  the set of all continuous functions *v* on  $[t_0, \infty)$  such that

$$\|v\|_h = \sup_{t \ge t_0} \frac{|v(t)|}{h(t)} < \infty$$

Clearly,  $C_h[t_0, \infty)$  is a Banach space equipped with the norm  $||v||_h$ . Let  $\Omega$  be a subset of  $C_h[t_0, \infty)$  defined by  $\Omega = \{v \in C_h[t_0, \infty) : |v(t)| \le (\alpha - 1)h(t), t \ge t_0\}$  and define the mapping  $\mathcal{T} : \Omega \to C_h[t_0, \infty)$  by

$$\mathcal{T}v(t) = (\alpha - 1) \int_t^\infty |v(s) + P(s)|^\beta \mathrm{d}s, \ t \ge t_0.$$
(3.9)

If  $v \in \Omega$ , then

$$|\mathcal{T}v(t)| \le (\alpha - 1)\alpha^{\beta} \int_{t}^{\infty} h^{\beta}(s) \, \mathrm{d}s \le \alpha^{\beta} a^{\beta - 1} h(t),$$

 $t \ge t_0$ , which implies that

$$\|\mathcal{T}v\|_{h} \le \alpha^{\beta} a^{\beta-1} < \alpha^{\beta} \left(\frac{(\alpha-1)^{\alpha-1}}{\alpha^{\alpha}}\right)^{\beta-1} = \alpha - 1.$$
(3.10)

Thus  $\mathcal{T}$  maps  $\Omega$  into itself. If  $v_1, v_2 \in \Omega$ , then, using the Mean Value Theorem, we see that

$$\begin{aligned} |\mathcal{T}v_{1}(t) - \mathcal{T}v_{2}(t)| &\leq (\alpha - 1)\int_{t}^{\infty} \left| |v_{1}(s) + P(s)|^{\beta} - |v_{2}(s) + P(s)|^{\beta} \right| \mathrm{d}s \\ &\leq (\alpha - 1)\beta \int_{t}^{\infty} (\alpha h(s))^{\beta - 1} |v_{1}(s) - v_{2}(s)| \, \mathrm{d}s \\ &= \alpha^{\beta} \int_{t}^{\infty} h^{\beta}(s) \frac{|v_{1}(s) - v_{2}(s)|}{h(s)} \, \mathrm{d}s \\ &\leq \alpha^{\beta} (\beta - 1)a^{\beta - 1} h(t) ||v_{1} - v_{2}||_{h}, \end{aligned}$$

 $t \ge t_0$ , from which it follows that

$$\|\mathcal{T}v_1 - \mathcal{T}v_2\|_h \le (\beta - 1)\alpha^{\beta}a^{\beta - 1}\|v_1 - v_2\|_h$$

In view of (3.5) (or (3.10)), this implies that  $\mathcal{T}$  is a contraction mapping on  $\Omega$ . Therefore, by the contraction mapping principle, there exists an element  $v \in \Omega$  such that  $v = \mathcal{T}v$ , that is, a solution of integral equation (3.7). Thus the function y(t) defined by (3.6) with this v(t) gives a solution of (3.2) on  $[t_0, \infty)$ . The fact that v satisfies (3.8) is a consequence of  $v \in \Omega$ . This completes the proof.

## 3.2.2 *RV* solutions with different indices

The following theorem is a generalization of the main part of Theorem 2.2. Proposition 3.1 is applied in the proof.

**Theorem 3.1.** Equation (3.2) is nonoscillatory and and has two solutions  $y_1$  and  $y_2$  such that  $y_1 \in NSV$ ,  $y_2 \in NRV(1)$  if and only if

$$\lim_{t \to \infty} t^{\alpha - 1} \int_t^\infty p(s) \, \mathrm{d}s = 0 \tag{3.11}$$

*Proof.* If. Suppose that the condition from the theorem holds. Put

$$\varphi(t) = \sup_{s \ge t} \left| s^{\alpha - 1} \int_s^\infty p(\tau) \, \mathrm{d}\tau \right|. \tag{3.12}$$

Then  $\varphi$  is nonincreasing and tends to zero as  $t \to \infty$ . Let  $t_0 > 0$  be such that

$$\varphi(t) < \frac{1}{\alpha} \left(\frac{\alpha - 1}{\alpha}\right)^{\alpha - 1}$$
 and  $|P(t)| \le \frac{\varphi(t)}{t^{\alpha - 1}}$ 

for  $t \ge t_0$ . Put  $h(t) = \varphi(t)t^{1-\alpha}$ . Then  $|P(t)| \le h(t)$  holds and

$$\int_t^\infty h^\beta(s) \, ds = \int_t^\infty \left(\frac{\varphi(s)}{s^{\alpha-1}}\right)^\beta \, \mathrm{d}s \le \frac{\varphi^\beta(t)}{(\alpha-1)t^{\alpha-1}} = \frac{1}{\alpha-1}\varphi^{\beta-1}(t)h(t)$$

 $t \ge t_0$ . Consequently, by Proposition 3.1, (3.2) has a nonoscillatory solution of the form (3.6) on  $[t_0, \infty)$  with v satisfying (3.8). Since

$$t^{\alpha-1}v(t) = O(t^{\alpha-1}h(t)) = o(1)$$
 and  $t^{\alpha-1}P(t) = O(t^{\alpha-1}h(t)) = o(1)$ 

as  $t \to \infty$ , we conclude that y is a normalized slowly varying function. The existence of a solution  $y_2 \in NRV(1)$  follows from the proof of Theorem 3.2.

*Only if.* It follows from the only if part of the proof of Theorem 3.2  $\Box$ 

The following theorem is a generalization of the main part of Theorem 2.3. As it can be easily seen, the previous theorem can be included in its statement, but we prefer to distinguish these two results because the part dealing with the existence of a SV solution uses different ideas. In fact, in the proof of Theorem 3.2 we consider only the nonzero roots of the associated algebraic equation, since the case with the zero root is treated in the proof of the previous theorem.

Let  $\lambda_1$  and  $\lambda_2$ ,  $\lambda_1 < \lambda_2$ , denote the two real roots of the equation

$$|\lambda|^{\beta} - \lambda + A = 0. \tag{3.13}$$

It is easy to see that (3.13) has two distinct real roots if and only if  $A < \frac{1}{\alpha} \left(\frac{\alpha-1}{\alpha}\right)^{\alpha-1}$ . Clearly,  $\lambda_1 < 0 < \lambda_2$  if A < 0, and  $0 < \lambda_1 < \lambda_2$  if  $0 < A < \frac{1}{\alpha} \left(\frac{\alpha-1}{\alpha}\right)^{\alpha-1}$ . It should be noticed that  $\alpha \Phi^{-1}(\lambda_1) < \alpha - 1 < \alpha \Phi^{-1}(\lambda_2)$ .

**Theorem 3.2.** Equation (3.2) is nonoscillatory and has two solutions  $y_1$  and  $y_2$  such that  $y_1 \in NRV(\Phi^{-1}(\lambda_1))$  and  $y_2 \in NRV(\Phi^{-1}(\lambda_2))$  if and only if

$$\lim_{t \to \infty} t^{\alpha - 1} \int_{t}^{\infty} p(s) \, \mathrm{d}s = A \in \left( -\infty, \frac{1}{\alpha} \left( \frac{\alpha - 1}{\alpha} \right)^{\alpha - 1} \right). \tag{3.14}$$

*Proof.* Only if. Let  $y_i$  be solutions belonging to  $\mathcal{NRV}(\Phi^{-1}(\lambda_i))$ , i = 1, 2. From the representation theorem it follows that

$$\lim_{t \to \infty} t \frac{y'_i(t)}{y_i(t)} = \Phi^{-1}(\lambda_i), \text{ so that } \lim_{t \to \infty} \frac{y'_i(t)}{y_i(t)} = 0, \ i = 1, 2.$$
(3.15)

Put  $w_i = \Phi(y'/y)$ , i = 1, 2. Then  $w_i$  satisfies generalized Riccati equation (3.3), from which, integrating on  $[t, \infty)$  and noting that  $\lim_{t\to\infty} w_i(t) = 0$ , we have

$$t^{\alpha-1}w_i(t) = (\alpha-1)t^{\alpha-1} \int_t^\infty \frac{|s^{\alpha-1}w_i(s)|^\beta}{s^\alpha} \,\mathrm{d}s + t^{\alpha-1} \int_t^\infty p(s) \,\mathrm{d}s, \ i = 1, 2, \qquad (3.16)$$

for all sufficiently large t. Let  $t \to \infty$  in (3.16). Using (3.15), we conclude that

$$\lim_{t\to\infty}t^{\alpha-1}\int_t^\infty p(s)\,\mathrm{d}s=\lambda_i-|\lambda_i|^\beta=A,\ i=1,2.$$

If. Assume that (3.14) holds. Put  $\omega(t) = t^{\alpha-1} \int_t^{\infty} p(s) ds - A$  and consider the functions

$$y_i(t) = \exp\left\{\int_{t_i}^t \Phi^{-1}\left(\frac{\lambda_i + \omega(s) + v_i(s)}{s^{\alpha - 1}}\right) \mathrm{d}s\right\}, \quad i = 1, 2.$$
(3.17)

Then the function  $y_i$  is a solution of (3.2) on  $[t_i, \infty)$  if  $v_i$  is chosen in such a way that  $w_i = (\lambda_i + \omega + v_i)/t^{\alpha-1}$  satisfies (3.3) on  $[t_i, \infty)$ , i = 1, 2. The differential equation for  $v_i$  then reads

$$v'_{i} - \frac{\alpha - 1}{t}v_{i} + \frac{\alpha - 1}{t}\left(|\lambda_{i} + \omega(t) + v_{i}|^{\beta} - |\lambda_{i}|^{\beta}\right) = 0, \quad i = 1, 2.$$
(3.18)

We rewrite (3.18) as

$$v'_{i} + \frac{\alpha \Phi^{-1}(\lambda_{i} + \omega(t)) - (\alpha - 1)}{t} v_{i} + \frac{\alpha - 1}{t} \Big[ |\lambda_{i} + \omega(t) + v_{i}|^{\beta} - \beta \Phi^{-1}(\lambda_{i} + \omega(t))v_{i} - |\lambda_{i}|^{\beta} \Big] = 0$$

and transform it into

$$(r_i(t)v_i)' + \frac{\alpha - 1}{t}r_i(t)F_i(t, v_i) = 0, \qquad (3.19)$$

where

$$r_i(t) = \exp\left\{\int_1^t \frac{\alpha \Phi^{-1}(\lambda_i + \omega(s)) - (\alpha - 1)}{s} \,\mathrm{d}s\right\}$$

and

$$F_i(t,v) = |\lambda_i + \omega(t) + v|^\beta - \beta \Phi^{-1}(\lambda_i + \omega(t))v - |\lambda_i|^\beta, \quad i = 1, 2.$$

It is convenient to express  $F_i(t, v)$  as  $F_i(t, v) = G_i(t, v) + z_i(t)$ , with  $G_i(t, v)$  and  $z_i(t)$  defined by

$$G_i(t,v) = |\lambda_i + \omega(t) + v|^\beta - \beta \Phi^{-1}(\lambda_i + \omega(t))v - |\lambda_i + \omega(t)|^\beta$$

and  $z_i(t) = |\lambda_i + \omega(t)|^{\beta} - |\lambda_i|^{\beta}$ , i = 1, 2. Now we suppose that  $A \neq 0$  in (3.14), which implies  $\lambda_i \neq 0$  for i = 1, 2. Let  $t_0 > 0$  be such that  $|\omega(t)| \le |\lambda_i|/4$  for  $t \ge t_0$ , i = 1, 2. This is possible because  $\omega(t) \to 0$  as  $t \to \infty$  by hypothesis. It follows that  $\frac{3}{4}|\lambda_i| \le |\lambda_i + \omega(t)| \le \frac{5}{4}|\lambda_i|$  for  $t \ge t_0$ , i = 1, 2. We observe that there exist positive constants  $K_i(\alpha), L_i(\alpha)$  and  $M_i(\alpha)$  such that  $|G_i(t, v)| \le K_i(\alpha)v^2$ ,

$$\left|\frac{\partial G_i}{\partial v}(t,v)\right| \le L_i(\alpha)|v| \tag{3.20}$$

and  $|z_i(t)| \le M_i(\alpha)|\omega(t)|$  for  $t \ge t_0$  and  $|v| \le |\lambda_i|/4$ , i = 1, 2. In fact, the last two estimations follow from the Mean Value Theorem, while the estimation for  $G_i$  is a consequence of the L'Hospital rule applied to  $G_i$ :

$$\lim_{v \to 0} \frac{G_i(t,v)}{v^2} = \frac{1}{2} \lim_{v \to 0} \frac{\partial^2 G_i(t,v)}{\partial v^2} = \frac{\beta}{2(\alpha-1)} |\lambda_i + \omega(t)|^{\beta-2}.$$

Let us examine equation (3.19) with i = 1. The following properties of  $r_1$  are needed:  $r_1 \in NRV(\Phi^{-1}(\alpha) - (\alpha - 1))$ ,  $\lim_{t\to\infty} r_1(t) = 0$ ,

$$\lim_{t \to \infty} \frac{\alpha - 1}{r_1(t)} \int_t^\infty \frac{r_1(s)}{s} \, \mathrm{d}s = (\alpha - 1) \lim_{t \to \infty} \frac{-r_1(t)}{tr_1'(t)} = \frac{\alpha - 1}{\alpha - 1 - \alpha \Phi^{-1}(\lambda_1)},\tag{3.21}$$

$$\lim_{t \to \infty} \frac{\alpha - 1}{r_1(t)} \int_t^\infty \frac{r_1(s)}{s} z(s) \, \mathrm{d}s = 0 \quad \text{if} \quad z \in C[0, \infty), \text{ and } \lim_{t \to \infty} z(t) = 0.$$
(3.22)

Let  $\varepsilon_1$  be a positive constant such that  $\varepsilon_1 < \min\{1, |\lambda_1|/4\}$  and

$$\frac{2(\alpha - 1)}{\alpha - 1 - \alpha \Phi^{-1}(\lambda_1)} [K_1(\alpha) + L_1(\alpha) + M_1(\alpha)] \varepsilon_1 \le 1,$$
(3.23)

and choose  $t_1 \ge t_0$  so that

$$|\omega(t)| \le \varepsilon_1^2, \ t \ge t_1, \tag{3.24}$$

and

$$\frac{\alpha - 1}{r_1(t)} \int_t^\infty \frac{r_1(s)}{s} \, \mathrm{d}s \le \frac{2(\alpha - 1)}{\alpha - 1 - \alpha \Phi^{-1}(\lambda_1)}, \quad t \ge t_1.$$
(3.25)

Note that (3.25) is an immediate consequence of (3.21).

Let  $C_0[t_1, \infty)$  denote the set of all continuous functions on  $[t_1, \infty)$  which tend to zero as  $t \to \infty$ . Then  $C_0[t_1, \infty)$  is a Banach space with the sup-norm ||v|| = $\sup\{|v(t)| : t \ge t_1\}$ . Consider the set  $\Omega_1 \subset C_0[t_1, \infty)$  defined by  $\Omega_1 = \{v \in C_0[t_1, \infty) :$  $|v(t)| \le \varepsilon_1, t \ge t_1\}$  and define the integral operator  $\mathcal{T}_1$  by

$$(\mathcal{T}_1 v)(t) = \frac{\alpha - 1}{r_1(t)} \int_t^\infty \frac{r_1(s)}{s} F_1(s, v(s)) \,\mathrm{d}s,$$

 $t \ge t_1$ . It can be shown that  $\mathcal{T}_1$  is a contraction mapping on  $\Omega_1$ . In fact, if  $v \in \Omega_1$ , then using the above inequalities we see that

$$\begin{aligned} |(\mathcal{T}_{1}v)(t)| &\leq \frac{\alpha-1}{r_{1}(t)} \int_{t}^{\infty} \frac{r_{1}(s)}{s} (|G_{1}(s,v(s))| + |z_{1}(s)|) \, \mathrm{d}s \\ &\leq \frac{\alpha-1}{r_{1}(t)} \int_{t}^{\infty} \frac{r_{1}(s)}{s} (K_{1}(\alpha)v^{2}(s) + M_{1}(\alpha)|\omega(s)|) \, \mathrm{d}s \\ &\leq \frac{\alpha-1}{r_{1}(t)} \int_{t}^{\infty} \frac{r_{1}(s)}{s} (K_{1}(\alpha) + M_{1}(\alpha))\varepsilon_{1}^{2} \, \mathrm{d}s \\ &\leq \frac{2(\alpha-1)}{\alpha-1-\alpha\Phi^{-1}(\lambda_{1})} [K_{1}(\alpha) + M_{1}(\alpha)]\varepsilon_{1}^{2}, \end{aligned}$$

 $t \ge t_1$ . Since  $F_1(t, v(t)) \to 0$  as  $t \to \infty$ , we have  $\lim_{t\to\infty} (\mathcal{T}_1 v)(t) = 0$  by (3.22). It follows that  $\mathcal{T}_1 v \in \Omega_1$ , and so  $\mathcal{T}_1$  maps  $\Omega_1$  into itself. Furthermore, if  $u, v \in \Omega_1$ ,

then using (3.20) and (3.23), we obtain

$$\begin{aligned} |(\mathcal{T}_{1}v)(t) - (\mathcal{T}_{1}u)(t)| &\leq \frac{\alpha - 1}{r_{1}(t)} \int_{t}^{\infty} \frac{r_{1}(s)}{s} |F_{1}(s, v(s)) - F_{1}(s, u(s))| \, \mathrm{d}s \\ &= \frac{\alpha - 1}{r_{1}(t)} \int_{t}^{\infty} \frac{r_{1}(s)}{s} |G_{1}(s, v(s)) - G_{1}(s, u(s))| \, \mathrm{d}s \\ &\leq \frac{2(\alpha - 1)}{\alpha - 1 - \alpha \Phi^{-1}(\lambda_{1})} L_{1}(\alpha) \varepsilon_{1} ||v - u||, \end{aligned}$$

 $t \ge t_1$ , which implies that

$$\|\mathcal{T}_1 v - \mathcal{T}_1 u\| \le \frac{2(\alpha - 1)L_1(\alpha)}{\alpha - 1 - \alpha \Phi^{-1}(\lambda_1)} \varepsilon_1 \|v - u\|.$$

In view of (3.23), this shows that  $\mathcal{T}_1$  is a contraction mapping on  $\Omega_1$ . The contraction mapping principle then ensures the existence of a unique fixed element  $v_1 \in \Omega_1$  such that  $v_1 = \mathcal{T}_1 v_1$ , which is equivalent to the integral equation

$$v_1(t) = \frac{\alpha - 1}{r_1(t)} \int_t^\infty \frac{r_1(s)}{s} F_1(s, v_1(s)) \,\mathrm{d}s, \tag{3.26}$$

 $t \ge t_1$ . Differentiation of (3.26) shows that  $v_1$  satisfies differential equation (3.19) with i = 1 on  $[t_1, \infty)$  and so substitution of this  $v_1$  into (3.17) gives rise to a solution  $y_1$  of half-linear equation (3.2) defined on  $[t_1, \infty)$ . Since  $\lim_{t\to\infty} v_1(t) = 0$ , from the representation theorem we have  $y_1 \in NRV(\Phi^{-1}(\lambda_1))$ .

Our next task is to solve equation (3.19) for i = 2 in order to construct a larger solution  $y_2$  of (3.2) via formula (3.17). It is easy to see that  $r_2 \in NRV(\alpha \Phi^{-1}(\lambda_2) - \alpha + 1)$ ,  $\lim_{t\to\infty} r_2(t) = \infty$  and that for any fixed  $t_2 > 0$ ,

$$\lim_{t \to \infty} \frac{\alpha - 1}{r_2(t)} \int_{t_2}^t \frac{r_2(s)}{s} \, \mathrm{d}s = (\alpha - 1) \lim_{t \to \infty} \frac{r_2(t)}{tr'_2(t)} = \frac{\alpha - 1}{\alpha \Phi^{-1}(\lambda_2) - \alpha + 1},$$
$$\lim_{t \to \infty} \frac{\alpha - 1}{r_2(t)} \int_{t_2}^t \frac{r_2(s)z(s)}{s} \, \mathrm{d}s = 0 \quad \text{if} \ z \in C[t_2, \infty) \ \text{and} \ \lim_{t \to \infty} z(t) = 0.$$

Let  $\varepsilon_2 > 0$  be small enough so that

$$\frac{2(\alpha-1)}{\alpha\Phi^{-1}(\lambda_2)-\alpha+1}[K_2(\alpha)+L_2(\alpha)+M_2(\alpha)]\varepsilon_2 \le 1,$$

and choose  $t_2 > 0$  so large that  $\omega(t) \le \varepsilon_2^2$ ,  $t \ge t_2$ , and

$$\frac{\alpha - 1}{r_2(t)} \int_{t_2}^t \frac{r_2(s)}{2} \, \mathrm{d}s \le \frac{2(\alpha - 1)}{\alpha \Phi^{-1}(\lambda_2) - \alpha + 1},$$

 $t \ge t_2$ . Define the set  $\Omega_2 \subset C_0[t_2, \infty)$  and the integral operator  $\mathcal{T}_2$  by  $\Omega_2 = \{v \in C_0[t_2, \infty) : |v(t)| \le \varepsilon_2, t \ge t_2\}$ , and

$$(\mathcal{T}_2 v)(t) = -\frac{\alpha - 1}{r_2(t)} \int_{t_2}^t \frac{r_2(s)}{s} F_2(s, v(s)) \, \mathrm{d}s,$$

 $t \ge t_2$ . It is a matter of easy calculation to verify that  $\mathcal{T}_2$  is a contraction mapping on  $\Omega_2$ . Therefore there exists a unique fixed element  $v_2 \in \Omega_2$  of  $\mathcal{T}_2$ , which satisfies the integral equation

$$v_2(t) = -\frac{\alpha - 1}{r_2(t)} \int_{t_2}^t \frac{r_2(s)}{s} F(s, v_2(s)) \, \mathrm{d}s,$$

 $t \ge t_2$ , and hence differential equation (3.19) with i = 2. Then the function  $y_2$  defined by (3.17) with this  $v_2$  is a nonoscillatory solution of (3.2) on  $[t_2, \infty)$ . The fact that  $y_2 \in N\mathcal{RV}(\Phi^{-1}(\lambda_2))$  follows from the representation theorem. This finishes the proof of the "if" part of the theorem for the case  $A \ne 0$ . If A = 0, then equation (3.13) has the two real roots  $\lambda_1 = 0$ ,  $\lambda_2 = 1$ . The solution  $y_1 \in N\mathcal{RV}(0) = NS\mathcal{V}$  of (3.2) corresponding to  $\lambda_1$  has already been constructed in Theorem 3.1. The existence of the solution  $y_2 \in N\mathcal{RV}(1)$  corresponding to  $\lambda_2$  can be proved in exactly the same manner as developed for the case  $A \ne 0$ .

#### 3.2.3 $\mathcal{RV}$ solutions in the border case

Let us consider equation (3.2) for which the condition

$$\lim_{t \to \infty} t^{\alpha - 1} \int_{t}^{\infty} p(s) \, \mathrm{d}s = \frac{1}{\alpha} \left( \frac{\alpha - 1}{\alpha} \right)^{\alpha - 1} \tag{3.27}$$

is satisfied. Such an equation can be regarded as a perturbation of the generalized Euler equation

$$(\Phi(y'))' + \frac{\gamma}{t^{\alpha}} \Phi(y) = 0$$
(3.28)

with  $\gamma = \overline{\gamma} = \left(\frac{\alpha-1}{\alpha}\right)^{\alpha}$ . Although (3.28) is nonoscillatory, because it has a solution  $y(t) = t^{(\alpha-1)/\alpha}$ , its perturbation may be oscillatory or nonoscillatory depending on the asymptotic behavior of the perturbed term as  $t \to \infty$ , see Došlý, Řehák [28]. Our purpose here is to show the existence of a class of perturbations which preserve the nonoscillation character of (3.28). The result can be seen as a generalization of Theorem 2.4.

**Theorem 3.3.** Suppose that (3.27) holds. Put

$$\Upsilon(t) = t^{\alpha - 1} \int_{t}^{\infty} p(s) \, \mathrm{d}s - \frac{1}{\alpha} \left(\frac{\alpha - 1}{\alpha}\right)^{\alpha - 1}$$

and suppose that

$$\int_{t}^{\infty} \frac{|\Upsilon(t)|}{t} \, \mathrm{d}t < \infty \tag{3.29}$$

and

$$\int_{t}^{\infty} \frac{\Psi(t)}{t} \, \mathrm{d}t < \infty, \quad \text{where } \Psi(t) = \int_{t}^{\infty} \frac{|\Upsilon(s)|}{s} \, \mathrm{d}s. \tag{3.30}$$

Then equation (3.2) is nonoscillatory and has a normalized regularly varying solution with index  $(\alpha - 1)/\alpha$  of the form  $y(t) = t^{(\alpha-1)/\alpha}L(t)$  with  $L \in NSV$  and  $\lim_{t\to\infty} L(t) = \ell \in (0, \infty)$ .

*Proof.* The solution is sought in the form

$$y(t) = \exp \int_{T}^{t} \Phi^{-1} \left( \frac{\bar{\gamma} + \Upsilon(s) + v(s)}{s^{\alpha - 1}} \right) \mathrm{d}s, \quad \bar{\gamma} = \left( \frac{\alpha - 1}{\alpha} \right)^{\alpha}, \tag{3.31}$$

for some T > 0 and  $v : [T, \infty) \to \mathbb{R}$ . The same argument as in the proof of the "if" part of the previous theorem leads to the differential equation for v

$$(r(t)v)' + \frac{\alpha - 1}{t}r(t)F(t, v) = 0, \qquad (3.32)$$

where

$$r(t) = \exp\left(\int_{1}^{t} \frac{\alpha \Phi^{-1}(\bar{\gamma} + \Upsilon(s)) - \alpha + 1}{s} \,\mathrm{d}s\right)$$

and

$$F(t,v) = |\bar{\gamma} + \Upsilon(t) + v|^{\beta} - \beta \Phi^{-1}(\bar{\gamma} + \Upsilon(t))v - \bar{\gamma}^{\beta}.$$
(3.33)

Choose  $t_0 > 0$  so that

$$|\Upsilon(t)| \le \frac{\bar{\gamma}}{4},\tag{3.34}$$

 $t \ge t_0$ . Since

$$|\Phi^{-1}(\bar{\gamma} + \Upsilon(t)) - \alpha + 1| = \alpha |(\bar{\gamma} + \Upsilon(t))^{\beta - 1} - \bar{\gamma}^{\beta - 1}| \le \alpha m(\alpha) |\Upsilon(t)|,$$

 $t \ge t_0$ , for some constant  $m(\alpha) > 0$ , we see in view of (3.29) that *r* is a slowly varying function and tends to a finite positive limit as  $t \to \infty$ . It follows that there exists  $t_1 \ge t_0$  such that

$$r(s)/r(t) \le 2 \tag{3.35}$$

for  $s \ge t \ge t_1$ . We rewrite the function F(t, v) defined by (3.33) as F(t, v) = G(t, v) + z(t), where

$$G(t,v) = |\bar{\gamma} + \Upsilon(t) + v|^{\beta} - \beta \Phi^{-1}(\bar{\gamma} + \Upsilon(t))v - |\bar{\gamma} + \Upsilon(t)|^{\beta}$$

and  $z(t) = |\bar{\gamma} + \Upsilon(t)|^{\beta} - \bar{\gamma}^{\beta}$ . As it is easily seen, there exist positive constants  $K(\alpha)$ ,  $L(\alpha)$  and  $M(\alpha)$  such that

$$|G(t,v)| \le K(\alpha)v^2, \tag{3.36}$$

$$\left|\frac{\partial G(t,v)}{\partial v}\right| \le L(\alpha)|v| \tag{3.37}$$

and  $|h(t)| \le M(\alpha)|\Upsilon(t)|$  for  $t \ge t_1$  and  $|v| \le \overline{\gamma}/4$ . Let  $T > t_1$  be large enough so that

$$4(p-1)M(\alpha)\Psi(t) \le \frac{\bar{\gamma}}{4},\tag{3.38}$$

 $t \geq T$ ,

$$16(\alpha-1)^2 K(\alpha) M(\alpha) \int_T^\infty \frac{\Psi(s)}{s} \, \mathrm{d}s \le 1,$$

and

$$16(\alpha - 1)^2 L(\alpha) M(\alpha) \int_T^\infty \frac{\Psi(s)}{s} \, ds \le 1.$$
(3.39)

We want to solve the integral equation

$$v(t) = \frac{\alpha - 1}{r(t)} \int_t^\infty \frac{r(s)}{s} F(s, v(s)) \,\mathrm{d}s, \qquad (3.40)$$

 $t \ge T$ , which follows from (3.32), subject to the condition  $\lim_{t\to\infty} v(t) = 0$ . Let  $C_{\Psi}[T, \infty)$  denote the set of all continuous functions v on  $[T, \infty)$  such that

$$\|v\|_{\Psi} = \sup_{t \ge T} \frac{|v(t)|}{\Psi(t)} < \infty.$$

Clearly,  $C_{\Psi}[T, \infty)$  is a Banach space equipped with the norm  $||v||_{\Psi}$ . Consider the set  $\Omega \subset C_{\Psi}[T, \infty)$  and the mapping  $\mathcal{T}\Omega \to C_{\Psi}[T, \infty)$  defined by

$$\Omega = \{ v \in C_{\Psi}[T, \infty) : |v(t)| \le 4(\alpha - 1)M(\alpha)\Psi(t), t \ge T \}$$
(3.41)

and

$$\mathcal{T}v(t) = \frac{\alpha - 1}{r(t)} \int_t^\infty \frac{r(s)}{s} F(s, v(s)) \,\mathrm{d}s = \frac{\alpha - 1}{r(t)} \int_t^\infty \frac{r(s)}{s} [G(s, v(s)) + z(s)] \,\mathrm{d}s,$$

 $t \ge T$ . Using (3.35), (3.36) and (3.37), we see that

$$\frac{\alpha-1}{r(t)}\int_t^\infty \frac{r(s)}{s}|z(s)|\,\mathrm{d} s \le 2(\alpha-1)\int_t^\infty \frac{M(\alpha)|\Upsilon(s)|}{s}\,\mathrm{d} s = 2(\alpha-1)M(\alpha)\Psi(t),$$

 $t \ge T$ , and that

$$\begin{split} \frac{\alpha-1}{r(t)} \int_t^\infty \frac{r(s)}{s} |G(s,v(s))| \, \mathrm{d}s &\leq 2(\alpha-1) \int_t^\infty \frac{K(\alpha)[4(\alpha-1)M(\alpha)\Psi(s)]^2}{s} \, \mathrm{d}s \\ &= 32(\alpha-1)^3 K(\alpha) M^2(\alpha) \int_t^\infty \frac{\Psi^2(s)}{s} \, \mathrm{d}s \\ &\leq 32(\alpha-1)^3 K(\alpha) M^2(\alpha) \Psi(t) \int_t^\infty \frac{\Psi(s)}{s} \, \mathrm{d}s \\ &\leq 32(\alpha-1)^3 K(\alpha) M^2(\alpha) \Psi(t) \int_T^\infty \frac{\Psi(s)}{s} \, \mathrm{d}s \leq 2(\alpha-1) M(\alpha) \Psi(t), \end{split}$$

 $t \ge T$ . This shows that  $v \in \Omega$  implies  $\mathcal{T}v \in \Omega$ , and hence  $\mathcal{T}$  maps  $\Omega$  into itself. If  $u, v \in \Omega$ , then using (3.37) we have

$$\begin{aligned} |\mathcal{T}v(t) - \mathcal{T}u(t)| &\leq \frac{\alpha - 1}{r(t)} \int_{t}^{\infty} \frac{r(s)}{s} |G(s, v(s)) - G(s, u(s))| \, \mathrm{d}s \\ &\leq 2(\alpha - 1) \int_{t}^{\infty} \frac{4(\alpha - 1)L(\alpha)M(\alpha)\Psi(s)|v(s) - u(s)|}{s} \, \mathrm{d}s \\ &= 8(\alpha - 1)^{2}L(\alpha)M(\alpha) \int_{t}^{\infty} \frac{\Psi^{2}(s)|v(s) - u(s)|}{s\Psi(s)} \, \mathrm{d}s \\ &\leq 8(\alpha - 1)^{2}L(\alpha)M(\alpha)\Psi(t)||v - u||_{\Psi} \int_{t}^{\infty} \frac{\Psi(s)}{s} \, \mathrm{d}s, \end{aligned}$$

 $t \ge T$ , from which, in view of (3.39), we conclude that  $\mathcal{T}$  is a contraction mapping:  $\|\mathcal{T}v - \mathcal{T}u\|_{\Psi} \le \|v - u\|_{\Psi}/2$ . Let  $v \in \Omega$  be a unique fixed element of  $\mathcal{T}$ . Then v satisfies (3.40), and hence (3.32), on  $[T, \infty)$ , and the function y defined by (3.31) provides a nonoscillatory solution of (3.2) on  $[T, \infty)$ . Since  $\Phi^{-1}(\bar{\gamma} + \Upsilon(t) + v(t)) \to (\alpha - 1)/p$  as  $t \to \infty, y \in \mathcal{NRV}((\alpha - 1)/\alpha)$ , and y is expressed as  $y(t) = t^{(\alpha - 1)/\alpha}L(t)$ , where

$$L(t) = \exp\left\{\int_1^t \frac{\Phi^{-1}(\bar{\gamma} + \Upsilon(s) + v(s)) - \bar{\gamma}^{\beta - 1}}{s} \,\mathrm{d}s\right\}.$$

 $t \ge T$ . Noting that  $|\Upsilon(t) + v(t)| \le \overline{\gamma}/2$ ,  $t \ge T$ , by (3.34) and (3.38), and applying the Mean Value Theorem, we see with the use of (3.41) that  $|\Phi^{-1}(\overline{\gamma} + \Upsilon(t) + v(t)) - \overline{\gamma}^{\beta-1}| \le N(\alpha)(|\Upsilon(t)| + |\Psi(t)|)$ ,  $t \ge T$ , for some constant  $N(\alpha) > 0$ . This, combined with the hypotheses (3.29) and (3.30), guarantees that L(t) tends to a finite positive limit as  $t \to \infty$ .

## 3.2.4 $\mathcal{RB}$ solutions

In the next theorem we somehow relax the condition on the existence of the limit in (3.11) and (3.14), and the existence of a  $\mathcal{RB}$  solution follows; an important role in the proof is played by Proposition 3.1. The result generalizes Theorem 2.5. But observe that here we have guaranteed the existence of (at least) one  $\mathcal{RB}$  solution, while in the linear case all eventually positive solutions are  $\mathcal{RB}$ . As already indicated above, the reason is that here we cannot use a reduction of order formula and the fact that any solution is a linear combination of elements of the fundamental set.

Theorem 3.4. If

$$\begin{aligned} &-\frac{1}{\alpha} \left(\frac{\alpha-1}{\alpha}\right)^{\alpha-1} < \liminf_{t \to \infty} t^{\alpha-1} \int_t^\infty p(s) \, \mathrm{d}s \\ &\leq \limsup_{t \to \infty} t^{\alpha-1} \int_t^\infty p(s) \, \mathrm{d}s < \frac{1}{\alpha} \left(\frac{\alpha-1}{\alpha}\right)^{\alpha-1} \end{aligned}$$

*holds, then equation* (3.2) *is nonoscillatory and has a normalized regularly bounded solution.*  *Proof.* If the assumption holds, then there exist positive constants  $t_0$  and a satisfying (3.5) such that  $|t^{\alpha-1}P(t)| \le a$ ,  $t \ge t_0$ . Put  $h(t) = at^{1-\alpha}$ . Then it can be easily verified that h satisfies  $|P(t)| \le h(t)$  and (3.4). So, by Proposition 3.1, (3.2) has a nonoscillatory solution of the form

$$y(t) = \exp\left\{\int_{t_0}^t \Phi^{-1}(v(s) + P(s)) \,\mathrm{d}s\right\}$$
  
=  $\exp\left\{\int_{t_0}^t \frac{\Phi^{-1}[s^{\alpha-1}(v(s) + P(s))]}{s} \,\mathrm{d}s\right\},$  (3.42)

 $t \ge t_0$ , with *v* satisfying (3.8). Since

$$\left| \Phi^{-1}[t^{\alpha - 1}(v(t) + \sigma(t))] \right| \le \alpha^{\beta - 1}(t^{\alpha - 1}h(t))^{\beta - 1} = (a\alpha)^{\beta - 1},$$

 $t \ge t_0$ , the solution *y* is a normalized  $\mathcal{RB}$  function by the representation theorem.  $\Box$ 

## 3.3 More precise information about *SV* solutions

The observations in this section are based on the paper [99] by Kusano, Marić, Tanigawa. We will need the statement of Proposition 3.1, with a slight modification, namely that condition (3.4) is replaced by the more general one:

$$\int_t^\infty h^\beta(s)\,\mathrm{d} s \leq \frac{1}{\alpha-1}a^{\beta-1}(t)h(t), \ t\geq t_0,$$

where a(t) is a continuous nonincreasing function satisfying

$$0 < a(t) \le a < \frac{1}{\alpha} \left(\frac{\alpha - 1}{\alpha}\right)^{\alpha - 1}$$

for some constant *a*. The proof of such modified proposition is almost the same as that of the original one, and so it is omitted.

**Theorem 3.5.** Suppose that the hypotheses of Proposition 3.1 with the above modification are satisfied. Let there exist a positive integer n such that

$$\int_{0}^{\infty} a^{n(\beta-1)}(t)h^{\beta-1}(t) \, \mathrm{d}t < \infty \quad if \quad 1 < \alpha \le 2,$$
(3.43)

$$\int^{\infty} a^{n(\beta-1)^2}(t)h^{\beta-1}(t)\,\mathrm{d}t < \infty \quad if \quad \alpha > 2. \tag{3.44}$$

Then, for the solution (3.6) of (3.2), the following asymptotic formula holds for  $t \to \infty$ 

$$y(t) \sim B \exp\left\{\int_{t_0}^t \Phi^{-1}[v_{n-1}(s) + P(s)] \,\mathrm{d}s\right\},$$
 (3.45)

where B is a positive constant. Here the sequence  $\{v_n(t)\}$  of successive approximations is defined by

$$v_0(t) = 0, \ v_n(t) = (\alpha - 1) \int_t^\infty |v_{n-1}(s) + P(s)|^\beta \, \mathrm{d}s, \ n = 1, 2, \dots$$
 (3.46)

*Proof.* Let *y* be the solution given by (3.6). Recall that the function *v* used in (3.6) has been constructed as the fixed element in  $C_h[t_0, \infty)$  of the contractive mapping  $\mathcal{T}$  defined by (3.9). The standard proof of the contraction mapping principle shows that the sequence  $\{v_n(t)\}$  defined by (3.46) converges to v(t) uniformly on  $[t_0, \infty)$ . To see how fast  $v_n(t)$  approaches v(t) we proceed as follows. First, note that  $|v_n(t)| \leq (\alpha - 1)h(t), t \geq t_0, n = 1, 2, \ldots$ . By definition, we have

$$|v_1(t)| = (\alpha - 1) \int_t^\infty |P(s)|^\beta \, \mathrm{d}s \le (\alpha - 1) \int_t^\infty h^\beta(s) \, \mathrm{d}s \le a^{\beta - 1}(t)h(t),$$

and

$$\begin{aligned} |v_{2}(t) - v_{1}(t)| &\leq (\alpha - 1) \int_{t}^{\infty} \left| |v_{1}(s) + P(s)|^{\beta} - |P(s)|^{\beta} \right| \mathrm{d}s \\ &\leq (\alpha - 1)\beta \int_{t}^{\infty} [\alpha h(s)]^{\beta} |v_{1}(s)| \,\mathrm{d}s \\ &\leq \alpha^{\beta} \int_{t}^{\infty} a^{\beta - 1}(s)h^{\beta}(s) \,\mathrm{d}s \leq \alpha^{\beta} a^{\beta - 1}(t) \int_{t}^{\infty} h^{\beta}(s) \,\mathrm{d}s \\ &\leq (\beta - 1)\alpha^{\beta} a^{2(\beta - 1)}(t)h(t) \leq \gamma_{\alpha}^{\beta - 1} \left(\frac{a(t)}{\gamma_{\alpha}}\right)^{2(\beta - 1)} h(t) \end{aligned}$$

for  $t \ge t_0$ , where  $\gamma_{\alpha} = \frac{1}{\alpha} \left(\frac{\alpha-1}{\alpha}\right)^{\alpha-1}$ . Assuming that

$$|v_n(t) - v_{n-1}(t)| \le \gamma_\alpha^\beta \left(\frac{a(t)}{\gamma_\alpha}\right)^{n(\beta-1)} h(t), \tag{3.47}$$

 $t \ge t_0$ , for some  $n \in \mathbb{N}$ , we compute

$$\begin{aligned} |v_{n+1}(t) - v_n(t)| &\leq (\alpha - 1) \int_t^{\infty} \left| |P(s) + v_n(s)|^{\beta} - |P(s) + v_{n-1}(s)|^{\beta} \right| \mathrm{d}s \\ &\leq (\alpha - 1)\beta \int_t^{\infty} [\alpha h(s)]^{\beta - 1} |v_n(s) - v_{n-1}(s)| \,\mathrm{d}s \\ &= \alpha^{\beta} \int_t^{\infty} \gamma_{\alpha}^{\beta - 1} \left(\frac{a(s)}{\gamma_{\alpha}}\right)^{n(\beta - 1)} h^{\beta}(s) \,\mathrm{d}s \\ &\leq \alpha^{\beta} \gamma_{\alpha}^{\beta - 1} \left(\frac{a(t)}{\gamma_{\alpha}}\right)^{n(\beta - 1)} \int_t^{\infty} h^{\beta}(s) \,\mathrm{d}s \\ &\leq \alpha^{\beta} \gamma_{\alpha}^{\beta - 1} \left(\frac{a(t)}{\gamma_{\alpha}}\right)^{n(\beta - 1)} (\beta - 1)a^{\beta - 1}(t)h(t) \\ &= \gamma_{\alpha}^{\beta - 1} \left(\frac{a(t)}{\gamma_{\alpha}}\right)^{n(\beta - 1)} h(t), \end{aligned}$$

 $t \ge t_0$ , which establishes the validity of (3.47) for all integers  $n \in \mathbb{N}$ . Now we have

$$v(t) = v_{n-1}(t) + \sum_{k=n}^{\infty} [v_k(t) - v_{k-1}(t)],$$

from which, due to (3.47), it follows that

$$\begin{aligned} |v(t) - v_{n-1}(t)| &\leq \gamma_{\alpha}^{\beta-1} \left(\frac{a(t)}{\gamma_{\alpha}}\right)^{k(\beta-1)} h(t) \\ &\leq \gamma_{\alpha}^{\beta-1} \left(\frac{a(t)}{\gamma_{\alpha}}\right)^{n(\beta-1)} \sum_{k=0}^{\infty} \left(\frac{a}{\gamma_{\alpha}}\right)^{k} h(t) \end{aligned} (3.48) \\ &= \gamma_{\alpha} \left(\frac{a(t)}{\gamma_{\alpha}}\right)^{n(\beta-1)} \frac{\gamma_{\alpha}}{\gamma_{\alpha} - a} h(t) = Ka^{n(\beta-1)}(t)h(t), \end{aligned}$$

where *K* is a constant depending only on  $\alpha$  and *n*. Using (3.6) and (3.48), we obtain

$$y(t)\left(\exp\left\{\int_{t_0}^t \Phi^{-1}[P(s) + v_{n-1}(s)]\,\mathrm{d}s\right\}\right)^{-1} = \exp\left\{\int_{t_0}^t \left(\Phi^{-1}[P(s) + v(s)] - \Phi^{-1}[P(s) + v_{n-1}(s)]\right)\,\mathrm{d}s\right\}.$$
 (3.49)

Let  $1 < \alpha \leq 2$ . Then, by the Mean Value Theorem and (3.48),

$$\begin{aligned} \left| \Phi^{-1}[P(t) + v(t)] - \Phi^{-1}[P(t) + v_{n-1}(t)] \right| &\leq (\beta - 1)[\alpha h(t)]^{\beta - 2} |v(t) - v_{n-1}(t)| \\ &\leq N a^{n(\beta - 1)}(t) h^{\beta - 1}(t), \end{aligned}$$
(3.50)

 $t \ge t_0$ , where *N* is a constant depending on  $\alpha$  and *n*. Let  $\alpha > 2$ . Then, using (3.48) and the inequality  $|a^{\lambda} - b^{\lambda}| \le 2|a - b|^{\lambda}$  holding for  $\lambda \in (0, 1)$  and  $a, b \in \mathbb{R}$ , we see that

$$\begin{aligned} \left| \Phi^{-1}[P(t) + v(t)] - \Phi^{-1}[P(t) + v_{n-1}(t)] \right| &\leq 2|v(t) - v_{n-1}(t)|^{\beta - 1} \\ &\leq Ma^{n(\beta - 1)^2}(t)h^{\beta - 1}(t), \end{aligned} (3.51)$$

 $t \ge t_0$ , where *M* is a constant depending on  $\alpha$  and *n*. Combining (3.49) with (3.50) or (3.51) according as  $1 < \alpha \le 2$  or  $\alpha > 2$ , and using (3.43) or (3.44), we conclude that the right-hand side of (3.49) tends to a constant B > 0 as  $t \to \infty$ , which implies that y(t) has the desired asymptotic behavior (3.45).

**Corollary 3.1.** Suppose that (3.11) holds and that the function  $a(t) = \varphi(t)$ , where  $\varphi(t)$  is defined by (3.12), satisfies

$$\int^{\infty} \frac{a^{(n+1)(\beta-1)}(t)}{t} \, \mathrm{d}t < \infty \quad if \quad 1 < \alpha \le 2,$$
(3.52)

$$\int_{0}^{\infty} \frac{a^{(n+\alpha-1)(\beta-1)^2}}{t} \, \mathrm{d}t < \infty \quad if \quad \alpha > 2.$$
(3.53)

Then the formula (3.45) holds for the slowly varying solution y(t) of (3.2).

*Proof.* The conclusion follows from the previous theorem combined with the observation that in this case

$$a^{n(\beta-1)}(t)h^{\beta-1}(t) = \frac{a^{(n+1)(\beta-1)}(t)}{t}$$
 and  $a^{n(\beta-1)^2}(t)h^{\beta-1}(t) = \frac{a^{(n+\alpha-1)(\beta-1)^2}(t)}{t}$ 

according to whether  $1 < \alpha \le 2$  or  $\alpha > 2$ .

## 3.4 More general case

It is a natural question whether a generalization of (some of) the results from Section 3.2 to equation (3.1) is possible.

First idea which can come into mind is a transformation into equation of the form (3.2). Unfortunately, in contrast to the linear case we do not have a (linear) transformation of dependent variable y = hu at disposal. Thus it is impossible to "kill" the coefficient r in general. Equation (3.1) can be transformed into an equation of the form (3.2) with preserving unboundedness and the form of the interval only when  $\int_a^{\infty} r^{1-\beta}(s) ds = \infty$ . Indeed, we introduce new independent variable by  $s = \xi(t) = \int_a^t r^{1-\beta}(s) ds$  and new function x(s) = y(t). Then (3.1) is transformed into

$$\frac{\mathrm{d}}{\mathrm{d}s}\left(r(\xi^{-1}(s))\Phi(\xi'(\xi^{-1}(s)))\Phi\left(\frac{\mathrm{d}}{\mathrm{d}s}x\right)\right) + \frac{p(\xi^{-1}(s))}{\xi'(\xi^{-1}(s))}\Phi(x) = 0,$$

where  $\xi^{-1}$  is the inverse of  $\xi$ , or

$$\frac{\mathrm{d}}{\mathrm{d}s} \left( \Phi\left(\frac{\mathrm{d}}{\mathrm{d}s}x\right) \right) + \frac{p(\xi^{-1}(s))}{r^{1-\beta}(\xi^{-1}(s))} \Phi(x) = 0.$$

Jaroš, Kusano, and Tanigawa in [66] used a different approach, based on the same idea as for the linear equation in Theorems 2.8 and 2.9, namely utilizing the concept of generalized varying functions (see Definition 1.10) with the combination of the approach described in Section 3.2. The two (logical) cases are distinguished:

$$\int_{a}^{\infty} r^{1-\beta}(s) \, \mathrm{d}s = \infty \tag{3.54}$$

and

$$\int_{a}^{\infty} r^{1-\beta}(s) \,\mathrm{d}s < \infty. \tag{3.55}$$

We denote  $R_{\alpha}(t) = \int_{a}^{t} r^{1-\beta}(s) ds$  in case (3.54) and  $\tilde{R}_{\alpha}(t) = \int_{t}^{\infty} r^{1-\beta}(s) ds$  in case (3.55).

Let  $\lambda_1 < \lambda_2$  denote the two real roots of the equation (3.13). A generalization of Theorem 3.1 and of Theorem 3.2 reads as follows; at the same time it can be viewed as a generalization of Theorem 2.8-(i).

**Theorem 3.6.** Let (3.54) hold. Equation (3.1) is nonoscillatory and has two solutions  $y_1, y_2$  such that  $y_i(t) \in NRV_{R_\alpha}(\Phi^{-1}(\lambda_i)), i = 1, 2, if and only if$ 

$$\lim_{t \to \infty} R_{\alpha}^{\alpha - 1}(t) \int_{t}^{\infty} p(s) \, \mathrm{d}s = A \in \left(-\infty, \frac{1}{\alpha} \left(\frac{\alpha - 1}{\alpha}\right)^{\alpha - 1}\right). \tag{3.56}$$

*Proof.* Since the proof uses essentially the same ideas as the proof of Theorem 3.1 and of Theorem 3.2, we mention only a few facts. Denote  $\omega(t) = R_{\alpha}^{\alpha-1}(t) \int_{t}^{\infty} p(s) ds - A$ . The solutions  $y_i$ , when  $\lambda_1 \neq 0$ , are sought in the form

$$y_i(t) = \exp\left\{\int_{t_i}^t \Phi^{-1}\left(\frac{\lambda_i + \omega(s) + v_i(s)}{r(s)R_\alpha^{\alpha-1}(s)}\right) \mathrm{d}s\right\},\,$$

*i* = 1,2. The function  $v_i$  is chosen in such a way that  $w_i = (\lambda_i + \omega(t) + v_i)/R^{\alpha-1}$  satisfies the generalized Riccati equation

$$w' + p(t) + (\alpha - 1)r^{1-\beta}(t)|w|^{\beta} = 0.$$
(3.57)

To find the desired  $v_i$ , the contraction mapping theorem is used.

In the case when  $\lambda_1 = 0$  (i.e., A = 0), the solution  $y_1$  is sought in the form

$$y_1(t) = \exp\left\{\int_{t_1}^t \Phi^{-1}\left(\frac{R_\alpha^{\alpha-1}(s)\int_s^\infty p(\tau)\,\mathrm{d}\tau + w_1(s)}{r(s)R_\alpha^{\alpha-1}(s)}\right)\,\mathrm{d}s\right\},\,$$

where the differential equation for  $w_1$  has the form

$$\left(\frac{w_1}{R_{\alpha}^{\alpha-1}(s)}\right)' + \frac{(\alpha-1)\left|R_{\alpha}^{\alpha-1}(t)\int_t^{\infty} p(\tau)\,\mathrm{d}\tau + w_1(s)\right|^{\beta}}{r^{\beta-1}(s)R_{\alpha}^{\alpha}(s)} = 0.$$

The next result is a generalization of Theorem 3.3 in case (3.54); at the same time it generalizes Theorem 2.8-(ii). We assume

$$\lim_{t \to \infty} R^{\alpha - 1}(t) \int_{t}^{\infty} p(s) \, \mathrm{d}s = \frac{1}{\alpha} \left(\frac{\alpha - 1}{\alpha}\right)^{\alpha - 1}.$$
(3.58)

Equation (3.1) with p satisfying this condition can be regarded as a perturbation of the generalized Euler equation

$$(r(t)\Phi(y'))' + \frac{\bar{\gamma}}{r^{\beta-1}(t)R^{\alpha}_{\alpha}(t)}\Phi(y) = 0, \quad \bar{\gamma} = \left(\frac{\alpha-1}{\alpha}\right)^{\alpha},$$

which is nonoscillatory. We show that (3.1) has a solution in  $NRV_{R_{\alpha}}$  provided the perturbation is small in some sense.

Theorem 3.7. Suppose that (3.54) and (3.58) hold. Put

$$\Upsilon(t) = R_{\alpha}^{\alpha-1}(t) \int_{t}^{\infty} p(s) \, \mathrm{d}s - \frac{1}{\alpha} \left(\frac{\alpha-1}{\alpha}\right)^{\alpha-1}$$

and suppose that

$$\int^{\infty} \frac{|\Upsilon(t)|}{r^{\beta-1}(t)R_{\alpha}(t)} \,\mathrm{d}t < \infty$$

and

$$\int_{0}^{\infty} \frac{\Psi(t)}{r^{\beta-1}(t)R_{\alpha}(t)} \, \mathrm{d}t < \infty, \quad where \quad \Psi(t) = \int_{t}^{\infty} \frac{|\Upsilon(s)|}{r^{\beta-1}(s)R_{\alpha}(s)} \, \mathrm{d}s.$$

Then equation (3.1) is nonoscillatory and has a solution  $y \in N\mathcal{RV}_{R_{\alpha}}((\alpha - 1)/\alpha)$  of the form  $y(t) = R_{\alpha}^{(\alpha-1)/\alpha}(t)L(t)$  with  $L \in NSV_{R_{\alpha}}$  and  $\lim_{t\to\infty} L(t) = \ell \in (0,\infty)$ .

*Proof.* Similarly as in the previous proof we omit details since it uses similar arguments as the proof of Theorem 3.3. Note just that we look for a solution of (3.1) expressed in the form

$$y(t) = \exp\left\{\int_{t_i}^t \Phi^{-1}\left(\frac{\bar{\gamma} + \Upsilon(s) + v(s)}{r(s)R_\alpha^{\alpha-1}(s)}\right) \mathrm{d}s\right\},\,$$

where v is desired to satisfy a certain Riccati type equation; the existence of v is proved by means of the contraction mapping theorem.

We now turn to the case where r in (3.1) satisfies (3.55). We give a halflinear extension of Theorem 2.9-(i), which is at the same time a counterpart of Theorem 3.6.

Let  $\sigma_1 < \sigma_2$  denote the two real roots of the equation  $|\sigma|^{\beta} + \sigma + B = 0$ .

**Theorem 3.8.** Let (3.55) hold. Equation (3.1) is nonoscillatory and has two solutions  $y_1, y_2$  such that  $y_i(t) \in NRV_{1/\tilde{R}_{\sigma}}(\Phi^{-1}(\sigma_i)), i = 1, 2, if and only if$ 

$$\lim_{t \to \infty} \frac{1}{\tilde{R}_{\alpha}(t)} \int_{t}^{\infty} \tilde{R}_{\alpha}^{\alpha}(s) p(s) \, \mathrm{d}s = B \in \left(-\infty, \left(\frac{\alpha - 1}{\alpha}\right)^{\alpha}\right)$$

*Proof.* The proof uses similar ideas as the proof of Theorem 3.7. Therefore we only mention a few facts. The solutions  $y_i$ , i = 1, 2, when  $\sigma_2 \neq 0$ , are sought in the form

$$y_i(t) = \exp\left\{\int_{t_i}^t \Phi^{-1}\left(\frac{\sigma_i + \omega(s) + v_i(s)}{r(s)\tilde{R}_{\alpha}^{\alpha-1}(s)}\right) \mathrm{d}s\right\},\,$$

i = 1, 2, where  $\omega(t) = \frac{1}{\tilde{R}_{\alpha}(t)} \int_{t}^{\infty} \tilde{R}_{\alpha}^{\alpha}(s)p(s) ds - B$ . The function  $v_i$  is chosen in such a way that  $w_i = (\sigma_i + \omega(t) + v_i)/\tilde{R}^{\alpha-1}$  satisfies the generalized Riccati equation (3.57). To find the desired  $v_i$ , the contraction mapping theorem is used again. The case

$$w_{2}' + \frac{\alpha - 1}{r^{\beta - 1}(t)\tilde{R}_{\alpha}(t)}w_{2} + \frac{\alpha - 1}{r^{\beta - 1}(t)\tilde{R}_{\alpha}(t)}[|\omega(t) + w_{2}|\beta + \beta\omega(t)] = 0.$$

The counterpart of Theorem 3.7 which is at the same time a generalization of Theorem 2.9-(ii) reads as follows. Equation (3.1) can be seen again as a perturbation of certain Euler type equation, this time of the form

$$(r(t)\Phi(y'))' + \frac{\bar{\gamma}}{r^{\beta-1}(t)\tilde{R}^{\alpha}_{\alpha}(t)}\Phi(y) = 0.$$

Theorem 3.9. Let (3.55) hold. Assume that

$$\lim_{t\to\infty}\frac{1}{\tilde{R}_{\alpha}(t)}\int_t^{\infty}\tilde{R}_{\alpha}^{\alpha}(s)p(s)\,\mathrm{d}s=\left(\frac{\alpha-1}{\alpha}\right)^{\alpha}.$$

Put

$$\Upsilon(t) = \frac{1}{\tilde{R}_{\alpha}(t)} \int_{t}^{\infty} \tilde{R}_{\alpha}^{\alpha}(s) p(s) \, \mathrm{d}s - \left(\frac{\alpha - 1}{\alpha}\right)^{\alpha}$$

and suppose that

$$\int^{\infty} \frac{|\Upsilon(t)|}{r^{\beta-1}(t)\tilde{R}_{\alpha}(t)} \,\mathrm{d}t < \infty$$

and

$$\int_{-\infty}^{\infty} \frac{\Psi(t)}{r^{\beta-1}(t)\tilde{R}_{\alpha}(t)} \, \mathrm{d}t < \infty, \quad \text{where } \Psi(t) = \int_{t}^{\infty} \frac{|\Upsilon(s)|}{r^{\beta-1}(s)\tilde{R}_{\alpha}(s)} \, \mathrm{d}s.$$

Then equation (3.1) is nonoscillatory and has a solution  $y \in \mathcal{NRV}_{1/\tilde{R}_{\alpha}}(-(\alpha-1)/\alpha)$  of the form  $y(t) = \tilde{R}_{\alpha}^{(\alpha-1)/\alpha}(t)L(t)$  with  $L \in \mathcal{NSV}_{1/\tilde{R}_{\alpha}}$  and  $\lim_{t\to\infty} L(t) = \ell \in (0,\infty)$ .

*Proof.* We omit details again. Note only that we seek a solution of (3.1) expressed in the form

$$y(t) = \exp\left\{\int_{t_i}^t \Phi^{-1}\left(\frac{\Upsilon(s) - \bar{\gamma} + v(s)}{r(s)\tilde{R}_{\alpha}^{\alpha-1}(s)}\right) \mathrm{d}s\right\},\,$$

where v is desired to satisfy a certain Riccati type equation; the existence of v is proved by means of the contraction mapping theorem.

# 3.5 Asymptotic formulas for nonoscillatory solutions of conditionally oscillatory half-linear equations

In this section which is based on the paper [137] by Pátíková we investigate asymptotic properties of nonoscillatory solutions of a special conditionally oscillatory half-linear second order differential equation, which was constructed in [29] as a perturbation of equation (3.1).

Throughout this section we suppose that (3.1) is nonoscillatory. Let d(t) be a positive continuous function, we say that the equation

$$(r(t)\Phi(x'))' + [p(t) + \mu d(t)]\Phi(x) = 0$$
(3.59)

is *conditionally oscillatory* if there exists a constant  $\mu_0 > 0$  such that (3.59) is oscillatory for  $\mu > \mu_0$  and nonoscillatory for  $\mu < \mu_0$ . Let h(t) be a positive solution of nonoscillatory equation (3.1) such that  $h'(t) \neq 0$  on some interval of the form  $[T_0, \infty)$  and denote

$$R(t) := r(t)h^{2}(t)|h'(t)|^{\alpha-2}, \quad G(t) := r(t)h(t)\Phi(h'(t)).$$
(3.60)

Under the assumptions

$$\int^{\infty} \frac{dt}{R(t)} = \infty, \quad \liminf_{t \to \infty} |G(t)| > 0,$$

the authors of [29] constructed a conditionally oscillatory equation seen as a perturbation of (3.1) in the form

$$(r(t)\Phi(x'))' + \left[p(t) + \frac{\mu}{h^{\alpha}(t)R(t)\left(\int^{t} R^{-1}(s)\,\mathrm{d}s\right)^{2}}\right]\Phi(x) = 0.$$
(3.61)

The critical oscillation constant of this equation is  $\mu_0 = \frac{1}{2\beta}$ , where  $\beta$  is the conjugate number to  $\alpha$ . In [29] it is also shown that (3.61) has for this constant  $\mu = \mu_0$  a solution with the asymptotic formula

$$x(t) = h(t) \left( \int^t R^{-1}(s) \, ds \right)^{\frac{1}{\alpha}} \left( 1 + O\left( \left( \int^t R^{-1}(s) \, ds \right)^{-1} \right) \right) \quad \text{as} \quad t \to \infty.$$
(3.62)

The aim of this section is to present more precise asymptotic formulas in terms of slowly and regularly varying functions in the case where the constant  $\mu$  is less than or equal to  $\frac{1}{26}$ .

The "perturbation approach", when the considered equation is regarded as a perturbation of another half-linear equation, has been also used in [135]. There, the asymptotics of nonoscillatory solutions of

$$(\Phi(x'))' + \frac{\gamma_{\alpha}}{t^{\alpha}} \Phi(x) + \tilde{p}(t)\Phi(x) = 0, \qquad (3.63)$$

where  $\gamma_{\alpha} = \left(\frac{\alpha-1}{\alpha}\right)^{\alpha}$ , was established under the assumption

$$\lim_{t \to \infty} \log t \int_t^\infty \tilde{p}(s) s^{\alpha - 1} \, \mathrm{d}s \in \left( -\infty, \frac{1}{2} \left( \frac{\alpha - 1}{\alpha} \right)^{\alpha - 1} \right].$$

Equation (3.63) has been seen as a perturbation of the half-linear Euler type equation

$$(\Phi(x'))' + \frac{\gamma_{\alpha}}{t^{\alpha}}\Phi(x) = 0 \tag{3.64}$$

with the critical constant  $\gamma_{\alpha}$ .

It is well-known (see, e.g., [28]), that similarly as in the linear oscillation theory, the nonoscillation of equation (3.1) is equivalent to the solvability of a Riccati type equation. In particular, if *x* is an eventually positive or negative solution of the nonoscillatory equation (3.1) on some interval of the form  $[T_0, \infty)$ , then  $w(t) = r(t)\Phi(\frac{x'}{x})$  solves the Riccati type equation

$$w' + p(t) + (\alpha - 1)r^{1-\beta}(t)|w|^{\beta} = 0.$$
(3.65)

Conversely, having a solution w(t) of (3.65) for  $t \in [T_0, \infty)$ , the corresponding solution of (3.1) can be expressed as

$$x(t) = C \exp\left\{\int^t r^{1-\beta}(s)\Phi^{-1}(w) \,\mathrm{d}s\right\},\,$$

where  $\Phi^{-1}$  is the inverse function of  $\Phi$  and *C* is a constant.

Using the concept of perturbations it appears useful to deal with the so called modified (or generalized) Riccati equation. Let *h* be a positive solution of (3.1) and  $w_h(t) = r(t)\Phi\left(\frac{h'}{h}\right)$  be the corresponding solution of the Riccati equation (3.65). Let us consider another nonoscillatory equation

$$(r(t)\Phi(x'))' + P(t)\Phi(x) = 0$$
(3.66)

and let w(t) be a solution of the Riccati equation associated with (3.66). Then  $v(t) = (w(t) - w_h(t))h^{\alpha}(t)$  solves the modified Riccati equation

$$v' + (P(t) - p(t))h^{\alpha} + \alpha r^{1-\beta} h^{\alpha} \tilde{P}(\Phi^{-1}(w_h), w) = 0, \qquad (3.67)$$

where

$$\tilde{P}(u,v) := \frac{|u|^{\alpha}}{\alpha} - uv + \frac{|v|^{\beta}}{\beta} \ge 0,$$

with the equality  $\tilde{P}(u, v) = 0$  if and only if  $v = \Phi(u)$ . We deal with this equation in a slightly different, but still equivalent, form

$$v' + (P(t) - p(t))h^{\alpha} + (\alpha - 1)r^{1 - \beta}h^{-\beta}|G|^{\beta}F\left(\frac{v}{G}\right) = 0,$$
(3.68)

where G(t) is defined by (3.60) and

$$F(u) = |u+1|^{\beta} - \beta u - 1.$$
(3.69)

Next we apply the perturbation principle combined with the (modified) Riccati technique to get asymptotical results for (3.61) with  $\mu < \frac{1}{2\beta}$  and  $\mu = \frac{1}{2\beta}$ .

**Theorem 3.10.** Suppose that (3.1) is nonoscillatory and possesses a positive solution h(t) such that  $h'(t) \neq 0$  for large t and let

$$\int^{\infty} \frac{\mathrm{d}t}{R(t)} = \infty, \tag{3.70}$$

and

$$\liminf_{t \to \infty} |G(t)| > 0. \tag{3.71}$$

If  $\mu < \frac{1}{2\beta}$ , then the conditionally oscillatory equation (3.61) has a pair of solutions given by the asymptotic formula

$$x_i = h(t) \left( \int^t R^{-1}(s) \, \mathrm{d}s \right)^{(\beta-1)\lambda_i} L_i(t),$$

where  $\lambda_i$  are zeros of the quadratic equation

$$\frac{\beta}{2}\lambda^2 - \lambda + \mu = 0 \tag{3.72}$$

and  $L_i(t)$  are generalized normalized slowly varying functions of the form

$$L_{i}(t) = \exp\left\{\int^{t} \frac{\varepsilon_{i}(s)}{R(s) \int^{s} R^{-1}(\tau) \, d\tau} \, ds\right\}$$

and  $\varepsilon_i(t) \to 0$  for  $t \to \infty$ .

*Proof.* We are looking for solutions of the modified Riccati equation associated with (3.61), which reads as

$$v'(t) + \frac{\mu}{R(t)\left(\int^t R^{-1}(s)\,\mathrm{d}s\right)^2} + (\alpha - 1)r^{1-\beta}(t)h^{-\beta}(t)|G(t)|^{\beta}F\left(\frac{v(t)}{G(t)}\right) = 0,\tag{3.73}$$

where *G* is defined by (3.60) and *F* by (3.69).

Assumptions (3.70) and (3.71) imply the convergence of the integral

$$\int^{\infty} r^{1-\beta}(t) h^{-\beta}(t) |G(t)|^{\beta} F\left(\frac{v(t)}{G(t)}\right) \mathrm{d}t,$$

from which it follows (see [29]) that  $v(t) \to 0$  and  $\frac{v(t)}{G(t)} \to 0$  as  $t \to \infty$ .

Let  $C_0[T, \infty)$  be the set of all continuous functions on the interval  $[T, \infty)$  (*T* will be specified later) which converge to zero a  $t \to \infty$  and let us consider a set of functions

$$V = \{ \omega \in C_0[T, \infty) : |\omega(t)| < \varepsilon, t \ge T \},\$$

where  $\varepsilon > 0$  is so small that

$$\frac{1}{\sqrt{1-2q\mu}}(q+1)\,\varepsilon < \frac{1}{2}.$$
(3.74)

Let us also observe that the inequality

$$\frac{1}{\sqrt{1-2q\mu}} \left(\frac{q}{2}+1\right) \varepsilon \le 1 \tag{3.75}$$

is implied. Denote the roots of the quadratic equation (3.72) for  $\mu < \frac{1}{2q}$  as

$$\lambda_1 = \frac{1 - \sqrt{1 - 2\beta\mu}}{\beta}, \quad \lambda_2 = \frac{1 + \sqrt{1 - 2\beta\mu}}{\beta}.$$

We assume that two solutions of the modified Riccati equation (3.73) are in the form 1 + z(t)

$$v_i(z,t) := \frac{\lambda_i + z(t)}{\int^t R^{-1}(s) \, \mathrm{d}s},$$

for  $t \in [T, \infty)$  and  $z \in V$ , i = 1, 2. Substituting this function and its derivative into (3.73), we have

$$z'(t) + \frac{\mu - \lambda_i - z(t)}{R(t) \int^t R^{-1}(s) \, \mathrm{d}s} + (\alpha - 1)r^{1-\beta}(t)h^{-\beta}(t)|G(t)|^{\beta}F\left(\frac{v_i(z,t)}{G(t)}\right) \int^t R^{-1}(s) \, \mathrm{d}s = 0$$

which can be rewritten as

$$z'(t) + \frac{(-1+\lambda_i\beta)z(t)}{R(t)\int^t R^{-1}(s)\,\mathrm{d}s} + \frac{1}{R(t)\int^t R^{-1}(s)\,\mathrm{d}s}E_i(z,t) = 0, \tag{3.76}$$

where

$$E_i(z,t) := \mu - \lambda_i - \lambda_i \beta z(t) + (\alpha - 1) \left( \int^t R^{-1}(s) \, \mathrm{d}s \right)^2 G^2(t) F\left(\frac{v_i(z,t)}{G(t)}\right).$$

This means that looking for solutions  $v_i$  of the modified Riccati equation (3.73) is equivalent to looking for solutions  $z_i$  of the equation (3.76). In the next we shall show that two solutions of (3.76) can be found through the Banach fixed-point theorem used onto suitable integral operators.

First, let us turn our attention to the behavior of the function F(u), which plays an important role in estimating of certain useful expressions. Studying the behavior of F(u) and F'(u) for u in a neighborhood of 0, we have

$$F(u) = \frac{F''(0)}{2}u^2 + \frac{F'''(\zeta)}{6}u^3$$
  
=  $\frac{\beta(\beta-1)}{2}u^2 + \frac{\beta(\beta-1)(\beta-2)}{6}|1+\zeta|^{\beta-3}\operatorname{sgn}(1+\zeta)u^3,$ 

where  $\zeta$  is between 0 and u. For  $|u| < \frac{1}{2}$  and hence also  $|\zeta| < \frac{1}{2}$  there exists a positive constant  $M_q$  such that

$$\frac{\beta(\beta-2)}{6}|1+\zeta|^{\beta-3}\operatorname{sgn}(1+\zeta)| \le M_{\beta} \quad \text{for} \quad \beta > 1.$$

Therefore

$$\left|F(u) - \frac{\beta(\beta - 1)}{2}u^2\right| \le (\beta - 1)M_\beta |u|^3.$$
(3.77)

Similarly,

$$\begin{aligned} F'(u) &= F''(0)u + \frac{F'''(\zeta')}{2}u^2 \\ &= \beta(\beta-1)u + \frac{\beta(\beta-1)(\beta-2)}{2}|1+\zeta'|^{\beta-3}\operatorname{sgn}(1+\zeta')u^2, \end{aligned}$$

where  $\zeta'$  is between 0 and *u*. Again, considering  $|\zeta'| < \frac{1}{2}$  we have

$$\left|F'(u) - \beta(\beta - 1)u\right| \le 3(\beta - 1)M_{\beta}|u|^{2}.$$
(3.78)

Now, let us denote  $J(t) := \int^t R^{-1}(s) ds$ , then the estimate of the function  $E_i(z, t)$  for  $z \in V$ , reads as

$$\begin{split} |E_{i}(z,t)| &= \left| \mu - \lambda_{i} - \lambda_{i}\beta z(t) + \frac{\beta}{2}J^{2}(t)v_{i}^{2}(z,t) + (\alpha-1)J^{2}(t)G^{2}(t)F\left(\frac{v_{i}(z,t)}{G(t)}\right) \right. \\ &- \frac{\beta}{2}J^{2}(t)v_{i}^{2}(z,t) \left| \leq |\mu - \lambda_{i} - \lambda_{i}\beta z(t) + \frac{\beta}{2}(\lambda_{i} + z(t))^{2}| \right. \\ &+ \left| (\alpha-1)J^{2}(t)G^{2}(t)\left[F\left(\frac{v_{i}(z,t)}{G(t)}\right) - \frac{\beta(\beta-1)}{2}\left(\frac{v_{i}(z,t)}{G(t)}\right)^{2}\right] \right| \\ &\leq \left. \frac{\beta}{2}|z(t)|^{2} + \frac{M_{\beta}|\lambda_{i} + z(t)|^{3}}{|G(t)||J(t)|} \leq \frac{\beta}{2}|z(t)|^{2} + \frac{KM_{\beta}|\lambda_{i} + z(t)|^{3}}{|J(t)|}, \end{split}$$

where (3.77) was used for  $u = \frac{v_i}{G}$  and  $K := \sup_{t \ge T} \frac{1}{|G(t)|}$  is a finite constant for T sufficiently large because of (3.71). According to (3.70) there exists  $T_1$  such that the last term in the previous inequality is less than  $\varepsilon^2$  and therefore

$$|E_i(z,t)| \le \frac{q}{2}\varepsilon^2 + \varepsilon^2 \le \varepsilon^2 \left(\frac{q}{2} + 1\right)$$
(3.79)

for  $t \ge T_1$ . Furthermore, for  $z_1, z_2 \in V$  we have

.

$$\begin{aligned} |E_i(z_1,t) - E_i(z_2,t)| &= \left| -\lambda_i \beta(z_1 - z_2) \right. \\ &+ (p-1)J^2(t)G^2(t) \left[ F\left(\frac{v_i(z_1,t)}{G(t)}\right) - F\left(\frac{v_i(z_2,t)}{G(t)}\right) \right] \right|, \end{aligned}$$

which, by the mean value theorem with a suitable  $z(t) \in V$ , becomes

$$= \left| -\lambda_{i}q(z_{1}-z_{2}) + \beta J(t)v_{i}(z,t)(z_{1}-z_{2}) + (\alpha-1)J(t)G(t)F'\left(\frac{v_{i}(z,t)}{G(t)}\right)(z_{1}-z_{2}) - \beta J(t)v_{i}(z,t)(z_{1}-z_{2})\right|$$

$$\leq \left| -\lambda_{i}\beta + \beta J(t)v_{i}(z,t) + (\alpha-1)J(t)G(t)\left(F'\left(\frac{v_{i}(z,t)}{G(t)}\right) - \beta(\beta-1)\frac{v_{i}(z,t)}{G(t)}\right)\right| \cdot ||z_{1}-z_{2}||$$

$$\leq \left(\beta|z(t)| + \left|\frac{3KM_{q}(\lambda_{i}+z(t))^{2}}{J(t)}\right|\right) \cdot ||z_{1}-z_{2}||,$$

where (3.78) was used. Similarly as in the previous estimate, there exists  $T_2$  such that the middle term in the last row of the inequality is less than  $\varepsilon$  and hence

$$|E_i(z_1, t) - E_i(z_2, t)| \le \varepsilon(\beta + 1) \cdot ||z_1 - z_2||$$
(3.80)

for  $t \in [T_2, \infty)$ .

Now, let us consider the pair of functions

$$r_i(t) := \exp\left\{\int^t \frac{-1 + \lambda_i \beta}{R(s) \int^s R^{-1}(\tau) \,\mathrm{d}\tau} \,\mathrm{d}s\right\}, \ i = 1, 2.$$

Then equation (3.76) is equivalent to

$$(r_i(t)z(t))' + r_i(t)\frac{1}{R(t)\int^t R^{-1}(s)\,\mathrm{d}s}E_i(z,t) = 0.$$
(3.81)

For i = 1, we have the function

$$r_{1}(t) = \exp\left\{\int^{t} \frac{-1 + \lambda_{1}\beta}{R(s)\int^{s} R^{-1}(\tau) \,\mathrm{d}\tau} \,\mathrm{d}s\right\} = \exp\left\{\int^{t} \frac{-\sqrt{1 - 2\beta\mu}}{R(s)\int^{s} R^{-1}(\tau) \,\mathrm{d}\tau} \,\mathrm{d}s\right\}$$

and it is easy to see that  $r_1(t) \to 0$  for  $t \to \infty$ .

Finally, let us define the integral operator  $F_1$  on the set of functions V by

$$(F_1 z)(t) = \frac{1}{r_1(t)} \int_t^\infty \frac{r_1(s)}{R(s) \int_s^s R^{-1}(\tau) \, \mathrm{d}\tau} E_1(z, s) \, \mathrm{d}s.$$

We observe that

$$\int_{t}^{\infty} \frac{r_{1}(s)}{R(s) \int^{s} R^{-1}(\tau) \, \mathrm{d}\tau} \, \mathrm{d}s = \frac{r_{1}(t)}{\sqrt{1 - 2\beta\mu}}$$

Taking  $T = \max\{T_1, T_2\}$ , by (3.79) and (3.74) we have

$$\begin{aligned} |(F_1 z)(t)| &\leq \frac{1}{r_1(t)} \int_t^\infty \frac{r_1(s)}{R(s) \int^s R^{-1}(\tau) \, \mathrm{d}\tau} |E_1(z,s)| \, \mathrm{d}s \\ &\leq \frac{1}{\sqrt{1 - 2\beta\mu}} \left(\frac{\beta}{2} + 1\right) \varepsilon^2 \leq \varepsilon, \end{aligned}$$

which means that  $F_1$  maps the set V into itself, and by (3.80) and (3.75) we see that

$$\begin{aligned} |(F_1 z_1)(t) - (F_1 z_2)(t)| &\leq \frac{1}{r_1(t)} \int_t^{\infty} \frac{r_1(s)}{R(s) \int_s^s R^{-1}(\tau) \, d\tau} |E_1(z_1, s) - E_1(z_2, s)| \, \mathrm{d}s \\ &\leq ||z_1 - z_2|| \frac{1}{\sqrt{1 - 2\beta\mu}} \varepsilon(\beta + 1) < \frac{1}{2} ||z_1 - z_2||, \end{aligned}$$

which implies that  $F_1$  is a contraction. Using the Banach fixed-point theorem we can find a function  $z_1(t)$ , that satisfies  $z_1 = F_1 z_1$ . That means that  $z_1(t)$  is a solution of (3.81) and also of (3.76) and  $v_1(t) = \frac{\lambda_1 + z_1(t)}{\int_{t}^{t} R^{-1} ds}$  is a solution of (3.73). For i = 2 we

have

$$r_2(t) = \exp\left\{\int^t \frac{-1 + \lambda_2 \beta}{R(s) \int^s R^{-1}(\tau) \,\mathrm{d}\tau} \,\mathrm{d}s\right\} = \exp\left\{\int^t \frac{\sqrt{1 - 2q\mu}}{R(s) \int^s R^{-1}(\tau) \,\mathrm{d}\tau} \,\mathrm{d}s\right\}$$

and we define an integral operator  $F_2$  by

$$(F_2 z)(t) = -\frac{1}{r_2(t)} \int^t \frac{r_2(s)}{R(s) \int^s R^{-1}(\tau) \, \mathrm{d}\tau} E_2(z, s) \, \mathrm{d}s.$$

Since

$$\int^t \frac{r_2(s)}{R(s) \int^s R^{-1}(\tau) \, \mathrm{d}\tau} \, \mathrm{d}s = \frac{r_2(t) - c}{\sqrt{1 - 2\beta\mu}}$$

where *c* is a positive suitable constant, the inequality

$$\frac{1}{r_2(t)} \int^t \frac{r_2(s)}{R(s) \int^s R^{-1}(\tau) \, \mathrm{d}\tau} \, \mathrm{d}s \le \frac{1}{\sqrt{1 - 2\beta\mu}}$$

holds for t sufficiently large, as  $r_2(t) \rightarrow \infty$  for  $t \rightarrow \infty$ . Taking  $T = \max\{T_1, T_2\}$ , the estimates for the operator  $F_2$  are the same as in the previous case and we can find a fixed point  $z_2(t)$  satisfying  $F_2 z_2 = z_2$ . Thus  $z_2(t)$  solves (3.81) and  $v_2(t) = \frac{\lambda_2 + z_2(t)}{\int_{t}^{t} R^{-1} ds}$ solves the modified Riccati equation (3.73).

Expressing the solutions of the "standard" Riccati equation for (3.61) corresponding to the solutions  $v_i(z_i, t)$  of the modified Riccati equation, we have

$$w_{i}(t) = h^{-\alpha}(t)v_{i}(z_{i}, t) + w_{h}(t) = w_{h}(t)\left(1 + \frac{v_{i}(z_{i}, t)}{h^{\alpha}(t)w_{h}(t)}\right)$$
  
$$= w_{h}(t)\left(1 + \frac{\lambda_{i} + z_{i}(t)}{h^{\alpha}(t)w_{h}(t)\int^{t}R^{-1} ds}\right) = w_{h}(t)\left(1 + \frac{\lambda_{i} + z_{i}(t)}{G(t)\int^{t}R^{-1}(s) ds}\right)$$

Since solutions of (3.61) are given by the formula

$$x(t) = \exp\left\{\int^t r^{1-\beta}(s)\Phi^{-1}(w)\,\mathrm{d}s\right\},\,$$

we need to express

$$\begin{aligned} r^{1-\beta}(t)\Phi^{-1}(w_i) &= \frac{h'(t)}{h(t)} \left( 1 + \frac{\lambda_i + z_i(t)}{G(t)\int^t R^{-1}(s)\,\mathrm{d}s} \right)^{\beta-1} \\ &= \frac{h'(t)}{h(t)} \left( 1 + (\beta-1)\frac{\lambda_i + z_i(t)}{G(t)\int^t R^{-1}(s)\,\mathrm{d}s} + o\left(\frac{\lambda_i + z_i(t)}{G(t)\int^t R^{-1}(s)\,\mathrm{d}s}\right) \right) \\ &= \frac{h'(t)}{h(t)} + \frac{(\beta-1)\lambda_i}{R(t)\int^t R^{-1}(s)\,\mathrm{d}s} \\ &+ \frac{(\beta-1)z_i(t)}{R(t)\int^t R^{-1}(s)\,\mathrm{d}s} + o\left(\frac{\lambda_i + z_i(t)}{R(t)\int^t R^{-1}(s)\,\mathrm{d}s}\right). \end{aligned}$$

Because

$$o\left(\frac{\lambda_i + z_i(t)}{R(t)\int^t R^{-1}(s)\,\mathrm{d}s}\right) = \frac{R(t)\int^t R^{-1}(s)\,\mathrm{d}s\,o\left(\frac{\lambda_i + z_i(t)}{R(t)\int^t R^{-1}(s)\,\mathrm{d}s}\right)}{R(t)\int^t R^{-1}(s)\,\mathrm{d}s}$$
$$= \frac{o(\lambda_i + z_i(t))}{R(t)\int^t R^{-1}(s)\,\mathrm{d}s}$$

holds for large *t*, the pair of solutions of (3.61) for i = 1, 2 is in the form

$$x_{i}(t) = \exp\left\{\ln h(t) + \ln\left(\int^{t} R^{-1}(s) \, \mathrm{d}s\right)^{(\beta-1)\lambda_{i}} + \int^{t} \frac{(\beta-1)z_{i}(s) + o(\lambda_{i} + z_{i}(s))}{R(s)\int^{s} R^{-1}(\tau) \, \mathrm{d}\tau}\right\}.$$

As  $z_i \in V$  and hence  $z_i(t) \to 0$  for  $t \to \infty$ , the statement of the theorem holds for  $\varepsilon_i(t) = (\beta - 1)z_i(t) + o(\lambda_i + z_i(t))$ .

Now let us present the asymptotic formula in case  $\mu = \frac{1}{2\beta}$ , which gives an improved version of (3.62).

**Theorem 3.11.** Let the assumptions of the previous theorem be satisfied and let  $\mu = \frac{1}{2q}$ . Then equation (3.61) has a solution of the form

$$x(t) = h(t) \left( \int^{t} R^{-1}(s) \, \mathrm{d}s \right)^{\frac{1}{p}} L(t), \tag{3.82}$$

where *L*(*t*) is a generalized normalized slowly varying function of the form

$$L(t) = \exp\left\{\int^t \frac{\varepsilon(s)}{R(s)\int^s R^{-1}(\tau) \,\mathrm{d}\tau} \,\mathrm{d}s\right\}$$

and  $\varepsilon(t) \to 0$  for  $t \to \infty$ .

*Proof.* For  $\mu = \frac{1}{2\beta}$  the quadratic equation (3.72) has the double root  $\lambda = \frac{1}{\beta}$ . We assume the solution of modified Riccati equation to be in the form (for  $z \in V$ )

$$v(z,t) = \frac{\frac{1}{\beta} + z(t)}{\int^t R^{-1}(s) \,\mathrm{d}s},$$

which gives, after substituting into the modified Riccati equation (3.73) for  $\mu = \frac{1}{2\beta}$ ,

$$z'(t) + \frac{-z(t) - \frac{1}{2\beta}}{R(t) \int^t R^{-1}(s) \, \mathrm{d}s} + (\alpha - 1)r^{1-\beta}(t)h^{-\beta}(t)|G(t)|^{\beta}F\left(\frac{v(z, t)}{G(t)}\right) \int^t R^{-1}(s) \, \mathrm{d}s = 0.$$

Let us denote

$$E(z,t) = -z(t) - \frac{1}{2\beta} + (\alpha - 1) \left( \int^t R^{-1}(s) \, \mathrm{d}s \right)^2 G^2(t) F\left(\frac{v(z,t)}{G(t)}\right)$$

and let us consider an integral operator  $F_3$ 

$$(F_3 z)(t) = \int_t^\infty \frac{1}{R(s) \int_s^s R^{-1}(\tau) \, \mathrm{d}\tau} E(z, s) \, \mathrm{d}s$$

on a set of continuous functions

$$V = \{ \omega \in C_0[T, \infty) : |\omega(t)| < \varepsilon, t \ge T \},\$$

where *T* and  $\varepsilon$  are to be established similarly as in the proof of the previous theorem. Then the solution of modified Riccati equation and also the solution of the studied equation can be found in almost the same manner as for the previous statement.

**Remark 3.1.** If  $r(t) \equiv 1$ ,  $p(t) = \gamma_{\alpha}t^{-\alpha}$  and  $h(t) = t^{\frac{\alpha-1}{\alpha}}$  then the conditionally oscillatory equation (3.61) with  $\mu = \frac{1}{2\beta}$ , seen as a perturbation of the Euler equation (3.64), becomes the Euler-Weber (or alternatively Riemann-Weber) half-linear differential equation

$$(\Phi(x'))' + \left[\frac{\gamma_{\alpha}}{t^{\alpha}} + \frac{\mu_{\alpha}}{t^{\alpha}\ln^{2}t}\right]\Phi(x) = 0$$

with the so-called critical coefficient  $\mu_p = \frac{1}{2} \left(\frac{\alpha-1}{\alpha}\right)^{\alpha-1}$ . The asymptotic formula (3.82) then reduces to the formula given in [135, Theorem 2].

**Remark 3.2.** For the Euler-Weber half-linear equation also the asymptotic formula for its second linearly independent solution is known (see [136]). An open question remains whether the second linearly independent solution of (3.61) with  $\mu = \frac{1}{2\beta}$  could be found in a similar form

$$x_{2}(t) = h(t) \left( \int^{t} R^{-1}(s) \, \mathrm{d}s \right)^{\frac{1}{\alpha}} \left( \ln \left( \int^{t} R^{-1}(s) \, \mathrm{d}s \right) \right)^{\frac{2}{\alpha}} L_{2}(t),$$

where

$$L_2(t) = \exp\left\{\int^t \frac{\varepsilon_2(s)}{R(s)\int^s R^{-1}(\tau)\,\mathrm{d}\tau\ln\left(\int^s R^{-1}(\tau)\,\mathrm{d}\tau\right)}\,\mathrm{d}s\right\}$$

and  $\varepsilon_2(t) \to 0$  as  $t \to \infty$ .

## 3.6 De Haan type solutions

## **3.6.1** Solutions in the class $\Gamma$

We consider the equations

$$(r(t)\Phi(y'))' = p(t)\Phi(y),$$
 (3.83)

and

$$(\Phi(y'))' = p(t)\Phi(y), \tag{3.84}$$

where r, p are positive continuous functions on  $[a, \infty)$ . In this subsection we deal with solutions of (3.84) which are shown to be in the class  $\Gamma$ , but  $\mathcal{RV}$  solutions are also examined. The results are taken from [147] by Řehák. Later we briefly discuss possible extensions (based on Řehák, Taddei [150]).

Because of the sign condition on p, equation (3.83) is nonoscillatory by the Sturm type comparison theorem; it suffices to compare (3.83) with  $(r(t)\Phi(y'))' = 0$ . For a (nonoscillatory) solution y of (3.83) it holds, y(t)y'(t) > 0 or y(t)y'(t) < 0 eventually. Without loss of generality we may work just with positive solutions. Denote

 $\mathbb{M}^+ = \{y : y \text{ is a solution of } (3.83), y(t) > 0, y'(t) > 0 \text{ for large } t\}.$ 

Under our sign condition, the class  $\mathbb{M}^+$  is nonempty. Moreover,

$$\forall (a_0, a_1) \in (0, \infty) \times (0, \infty) \forall t_0 \text{ sufficiently} \text{ large } \exists y \in \mathbb{M}^+ : y(t_0) = a_0, y'(t_0) = a_1.$$

$$(3.85)$$

Further, denote  $\mathbb{M}_B^+ = \{y \in \mathbb{M}^+ : \lim_{t \to \infty} y(t) < \infty\}$  and  $\mathbb{M}_\infty^+ = \{y \in \mathbb{M}^+ : \lim_{t \to \infty} y(t) = \infty\}$ . Clearly,  $\mathbb{M}^+ = \mathbb{M}_B^+ \cup \mathbb{M}_\infty^+$ . Set

$$J = \int_{a}^{\infty} r^{1-\beta}(t) \left( \int_{a}^{t} p(s) \, \mathrm{d}s \right)^{\beta-1} \mathrm{d}t \text{ and } J_{r} = \int_{a}^{\infty} r^{1-\beta}(s) \, \mathrm{d}s$$

There hold  $J < \infty \Rightarrow J_r < \infty$  and  $\mathbb{M}_{\mathbb{R}}^+ \neq \emptyset \Leftrightarrow J < \infty$ . Hence,

$$J_r = \infty \Rightarrow (\emptyset \neq) \mathbb{M}^+ = \mathbb{M}^+_{\infty}. \tag{3.86}$$

In the special case of equation (3.84), that is r(t) = 1, the discussed classes will be denoted as  $\mathbb{M}^+(1)$  and  $\mathbb{M}^+_{\infty}(1)$ . For basic properties (classification and existence) of nonoscillatory solutions see e.g. [28, Section 4.1].

**Theorem 3.12.** If 
$$p^{-\frac{1}{\alpha}} \in \mathcal{BSV}$$
, then  $\emptyset \neq \mathbb{M}^+(1) = \mathbb{M}^+_{\infty}(1) \subseteq \Gamma\left(\left(\frac{\alpha-1}{p}\right)^{\frac{1}{\alpha}}\right)$ .

*Proof.* In the first part of the proof, we assume that  $p \in C^1$ . From the assumption of the theorem, in view of properties of  $\mathscr{BSV}$  functions, we have  $(p^{-\frac{1}{\alpha}})'(t) = -p'(t)/(\alpha p^{\frac{\alpha+1}{\alpha}}(t)) \to 0$  as  $t \to \infty$ . We know that  $\mathbb{M}^+(1) \neq \emptyset$ . Take  $y \in \mathbb{M}^+(1)$ . Then  $y \in \mathbb{M}^+_{\infty}(1)$ . Alternatively we can start with  $y(t) \to \infty$ , and then we find  $y \in \mathbb{M}^+(1) = \mathbb{M}^+_{\infty}(1)$ . If we set  $w = \frac{f\Phi(y')}{\Phi(y)}$ , where f is  $C^1$  nonzero function, then wsatisfies the generalized Riccati equation

$$w' - \frac{f'(t)}{f(t)}w - f(t)p(t) + (\alpha - 1)f^{1-\beta}(t)|w|^{\beta} = 0$$

for large *t*, see [28, Section 2.2.2]. Put  $f = p^{-\frac{1}{\beta}}$ . Then w > 0 and the Riccati like equation becomes

$$\frac{w'}{p^{\frac{1}{\alpha}}(t)} = 1 - (\alpha - 1)w \left(\frac{p'(t)}{\alpha p^{\frac{\alpha+1}{\alpha}}(t)} + w^{\beta - 1}\right)$$
(3.87)

for large *t*. We want to prove that  $\lim_{t\to\infty} w(t) = (\alpha - 1)^{-\frac{1}{\beta}}$ . We distinguish three cases according to the eventual sign of w'. First, let w'(t) > 0 for large *t*. Then  $\lim_{t\to\infty} w(t) = A \in (0,\infty) \cup \{\infty\}$ . If  $A = \infty$ , then from (3.87) and  $p'(t)/(p^{\frac{\alpha+1}{\alpha}}(t)) \to 0$ , we have  $\lim_{t\to\infty} w'(t)p^{-\frac{1}{\alpha}}(t) = -\infty$ , a contradiction with w'(t) > 0. Let  $A \in (0,\infty)$ . Then from (3.87) and  $p'(t)/(p^{\frac{\alpha+1}{\alpha}}(t)) \to 0$ , we have  $\lim_{t\to\infty} w'(t)p^{-\frac{1}{\alpha}}(t) = 1 - (\alpha - 1)A^{\beta}$ . If  $A \neq (\alpha - 1)^{-\frac{1}{\beta}}$ , then  $w'(t) \sim (1 - (\alpha - 1)A^{\beta})p^{\frac{1}{\alpha}}(t)$  as  $t \to \infty$ , hence

$$m_1 \int_{t_0}^t p^{\frac{1}{\alpha}}(s) \, \mathrm{d}s \le w(t) - w(t_0) \le M_1 \int_{t_0}^t p^{\frac{1}{\alpha}}(s) \, \mathrm{d}s, \tag{3.88}$$

 $t \ge t_0$ ,  $t_0$  being large, for some  $0 < m_1 < M_1 < \infty$ . From  $w = p^{-\frac{1}{\beta}} (y'/y)^{\alpha-1}$ , we have  $w^{\beta-1} = p^{-\frac{1}{\alpha}} y'/y$ . Since  $w^{\beta-1}(t) \le M$ ,  $t \ge t_0$ , for some  $M \in (0, \infty)$ , we get  $p^{\frac{1}{\alpha}}(t) \ge \frac{y'(t)}{y(t)M}$ ,  $t \ge t_0$ . Hence,

$$\int_{t_0}^t p^{\frac{1}{\alpha}}(s) \, \mathrm{d}s \ge \frac{1}{M} \int_{t_0}^t \frac{y'(s)}{y(s)} \, \mathrm{d}s = \frac{1}{M} \ln \frac{y(t)}{y(t_0)},$$

 $t \ge t_0$ , which implies  $\int_{t_0}^t p^{\frac{1}{\alpha}}(s) \, ds \to \infty$  as  $t \to \infty$ . In view of (3.88), we obtain  $\lim_{t\to\infty} w(t) = \infty$ , a contradiction with  $A < \infty$ . Thus,  $\lim_{t\to\infty} w(t) = (\alpha - 1)^{-\frac{1}{\beta}}$ . If

w'(t) < 0 for large t, then we use similar arguments; in particular, we distinguish the cases when  $\lim_{t\to\infty} w(t) = 0$  and  $\lim_{t\to\infty} w(t) > 0$ , and reach a contradiction in undesirable cases. Finally we assume  $w'(t_n) = 0$  for a sequence  $\{t_n\}$  with  $t_n \to \infty$  as  $n \to \infty$ . We take here zeroes of w' being consecutive; the finite cluster point cannot exist. From (3.87), we have

$$w^{\beta}(t_n) - \frac{1}{\alpha - 1} = -\frac{p'(t_n)}{\alpha} p^{-\frac{\alpha + 1}{\alpha}}(t_n) w(t_n).$$
(3.89)

Hence, w(t) hits the positive real root of the equation

$$|\lambda|^{\beta} - \frac{1}{\alpha - 1} = -\frac{p'(t_n)}{\alpha} p^{-\frac{\alpha + 1}{\alpha}}(t_n)\lambda$$

at each  $t = t_n$ . The positive root indeed exists; it is sufficient to realize that the left hand side of the equation represents a parabola like curve symmetric w.r.t. the vertical axis, and with the negative minimum, while the right hand side is a line which goes through the origin. The function w is monotone between zeroes of w', thus  $w(t_n) \le w(t) \le w(t_{n+1})$  or  $w(t_{n+1}) \le w(t) \le w(t_n)$ ,  $t_n \le t \le t_{n+1}$ . Thanks to  $p'(t)/(p^{\frac{\alpha+1}{\alpha}}(t)) \to 0$  as  $t \to \infty$ , from (3.89) we get  $\lim_{n\to\infty} w(t_n) = (\alpha - 1)^{-\frac{1}{\beta}}$ , and so  $\lim_{t\to\infty} w(t) = (\alpha - 1)^{-\frac{1}{\beta}}$ . Thus altogether we have  $\lim_{t\to\infty} p^{-\frac{1}{\beta}}(t)(y'(t)/y(t))^{\alpha-1} = (\alpha - 1)^{-\frac{1}{\beta}}$ , which implies

$$\left(\frac{y(t)}{y'(t)}\right)^{\alpha} \sim \left(\frac{\alpha - 1}{p(t)}\right)^{\frac{1}{\beta} \cdot \frac{\alpha}{\alpha - 1}} = \frac{\alpha - 1}{p(t)} \quad \text{as } t \to \infty.$$
(3.90)

Further, because of the identity  $(\Phi(y'))' = (\alpha - 1)y''|y'|^{\alpha-2}$ , from (2.1) we get

$$\frac{y''y}{y'^2} = \frac{y''y(y')^{\alpha-2}}{y'^{\alpha}} = \frac{(\Phi(y'))'y}{(\alpha-1)y'^{\alpha}} = \frac{p\Phi(y)y}{(\alpha-1)y'^{\alpha}} = \frac{p}{\alpha-1}\left(\frac{y}{y'}\right)^{\alpha}.$$
 (3.91)

From (3.90) and (3.91), we obtain

$$\frac{y^{\prime\prime}(t)y(t)}{y^{\prime2}(t)} = \frac{p(t)}{\alpha - 1} \left(\frac{y(t)}{y^{\prime}(t)}\right)^{\alpha} \sim \frac{p(t)}{\alpha - 1} \cdot \frac{\alpha - 1}{p(t)} = 1$$

as  $t \to \infty$ . Hence,  $y \in \Gamma(h)$ , where h = y/y'. From (3.90),  $h(t) \sim \left(\frac{\alpha-1}{p(t)}\right)^{\frac{1}{\alpha}}$  as  $t \to \infty$ , thus  $y \in \Gamma\left(\left(\frac{\alpha-1}{p}\right)^{\frac{1}{\alpha}}\right)$ .

Now we drop the assumption of differentiability of p. Since  $p^{-\frac{1}{\alpha}} \in \mathcal{BSV}$ , there exists  $\hat{p} \in C^1$  such that  $\hat{p}(t) \sim p(t)$  and  $(\hat{p}^{-\frac{1}{\alpha}})'(t) \to 0$  as  $t \to \infty$ . For every  $\varepsilon \in (0, 1)$  there exists a (large)  $t_0$  such that  $(1 - \varepsilon)\hat{p}(t) \leq p(t) \leq (1 + \varepsilon)\hat{p}(t), t \geq t_0$ . Consider the two auxiliary equations  $(\Phi(u'))' = (1 + \varepsilon)\hat{p}(t)\Phi(u)$  resp.  $(\Phi(v'))' = (1 - \varepsilon)\hat{p}(t)\Phi(v)$ . Take their solutions u resp. v which are in the relevant  $\mathbb{M}^+_{\infty}(1)$  type classes, and satisfy  $u(t_0) = u_0 > 0, u'(t_0) = u_1 > 0$  resp.  $v(t_0) = v_0 > 0, v'(t_0) = v_1 > 0$ , with

 $u_0, u_1, v_0, v_1$  being determined more precisely later. Such solutions indeed exist by (3.85) and (3.86). Set  $w_u = \Phi(u'/u)$ , resp.  $w_v = \Phi(v'/v)$ , resp.  $w_y = \Phi(y'/y)$ . These (eventually positive) functions then satisfy the generalized Riccati equations  $w'_u = (1+\varepsilon)\hat{p}(t)-(\alpha-1)|w_u|^{\beta}$ , resp.  $w'_v = (1-\varepsilon)\hat{p}(t)-(\alpha-1)|w_v|^{\beta}$ , resp.  $w'_y = p(t)-(\alpha-1)|w_y|^{\beta}$ ,  $t \ge t_0$ . Because of arbitrariness of  $v_0, v_1$  (see (3.85)) we may take v in the  $\mathbb{M}^+_{\infty}(1)$  type class in such a way that  $w_v(t_0) \le w_y(t_0)$ . From the standard result on differential inequalities ([53, Chapter III, Section 4]), we have  $w_v(t) \le w_y(t), t \ge t_0$ . Hence,

$$\left(\frac{v'(t)}{v(t)}\right)^{\alpha-1}p^{-\frac{1}{\beta}}(t) \le \left(\frac{y'(t)}{y(t)}\right)^{\alpha-1}p^{-\frac{1}{\beta}}(t),$$

 $t \ge t_0$ . As in the first part of the proof, we get

$$\left(\frac{v'(t)}{v(t)}\right)^{\alpha-1}\hat{p}^{-\frac{1}{\beta}}(t)\sim \left(\frac{1-\varepsilon}{\alpha-1}\right)^{\frac{1}{\beta}},$$

which yields, in view of  $p(t) \sim \hat{p}(t)$ ,

$$\left(\frac{v'(t)}{v(t)}\right)^{\alpha-1} p^{-\frac{1}{\beta}}(t) \sim \left(\frac{1-\varepsilon}{\alpha-1}\right)^{\frac{1}{\beta}}$$

as  $t \to \infty$ . Thus we get

$$\liminf_{t\to\infty} \left(\frac{y'(t)}{y(t)}\right)^{\alpha-1} p^{-\frac{1}{\beta}}(t) \ge \left(\frac{1-\varepsilon}{\alpha-1}\right)^{\frac{1}{\beta}}.$$

Analogously, examining  $w_u$ , we get

$$\limsup_{t \to \infty} \left( \frac{y'(t)}{y(t)} \right)^{\alpha - 1} p^{-\frac{1}{\beta}}(t) \le \left( \frac{1 + \varepsilon}{\alpha - 1} \right)^{\frac{1}{\beta}}.$$

Since  $\varepsilon \in (0, 1)$  was arbitrary, we obtain

$$\lim_{t \to \infty} (y'(t)/y(t))^{\alpha - 1} p^{-\frac{1}{\beta}}(t) = (\alpha - 1)^{-\frac{1}{\beta}}.$$

The rest of the proof is the same as in the previous part.

Results in the spirit of Theorem 3.12 for linear equation are presented in Subsection 2.6.1 Recall that  $\Gamma \subset \mathcal{RPV}(\infty)$ . Thus we should mention also Theorem 2.1, where the condition

$$t\int_t^{\lambda t} p(s)\,\mathrm{d}s \to \infty$$

as  $t \to \infty$  is proved to be necessary and sufficient for decreasing and increasing solutions of (2.2) to be rapidly varying; the proof is presented just for decreasing solutions. Concerning an extension of this result to half-linear equation (3.84),

the author has not found any paper on this topic, except of [121], where the discrete counterpart is considered and only decreasing solutions are examined. As for differential equations, there is only a mention made by prof. Kusano at CDDE conference (Brno, 2000) about an allegedly existing statement which extends Marić's theorem for the case of decreasing solutions to (3.84), with emphasizing that the case of increasing solutions of (3.84) is an open problem (both, the existence of an  $\mathcal{RPV}$  increasing solution as well as the fact that all increasing solutions are  $\mathcal{RPV}$ ). Theorem 3.12, in fact, offers a condition guaranteeing that any eventually positive increasing solution (which indeed exists) is rapidly varying and, moreover, its asymptotic behavior is more specified. Observe that this sufficient condition, for a differentiable *p* having the form  $(p^{-\frac{1}{\alpha}})'(t) \to 0$  as  $t \to \infty$ , implies the one of Marić type modified for (3.84), namely

$$t^{\alpha-1}\int_t^{\lambda t} p(s)\,\mathrm{d}s\to\infty.$$

Theorem 3.12 can be seen also as an extension of the classical result by Hartman-Wintner [54, Paragraph 24]. Finally, consider the special case when  $p(t) \rightarrow C \in (0, \infty)$  as  $t \rightarrow \infty$ . Then, from the proof of Theorem 3.12 we see that  $y \in \mathbb{M}^+$  satisfies  $\lim_{t\to\infty} y'(t)/y(t) = ((\alpha - 1)/C)^{-\frac{1}{\alpha}}$ . Thus, from this point of view, the solution *y* can be understood as of Poincaré-Perron type. Such a behavior, along with numerous refinements, is studied quite extensively for linear equations or systems also in the present; as one of the pioneering works can be regarded the classical paper [140]. So as a by-product of our results, we offer — to some extent — a half-linear extension of this result in the second order case.

Now we somehow modify the above ideas which will lead to regularly varying behavior. Let *p* be differentiable. In the previous considerations we assumed that  $\lim_{t\to\infty} (p^{-\frac{1}{\alpha}})'(t) = 0$ . Now assume that the limit is nonzero, i.e.,  $\lim_{t\to\infty} p'(t)/p^{-\frac{\alpha+1}{\alpha}} = C \neq 0$ . Denote by  $\hat{\varrho}$  the positive root of

$$|\varrho|^{\beta} + \frac{C}{\alpha}\varrho - \frac{1}{\alpha - 1} = 0.$$

It is easy to see that the root indeed exists and  $\hat{\varrho} \neq (\alpha - 1)^{-\frac{1}{\beta}}$ . Take  $y \in \mathbb{M}^+(1)$ . Similar arguments as in the first part of the proof of Theorem 3.12 yield

$$\left(\frac{y'(t)}{y(t)}\right)^{\alpha-1}p^{-\frac{1}{\beta}}(t)\sim\hat{\varrho}$$

as  $t \to \infty$ . Using this relation along with (3.91), we get

$$\lim_{t\to\infty}\frac{y^{\prime\prime}(t)y(t)}{y^{\prime 2}(t)}=\frac{1}{(\alpha-1)\hat{\varrho}^{\beta}}\neq 1.$$

We (naturally) assume that C < 0. Then  $\hat{\varrho} > (\alpha - 1)^{-\beta}$ , and so

$$\lim_{t\to\infty}\frac{y''(t)y(t)}{y'^2(t)}=\sigma<1,$$

where  $\sigma = 1/((\alpha - 1)\hat{\varrho}^{\beta})$ . Hence,  $y \in \mathcal{RV}(1/(1 - \sigma))$ , or  $y \in \mathcal{RV}(-\alpha \hat{\varrho}^{\beta-1}/C)$ . Thanks to convexity of solutions to (2.1), we get normalized regular variation. Thus we have proved the following theorem.

**Theorem 3.13.** Let *p* be differentiable and  $\lim_{t\to\infty} p'(t)/p^{-\frac{\alpha+1}{\alpha}}(t) = C < 0$ . Then

 $\emptyset \neq \mathbb{M}^+(1) = \mathbb{M}^+_{\infty}(1) \subseteq \mathcal{NRV}(-\alpha \hat{\varrho}^{\beta-1}/C),$ 

where  $\hat{\varrho}$  is the positive root of

$$|\varrho|^{\beta} + \frac{C}{\alpha}\varrho - \frac{1}{\alpha - 1} = 0.$$

From the results of Section 3.2 it follows that (3.84) possesses solutions  $y_i$  with  $y_i \in \mathcal{RV}(\Phi^{-1}(\lambda_i))$ , i = 1, 2, where  $\lambda_1 < 0 < \lambda_2$  are the roots of  $|\lambda|^{\beta} - \lambda - A = 0$  if and only if  $\lim_{t\to\infty} t^{\alpha-1} \int_t^{\infty} p(s) \, ds = A(<0)$ . Observe that  $p'(t)p^{-\frac{\alpha+1}{\alpha}}(t) \sim C(<0)$  implies  $t^{\alpha-1} \int_t^{\infty} p(s) \, ds \sim (-\alpha/C)^{\alpha}/(\alpha-1)$  as  $t \to \infty$ . Further, it is easy to see that  $\lambda_2 = (-\alpha/C)^{\alpha-1}\hat{\varrho}$ , and so — as expected — the indices of regular variation of increasing solutions in both the results coincide. Note that the integral condition from Section 3.2 is more general than the condition in the previous theorem. On the other hand, the fixed point approach used in Section 3.2 guarantees the existence of at least one positive increasing  $\mathcal{RV}$  solution, while the result in this section says that all positive increasing solutions are regularly varying.

Now we show a connection with generalized regular variation. First observe, that if a positive function  $f \in C^1$  satisfies  $\tau(t)f'(t)/f(t) \sim \vartheta \in \mathbb{R}$  as  $t \to \infty$ , where  $\tau$  is positive continuous with  $\int_a^{\infty} (1/\tau(s)) ds = \infty$ , then  $f \in N\mathcal{RV}_{\omega}(\vartheta)$ , where  $\omega(t) = \exp\left\{\int_a^t 1/\tau(s) ds\right\}$ . This fact easily follows from the representation theorem. Assume now  $p^{-\frac{1}{\alpha}} \in \mathcal{BSV}$ , and take  $y \in \mathbb{M}^+(1)$ . From the proof of Theorem 3.12, we have  $y'(t)/y(t) \sim (p(t)/(\alpha - 1))^{\frac{1}{\alpha}}$  as  $t \to \infty$ . Set  $\tau(t) = p^{-\frac{1}{\alpha}}(t)$ . Then  $\ln y(t) \sim N \int_{t_0}^t (1/\tau(s)) ds$  as  $t \to \infty$ , for some  $N \in (0, \infty)$ , which implies  $\int_{t_0}^{\infty} (1/\tau(s)) ds = \infty$ . If we set  $\vartheta = (\alpha - 1)^{-\frac{1}{\alpha}}$ , then y satisfies  $\tau(t)y'(t)/y(t) \sim \vartheta$  as  $t \to \infty$  and now it is easy to see that

$$y \in \mathcal{NRV}_{\omega}\left((\alpha-1)^{-\frac{1}{\alpha}}\right), \text{ where } \omega(t) = \exp\left\{\int_{a}^{t} p^{\frac{1}{\alpha}}(s) \, \mathrm{d}s\right\}.$$

The statement of Theorem 3.12 can therefore be reformulated in terms of generalized  $\mathcal{RV}$  functions. In particular, we get

$$\mathbb{M}^+(1) \subseteq \mathcal{NRV}_{\omega}\left((\alpha - 1)^{-\frac{1}{\alpha}}\right)$$

with the above defined  $\omega$ .

Consider now more general equation (3.83), where the coefficient *r* satisfies  $\int_{a}^{\infty} r^{1-\beta}(s) ds = \infty$ . Set  $R(t) = \int_{a}^{t} r^{1-\beta}(s) ds$  and denote by  $R^{-1}$  its inversion. Let us

introduce new independent variable  $s = \xi(t)$  and new function  $u(s) = y(\xi^{-1}(s)) = y(t)$ , where  $\xi \in C^1$ ,  $\xi(t) > 0$ ,  $\xi'(t) > 0$ ,  $t \in [a, \infty)$ , and  $\xi(t) \to \infty$  as  $t \to \infty$ . Then  $\frac{dt}{dt} = \xi'(\xi^{-1}(s))\frac{d}{ds}$ , and (3.83) is transformed into

$$\frac{\mathrm{d}}{\mathrm{d}s}\left(r(\xi^{-1}(s))\Phi(\xi'(\xi^{-1}(s)))\Phi\left(\frac{\mathrm{d}u}{\mathrm{d}s}\right)\right) = \frac{p(\xi^{-1}(s))}{\xi'(\xi^{-1}(s))}\Phi(u).$$
(3.92)

Set  $\xi(t) = R(t)$ . Then  $\xi'(\xi^{-1}(s)) = r^{1-\beta}(R^{-1}(s))$  and (3.92) becomes

$$\frac{\mathrm{d}}{\mathrm{d}s} \left( \Phi\left(\frac{\mathrm{d}u}{\mathrm{d}s}\right) \right) = \tilde{p}(s) \Phi(u), \tag{3.93}$$

where  $\tilde{p}(s) = \frac{p(R^{-1}(s))}{r^{1-\beta}(R^{-1}(s))}$ . Take a solution y of (3.83) such that  $y \in \mathbb{M}^+$ . Then  $y \in \mathbb{M}^+_{\infty}$ , and — after the transformation — the corresponding solution u of (3.93) satisfies  $u(s) \to \infty$  as  $s \to \infty$  with  $\frac{d}{ds}u(s) > 0$  eventually. If we now assume  $\tilde{p}^{-\frac{1}{\alpha}} \in \mathcal{BSV}$ , then as in the proof of Theorem 4.14 we get

$$\frac{\frac{\mathrm{d}}{\mathrm{d}s}u(s)}{u(s)} \sim \left(\frac{\tilde{p}(s)}{\alpha-1}\right)^{\frac{1}{\alpha}}$$

as  $s \to \infty$ . Since  $\frac{d}{ds} = r^{\beta-1}(t)\frac{d}{dt}$ , the last asymptotic relation yields

$$\frac{r^{\beta-1}(t)y'(t)}{y(t)} \sim \left(\frac{r^{\beta-1}(t)p(t)}{\alpha-1}\right)^{\frac{1}{\alpha}}$$

as  $t \to \infty$ , which yields  $y'(t)/y(t) \sim 1/Q(t)$  as  $t \to \infty$ , where  $Q(t) = [p(t)/((\alpha - 1)r(t))]^{-\frac{1}{\alpha}}$ . Take  $\varepsilon > 0$  and  $t_0$  such that

$$\frac{1-\varepsilon}{Q(t)} \le \frac{y'(t)}{y(t)} \le \frac{1+\varepsilon}{Q(t)},$$

 $t \ge t_0$ . Assume  $Q \in BSV$ . Then  $Q \in SN$ , because of continuity. Let  $\lambda > 0$  and take the integral between t and  $t + \lambda Q(t)$ , for which by the local uniform convergence the following relation holds,

$$\int_{t}^{t+\lambda Q(t)} \frac{1}{Q(v)} \, \mathrm{d}v = \int_{0}^{\lambda} \frac{Q(t)}{Q(t+zQ(t))} \, \mathrm{d}z \to \lambda$$

as  $t \to \infty$ . Thus we get

$$(1-\varepsilon)\lambda \leq \liminf_{t\to\infty} \ln \frac{y(t+\lambda Q(t))}{y(t)} \leq \limsup_{t\to\infty} \ln \frac{y(t+\lambda Q(t))}{y(t)} \leq (1+\varepsilon)\lambda.$$

Letting  $\varepsilon \to 0$  we come to  $y \in \Gamma(Q)$ . The above considerations are made under the assumptions  $\tilde{p} \in \mathcal{BSV}$  and  $Q \in \mathcal{BSV}$ . If we assume p, r differentiable, then after some rearrangement in the former equality — we get

$$\frac{\mathrm{d}}{\mathrm{d}s}\tilde{p}^{-\frac{1}{\alpha}}(s) = \left(-\frac{1}{\alpha}\left(\frac{r}{p}\right)^{\frac{\alpha+1}{\alpha}}\frac{p'r-pr'}{r^2} + (1-\beta)\left(\frac{r}{p}\right)^{\frac{1}{\alpha}}\frac{r'}{r}\right) \circ R^{-1}(s)$$

and

$$(\alpha - 1)^{-\frac{1}{\alpha}}Q'(t) = \left(-\frac{1}{\alpha}\left(\frac{r}{p}\right)^{\frac{\alpha+1}{\alpha}}\frac{p'r - pr'}{r^2}\right)(t).$$

Thus we see that in the case of differentiable coefficients it is sufficient to require the expressions on the right-hand sides to tend to zero as *s* resp. *t* tends to infinity. So we have obtained the following result, which extends Theorem 3.12.

**Theorem 3.14.** Let  $\int_a^{\infty} r^{1-\beta}(s) ds = \infty$ . Denote  $R(t) = \int_a^t r^{1-\beta}(s) ds$ . Then any of the two sets of conditions

(i) 
$$\left(\frac{p}{r^{1-\beta}}\right)^{-\frac{1}{\alpha}} \circ R^{-1} \in \mathcal{BSV}, \left(\frac{p}{r}\right)^{-\frac{1}{\alpha}} \in \mathcal{BSV},$$

or

(*ii*) 
$$p, r \text{ are differentiable, } \left( \left( \frac{p}{r} \right)^{-\frac{1}{\alpha}} \right)'(t) \to 0 \text{ and } \left( \left( \frac{p}{r} \right)^{-\frac{1}{\alpha}} \frac{r'}{r} \right)(t) \to 0 \text{ as } t \to \infty$$

guarantees

$$\emptyset \neq \mathbb{M}^+ = \mathbb{M}^+_{\infty} \subseteq \Gamma\left(\left(\frac{(\alpha-1)r}{p}\right)^{\frac{1}{\alpha}}\right).$$

A closer examination of the above observations reveals that the part of the proof of Theorem 3.12 starting with (3.90) can alternatively be shown via a different method, namely the one which was used to obtain Theorem 3.14.

The asymptotic conditions in the second part of Theorem 3.14 can be relaxed to 
$$\left(\left(\frac{p}{r}\right)^{-\frac{1}{\alpha}}\right)'(t) \to 0$$
 and  $\left(\left(\frac{p}{r^{1-\beta}}\right)^{-\frac{1}{\alpha}}\right)'(t) \cdot r^{\beta-1}(t) \to 0$  as  $t \to \infty$ .

The results of this subsection were further extended in [150] by Řehák and Taddei. Equation (3.83) is examined there directly (i.e., not via a transformation) no matter which of the conditions  $\int_{0}^{\infty} r^{1-\beta}(s) ds = \infty$  and  $\int_{0}^{\infty} r^{1-\beta}(s) ds < \infty$  happens. Moreover, not only increasing solutions but also decreasing solutions are considered.

### **3.6.2** Solutions in the class $\Pi$

In the paper [150] by Řehák and Taddei, half-linear extensions of the Geluk type results (see e.g. Theorem 2.11) were obtained. The paper is in preparation. We give — for illustration — only one statement without proof. By  $\mathbb{M}^-$  we mean the set of all eventually positive solutions of (3.83).

**Theorem 3.15.** Let  $p \in \mathcal{RV}(\delta)$  and  $r \in \mathcal{RV}(\delta + \alpha)$  with  $\delta < -1$ . If  $L_p(t)/L_r(t) \to 0$  as  $t \to \infty$ , where  $L_p$  and  $L_r$  are  $\mathcal{SV}$  components of p and r, respectively, then  $\mathbb{M}^- \subset \mathcal{NSV}$ . If  $y \in \mathbb{M}^-$ , then  $-y \in \Pi(-ty'(t))$ . Moreover,

(i) If 
$$\int_{a}^{\infty} \left(\frac{sp(s)}{r(s)}\right)^{\frac{1}{\alpha-1}} ds = \infty$$
, then  
$$y(t) = \exp\left\{-\int_{a}^{t} \left(\frac{sp(s)}{-(\delta+1)r(s)}\right)^{\frac{1}{\alpha-1}} (1+o(1)) ds\right\}$$

and  $y(t) \to 0$  as  $t \to \infty$ .

(*ii*) If 
$$\int_{a}^{\infty} \left(\frac{sp(s)}{r(s)}\right)^{\frac{1}{\alpha-1}} \mathrm{d}s < \infty$$
, then

$$y(t) = y(\infty) \exp\left\{\int_t^\infty \left(\frac{sp(s)}{-(\delta+1)r(s)}\right)^{\frac{1}{\alpha-1}} (1+o(1)) \,\mathrm{d}s\right\}$$

and  $y(t) \to y(\infty) \in (0, \infty)$  as  $t \to \infty$ .

Note that if we want to deal with SV solutions in the complementary case  $\delta > -1$ , we must look for them in the class  $\mathbb{M}^+$  (positive increasing solutions).

# 3.7 Half-linear differential equations having regularly varying solutions

Kusano and Marić in [94] deals with the question whether for any distinct real constants  $\vartheta_1$  and  $\vartheta_2$  there exists a differential equation of the form (3.1) which possesses a pair of solutions  $y_i \in \mathcal{RV}(\vartheta_i)$ , i = 1, 2. The function  $\omega$  which appears in the theorem is assumed to satisfy conditions from the definition of regularly varying functions with respect to  $\omega$ , see Definition 1.10.

For any  $|\vartheta_1| \neq |\vartheta_2|$  define

$$M(\vartheta_1, \vartheta_2) = \frac{|\vartheta_1|^{\beta} - |\vartheta_2|^{\beta}}{\vartheta_1 - \vartheta_2}, \quad N(\vartheta_1, \vartheta_2) = \frac{\vartheta_2 |\vartheta_1|^{\beta} - \vartheta_1 |\vartheta_2|^{\beta}}{\vartheta_1 - \vartheta_2}$$

and observe that

$$M(\vartheta_1,\vartheta_2) \leq 0 \Leftrightarrow \vartheta_1 + \vartheta_2 \leq 0.$$

**Theorem 3.16.** Let  $\vartheta_1$  and  $\vartheta_2$  be any given real constants such that  $|\vartheta_1| \neq |\vartheta_2|$ .

*(i)* Suppose that r satisfies

$$r^{1-\beta}(t) \sim K\omega^{M(\vartheta_1,\vartheta_2)-1}(t)\omega'(t) \tag{3.94}$$

as  $t \to \infty$  for some positive constant K. Let  $M(\vartheta_1, \vartheta_2) > 0$  and p be conditionally integrable on  $[a, \infty)$ . Then equation (3.1) possesses a pair of solutions  $y_i \in NRV_{\omega}(\Phi^{-1}(\vartheta_i)), i = 1, 2,$ if and only if

$$\lim_{t \to \infty} K^{\alpha - 1} \omega^{(\alpha - 1)M(\vartheta_1, \vartheta_2)}(t) \int_t^\infty p(s) \, \mathrm{d}s = \frac{N(\vartheta_1, \vartheta_2)}{M(\vartheta_1, \vartheta_2)}.$$
(3.95)

*(ii)* Suppose that r satisfies

$$r^{1-\beta}(t) = K\omega^{M(\vartheta_1,\vartheta_2)-1}(t)\omega'(t)$$

for some positive constant K. Let  $M(\vartheta_1, \vartheta_2) < 0$  and  $\omega^{\alpha} M(\vartheta_1, \vartheta_2) p$  be conditionally integrable on  $[a, \infty)$ . Then equation (3.1) possesses a pair of solutions  $y_i \in NRV_{\omega}(\Phi^{-1}(\vartheta_i))$ , i = 1, 2, if and only if

$$\lim_{t\to\infty} K^{\alpha-1} \omega^{|M(\vartheta_1,\vartheta_2)|}(t) \int_t^\infty \omega^{\alpha M(\vartheta_1,\vartheta_2)} p(s) \,\mathrm{d}s = \frac{(\alpha-1)N(\vartheta_1,\vartheta_2)}{|M(\vartheta_1,\vartheta_2)|}.$$

*Proof.* (i) Suppose  $M(\vartheta_1, \vartheta_2) > 0$ . For arbitrary real numbers  $\vartheta_1, \vartheta_2$  choose A in equation (3.13) as

$$A = \frac{N(\vartheta_1, \vartheta_2)}{M^{\alpha}(\vartheta_1, \vartheta_2)}.$$

Then, its only two real roots are

$$\lambda_i = \frac{\vartheta_i}{M^{\alpha - 1}(\vartheta_1, \vartheta_2)},$$

i = 1, 2. Further, choose function r in equation (3.1) such that for some constant K > 0,

$$r^{1-\beta}(t) \sim K \omega^{M(\vartheta_1, \vartheta_2)-1}(t) \omega'(t)$$

as  $t \to \infty$ . Whence due to  $R_{\alpha}(t) = \int_0^t r^{1-\beta}(s) \, ds$  and (3.94), condition (3.56) becomes (3.95) and an application of Theorem 3.6 asserts that equation (3.1) has two solutions

$$y_i \in \mathcal{NRV}_{R_{\alpha}}\left(\frac{\Phi^{-1}(\vartheta_i)}{M(\vartheta_1,\vartheta_2)}\right),$$

i = 1, 2. By applying the basic properties of generalized  $\mathcal{RV}$  functions,

$$\mathcal{RV}_{R_{\alpha}}\left(\frac{\Phi^{-1}(\vartheta_i)}{M(\vartheta_1,\vartheta_2)}\right) = \mathcal{RV}_{\omega M(\vartheta_1,\vartheta_2)}\left(\frac{\Phi^{-1}(\vartheta_i)}{M(\vartheta_1,\vartheta_2)}\right) = \mathcal{RV}_{\omega}(\Phi^{-1}(\vartheta_i)).$$

i = 1, 2, whence we conclude that the solutions  $y_i$ , i = 1, 2 of (3.1) belong to  $NRV_{\omega}(\Phi^{-1}(\vartheta_i))$ , i = 1, 2, as desired.

(ii) The proof of this part is similar to that of (i); this time we utilize Theorem 3.8.


# Emden-Fowler type equations and systems

# 4.1 Introductory and historical remarks

One of typical examples of the objects studied in this chapter is the second order *Emden-Fowler type* (or the *generalized Emden-Fowler* or the *quasilinear*) equation

$$(r(t)\Phi_{\alpha}(y'))' + p(t)\Phi_{\gamma}(y) = 0, \tag{4.1}$$

 $\Phi_{\lambda}(u) := |u|^{\lambda}$ , where  $\alpha, \gamma \in (0, \infty)$  and r > 0, p are continuous on  $[a, \infty)$ . If  $\alpha = \gamma$ , then (4.1) reduces to the half-linear equation of the form (3.2). Note that the power in  $\Phi_{\lambda}$  is shifted by one in comparison with the power in  $\Phi$  which was used in the previous chapter. But since here we start with zero, while formerly it was with one, the both nonlinearities are practically the same. We decided for these conventions because they are very usual in the literature when studying half-linear and quasi-linear differential equations.

First we give few historical remarks. In the study of stellar structure, the *Lane-Emden* equation

$$y''(t) + \frac{2}{t}y'(t) + y^{\gamma}(t) = 0$$
(4.2)

was considered. This equation was proposed by Lane [102] and studied in detail by Emden [31]. Fowler [41, 42] considered a generalization of this equation, called *Emden-Fowler* equation

$$ty'' + Cy'(t) + Dt^{\sigma}y^{\gamma} = 0.$$

Note that (4.2) has the self-adjoint form  $(t^2y')' + t^2y^{\gamma} = 0$ . By the change of variable  $t = 1/\xi$ , (4.2) becomes  $\frac{d^2y}{d\xi^2} + \frac{y^{\gamma}}{\xi^4} = 0$ , and by the change of variable  $y = \eta/t$ ,  $\frac{d^2\eta}{dt^2} + \frac{\eta^{\gamma}}{t^{\gamma-1}} = 0$ . The work of Emden also got the attention of physicists outside the field of astrophysics. For instance, the works of Thomas [164] and Fermi

[40] resulted in the *Thomas-Fermi* equation, used in atomic theory (see below for more details). The terminology is somehow confusing. Occasionally (4.2) is also called the *Lienard-Emden* equation and equation (4.4) mentioned below is called the *Emden-Fowler* equation (no matter what the sign is). Equation (4.4) with "+" is sometimes called the Fowler equation. All these equations are still extensively being investigated by physicists and mathematicians, and lot of other applications is known. Some people in the current literature (which is related to our topic) call the equation

$$(r(t)\Phi_{\alpha}(y'))' + \delta p(t)\Phi_{\gamma}(y) = 0,$$
(4.3)

where p(t) > 0 and  $\delta \in \{\pm 1\}$ , as of the *Emden-Fowler type* when  $\delta = 1$  and as of the *Thomas-Fermi type* when  $\delta = -1$ . The same terminology is used for corresponding higher order equations or systems.

As already indicated, equation (4.1) can be understood also as a natural generalization of half-linear equation (3.1).

If, in (4.3),  $\alpha = 1$  and  $0 < \gamma < 1$ , then we speak about *sub-linear* equation (at infinity), while if  $\alpha = 1$  and  $\gamma > 1$ , then we speak about *super-linear* equation. For general  $\alpha, \gamma \in (0, \infty)$  we call (4.3) *sub-half-linear* or *sub-homogeneous* provided  $\alpha > \gamma$ , while we call it *super-half-linear* or *super-homogeneous* when  $\alpha < \gamma$ . Analogously we use this terminology for related higher order equations and systems, and also for the objects where nonlinearities are somehow close to power functions.

As one of the motivations for considerations in this chapter can be the Thomas-Fermi atomic model described by the following nonlinear singular boundary value problem

$$y'' = \frac{1}{\sqrt{t}}y^{3/2},$$
  
 $y(0) = 1, \ y(\infty) = 0$ 

see Thomas [164], Fermi [40]; it is in fact a dimensionless form of the radially symmetric Poisson equation. As already mentioned, more general equation of physical interest is the Emden-Fowler one

$$\frac{\mathrm{d}}{\mathrm{d}s} \left( s^{\varrho} \frac{\mathrm{d}u}{\mathrm{d}s} \right) \pm s^{\sigma} u^{\gamma} = 0, \tag{4.4}$$

where  $\rho, \sigma, \gamma \in \mathbb{R}$ . For  $\rho \neq 1$ , it is reducible to the form  $y'' \pm t^{\tau}y^{\gamma} = 0$ , where  $\tau$  depends on  $\rho, \sigma, \gamma$ , and for  $\rho = 1$ , to the form  $y'' \pm e^{(\sigma+1)t}y^{\gamma} = 0$ , see Bellman [12, Chapter 7]. It is perhaps worthwhile mentioning Fowler's statement that even for  $p(t) \sim t^{\sigma}$  as  $t \to \infty$  (instead of  $p(t) = t^{\sigma}$ ) his method is not applicable.

Before we come to connections with regular variation, note for instance that Kamo and Usami in [70, 71] consider the equation

$$(\Phi_{\alpha}(y'))' = p(t)\Phi_{\gamma}(y), \tag{4.5}$$

where  $\alpha, \gamma > 0$  and  $p(t) \sim t^{\sigma}$  as  $t \to \infty$ . Under (natural) additional conditions on  $\alpha, \gamma, \sigma$ , they show that solutions *y* of (4.5) in a certain basic asymptotic class have

the form  $y(t) \sim Kt^{\delta}$ , where  $K = K(\alpha, \gamma, \sigma)$ ,  $\delta = \delta(\alpha, \gamma, \sigma)$ . The key role is played by the asymptotic equivalence theorem which says, roughly speaking: If the coefficients of two equations of the form (4.5) are asymptotically equivalent, then their solutions which are "of the same type" are also asymptotically equivalent. Of course, it is the equation with  $t^{\sigma}$  as the coefficient which is used for comparison purposes.

Realizing now that  $\mathcal{RV}$  functions can be understood as a (nontrivial) extension of functions asymptotically equivalent to power ones, we can ask natural questions: How about an extension in the sense that the coefficient in the Emden-Fowler equation is a regularly varying function? Or, how about an extension in the sense of replacing the nonlinearity in the Emden-Fowler equation by a regularly varying function? All these and also further related problems are discussed in this chapter. Note for example, that such results as the above asymptotic equivalence theorem cannot be used in a general case. We will describe various approaches, which are used either in the same settings or in different settings.

### 4.2 Former results

Before we present a survey of recent results, let us recall some classical works made in particular by Avakumović, Geluk, Marić, and Tomić.

#### 4.2.1 Superlinear second order equations in the work by Avakumović and related results

A substantial generalization of considerations about equation (4.4) was made by Avakumović in 1947, [9]. In fact, it is the first paper connecting regular variation and differential equations. It deals with the special type of (4.1), namely

$$y^{\prime\prime} = p(t)y^{\gamma},\tag{4.6}$$

where  $p \in \mathcal{RV}(\sigma)$  is a continuous function and  $\gamma > 1$ . He proved the following statement.

**Theorem 4.1** (Avakumović [9]). Let  $p \in \mathcal{RV}(\sigma)$  with  $\sigma > -2$  and  $\gamma > 1$ . If y is an eventually positive solution of (4.6) such that  $\lim_{t\to\infty} y(t) = 0$ , then

$$y(t) \sim \left(\frac{(1+\gamma+\sigma)(\sigma+2)}{(\gamma-1)^2}\right)^{\frac{1}{\gamma-1}} \left(t^2 p(t)\right)^{-\frac{1}{\lambda-1}}.$$

The proof is rather involved and proceeds by considering a suitable function h(t), satisfying the relations  $t^{\sigma+\gamma}L_p(t)h^{\gamma-1}(t) \sim (1+\gamma+\sigma)(2+\sigma)$  (with  $L_p(t) = p(t)/t^{\sigma}$ ) and  $h''(t) \sim p(t)h^{\gamma}(t)$ , and applying a "variation of constants"  $y(t) = h(t)z(\varphi(t))$ , where  $\varphi$  is a solution of  $h\varphi'' + 2h'\varphi' + \delta h\varphi'^2 = 0$ ,  $\delta > 0$ . The resulting differential equation for z is of the form  $z_{\varphi\varphi} - \delta z_{\varphi} = f(t)z(g(t)z^{\gamma-1} - 1)$ , where  $f(t) \ge \zeta > 0$  and  $g(t) \sim 1$ .

A refinement of the last asymptotic formula is made in Avakumović [10] along with applications which lead to refined asymptotic formulae for o(1) solutions of (4.6).

Only in 1991, Geluk in [46] presented a simple and elegant proof using a result on smoothly varying functions (see (1.11)) proved by Balkema, de Haan, and Geluk.

Geluk proved, in fact, the following statement.

**Theorem 4.2.** Let  $p \in \mathcal{RV}(\sigma)$  with  $\sigma > -2$  and  $\gamma > 1$ . If y is an eventually positive and bounded solution of (4.6), then  $y \in \mathcal{RV}(-(\sigma + 2)/(\gamma - 1))$ .

*Proof.* Substitutions  $u = y^{1-\gamma}$  and  $v(t) = \ln u(e^t)$  show that v satisfies the equation

$$v^{\prime\prime} - v^{\prime} - \beta v^{\prime 2} = -e^{\psi - v},$$

where  $\psi = \ln[(\gamma - 1)e^{2t}p(e^t)]$ ,  $\beta = 1/(\gamma - 1)$ . By the observations at (1.11), applied to the function  $(\gamma - 1)t^2p(t)$ , there exists a function  $\psi_1(t)$  such that  $\psi(t) - \psi_1(t) \rightarrow 0$ ,  $\psi'_1(t) \rightarrow \sigma + 2$ ,  $\psi''_1(t) \rightarrow 0$  as  $t \rightarrow \infty$ , and  $\psi_1(t) \leq \psi(t)$  for *t* sufficiently large. By substituting  $v(t) = \psi_1(t) + h(t)$ , the previous equation is reduced to

$$h'' - \omega h' - \beta h'^2 = -(1 + o(1))e^{-h} + (\sigma + 2)(1 + \beta \sigma + 2\beta) + o(1)$$

with  $\omega(t) \to 2\beta(\sigma + 2) + 1$  as  $t \to \infty$ . We claim that h(t) tends to a finite limit as  $t \to \infty$ . The following three cases are possible.

(i) h'(t) > 0 for  $t > t_0$ . Then *h* is ultimately increasing and its limit exists. If  $h(t) \to \infty$ , then by the preceding asymptotic equation, for  $t \ge t_0$ , one has  $h'' > \omega h' + \beta h'^2 > h'/2$ . This implies  $h'(t) \to \infty$ , and so, due to the mentioned equation,  $(-1/h'(t))' \to \beta$  as  $t \to \infty$ . Hence, integrating,  $-1/h'(t) \sim \beta t$ , which contradicts the assumption h'(t) > 0 for large *t*.

(ii) h'(t) < 0 for  $t > t_0$ . Then h is ultimately decreasing and its limit exists. The case when  $h(t) \to -\infty$  as  $t \to \infty$  is again disposed of. Because of  $\psi_1 \le \psi$ , the equation for v gives

$$-v'' + v' + \beta v'^2 = e^{\psi - v} \ge e^{\psi_1 - v} = e^{-h}.$$

Since  $-h(t) \to \infty$  as  $t \to \infty$ , there exists a sequence  $\{t_n\}$  such that  $v'(t_n) \to \pm \infty$  as  $n \to \infty$ . If  $v'(t_n) \to \infty$ , then  $h'(t_n) \to \infty$ , a contradiction. The case  $v'(t_n) \to -\infty$  implies  $u'(\exp t_n) < 0$ , hence  $y'(\exp t_n) > 0$  for *n* sufficiently large. Since y''(t) > 0, this contradicts the boundedness of *y*.

(iii) h'(t) oscillates. This implies the existence of a sequence  $\{t_n\}$  such that  $h'(t_n) = 0$  and  $t_n \to \infty$  as  $n \to \infty$ . If  $h''(t_n) < 0$  (i.e., h(t) attains its maximum for  $t_n$ ), then for large n one has  $h(t_n) < -\ln[(\sigma + 2)(1 + \beta\sigma + 2\beta)]$ . Similarly if  $h''(t_n) > 0$ , we find  $h(t_n) > -\ln[(\sigma + 2)(1 + \beta\sigma + 2\beta)]$ , a contradiction.

Thus h(t) tends to a finite constant as  $t \to \infty$  which then implies  $t^2 p(t) \sim k y^{1-\gamma}$  as  $t \to \infty$ . Hence, *y* is regularly varying of index  $-(\sigma + 2)/(\gamma - 1)$ .

**Remark 4.1.** The conclusion  $y \in \mathcal{RV}(-(\sigma + 2)/(\gamma - 1))$  implies that  $y(t) \to 0$ . Moreover, y'' is regularly varying as the product of two  $\mathcal{RV}$  functions. Application of Karamata's theorem then gives the exact value of the constant k, and thus we obtain the asymptotic formula from Theorem 4.1.

#### 4.2.2 Superlinear second order equations in the former works by Marić and Tomić

In this subsection we present results based on Marić [105] and Marić, Tomić [110, 111, 112].

We continue in considerations from the previous subsection. Neither Avakumović nor Geluk consider the border case  $\sigma = -2$  when the solutions tending to zero may still exists (see e.g. Wong [168], Taliaferro [158]). Consider the equation

$$y'' = p(t)F(y),$$
 (4.7)

where p(t) and F(y) are continuous and positive for t > 0 and y > 0. Wong's result reads as follows. Let F(t)/t increase. Then (4.7) has a positive solution which tends to zero if and only if

$$\int_{0}^{\infty} sp(s) \, \mathrm{d}s = \infty.$$

The above monotonicity condition is changed to

$$\limsup_{u \to 0} \sup_{0 < \lambda < d} \frac{F(\lambda u)}{\lambda F(u)} < 1$$

for some d > 0, by Taliaferro, not affecting the statement of the theorem.

Marić and Tomić (see [105, 110, 111, 112]) in their consideration neither the monotonicity nor the Taliaferro condition need always to hold, in which case the condition  $\int_{\infty}^{\infty} sp(s) ds = \infty$  is assumed to hold independently (and the existence of relevant solution as well). First we present the result concerning the estimates of solutions.

**Theorem 4.3.** Let  $p \in \mathcal{RB}$  and such that

$$\int_{-\infty}^{\infty} sp(s) \, \mathrm{d}s = \infty. \tag{4.8}$$

Assume that  $F \in \mathcal{RB}_0$  and such that

 $u^{-\gamma}F(u)$  almost decreases for some  $\gamma > 1$  (4.9)

as  $u \to 0$ . Then for every eventually positive solution y of (4.7) tending to zero there holds

$$\frac{F(y(t))}{y(t)} \approx \frac{1}{\int_{a}^{t} sp(s) \mathrm{d}s}$$
(4.10)

as  $t \to \infty$ . Moreover,  $F(u(t))/u(t) \in \mathcal{RB}$ .

*Proof.* Since the functions p and F are regularly bounded, in addition to hypotheses (4.8) and (4.9), due to a relation of  $\mathcal{RB}$  with almost monotonicity, there hold

 $t^{\delta}p(t)$  almost increases for some  $\delta$ , (4.11)

$$t^{\omega}p(t)$$
 almost decreases for some  $\omega < \delta$  (4.12)

for large *t*. Inequalities which yield (4.10) are then the results of a series of various estimates. This part of the proof is rather technical; for details see [105].  $\Box$ 

Notice that in view of the properties of  $\mathcal{RV}$  functions, if  $p \in \mathcal{RV}(-2)$ , then the integral  $I(t) = \int_a^t sp(s) ds$  is a new  $\mathcal{SV}$  function which cannot be disposed of in general by estimating it in a unique way. If e.g.  $p(t) = t^{-2} \ln t$ , then  $I(t) \sim \frac{1}{2} \ln^2 t$  as  $t \to \infty$ , whereas if  $f(t) = 1/(t^2 \ln t)$ , then  $I(t) \sim \ln \ln t$  as  $t \to \infty$ . However this is possible by restricting the rate of decay of function p. More precisely, the following statement holds.

**Corollary 4.1.** *Let*  $p \in RB$  *and such that for large t* 

 $t^{\delta}p(t)$  almost increases for some  $\delta < 2$ .

Assume that  $F \in \mathcal{RB}_0$  and such that

 $u^{-\gamma}F(u)$  almost decreases for some  $\gamma > 1$ 

as  $u \to 0$ . Then for every eventually positive solution y of (4.7) tending to zero there holds

$$\frac{F(y(t))}{y(t)} \asymp \frac{1}{t^2 p(t)}$$

as  $t \to \infty$ .

*Proof.* Since  $\delta < 2$ , condition (4.8) is fulfilled and the previous theorem applies. Furthermore, due to (4.11),

$$\int_{a}^{t} sp(s) \, \mathrm{d}s \le M t^{\delta} p(t) \int_{a}^{t} s^{1-\delta} \mathrm{d}s \le t^{2} p(t),$$

which gives the left-hand side of the required inequality. The right-hand one is obtained likewise by using (4.11) instead of (4.11).

Next we describe the asymptotic behavior of solutions to (4.7), where we assume that instead of being regularly bounded, the functions p and F are in  $\mathcal{RV}$  and  $\mathcal{RV}_0$ , respectively. Then instead of estimates for large t we obtain a precise asymptotic behavior of the function F(y(t))/y(t) as  $t \to \infty$ . We also tacitly assume that Wong's condition (4.8) holds in order not to violate the existence of the considered solutions y tending to zero even in the simplest cases e.g. such as  $y'' = t^2 L_p(t)y^{\gamma}$ .

#### Theorem 4.4. Assume

$$p(t) = t^{\sigma}L_p(t), \ \sigma \geq -2, \ F(u) = u^{\gamma}L_F(u), \ \gamma > 1,$$

where  $L_p \in SV$  and  $L_F \in SV_0$ . Then for every solution y(t) of (4.7) tending to zero as  $t \to \infty$  there holds for  $t \to \infty$ :

(*i*) for  $\sigma > -2$ 

$$y^{\gamma-1}(t)L_F(y(t)) \sim \frac{(1+\sigma+\gamma)(2+\sigma)}{(\gamma-1)^2} \cdot \frac{1}{t^{2+\sigma}L_p(t)}$$

and  $y \in \mathcal{RV}((\sigma + 2)/(1 - \gamma));$ (*ii*) for  $\sigma = -2$ 

$$y^{\gamma-1}(t)L_F(y(t)) \sim \frac{1}{(\gamma-1)\int_a^t L_p(s)/s \, \mathrm{d}s}$$

and  $y \in SV$ .

*Proof.* The proof is rather technically complicated. We briefly mention only the main ideas. The details can be found in [105, 112]. Put

$$I(t) := \int_{a}^{t} \frac{1}{s^{2}} \int_{a}^{s} u^{2} p(u) \, \mathrm{d}u \, \mathrm{d}s,$$
  

$$G(y) := \int_{0}^{y} \frac{1}{t} \int_{0}^{t} \frac{F(u)}{u^{2}} \, \mathrm{d}u \, \mathrm{d}t,$$
  

$$Z(t) := I(t)G(y(t)).$$

The function Z satisfies the differential equation

$$\frac{Z''}{Z} \cdot \frac{I}{I'} = 2\frac{Z'}{Z} + \frac{I''}{I'} - \frac{2I'}{I} + Z\frac{p}{I'} \cdot \frac{FG'}{G^2} + \left(\frac{Z'}{Z} - \frac{I'}{I}\right)^2 \frac{I}{I'} \cdot \frac{GG''}{G'^2}.$$

The behavior of all intervening functions (coefficients) is then determined. Among others, it is shown that

$$G(y) \sim \frac{F(y)}{(\gamma - 1)^2 y}$$

as  $y \to 0$ ,

$$I(t) \sim t^2 p(t) / ((\sigma + 3)(\sigma + 2))$$

as  $t \to \infty$ , and

$$\lim_{t\to\infty} Z(t) = c > 0.$$

In addition to various estimates and properties of  $\mathcal{RV}$  functions, also the Avakumović theorem (Theorem 4.1) and estimates (4.10) find application in proving the above estimations. The asymptotic formula in the part (i) is then the result of these relations and some additional observations; among others we apply the following claim: Since  $H(y) = y^{\gamma-1}L_F(y)$  is in  $\mathcal{RV}_0(\gamma - 1)$ , we find  $\tilde{H} \in \mathcal{RV}_0(1/(\gamma - 1))$  such that  $\tilde{H}(H(y)) \sim y$ . Concerning the part (ii), in contrast to the part (i), first it is proved that  $y \in \mathcal{SV}$  and then the asymptotic formula is derived (using analogous arguments to those in the part (i)).

**Remark 4.2.** The natural question of how to obtain asymptotic behavior of solutions *y* by inverting formulae in Theorem 4.4 turns out to be in general a very difficult one (but not related to differential equations). In fact, that theorem gives only that solutions are of the form

$$y(t) = \begin{cases} t^{(\sigma+2)/(1-\gamma)}L_1(t) & \text{for } \sigma > -2\\ L_2(t) & \text{for } \sigma = -2 \end{cases}$$

where SV functions  $L_1, L_2$  satisfy the following implicit asymptotic relations

$$L_1^{\gamma-1}(t)L_F(t^{(\sigma+2)/(1-\gamma)}L_1(t)) \sim \frac{(1+\sigma+\gamma)(2+\sigma)}{(\gamma-1)^2} \cdot \frac{1}{L_p(t)},$$
$$L_2^{\gamma-1}(t)L_F(L_2(t)) \sim \frac{1}{(\gamma-1)\int_a^t L_p(s)/s \, \mathrm{d}s}$$

as  $t \to \infty$ .

#### 4.2.3 Rapidly varying coefficient and nonlinearity

In the previous subsection it is assumed for the functions p and F in (4.7) to be  $\mathcal{RV}$ and  $\mathcal{RV}_0$ , respectively. Therefore that analysis did not include even such simple cases as  $p(t) = e^t$  or  $F(u) = e^{-1/u}$ . These examples indicate that we intend to keep as the main feature of equation (4.7) — to be superlinear and such that  $F(u) \rightarrow 0$ as  $u \rightarrow 0$ . This type of problem was considered in Marić [105], see also Marić, Radaśin [106, 107, 108].

Let us write equation (4.7) as

$$y'' = f(g(t))\varphi(\psi(t)).$$
 (4.13)

We assume that g(t) is positive, increasing to infinity as  $t \to \infty$ , twice differentiable, and such that there exists the limit

$$\lim_{t\to\infty}\frac{g(t)g''(t)}{{g'}^2(t)}=A.$$

For  $\psi(u)$  we assume that it is positive, decreasing to zero as  $u \to 0$ , twice differentiable, and such that there exists the limit

$$\lim_{u\to 0}\frac{\psi(u)\psi''(u)}{\psi'^2(u)}=B.$$

Recall that then  $A, B \le 1$ ; if A < 1, then  $g \in \mathcal{RV}(1/(1 - A))$ ; if A = 1, then  $g \in \mathcal{RPV}$ ; if B < 1, then  $\psi \in \mathcal{RV}_0(1/(1 - B))$ ; if B = 1, then  $\psi \in \mathcal{RPV}_0$ .

**Theorem 4.5.** Let  $f \in \mathcal{RV}(\sigma)$  with  $\sigma + 2 - 2A \ge 0$  for  $A \ne 1$  and  $\sigma > 0$  for A = 1, and  $\varphi \in \mathcal{RV}_0(\gamma)$  with  $\gamma > 1 - B$ . Assume that A and B are not both equal to one. Then for all positive solutions y(t) of (4.13) which tend to zero as  $t \rightarrow \infty$ , there holds

$$\frac{\psi'(y(t))}{\psi(y(t))}\varphi[\psi(y(t))] \sim \frac{(1-A)(\gamma+B-1)+(1-B)(\sigma+2-2A)}{(\gamma+B-1)^2 \int_a^t \frac{g(s)}{g'(s)} f(g(s)) \,\mathrm{d}s}$$

as  $t \to \infty$ .

*Proof.* Basically, the proof runs along the same line as the proof of Theorem 4.4, except that here we do not have estimates as (4.10), which requires some additional arguments. We put

$$I(t) := \int_{a}^{t} \left(\frac{g'(s)}{g(s)}\right)^{2} \int_{a}^{s} \left(\frac{g(u)}{g'(u)}\right)^{2} f(g(u)) \, \mathrm{d}u \, \mathrm{d}s,$$
  

$$G(y) := \int_{0}^{y} \frac{\psi'(s)}{\psi(s)} \int_{0}^{s} \left(\frac{\psi(u)}{\psi'(u)}\right)^{2} \varphi(\psi(u)) \, \mathrm{d}u \, \mathrm{d}s,$$
  

$$Z(t) := I(t)G(y(t)).$$

The function Z satisfies the differential equation

$$\frac{Z''}{Z} \cdot \frac{I}{I'} = 2\frac{Z'}{Z} + \frac{I''}{I'} - \frac{2I'}{I} \cdot \frac{FG'}{G^2} + \left(\frac{Z'}{Z} - \frac{I'}{I}\right)^2 \frac{I}{I'} \cdot \frac{GG''}{G'^2}.$$

For details see [105].

**Remark 4.3.** Somehow related results were obtained by Taliaferro (see [158, 160] and also [105, Section 3.6]) for the more general equation y'' = F(t, y, y'). He uses some definitions which are — as stated there — "partially motivated" by  $\mathcal{RV}$  functions, but his methods make no use of Karamata functions which is the subject of this treatise.

# 4.3 Selection of recent results

#### 4.3.1 Introductory remarks

In the last decade, many papers have appeared (and still are appearing) which are devoted to the investigation of various forms of Emden-Fowler type equations (incl. systems and higher order equations) in the framework of regular variation. It is impossible to present here all these results. Instead of this we prefer to make a reasonable selection of typical results which will show a wide variety of methods which are employed — this is in accordance with the principal aim of our treatise.

It is worthy of note that there are some overlaps in the papers in spite of different methods. However, this fact is quite natural, when one realizes that a typical result is of the form: It the coefficients in an equation are  $\mathcal{RV}$ , then (at

least on or all) solutions in certain basic asymptotic classes are  $\mathcal{RV}$  (with the index which depends on the indices of coefficients and powers of nonlinearities) and their behavior is governed by a specific asymptotic formula.

Before we present selected results, let us try to give at least a brief (but comprehensive) survey of the literature devoted to this topic, that is investigation of quasilinear equations in the framework of regular variation.

First note that when we study  $\mathcal{RV}$  (or somehow similar) solutions we consider nonoscillatory solutions. There is an extensive literature devoted to classifications of nonoscillatory solutions and investigation of the existence and behavior of solutions in the individual classes. One of the most important publications along this line is the book [79] by Kiguradze and Chanturia. Instead of quoting other works, note that some related references are spread in this text, in the places where we utilize the relevant results.

There are various approaches to examine Emden-Fowler type equations in the framework of regular variation and several independent groups which work on it.

Existence and asymptotic behavior of  $\mathcal{RV}(1)$  increasing solutions of the second order Thomas-Fermi type equation

$$y^{\prime\prime} = p(t)F(x),$$

p(t) > 0, in the sub-linear case were studied by Kusano, Manojlović, Marić in [83]; other  $\mathcal{RV}$  solutions of the same problem are examined in [104] by Manojlović and Marić. The generalized Thomas-Fermi equation

$$(\Phi_{\alpha}(y'))' = p(t)\Phi_{\gamma}(y),$$

p(t) > 0, is considered in the sub-homogeneous case ( $\alpha > \gamma$ ) by Kusano, Marić, Tanigawa in [100]; existence conditions and formulas for SV, RV(1) solutions and for nearly RV solutions are established. An asymptotic analysis of RV solutions to the Emden-Fowler type equation

$$y'' + p(t)\Phi_{\gamma}(y) = 0,$$

p(t) > 0, is made in the sub-linear case in [84] and existence of SV, RV(1) solutions is discussed for the same equation in both the sub- and super-linear cases in [85] by Kusano and Manojlović. The same authors study the sublinear Emden-Fowler type equation

$$y^{\prime\prime} + p(t)F(x) = 0,$$

p(t) > 0, in [86]; existence and asymptotic behavior of SV, RV(1) solutions are examined, and necessary conditions for the existence along with asymptotic formulas for intermediate solutions are established. A generalized Emden-Fowler type equation of the form

$$(r(t)\Phi_{\alpha}(y'))' + p(t)\Phi_{\gamma}(y) = 0,$$

r(t), p(t) > 0,  $\alpha > \gamma$ , is considered in [91] by Kusano, Manojlović, and Milošević and in [64] by Jaroš, Kusano and Manojlović; its intermediate generalized  $\mathcal{RV}$ solutions are discussed. The fourth order sublinear equation

$$y^{(4)} + \delta p(t) \Phi_{\gamma}(y) = 0,$$

 $p(t) > 0, 0 < \gamma < 1$ , is considered by Kusano and Manojlović in [87] under the condition  $\delta = 1$  and in [88] under the condition  $\delta = -1$ ; existence of (all possible)  $\mathcal{RV}$  solutions and their accurate behavior are examined. This equation with  $\delta = 1$  is examined also by Kusano, Manojlović, and Tanigawa in [63]. The more general equation

$$(\Phi_{\alpha}(y''))'' + p(t)\Phi_{\gamma}(y) = 0,$$

p(t) > 0,  $\alpha > \gamma > 0$ , is considered by the same authors in [93]; all possible types of positive solutions are examined. Jaroš, Kusano, and Tanigawa in [67] discuss the existence and asymptotic behavior of  $\mathcal{RV}$  solutions of the third order sublinear equation

$$y^{\prime\prime\prime}+p(t)\Phi_{\gamma}(y)=0,$$

p(t) > 0, see also the papers [68, 69] by the same authors. The two-dimensional system

$$x' = \delta p(t) y^{\alpha}, \quad y' = \delta q(t) x^{\beta},$$

p(t), q(t) > 0,  $\alpha$ ,  $\beta > 0$ ,  $\alpha\beta < 1$ ,  $\delta = \pm 1$ , is considered by Jaroš and Kusano in [61]; strongly monotone  $\mathcal{RV}$  solutions are analyzed, see also [63]. The same authors study strongly monotone  $\mathcal{RV}$  solutions of the second order system

$$x^{\prime\prime} = p(t)y^{\alpha}, \ y^{\prime\prime} = q(t)x^{\beta},$$

p(t), q(t) > 0,  $\alpha, \beta > 0$ ,  $\alpha\beta < 1$ , see [62], and strongly decreasing solutions of the system

$$x'' = p_1(t)x^{\alpha_1} + q_1(t)y^{\beta_1}, \quad y'' = p_2(t)x^{\alpha_2} + q_2(t)y^{\beta_2},$$

where  $p_1, q_1, p_2, q_2 \in \mathcal{RV}$  and  $\alpha_1, \beta_1, \alpha_2, \beta_2 > 0$ , as presented by Tanigawa at the conference Equadiff 13. Kusano and Manojlović consider the odd-order equation

$$y^{(2n+1)} + \delta p(t)\Phi(\gamma) = 0,$$

p(t) > 0,  $0 < \gamma < 1$ . Existence and asymptotic behavior of all possible types of positive solutions to this equation is studied in the framework of regular variation; the case  $\delta = 1$  in [89] and the case  $\delta = -1$  in [90].

In many results of the above mentioned papers, various modifications of the following technique is utilized: Certain asymptotic relation is investigated which can be considered as an "approximation" of the given differential equation rewritten to a certain integral form; properties of  $\mathcal{RV}$  functions — mainly the Karamata integration theorem — are then extensively used there. A priori bounds are obtained. Further, the Schauder-Tychonoff fixed point theorem in locally convex

spaces plays an important role. For some of the equations there has been made a complete analysis of the existence and asymptotic behavior of regularly varying solutions which belong to all possible Kiguradze type classes. Various utilizations of this technique are presented in Subsections 4.3.2, 4.3.3, 4.3.4, and 4.3.6.

A somehow different approach is represented by the works of Evtukhov and others. Note that — in some of these works — the conditions which are usually imposed on the coefficients and nonlinearities in the equation are not directly in terms of regular variation (or rapid variation), but in fact they belong to these classes (or are close to them) due to known results (like the Karamata theorem) which enables an alternative expression. Similar observations hold for some of the classes of solutions which are studied. It is worthy of note that some of these papers (especially the older ones) do not use (or even mention) the concept of regular variation at all, even though the considerations are "close" to regular variation. Typically, both types of equations (Emden-Fowler and Thomas Fermi) are simultaneously studied and also both the cases (sub- and super-linearity) are considered. Even a negative power or index in a nonlinearity is sometimes allowed. The conditions for the existence of solutions in various asymptotic classes (they are called  $P_{\omega}$  and have various modifications) are derived and asymptotic representations are established. Evtukhov and Samoilenko in [36] consider the equation

$$y^{(n)} = \delta p(t) F(t),$$

 $p(t) > 0, \delta = \pm 1$ . The second order case is studied by Evtukhov and Kharkov in [34] and by Evtukhov and Abu Elshour in [32], and the third order case by Evtukhov and Stekhun [37]. Bilozerowa and Evtukhov in [13] examine the generalization of the Emden-Fowler equation in the following form

$$y^{(n)} = \delta p(t) \prod_{i=0}^{n-1} F_i(y^{(i)}),$$

 $p(t) > 0, \delta = \pm 1$ ; the second order case is considered in [33]. The equation

$$y'' = \sum_{i=1}^{m} \delta_i p_i(t) (1 + q_i(t)) F_{i0}(y) F_{i1}(y')$$

is studied by Kozma in [82]. Evtukhov and Vladova in [39] concentrate on the cyclic system

$$y_i' = \delta_i p_i(t) F_i(y_{i+1}),$$

 $i = 1, ..., n, p_i(t) > 0, \delta_i = \pm 1, y_{n+1}$  means  $y_1$ ; the two dimensional case is studied in [38]. Subsection 4.3.7 offers a more detailed description of the method used by Evtukhov et al.

Another direction of the approach in the study of objects related to Emden-Fowler type equations and regular variation is represented by the following three works. A substantial part of these works is devoted to a generalization of the results described after (4.5). In other words, typical feature is that not only the existence of  $\mathcal{RV}$  solutions is studied and asymptotic formulas are established, but, in addition,  $\mathcal{RV}$  behavior of all solutions in a given basic asymptotic class of solutions is guaranteed. Only some of the Kiguradze type subclasses have been examined in such a way, namely the most extreme ones. Matucci and Řehák in [123] study decreasing  $\mathcal{RV}$  solutions of the coupled system

$$\begin{cases} (p(t)\Phi_{\alpha}(x'))' = \varphi(t)\Phi_{\lambda}(y), \\ (q(t)\Phi_{\beta}(y'))' = \psi(t)\Phi_{\mu}(x), \end{cases}$$

 $p(t), r(t), \varphi(t), \psi(t) > 0, \alpha\beta > \lambda\mu$ , see Subsection 4.3.5 for couple of notes. Strongly monotone solutions of the system

$$y_i' = \delta p_i(t) F_i(y_{i+1}),$$

 $i = 1, ..., n, p_i(t) > 0, \delta = \pm 1, y_{n+1}$  means  $y_1$ , are investigated by Řehák in [146] and by Matucci and Řehák in [124]. See Subsection 4.3.8 where this approach is discussed in details. A slight modification of the setting of the latter paper enables us to include also equations with a general  $\Phi$ -Laplacian, see Subsection 5.1.2.

# **4.3.2** Asymptotic behavior of *SV* and *RV*(1) solutions of sublinear second order equations

The result in this section is selected from the paper [86] by Kusano and Manojlović as a representant of the below described method. Consider the second order Emden-Fowler type equation

$$y'' + p(t)F(y) = 0, (4.14)$$

where  $p : [a, \infty) \to (0, \infty)$  is a continuous function with  $p \in \mathcal{RV}(\sigma)$ ,  $\sigma \in \mathbb{R}$ , and  $F : (0, \infty) \to (0, \infty)$  is an increasing continuous function with  $F \in \mathcal{RV}(\gamma)$ . The sublinearity condition  $0 < \gamma < 1$  is assumed. Recall that the following generalization of the known Belohorec theorem holds: Equation (4.14) has a positive solution if and only if  $\int_{a}^{\infty} p(s)F(s) ds < \infty$ .

Suppose that (4.14) has a positive solution y (called *intermediate*) such that

$$\lim_{t \to \infty} \frac{y(t)}{t} = 0 \text{ and } \lim_{t \to \infty} y(t) = \infty.$$
(4.15)

From the conventional classification of eventually positive solutions, in addition to intermediate solutions, only the two following (and somehow easier) classes are possible:  $\lim_{t\to\infty} y(t) = \text{const} > 0$  (the so-called *minimal* solutions) or  $\lim_{t\to\infty} x(t)/t = \text{const} > 0$  (the so-called *maximal* solutions); the terminology may come from the fact that a positive solution *y* always satisfies  $c_1 \le x(t) \le c_2 t$ ,  $t \ge T_x$ , for some positive constants  $c_1, c_2 \in \mathbb{R}$ . It is clear that a minimal solution is in tr-SV and a maximal solution is in tr-RV(1). However, as we will see, there can also be nontrivial SV or nontrivial RV(1) solutions which, of course, must be sought among intermediate solutions. Note that in the sub-linear case, the sufficient and necessary condition for the existence of a solution *y* satisfying (4.15) reads as

$$\int_{a}^{\infty} p(s)F(s)\,\mathrm{d} s < \infty \text{ and } \int_{a}^{\infty} sp(s)\,\mathrm{d} s = \infty,$$

see Kusano, Ogata, Usami [101]. Since  $y'(\infty) = 0$ , integrating (4.14) first from *t* to  $\infty$  and then from *a* to *t*, we have

$$y(t) = y(a) + \int_a^t \int_s^\infty p(\tau) F(y(\tau)) \,\mathrm{d}\tau \,\mathrm{d}s, \tag{4.16}$$

 $t \ge a$ . It is somehow natural to search for a solution of (4.14) with specific asymptotic behavior at infinity as a fixed point of the integral operator

$$\mathcal{F}y(t) = C + \int_a^t \int_s^\infty p(\tau) F(y(\tau)) \,\mathrm{d}\tau \,\mathrm{d}s,$$

 $t \ge a$ , in some suitably chosen set X of  $C[a, \infty)$ . A thorough analysis can be made of the existence and the precise asymptotic behavior of  $\mathcal{RV}$  solutions of the integral asymptotic relation

$$y(t) \sim \int_{a}^{t} \int_{s}^{\infty} p(\tau) F(y(\tau)) \, \mathrm{d}\tau \, \mathrm{d}s, \quad t \to \infty,$$
(4.17)

which can be considered as an "approximation" of (4.16). Note that (4.17) follows from (4.16), using that  $y(\infty) = \infty$ . Then the set X with the required properties can be found by means of  $\mathcal{RV}$  solutions of the integral asymptotic relation (4.17).

It is worthy of note that an important role in the proof is played by the fact that the auxiliary linear second order equation possesses SV and RV(1) solutions. More precisely, we will utilize the following statement which follows from Theorem 2.2 and a simple application of the properties of RV functions to the (positive) solution  $y(t) = c_1y_1(t) + c_2y_2(t)$ ,  $c_1, c_2 \in \mathbb{R}$ ,  $\{y_1, y_2\}$  being a fundamental set of RV solutions.

Lemma 4.1. If

$$\lim_{t\to\infty}t\int_t^\infty q(s)\,\mathrm{d}s=0,$$

then every (eventually) positive solution of the equation y'' + q(t)y = 0, q(t) > 0, is SV or RV(1).

We now give the conditions guaranteeing the existence of nontrivial SV and RV(1) solutions and establish asymptotic formulae.

**Theorem 4.6.** Suppose that  $p \in \mathcal{RV}(\sigma)$  and  $F \in \mathcal{RV}(\gamma)$  with  $\gamma \in (0, 1)$ .

(*i*) Equation (4.14) possesses nontrivial SV solutions if and only if  $\sigma = -2$  and  $\int_a^{\infty} sp(s) ds = \infty$ . The asymptotic behavior of each nontrivial SV solution y is governed by the unique formula

$$y(t) \sim F^{-1}\left(\int_a^t sp(s)\,\mathrm{d}s\right)$$

as  $t \to \infty$ .

(ii) Let, moreover,

$$F(tu(t)) \sim F(t)u^{\gamma}(t) \tag{4.18}$$

as  $t \to \infty$ , for every  $u \in SV \cap C^1$ . Equation (4.14) possesses nontrivial RV(1) solutions if and only if  $\sigma = -\gamma - 1$  and  $\int_a^{\infty} p(s)F(s) ds < \infty$ . The asymptotic behavior of any nontrivial RV(1) solution y is governed by the unique formula

$$y(t) \sim t \left( (1 - \gamma) \int_t^\infty p(s) F(s) \, \mathrm{d}s \right)^{\frac{1}{1 - \gamma}}$$

as  $t \to \infty$ .

*Proof.* We give the proof only of the "if" part. The "only if" part is proved in [86] and extensively uses the Karamata integration theorem and some basic properties of  $\mathcal{RV}$  functions. Let  $x_0(t), x_1(t)$  be the functions on  $[a, \infty)$  defined by

$$x_0(t) = F^{-1}\left(\int_a^t sp(s) \,\mathrm{d}s\right),$$
$$x_1(t)t\xi_1^{\frac{1}{1-\gamma}}(t), \text{ where } \xi_1(t) = (1-\gamma)\int_t^\infty p(s)F(s) \,\mathrm{d}s \in \mathcal{SV}.$$

The proof will be performed simultaneously and in two steps. In the first step we show that (4.14) possesses intermediate solution y(t) satisfying  $kx_0(t) \le y(t) \le Kx_0(t)$  or  $kx_1(t) \le y(t) \le Kx_1(t)$ ,  $t \ge T$ , for some positive constants T > a, k < 1, K > 1 depending on whether the assumptions in (i) or (ii) are satisfied, respectively. Then, in the second step, with the help of Lemma 4.1, we show that solutions constructed in the first step are nontrivial SV, resp. RV(1). Since we perform the proofs for  $x_i(t)$ . i = 0, 1, simultaneously, the subscripts i = 0, 1 will be deleted in the rest of the proof.

Step 1. It is not difficult to show — the Karamata integration theorem plays an important role and we use (4.18) — that *x* satisfies the asymptotic relation

$$x(t) \sim \int_{a}^{t} \int_{s}^{\infty} p(\tau) F(x(\tau)) \,\mathrm{d}\tau \,\mathrm{d}s \tag{4.19}$$

as  $t \to \infty$ . Let *K*, *k* be fixed positive constants such that  $K^{1-\gamma} \ge 4$  and  $k^{1-\gamma} \le 1/2$ . Note that k < 1 and K > 2. Using that  $F(Kx(t)) \sim K^{\gamma}F(x(t))$ , from (4.19) we have

$$\int_a^t \int_s^\infty p(\tau) F(Kx(\tau)) \, \mathrm{d}\tau \, \mathrm{d}s \sim K^\gamma x(t)$$

as  $t \to \infty$ , which implies the existence of  $T_0 > a$  depending only on *K* such that

$$\int_{T_0}^t \int_s^\infty p(\tau) F(Kx(\tau)) \, \mathrm{d}\tau \, \mathrm{d}s \le 2K^\gamma x(t),$$

 $t \ge T_0$ . Let such a  $T_0$  be fixed. We may assume that x(t) is increasing on  $[T_0, \infty)$ . Since  $x(t) \to \infty$  and  $F(kx(t)) \sim k^{\gamma} F(x(t))$  as  $t \to \infty$ , from (4.19) we have

$$\int_{T_0}^t \int_s^\infty p(\tau) F(kx(\tau)) \, \mathrm{d}\tau \, \mathrm{d}s \sim k^\gamma x(t)$$

as  $t \to \infty$ , and so there exists  $T_1 > T_0$  depending only on *k* such that

$$\int_{T_0}^t \int_s^\infty p(\tau) F(kx(\tau)) \,\mathrm{d}\tau \,\mathrm{d}s \sim \frac{k^\gamma}{2} x(t),$$

 $t \ge T_1$ . Let such a  $T_1$  be fixed. Let us define X to be the set of continuous functions y(t) on  $[T_0, \infty)$  satisfying

$$\begin{cases} x(T_0) \le y(t) \le Kx(t) & \text{for } T_0 \le t \le T_1, \\ kx(t) \le y(t) \le Kx(t) & \text{for } t \ge T_1. \end{cases}$$

$$(4.20)$$

It is clear that X is a closed convex subset of the locally convex space  $C[T_0, \infty)$  equipped with the topology of uniform convergence on compact subintervals of  $[T_0, \infty)$ . We now define the integral operator

$$\mathcal{F}y(t) = x(T_0) + \int_{T_0}^t \int_s^\infty p(\tau)F(y(\tau))\,\mathrm{d}\tau\,\mathrm{d}s,$$

 $t \ge T_0$ , and let it act on the set X defined above. It is not difficult to verify that  $\mathcal{F}$  is a self-map on X and sends X continuously to a relatively compact subset of  $C[T_0, \infty)$ . Therefore we are able to apply the Schauder-Tychonoff fixed point theorem to conclude that there exists  $y \in X$  such that  $y(t) = \mathcal{F}y(t), t \ge T_0$ . It is clear from (4.20) that y satisfies

$$kx(t) \le y(t) \le Kx(t), \tag{4.21}$$

 $t \ge T_0$ , which completes the proof of the first step.

*Step* 2. Let *y* be a solution obtained in the first step. It can be regarded as a solution of the linear differential equation y'' + q(t)y = 0 with  $q(t) = p(t)\frac{F(y(t))}{y(t)}$ . We will show that

$$\lim_{t \to \infty} t \int_t^\infty p(s) \frac{F(y(s))}{y(s)} \, \mathrm{d}s = 0.$$
(4.22)

Since y satisfies (4.21) it suffices to show that

$$\lim_{t \to \infty} t \int_t^\infty p(s) \frac{F(x(s))}{x(s)} \, \mathrm{d}s = 0.$$

Suppose first that the assumption of part (i) are satisfied. The using that  $x(t) = x_0(t) \in SV$  is a solution of the differential equation  $x'_0 = tp(t)F(x_0)$  and applying the L'Hospital rule we obtain

$$\lim_{t \to \infty} t \int_t^\infty p(s) \frac{F(x(s))}{x(s)} \, \mathrm{d}s = \lim_{t \to \infty} \frac{t x_0'(t)}{x_0(t)},$$

since for our  $SV x_0$  the last limit is equal to 0. If the assumptions of part (ii) are satisfied, then  $y(t) = x_1(t) \in RV(1)$ . Using (4.18) with application of the L'Hospital rule, we get

$$\lim_{t \to \infty} t \int_{t}^{\infty} p(s) \frac{F(x(s))}{x(s)} ds = \lim_{t \to \infty} t \int_{t}^{\infty} s^{-\gamma - 1} L_p(s) \frac{F(s)\xi_1^{\frac{1}{1-\gamma}}(s)}{s\xi_1^{\frac{1}{1-\gamma}}(s)} ds$$
$$= \lim_{t \to \infty} t \int_{t}^{\infty} \frac{s^{-2} L_p(s) L_F(s)}{\xi_1(s)} ds$$
$$= \lim_{t \to \infty} \frac{L_p(t) L_F(t)}{\xi_1(t)}.$$

Since

$$\int_{a}^{\infty} \frac{L_{p}(s)L_{F}(s)}{s} \, \mathrm{d}s = \int_{a}^{\infty} p(s)F(s) \, \mathrm{d}s < \infty,$$

by the Karamata theorem we get

$$\lim_{t \to \infty} \frac{L_p(t)L_F(t)}{\int_t^\infty L_p(s)L_F(s)/s \, \mathrm{d}s} = 0,$$

and so (4.22) follows in both the cases (i) and (ii). It follows from Lemma 4.1 that  $y \in SV \cup RV(1)$ . But if the assumptions in (i) hold, noting that  $x(t) = x_0(t) \in SV$ , we conclude that y must be SV, and if the assumptions in (ii) hold, since  $x(t) = x_1(t) \in RV(1)$ , solution y must be RV(1).

Condition (4.18) amounts to requiring that SV part  $L_F$  of F satisfies

$$L_F(tu(t)) \sim L_F(t) \tag{4.23}$$

γ

as  $t \to \infty$  for every  $u \in SV \cap C^1$ . Compare this condition with (4.76), (4.95), and (4.96). Many (usual) SV functions satisfy (4.23). For instance,  $L_F(t) = \prod_{k=1}^N (\ln_k t)^{\alpha_k}$ ,  $\alpha_k \in \mathbb{R}$ , or  $L_F$  such that  $L_F(t) \to c \in (0, \infty)$  as  $t \to \infty$ , etc. Condition (4.23) is not satisfied e.g. by  $L_F(t) = \exp\left(\prod_{k=1}^N (\ln_k t)^{\beta_k}\right), \beta_k \in (0, 1).$ 

#### 4.3.3 *SV* solutions of second order super-linear equations

In this section which is based on Kusano, Manojlović [85] we discuss the existence and behavior of SV solutions to the equation

$$y'' + p(t)\Phi_{\gamma}(y) = 0, \qquad (4.24)$$

where  $p : [a, \infty) \rightarrow (0, \infty)$  is a continuous function. The super-linearity condition  $\gamma > 1$  is assumed.

By the famous Atkinson result, (4.24) is oscillatory (i.e., all of its nontrivial solutions are nonoscillatory) if and only if

$$\int_a^\infty sp(s)\,\mathrm{d}s=\infty.$$

Recall that under the assumption of sub-linearity (i.e.,  $0 < \gamma < 1$ ), similar type of result was established by Belohorec, and reads as follows: Equation (4.24) is oscillatory if and only if

$$\int_a^\infty s^\gamma p(s)\,\mathrm{d}s=\infty.$$

Eventually positive solutions can be classified into the three types in the same way as in the previous section.

The proof of the main result heavily depends on a similar result for the linear differential equation

$$y'' + p(t)y = 0. (4.25)$$

More precisely, first we form an infinite family of linear equations of the form (4.25) and then select with the help of fixed point techniques one equation from the family whose solution would give birth to the desired  $\mathcal{RV}$  solution of Emden-Fowler equation (4.24). The feasibility of such a procedure is assured by the extensive use of the following statement which can be proved by means of the contraction mapping theorem. We define the mapping

$$\mathcal{F}v(t) = t \int_{t}^{\infty} \left(\frac{v(s) + Q(s)}{s}\right)^{2} \mathrm{d}s$$

and consider it on the set

$$\{v \in C_0[T,\infty) : 0 \le v(t) \le \varphi(t), t \ge T\},\$$

where  $Q, \varphi$  are defined below in the theorem, and  $C_0[T, \infty)$  denotes the set of all continuous functions on  $[T, \infty)$  which tend to zero.

**Lemma 4.2** ([85]). Assume that there is a continuous function  $\varphi(t) : [a, \infty) \to (0, \infty)$ which is decreasing to 0 as  $t \to \infty$  and such that  $Q(t) \le \varphi(t)$  for large t, where Q is defined by

$$Q(t) = t \int_t^\infty p(s) \, \mathrm{d}s.$$

*Then equation* (4.25) *has a* SV *solution y in the form* 

$$y(t) = \exp\left\{\int_T^t \frac{v(s) + Q(s)}{s} \,\mathrm{d}s\right\},\,$$

 $t \ge T$ , for some T > a, in which v(t) is a unique solution of the integral equation

$$v(t) = t \int_t^\infty \left(\frac{v(s) + Q(s)}{s}\right)^2 \mathrm{d}s,$$

 $t \ge T$ , and  $0 \le v(t) \le 4\varphi^2(t)$  for only  $t \ge T$ .

We will use the notation

$$Q_L(t) = t \int_t^\infty p(s) L^{\gamma-1}(s) \,\mathrm{d}s.$$

**Theorem 4.7.** Let  $\gamma > 1$ . Assume that  $L \in SV$  is such that

$$L(t) = L(a) \exp\left\{\int_{a}^{t} \frac{\delta_{L}(s)}{s} \mathrm{d}s\right\} \nearrow \infty \quad \text{with } \delta_{L}(t) \searrow 0 \tag{4.26}$$

as  $t \to \infty$ . Suppose that there exists a constant K > 0 such that

$$Q_L(t) \le K\delta_L(t)$$
 for large t. (4.27)

*Then equation* (4.24) *possesses a* SV *solution y such that*  $y(t) \le L(t)$  *for all large t.* 

*Proof.* First observe that due to (4.26) we have

$$\delta_L(t) = t \frac{L'(t)}{L(t)} > 0, \quad \int_a^\infty \frac{\delta_L(s)}{s} \mathrm{d}s = \infty.$$

Let  $\mu \in (0, 1)$  be a given constant. There exists T > a such that (4.27) holds for  $t \ge T$  and

$$2\delta_L(t) \le \mu \text{ for } t \ge T, \quad \frac{2K}{L^{\gamma-1}(T)}.$$
(4.28)

Define  $\Xi$  to be the set of positive nondecreasing functions  $\xi(t)$  on  $[T, \infty)$  satisfying  $1 \le \xi(t) \le L(t)/L(T)$  for  $t \ge T$ . It is clear that  $\Xi$  is a closed convex subset of the locally convex space  $C[T, \infty)$ . For  $\xi \in \Xi$  put

$$p_{\xi}(t) = p(t)\xi^{\gamma-1}(t), \ Q_{\xi}(t) = t \int_{t}^{\infty} p_{\xi}(s) \,\mathrm{d}s,$$

and consider the family of ordinary linear differential equations

$$x'' + p_{\xi}(t)x = 0, \quad \xi \in \Xi.$$
 (4.29)

For each  $\xi \in \Xi$  we have by (4.27) and (4.28)

$$Q_{\xi}(t) \le t \int_{t}^{\infty} p(s) \left(\frac{L(s)}{L(T)}\right)^{\gamma-1} \mathrm{d}s \le \frac{K\delta_{L}(t)}{L^{\gamma-1}(T)} \le \frac{\delta_{L}(t)}{2} = \varphi(t), \tag{4.30}$$

 $t \ge T$ , and so from Lemma 4.2, we see that for each  $\xi \in \Xi$  equation (4.29) has a SV solution  $x_{\xi}$ , which is increasing and expressed in the form

$$x_{\xi}(t) = \exp\left\{\int_{T}^{t} \frac{v_{\xi}(s) + Q_{\xi}(s)}{s} \mathrm{d}s\right\},\tag{4.31}$$

 $t \geq T$ , where  $v_{\xi}$  is a unique solution of the integral equation

$$v_{\xi}(t) = t \int_{t}^{\infty} \left(\frac{v_{\xi}(s) + Q_{\xi}(s)}{s}\right)^{2} \mathrm{d}s,$$

 $t \ge T$ , and

$$v_{\xi}(t) \le \delta_L^2(t) \le \frac{\delta_L(t)}{2}.$$
(4.32)

Using (4.30) and (4.32) in (4.31), we conclude that

$$1 \le x_{\xi}(t) \le \exp\left\{\int_{T}^{t} \frac{\delta_{L}(s)}{s} \,\mathrm{d}s\right\} = \frac{L(t)}{L(T)},$$

 $t \geq T$ . Let us now define the mapping  $\mathcal{T} : \Xi \to C([T, \infty)$  by

$$\mathcal{T}\xi(t)=x_{\xi}(t),$$

 $t \ge T$ . It is immediate that  $\mathcal{T} \ge \subset \Xi$ . Further, relative compactness of  $\mathcal{T} \ge$  follows from the Arzela-Ascoli lemma. To show that  $\mathcal{T}$  is continuous is a little bit laborious, and we just note here that the Lebesgue dominated convergence theorem plays an important role. For details see [85]. Thus, all the hypotheses of the Schauder-Tychonoff fixed point theorem are fulfilled, and so there exists  $\xi \in \Xi$  such that  $\xi(t) = \mathcal{T}\xi(t) = x_{\xi}(t)$  for  $t \ge T$ . Since  $x_{\xi}(t)$  is a solution of linear differential equation (4.29), we have for  $t \ge T$  that

$$0 = x_{\xi}''(t) + p_{\xi}(t)x_{\xi}(t) = \xi''(t) + p(t)\xi^{\gamma-1}(t)\xi(t) = \xi''(t) + p(t)\xi^{\gamma}(t),$$

which implies that  $\xi$  is a solution of equation (4.24). It is obvious that  $\xi \in SV$ .  $\Box$ 

Example 1 in [85] suggests that super-linear Emden-Fowler equations may have both trivial and nontrivial SV solutions at the same time. It is therefore important to establish a means by which one can distinguish nontrivial SV solutions from trivial ones for a given Emden-Fowler equation. Under the stronger assumption than (4.27), one can determine the exact asymptotic behavior of any nontrivial SVsolution of super-linear Emden-Fowler equations as the following theorem shows.

**Theorem 4.8.** Let  $\gamma > 1$  and (4.26) hold. Suppose that  $p \in \mathcal{RV}(-\nu)$  for some  $\nu > 1$  and that

$$Q_L(t) \sim t \frac{L'(t)}{L(t)} = \delta_L(t)$$

as  $t \to \infty$ , where  $\delta_L$  is required to be SV function. Then equation (4.24) may possess a SV solution y only if v = 2, and in this case any nontrivial SV solution of equation (4.24) has one and the same asymptotic behavior

$$y(t) \sim L(t) \tag{4.33}$$

as  $t \to \infty$ .

*Proof.* Since  $p \in \mathcal{RV}(-\nu)$ , we have that  $p(t) = t^{-\nu}L_p(t)$ ,  $L_p \in \mathcal{SV}$ , and application of the Karamata integration theorem gives

$$Q_L(t) = t \int_t^\infty s^{-\nu} L_p(s) L^{\gamma-1}(s) \, \mathrm{d}s \sim \frac{t^{2-\nu}}{\nu-1} L_p(t) L^{\gamma-1}(t) \tag{4.34}$$

as  $t \to \infty$ . Thus, due to the assumption that  $Q_L(t) \sim \delta_L(t) \in SV$  as  $t \to \infty$ , from (4.34) we conclude that v = 2 and

$$L_p(t) \sim t \frac{L'(t)}{L^{\gamma}(t)}$$

as  $t \to \infty$ . Suppose that (4.24) has a nontrivial SV solution y. Integrating (4.24) from t to  $\infty$  and using the Karamata integration theorem, we have

$$y'(t) = \int_{t}^{\infty} p(s)y^{\gamma}(s) \, \mathrm{d}s = \int_{t}^{\infty} \frac{L_{p}(s)}{s^{2}} y^{\gamma}(s) \, \mathrm{d}s \sim \frac{L_{p}(t)}{t} y^{\gamma}(t) \sim \frac{L'(t)}{L^{\gamma}(t)} y^{\gamma}(t)$$

as  $t \to \infty$ , which implies that

$$\frac{y'(t)}{y^{\gamma}(t)} \sim \frac{L'(t)}{L^{\gamma}(t)} \tag{4.35}$$

as  $t \to \infty$ . Integrating (4.35) over  $[t, \infty)$  and noting that  $y(t) \to \infty$  and  $L(t) \to \infty$  as  $t \to \infty$ , we obtain

$$\frac{y^{1-\gamma}(t)}{\gamma-1} \sim \frac{L^{1-\gamma}(t)}{\gamma-1}$$

as  $\rightarrow \infty$ , which immediately yields (4.33).

# 4.3.4 Intermediate generalized $\mathcal{RV}$ solutions of second order sub-half-linear equations

In this section which is based on the paper [91] by Kusano, Manojlović, and Milošević we want to concisely describe how the approach based on the approximation of certain Emden-Fowler equation in an integral form by means of an asymptotic relation and on the use of the fixed point theorem can be combined with the concept of general regularly varying functions.

Consider the equation

$$(r(t)\Phi_{\alpha}(y'))' + p(t)\Phi_{\beta}(y) = 0, \qquad (4.36)$$

where  $\alpha > \beta > 0$  (which is the sub-half-linearity condition), r, p are continuous functions, and r satisfies  $\int_{a}^{\infty} r^{-\frac{1}{\alpha}}(t) dt < \infty$ . Note that a similar analysis in the case where this integral diverges was made by Jaroš, Kusano, Manojlović in [64]. Denote

$$R(t) = \int_t^\infty r^{-\frac{1}{\alpha}}(s) \, \mathrm{d}s.$$

We have the following conventional classification of positive solutions :

(i)  $\lim_{t\to\infty} y(t) = \text{const} > 0$ ,

(ii)  $\lim_{t\to\infty} y(t) = 0$ ,  $\lim_{t\to\infty} y(t)/R(t) = \infty$ ,

(iii)  $\lim_{t\to\infty} y(t)/R(t) = \text{const} > 0.$ 

Solutions of type (ii) are called intermediate solutions. Sharp conditions for the existence of such solutions was obtained by Kamo and Usami in [72].

Let *y* be an intermediate solution of (4.36) on  $[a, \infty)$ . Then  $\int_{a}^{\infty} p(s)y^{\beta}(s) ds = \infty$  and

$$y(t) = \int_{t}^{\infty} \frac{1}{r_{\alpha}^{\frac{1}{\alpha}}} \left( r(a)(-y'(s))^{\alpha} + \int_{a}^{s} p(\tau) y^{\beta}(\tau) \, \mathrm{d}\tau \right)^{\frac{1}{\alpha}} \mathrm{d}s, \tag{4.37}$$

 $t \ge a$ . It follows therefore that *y* satisfies the integral asymptotic relation

$$y(t) \sim \int_t^\infty \frac{1}{r^{\frac{1}{\alpha}}} \left( \int_a^s p(\tau) y^\beta(\tau) \, \mathrm{d}\tau \right)^{\frac{1}{\alpha}} \mathrm{d}s, \tag{4.38}$$

as  $t \to \infty$ , which is regarded as an "approximation" of (4.37) at infinity. The proofs of the main results are essentially based on the fact that a thorough knowledge of the existence and asymptotic behavior of generalized  $\mathcal{RV}$  solutions of (4.38) can be acquired. As a matter of fact, the "only if" part of the below presented theorem is an immediate consequence of manipulation of (4.38) by means of regular variation. The "if" part is proved by solving the integral equation

$$y(t) = \int_t^\infty \frac{1}{r^{\frac{1}{\alpha}}} \left( \int_a^s p(\tau) y^\beta(\tau) \, \mathrm{d}\tau \right)^{\frac{1}{\alpha}} \mathrm{d}s,$$

with the help of fixed point technique, the essence of which is based on detecting fixed points of the integral operator

$$\mathcal{F}y(t) = \int_t^\infty \frac{1}{r^{\frac{1}{\alpha}}} \left( \int_a^s p(\tau) y^\beta(\tau) \, \mathrm{d}\tau \right)^{\frac{1}{\alpha}} \mathrm{d}s,$$

lying in the set

$$\mathcal{X}=\{y\in C[a,\infty): mx(t)\leq y(t)\leq Mx(t), t\geq a\},$$

*m*, *M* being a suitable numbers, and

$$x(t) = \left(\frac{R^{\alpha+1}(t)r^{\frac{1}{\alpha}}(t)p(t)}{\alpha(-\rho)^{\alpha}(\rho+1)}\right)^{\frac{1}{\alpha-\beta}},$$

where  $\rho$  is given below in the theorem. The operator  $\mathcal{F}$  is continuous self-map on X and sends it into a relatively compact subset of  $C[a, \infty)$ . To show that the obtained solution y is indeed  $\mathcal{RV}$  with respect to 1/R, we apply the generalized L'Hospital rule which yields

$$y(t) \sim \int_t^\infty \frac{1}{r^{\frac{1}{\alpha}}} \left( \int_a^s p(\tau) x^\beta(\tau) \, \mathrm{d}\tau \right)^{\frac{1}{\alpha}} \, \mathrm{d}s$$

as  $t \to \infty$ .

Thus we have briefly described the proof of the following theorem.

**Theorem 4.9.** Let  $r \in \mathcal{RV}_{1/R}(\eta)$  and  $p \in \mathcal{RV}_{1/R}(\sigma)$ . Equation (4.36) has intermediate solutions  $y \in \mathcal{RV}_{1/R}(\rho)$  with  $\rho \in (-1, 0)$  if and only if

$$\beta - \frac{\eta}{\alpha} + 1 < \sigma < \alpha - \frac{\eta}{\alpha} + 1, \tag{4.39}$$

in which case  $\rho$  is given by

$$\rho = \frac{\sigma - \alpha - 1 + \frac{\eta}{\alpha}}{\alpha - \beta}$$

and asymptotic behavior of any such solution y is governed by the unique formula

$$y(t) \sim \left(\frac{R^{\alpha+1}(t)r^{\frac{1}{\alpha}}(t)p(t)}{\alpha(-\rho)^{\alpha}(\rho+1)}\right)^{\frac{1}{\alpha-\beta}}$$

as  $t \to \infty$ .

Note that the cases where  $\sigma$  is equal to the border values in (4.39) are also discussed in [91] and lead to the existence of nontrivial  $SV_{1/R}$  and  $RV_{1/R}(-1)$  solutions.

#### 4.3.5 Strongly monotone solutions of coupled systems

In the first part of this section we present the results concerning the so-called strongly decreasing solutions of the coupled system

$$\begin{cases} (p(t)\Phi_{\alpha}(x'))' = \varphi(t)\Phi_{\lambda}(y), \\ (q(t)\Phi_{\beta}(y'))' = \psi(t)\Phi_{\mu}(x), \end{cases}$$
(4.40)

which were established by Matucci, Řehák in [123]. We do not give the proof since its main ideas can be extracted from the proof of subsequent Theorem 4.15, which deals with a more general case. Note that originally, there are some differences between the proofs. Rather we focus on comments and applications. In the second part, we mention the work of Jaroš, Kusano [62] in which the coupled system in a more special form is considered; the approach used there is based on similar ideas as in the previous subsections. We present a result which deals with mixed strongly monotone solutions.

In (4.40) we assume that  $\alpha$ ,  $\beta$ ,  $\lambda$ ,  $\mu$  are positive constants, and p, q,  $\varphi$ ,  $\psi$  are positive continuous functions defined on  $[a, \infty)$ ,  $a \ge 0$ . Further we suppose that system (4.40) is *subhomogeneous* (at  $\infty$ ), i.e.,

$$\alpha\beta > \lambda\mu.$$

In contrast to the most of related works, we do not pose in general any condition on divergence or convergence of the integrals

$$P = \int_{a}^{\infty} p^{-\frac{1}{\alpha}}(s) \, \mathrm{d}s, \quad Q = \int_{a}^{\infty} q^{-\frac{1}{\beta}}(s) \, \mathrm{d}s, \tag{4.41}$$

and we do not explicitly distinguish among particular cases. In fact, all possible cases (including the mixed ones) are covered by the results. If (x, y) is a solution of (4.40), then by *quasiderivatives* denoted as  $x^{[1]}, y^{[1]}$ , we mean

$$x^{[1]} = p \Phi_{\alpha}(x'), \quad y^{[1]} = q \Phi_{\beta}(y').$$

Let  $\mathcal{DS}$  denote the set of all positive decreasing solutions of (4.40), i.e., all solutions whose components are both eventually positive and decreasing. Note that, due to the sign conditions on the coefficients, any solution of (4.40) has necessarily both components eventually of one sign and monotone. If  $P = Q = \infty$ , then  $\lim_{t\to\infty} x^{[1]}(t) = \lim_{t\to\infty} y^{[1]}(t) = 0$  for any  $(x, y) \in \mathcal{DS}$ . Indeed,  $-x^{[1]}$  is eventually positive decreasing. If  $\lim_{t\to\infty} -x^{[1]}(t) = c > 0$ , then  $p(t)(-x'(t))^{\alpha} \ge c$  or  $-x'(t) \ge c^{\frac{1}{\alpha}}p^{-\frac{1}{\alpha}}(t)$ ,  $t \ge t_0$  with some  $t_0 \ge a$ . Integrating the latter inequality, we get  $x(t) \le x(t_0) - c^{\frac{1}{\alpha}} \int_{t_0}^t p^{-\frac{1}{\alpha}}(s) \, ds \to -\infty$  as  $t \to \infty$ , a contradiction. Similarly we prove  $\lim_{t\to\infty} y^{[1]}(t) = 0$ . However, in general, for positive decreasing solutions, the limits of quasiderivatives do not need to be zero. Observe that any positive decreasing solution have both the components and their quasiderivatives tending to nonnegative resp. nonpositive numbers. Among all these solutions we are interested in the so called *strongly decreasing solutions*, which we denote as

$$\mathcal{SDS} = \left\{ (x, y) \in \mathcal{DS} : \lim_{t \to \infty} x(t) = \lim_{t \to \infty} y(t) = \lim_{t \to \infty} x^{[1]}(t) = \lim_{t \to \infty} y^{[1]}(t) = 0 \right\};$$

notice that other types of  $\mathcal{DS}$  solutions are somehow easier.

The integral expressions below play important roles in existence results for

strongly decreasing solutions:

$$\begin{split} I_1(t) &:= \int_t^{\infty} \left( \frac{1}{p(u)} \int_u^{\infty} \varphi(s) \left( \int_s^{\infty} \left( \frac{1}{q(r)} \int_r^{\infty} \psi(\tau) \, \mathrm{d}\tau \right)^{\frac{1}{\beta}} \, \mathrm{d}r \right)^{\lambda} \, \mathrm{d}s \right)^{\frac{1}{\alpha}} \, \mathrm{d}u, \\ I_2(t) &:= \int_t^{\infty} \left( \frac{1}{q(u)} \int_u^{\infty} \psi(s) \left( \int_s^{\infty} \left( \frac{1}{p(r)} \int_r^{\infty} \varphi(\tau) \, \mathrm{d}\tau \right)^{\frac{1}{\alpha}} \, \mathrm{d}r \right)^{\mu} \, \mathrm{d}s \right)^{\frac{1}{\beta}} \, \mathrm{d}u, \\ I_3(t) &:= \int_t^{\infty} \varphi(u) \left( \int_u^{\infty} \left( \frac{1}{q(s)} \int_s^{\infty} \psi(r) \left( \int_r^{\infty} \frac{1}{p^{\frac{1}{\alpha}}(\tau)} \, \mathrm{d}\tau \right)^{\mu} \, \mathrm{d}r \right)^{\frac{1}{\beta}} \, \mathrm{d}s \right)^{\lambda} \, \mathrm{d}u, \\ I_4(t) &:= \int_t^{\infty} \psi(u) \left( \int_u^{\infty} \left( \frac{1}{p(s)} \int_s^{\infty} \varphi(r) \left( \int_r^{\infty} \frac{1}{q^{\frac{1}{\beta}}(\tau)} \, \mathrm{d}\tau \right)^{\lambda} \, \mathrm{d}r \right)^{\frac{1}{\alpha}} \, \mathrm{d}s \right)^{\mu} \, \mathrm{d}u. \end{split}$$

The following existence result holds. Its proof can be found in the paper [123] by Matucci, Řehák; the Schauder-Tychonoff fixed point theorem is the main tool.

**Theorem 4.10.** If  $I_i(a) < \infty$  for at least one index  $i \in \{1, 2, 3, 4\}$ , then  $SDS \neq \emptyset$ .

Next we describe the so-called *reciprocity principle*. This tool is very useful in the proof of the existence result, but also in the proof of the main result devoted to exact asymptotic behavior. In fact, it enables us to extend the existence and asymptotic results made under certain setting to "new" situations. Let  $(x, y) \in DS$ . Set  $u = -x^{[1]}, v = -y^{[1]}$ . Then (u, v) is an eventually positive decreasing solution of the *reciprocal system* 

$$\begin{cases} \left(\frac{1}{\varphi^{\frac{1}{\lambda}}(t)}\Phi_{\frac{1}{\lambda}}(u')\right)' = \frac{1}{q^{\frac{1}{\beta}}}\Phi_{\frac{1}{\beta}}(v),\\ \left(\frac{1}{\psi^{\frac{1}{\mu}}(t)}\Phi_{\frac{1}{\mu}}(v')\right)' = \frac{1}{p^{\frac{1}{\alpha}}}\Phi_{\frac{1}{\alpha}}(u). \end{cases}$$
(4.42)

Observe that (4.42) has the same structure as (4.40), and is subhomogeneous. Indeed,  $1/(\lambda \mu) > 1/(\alpha \beta)$ . Moreover, the quasiderivatives of u and v, i.e.,  $\varphi^{-\frac{1}{\lambda}} \Phi_{\frac{1}{\lambda}}(u')$  and  $\psi^{-\frac{1}{\mu}} \Phi_{\frac{1}{\mu}}(v')$ , are equal to -y and -x, respectively. Conversely, if (u, v) is an eventually positive decreasing solution of (4.42) and we set  $x = -\psi^{-\frac{1}{\mu}} \Phi_{\frac{1}{\mu}}(v')$  and  $y = -\varphi^{-\frac{1}{\lambda}} \Phi_{\frac{1}{\lambda}}(u')$ , then  $(x, y) \in \mathcal{DS}$  and  $x^{[1]} = -u$ ,  $y^{[1]} = -v$ . Hence, with the use of these relations, it holds

$$(x, y)$$
 is a strongly decreasing solution of (4.40)  
 $(u, v)$  is a strongly decreasing solution of (4.42).  
(4.43)

It is easy to see that the roles which are played by the integrals  $I_1$  and  $I_2$  for system (4.40) are played by the integrals  $I_3$  and  $I_4$ , respectively, for system (4.42).

From now on we assume

$$p \in \mathcal{RV}(\gamma), q \in \mathcal{RV}(\delta), \varphi \in \mathcal{RV}(\sigma), \psi \in \mathcal{RV}(\varrho).$$

We set

$$\Lambda = \frac{1}{\alpha\beta - \lambda\mu}$$

and

$$\nu = \Lambda \Big( \beta(\alpha - \gamma + 1 + \sigma) + \lambda(\beta - \delta + 1 + \varrho) \Big),$$
  

$$\omega = \Lambda \Big( \alpha(\beta - \delta + 1 + \varrho) + \mu(\alpha - \gamma + 1 + \sigma) \Big).$$
(4.44)

Further we denote

$$\nu^{[1]} = (\nu - 1)\alpha + \gamma, \quad \omega^{[1]} = (\omega - 1)\beta + \delta.$$
(4.45)

Theorem 4.11. Assume

$$\nu < 0, \ \omega < 0, \ \nu^{[1]} < 0, \ \omega^{[1]} < 0.$$
 (4.46)

*Then*  $SDS \neq \emptyset$ *. Further, for every*  $(x, y) \in SDS$  *there hold*  $(x, y) \in RV(v) \times RV(\omega)$  *and* 

$$x(t) \sim \left(K_1 K_2^{\frac{\lambda}{\alpha}}\right)^{\alpha\beta\Lambda} t^{\nu} L_1(t), \quad y(t) \sim \left(K_2 K_1^{\frac{\mu}{\beta}}\right)^{\alpha\beta\Lambda} t^{\omega} L_2(t)$$
(4.47)

as  $t \to \infty$ , where

$$K_1 = \frac{-1}{\nu \left(-\nu^{[1]}\right)^{\frac{1}{\alpha}}}, \quad K_2 = \frac{-1}{\omega \left(-\omega^{[1]}\right)^{\frac{1}{\beta}}}, \tag{4.48}$$

and

$$L_{1} = \left(\frac{L_{\varphi}^{\beta}L_{\psi}^{\lambda}}{L_{p}^{\beta}L_{q}^{\lambda}}\right)^{\Lambda} \in S\mathcal{V}, \quad L_{2} = \left(\frac{L_{\varphi}^{\mu}L_{\psi}^{\alpha}}{L_{p}^{\mu}L_{q}^{\alpha}}\right)^{\Lambda} \in S\mathcal{V}.$$

**Remark 4.4.** (i) Under the assumptions of Theorem 4.11, the quasiderivatives of a strongly decreasing solution (x, y) of (4.40) satisfy

$$\begin{split} -x^{[1]}(t) &\sim \frac{\left(K_2 K_1^{\frac{\mu}{\beta}}\right)^{\alpha\beta\Lambda\lambda}}{-\nu^{[1]}} t^{\nu^{[1]}} L_{\varphi}(t) L_2^{\lambda}(t) \in \mathcal{RV}\left(\nu^{[1]}\right) \\ -y^{[1]}(t) &\sim \frac{\left(K_1 K_2^{\frac{\lambda}{\alpha}}\right)^{\alpha\beta\Lambda\mu}}{-\omega^{[1]}} t^{\omega^{[1]}} L_{\psi}(t) L_1^{\mu}(t) \in \mathcal{RV}\left(\omega^{[1]}\right). \end{split}$$

(ii) The asymptotic formula (4.47) can alternatively be written in terms of the coefficients,

$$x(t) \sim C_1 \left(\frac{t^{\alpha+1}\varphi(t)}{p(t)}\right)^{\beta/(\alpha\beta-\lambda\mu)} \left(\frac{t^{\beta+1}\psi(t)}{q(t)}\right)^{\lambda/(\alpha\beta-\lambda\mu)},$$
$$y(t) \sim C_2 \left(\frac{t^{\beta+1}\psi(t)}{q(t)}\right)^{\alpha/(\alpha\beta-\lambda\mu)} \left(\frac{t^{\alpha+1}\varphi(t)}{p(t)}\right)^{\mu/(\alpha\beta-\lambda\mu)}$$

as  $t \to \infty$ , where  $C_1 = (K_1 K_2^{\lambda/\alpha})^{\alpha\beta/(\alpha\beta-\lambda\mu)}$  and  $C_2 = (K_2 K_1^{\mu/\beta})^{\alpha\beta/(\alpha\beta-\lambda\mu)}$ .

(iii) If the components of a (decreasing) regularly varying solution to (4.40) are eventually convex, then they are both normalized regularly varying. Note that the eventual convexity of decreasing solutions of (4.40) can be guaranteed, e.g., by p(t) = q(t) = 1.

(iv) The assumptions of Theorem 4.11 can be replaced by any of the following inequalities

$$\varrho + 1 < \min\left\{0, \delta - \beta, \delta - \beta - \frac{\beta}{\lambda}(\sigma + 1), \delta - \beta - \frac{\beta}{\lambda}(\sigma + 1 + \alpha - \gamma)\right\}$$
(4.49)

or

$$\sigma + 1 < \min\left\{0, \gamma - \alpha, \gamma - \alpha - \frac{\alpha}{\mu}(\varrho + 1), \gamma - \alpha - \frac{\alpha}{\mu}(\varrho + 1 + \beta - \delta)\right\}$$
(4.50)

or

$$\alpha - \gamma < \min\left\{0, -\frac{\alpha}{\mu}(\varrho+1), -\frac{\alpha}{\mu}(\varrho+1+\beta-\delta), -\frac{\alpha}{\mu}(\varrho+1+\beta-\delta) - \frac{\alpha\beta}{\lambda\mu}(\sigma+1)\right\} \quad (4.51)$$

or

$$\beta - \delta < \min\left\{0, -\frac{\beta}{\lambda}(\sigma+1), -\frac{\beta}{\lambda}(\sigma+1+\alpha-\gamma), -\frac{\beta}{\lambda}(\sigma+1+\alpha-\gamma) - \frac{\alpha\beta}{\lambda\mu}(\varrho+1)\right\}, \quad (4.52)$$

and the statement remains valid. Moreover, the following equivalence among the sufficient conditions hold:

$$(4.46) \Leftrightarrow (4.49) \text{ or } (4.50) \text{ or } (4.51) \text{ or } (4.52).$$
 (4.53)

As an example of typical setting, assume  $P = Q = \infty$ . Then necessarily  $\alpha \ge \gamma$  and  $\beta \ge \delta$ ; the sufficient condition (4.46) reduces to

$$\nu < 0, \, \omega < 0. \tag{4.54}$$

Indeed,  $\nu^{[1]} = \nu \alpha - \alpha + \gamma \leq \nu \gamma < 0$ . Similarly we obtain  $\omega^{[1]} < 0$ .

Another example of a typical situation is when

$$\int_{a}^{\infty} \varphi(t) \, \mathrm{d}t = \int_{a}^{\infty} \psi(t) \, \mathrm{d}t = \infty$$

is assumed. This condition implies  $\sigma + 1 \ge 0$ ,  $\varrho + 1 \ge 0$ , and (4.46) reduces to  $\nu^{[1]} < 0$ ,  $\omega^{[1]} < 0$ . It can be also quite easily observed how the "mixed" cases, for instance,

$$\int_{a}^{\infty} p^{-\frac{1}{\alpha}}(s) \, \mathrm{d}s = \infty, \int_{a}^{\infty} q^{-\frac{1}{\beta}}(s) \, \mathrm{d}s < \infty,$$
$$\int_{a}^{\infty} \varphi(t) \, \mathrm{d}t = \infty, \int_{a}^{\infty} \psi(t) \, \mathrm{d}t < \infty$$

can be covered by the above results.

The main theorem can be applied also in some special situations where the coefficients of (4.40) are not regularly varying; we use a change of the independent variable. Let  $s = \zeta(t)$ , where  $\zeta$  is a differentiable function such that  $\zeta'(t) \neq 0$  on  $[a, \infty)$ . Set  $(w, z)(s) = (x, y)(\zeta^{-1}(s)), \zeta^{-1}$  being the inverse of  $\zeta$ . Since  $d/dt = \zeta'(t)d/ds$ , system (4.40) is transformed into the system

$$\begin{cases} \frac{\mathrm{d}}{\mathrm{d}s} \left( \hat{p}(s) \Phi_{\alpha} \left( \frac{\mathrm{d}w}{\mathrm{d}s} \right) \right) = \hat{\varphi}(s) \Phi_{\lambda}(z), \\ \frac{\mathrm{d}}{\mathrm{d}s} \left( \hat{q}(s) \Phi_{\beta} \left( \frac{\mathrm{d}z}{\mathrm{d}s} \right) \right) = \hat{\psi}(s) \Phi_{\mu}(w), \end{cases}$$
(4.55)

where

$$\begin{split} \hat{p} &:= (p \circ \zeta^{-1}) \cdot \Phi_{\alpha}(\zeta' \circ \zeta^{-1}), \ \hat{q} := (q \circ \zeta^{-1}) \cdot \Phi_{\beta}(\zeta' \circ \zeta^{-1}), \\ \hat{\varphi} &:= \frac{\varphi \circ \zeta^{-1}}{\zeta' \circ \zeta^{-1}}, \ \hat{\psi} := \frac{\psi \circ \zeta^{-1}}{\zeta' \circ \zeta^{-1}}. \end{split}$$

Clearly, (4.55) is of the form (4.40) and is subhomogeneous. If  $\zeta$  is unbounded with  $\zeta' > 0$  and such that  $\hat{p}, \hat{q}, \hat{\psi} \in \bigcup_{\vartheta \in \mathbb{R}} \mathcal{RV}(\vartheta)$ , then the results can be applied to (4.55). To illustrate a possible application, take, for example,

$$p(t) = e^{\gamma t} h_1(t), \ q(t) = e^{\delta t} h_2(t), \ \varphi(t) = e^{\sigma t} h_3(t), \ \psi(t) = e^{\rho t} h_4(t),$$

where  $\gamma, \delta, \sigma, \varrho \in \mathbb{R}$  and  $h_i \in \bigcup_{\vartheta \in \mathbb{R}} \mathcal{RV}(\vartheta)$ , i = 1, 2, 3, 4. In such a case we can set  $\zeta(t) = e^t$ . Thus  $t = \ln s$  and  $[a, \infty)$  is transformed into another right half-line. We then get

$$\hat{p}(s) = s^{\gamma+\alpha}H_1(s) \in \mathcal{RV}(\gamma+\alpha)$$
, where  $H_1 := h_1 \circ \ln \beta$ 

since  $H_1 \in \mathcal{RV}(\gamma_1 \cdot 0) = \mathcal{RV}(0) = \mathcal{SV}$ ,  $\gamma_1$  being the index of regular variation of  $h_1$ . Similarly,

$$\hat{q} \in \mathcal{RV}(\delta + \beta), \ \hat{\varphi} \in \mathcal{RV}(\sigma - 1), \ \hat{\psi} \in \mathcal{RV}(\varrho - 1).$$

or

Consequently, Theorem 4.11 can directly be applied to system (4.55), and hereby to the original system through the transformation.

We conclude this part by the application of Theorem 4.11 to the widely studied 4th-order equation

$$x'''' = \psi(t)\Phi_{\mu}(x).$$
(4.56)

The below mentioned observations can be easily extended to some more general 4th-order equations, but for comparison purposes we take the form (4.56), which appears quite frequently in the literature. We assume  $\psi \in \mathcal{RV}(\varrho)$  and  $\mu > 0$ . This equation is equivalent to system (4.40), where we set  $\alpha = \beta = \lambda = 1$  and  $p(t) = q(t) = \varphi(t) = 1$ . Then  $\gamma = \delta = \sigma = 0$ ,  $L_p(t) = L_q(t) = L_{\varphi}(t) = 1$ , and  $\Lambda = 1/(1 - \mu)$ . The subhomogeneity assumption reads as  $\mu < 1$ . Further,

$$\nu = \frac{\varrho + 4}{1 - \mu}, \quad \omega = \frac{\varrho + 2 + 2\mu}{1 - \mu}, \quad \nu^{[1]} = \frac{\varrho + 3 + \mu}{1 - \mu}, \quad \omega^{[1]} = \frac{\varrho + 1 + 3\mu}{1 - \mu}.$$

A strongly decreasing solution x of (4.56) is a positive solution such that

$$\lim_{t \to \infty} x(t) = \lim_{t \to \infty} x'(t) = \lim_{t \to \infty} x''(t) = \lim_{t \to \infty} x'''(t) = 0$$

It is easy to see that in order (4.46) to be fulfilled, it is sufficient to take  $\rho < -4$ . Alternatively, we can check how (4.49) is verified (this is the only one among conditions (4.49), (4.50), (4.51), (4.52), that can be satisfied in this situation). Thus, under the assumptions  $0 < \mu < 1$  and  $\rho < -4$ , Theorem 4.11 assures that (4.56) possesses a strongly decreasing solution and for any such a solution *x* it holds

$$x(t) \sim \left(\frac{(1-\mu)^4 t^{\varrho+4} L_{\psi}(t)}{\prod_{i=1}^4 [\varrho+i+(4-i)\mu]}\right)^{\frac{1}{1-\mu}}$$

as  $t \to \infty$ .

A special case of (4.40) in the form

$$x'' = \varphi(t)y^{\lambda}, \quad y'' = \psi(t)x^{\mu},$$
 (4.57)

 $\varphi(t) > 0, \psi(t) > 0, \lambda, \mu > 0, \lambda\mu < 1$ , was investigated in [62] by Jaroš and Kusano. More precisely, strongly monotone solutions were studied there. Since the approach is based on similar ideas to those described in Subsections 4.3.2–4.3.4 (approximation by means of a suitable integral asymptotic relation, a-priori bounds, and the Schauder fixed point theorem) we present one selected result without proof. It concerns existence and asymptotic behavior of mixed strongly monotone solutions to (4.57), i.e., the positive solutions where one of the components is strongly decreasing, while the other is strongly increasing.

**Theorem 4.12.** Suppose that  $\varphi \in \mathcal{RV}(\sigma)$ ,  $\psi \in \mathcal{RV}(\varrho)$ . System (4.57) possesses regularly varying solutions (x, y) such that

$$x \in \mathcal{RV}(\nu), y \in \mathcal{RV}(\omega), \nu < 1, \omega > 1,$$

if and only if

$$\sigma + 2 + \lambda(\rho + 2) < 0, \quad \mu(\lambda + \sigma + 2) + \rho + 1 > 0$$

in which case v and  $\omega$  are given by

$$\nu = \frac{\sigma + 2 + \lambda(\varrho + 2)}{1 - \lambda\mu}, \quad \omega = \frac{\mu(\sigma + 2) + \varrho + 2}{1 - \lambda\mu},$$

and the asymptotic behavior of any such solution is governed by the formulas

$$x(t) \sim \left(\frac{t^{2(\lambda+1)}\varphi(t)\psi^{\lambda}(t)}{\nu(\nu-1)[\omega(\omega-1)]^{\lambda}}\right)^{\frac{1}{1-\lambda\mu}}, \ y(t) \sim \left(\frac{t^{2(\mu+1)}\varphi^{\mu}(t)\psi(t)}{[\nu(\nu-1)]^{\mu}\omega(\omega-1)}\right)^{\frac{1}{1-\lambda\mu}}$$

as  $t \to \infty$ .

#### 4.3.6 Overall structure of $\mathcal{RV}$ solutions to odd-order equations

In this section we present a description of overall structure of  $\mathcal{RV}$  solutions to the equation

$$y^{2n+1} + p(t)\Phi_{\gamma}(y) = 0, \tag{4.58}$$

where  $0 < \gamma < 1$  and  $p : [a, \infty) \rightarrow (0, \infty)$  is a continuous function. Such a description is the result of considerations made by Kusano and Manojlović in [89]. Some of the ideas in the proofs are similar to those used e.g. in Subsection 4.3.3. In particular, here again an integral asymptotic relation is investigated, which can be seen as an approximation of an integral form of equation (4.58); this time the relation reads as

$$y(t) \sim \int_{a}^{t} \frac{(t-s)^{k-1}}{(k-1)!} \int_{s}^{\infty} \frac{(r-s)^{2n-k}}{(2n-k)!} p(\tau) y^{\gamma}(\tau) d\tau ds,$$

*k* is assumed to be even integer such that  $2 \le k \le 2n$ . In addition, the tools like the Schauder-Tychonoff fixed point theorem and the Karamata integration theorem again play important roles in the proofs. In classifying eventually positive solutions of (4.58), the well-known Kiguradze lemma is utilized.

Assume that  $p \in \mathcal{RV}(\sigma)$ . Denote with  $\mathcal{RS}$  the set of all  $\mathcal{RV}$  solutions of (4.58), and define the subsets

$$\mathcal{RS}(\varrho) = \mathcal{RS} \cap \mathcal{RV}(\varrho),$$
  
tr- $\mathcal{RS}(\varrho) = \mathcal{RS} \cap \text{tr-}\mathcal{RV}(\varrho),$   
ntr- $\mathcal{RS}(\varrho) = \mathcal{RS} \cap \text{ntr-}\mathcal{RV}(\varrho).$ 

Using notation  $\gamma_m = m(1 - \gamma) - 2n - 1$ ,  $m \in \{0, 1, 2, ..., 2n\}$ , to make the full analysis we separately consider the case  $\sigma < \gamma_0 = -2n - 1$  together with the central cases

$$\sigma \in (\gamma_0, \gamma_1) \cup (\gamma_2, \gamma_3) \cup \cdots \cup (\gamma_{2n-2}, \gamma_{2n-1})$$

or

$$\sigma \in (\gamma_1, \gamma_2) \cup (\gamma_3, \gamma_4) \cup \cdots \cup (\gamma_{2n-1}, \gamma_{2n}),$$

and the border cases

$$\sigma \in \{\gamma_0, \gamma_2, \dots, \gamma_{2n}\}$$
 or  $\sigma \in \{\gamma_1, \gamma_3, \dots, \gamma_{2n-1}\}$ .

Denote

$$Q_j = \int_a^\infty t^{2n-j(1-\gamma)} p(s) \,\mathrm{d}s,$$

j = 0, 1, ..., 2n. The structure of  $\mathcal{RV}$  solutions to equation (4.58) reads then as follows:

(i) If  $\sigma < \gamma_0$ , then

$$\mathcal{RS} = \bigcup_{j=0}^{2n} \operatorname{tr-}\mathcal{RS}(j) \cup \mathcal{RS}\left(\frac{\sigma+2n+1}{1-\gamma}\right).$$

(ii) If  $\sigma = \gamma_0$  and  $Q_0 < \infty$ , then

$$\mathcal{RS} = \bigcup_{j=0}^{2n} \operatorname{tr} \mathcal{RS}(j) \cup \operatorname{ntr} \mathcal{RS}(0).$$

(iii) If  $\sigma = \gamma_0$  and  $Q_0 = \infty$ , then

$$\mathcal{RS} = \bigcup_{j=1}^{2n} \operatorname{tr-}\mathcal{RS}(j).$$

(iv) If  $\sigma = \gamma_m$  for some  $m \in \{1, 3, ..., 2n - 1\}$  and  $Q_m < \infty$ , then

$$\mathcal{RS} = \bigcup_{j=m}^{2n} \operatorname{tr-}\mathcal{RS}(j).$$

(v) If  $\sigma = \gamma_m$  for some  $m \in \{1, 3, ..., 2n - 1\}$  and  $Q_m = \infty$ , then

$$\mathcal{RS} = \bigcup_{j=m+1}^{2n} \operatorname{tr-}\mathcal{RS}(j) \cup \operatorname{ntr-}\mathcal{RS}(m).$$

(vi) If  $\sigma \in (\gamma_{m-1}, \gamma_m)$  for some  $m \in \{2, 4, \dots, 2n-2\}$ , then

$$\mathcal{RS} = \bigcup_{j=m}^{2n} \operatorname{tr-}\mathcal{RS}(j) \cup \operatorname{ntr-}\mathcal{RS}\left(\frac{\sigma+2n+1}{1-\gamma}\right).$$

(vii) If  $\sigma \in (\gamma_m, \gamma_{m+1})$  for some  $m \in \{0, 2, \dots, 2n - 2\}$ , then

$$\mathcal{RS} = \bigcup_{j=m+1}^{2n} \operatorname{tr-}\mathcal{RS}(j).$$

(viii) If  $\sigma = \gamma_m$  for some  $m \in \{2, 4, ..., 2n\}$  and  $Q_m < \infty$ , then

$$\mathcal{RS} = \bigcup_{j=m}^{2n} \operatorname{tr} \mathcal{RS}(j) \cup \operatorname{ntr} \mathcal{RS}(m).$$

(ix) If  $\sigma = \gamma_m$  for some  $m \in \{2, 4, \dots, 2n-2\}$  and  $Q_m = \infty$ , then

$$\mathcal{RS} = \bigcup_{j=m+1}^{2n} \operatorname{tr-}\mathcal{RS}(j).$$

(x) If  $\sigma = \gamma_{2n}$  and  $Q_{2n} = \infty$ , then

$$\mathcal{RS} = \emptyset.$$

(xi) If  $\sigma > \gamma_{2n}$ , then

$$\mathcal{RS} = \emptyset.$$

A similar description was obtained for the equation

$$y^{2n+1} = p(t)\Phi_{\gamma}(y),$$

where  $0 < \gamma < 1$  and  $p : [a, \infty) \to (0, \infty)$  is a continuous  $\mathcal{RV}$  function, in the paper [90] by Kusano and Manojlović.

For further related information see Remark 4.5-(v).

## **4.3.7** $\mathcal{P}_{\omega}$ solutions of *n*-th order equations

In this section we present a selected result from Evtukhov, Samoilenko [36] which represents the approach of Evtukhov et al. to the investigation of Emden-Fowler equation in the framework close to regular variation.

Consider the *n*-th order differential equation

$$y^{(n)} = \delta p(t)F(y), \tag{4.59}$$

where  $\delta \in \{-1, 1\}$ ,  $p : [a, \omega) \to (0, \infty)$  is a continuous function,  $-\infty < a < \omega \le \infty$ , and  $F : \Delta_T \to (0, \infty)$  is a continuous  $\mathcal{RV}_T$  function with the index  $\alpha \neq 1$ ; here *T* is zero or  $\pm \infty$ , and  $\Delta_T$  is an one-sided neighborhood of *T*. By  $F \in \mathcal{RV}_T(\alpha)$  we mean here that

$$F(u) = |u|^{\alpha} L_F(u)$$
 with  $\lim_{u \to T, u \in \Delta_T} \frac{L_F(\lambda u)}{L_F(u)}$ 

for each  $\lambda > 0$ .

A solution *y* of (4.59) is called a  $\mathcal{P}_{\omega}(T, \lambda_0)$  solution,  $-\infty \leq \lambda_0 \leq \infty$ , if it is defined on an interval  $[t_0, \omega) \subset [a, \omega)$  and satisfies the conditions

$$y: [t_0, \omega) \to \Delta_T, \quad \lim_{t \to \omega} y(t) = T,$$

$$\lim_{t \to \omega} y^{(k)}(t) = \begin{cases} \text{either } 0\\ \text{or } \pm \infty \end{cases} \quad (k = 1, \dots, n),$$

$$\lim_{t \to \omega} \frac{(y^{(n-1)}(t))^2}{y^{(n)}(t)y^{(n-2)}(t)} = \lambda_0.$$

$$(4.61)$$

Take a number  $b \in \Delta_T$  such that |b| < 1 if T = 0, b > 1 if  $T = \infty$ , b < -1 if  $T = -\infty$ and set  $\Delta_T(b) = [b, T)$  if  $\Delta_T$  is a left neighborhood of T,  $\Delta_T(b) = (T, b]$  if  $\Delta_T$  is a right neighborhood of T. It follows from the definition that each  $\mathcal{P}_{\omega}(T, \lambda_0)$  solution of equation (4.59) and all of its derivatives of order less than or equal to n are nonzero on some interval  $[t_1, \omega) \subset [t_0, \omega)$ ; moreover, the first derivative of the solution is positive on this interval if  $\Delta_T$  is a left neighborhood of T and negative otherwise. With regard for this fact and the choice of b, we introduce two numbers

$$\mu_0 = \operatorname{sgn} b, \quad \mu_1 = \begin{cases} 1 & \text{if } \Delta_T \text{ is a left neighborhood of } T \\ -1 & \text{if } \Delta_T \text{ is a right neighborhood of } T, \end{cases}$$

which determine the signs of the  $\mathcal{P}_{\omega}(T, \lambda_0)$  solution and its first derivative, respectively, on the interval  $[t_1, \omega)$ . In view of (4.60), we assume that  $y(t) \in \Delta_T(b)$  for  $t \in [t_1, \omega)$ . Further, we put

$$t_{\omega} = \begin{cases} t & \text{for } \omega = \infty \\ t - \omega & \text{for } \omega < \infty. \end{cases}$$

Set

$$J(t) = \int_{K}^{t} s_{\omega}^{n-1} p(s) \, \mathrm{d}s, \text{ where } K = \begin{cases} a & \text{if } \int_{a}^{\omega} s_{\omega}^{n-1} p(s) \, \mathrm{d}s = \infty \\ \omega & \text{if } \int_{a}^{\omega} s_{\omega}^{n-1} p(s) \, \mathrm{d}s < \infty \end{cases}$$

Theorem 4.13. Let

$$\lambda_0 = \mathbb{R} \setminus \left\{ 0, \frac{1}{2}, \frac{2}{3}, \dots, \frac{n-2}{n-1}, 1 \right\}.$$

Then for the existence of  $\mathcal{P}_{\omega}(T, \lambda_0)$  solutions of (4.59), it is necessary and, if the algebraic equation

$$(1+\varrho)\prod_{i=1}^{n-1}(A_i+\varrho) = \alpha \prod_{i=1}^{n-1}A_i$$
(4.62)

for  $\varrho$ , where  $A_i = (n - i)\lambda_0 - (n - i - 1)$ , i = 1, ..., n - 1, has no roots with zero real part, sufficient that

$$\lim_{t \to \omega} \frac{t_{\omega} J'(t)}{J(t)} = \frac{(1 - \alpha)A_1}{\lambda_0 - 1}$$
(4.63)

and the following inequalities hold

$$\delta\mu_0[(\lambda_0 - 1)t_{\omega}]^n \prod_{i=1}^{n-1} A_i > 0, \quad \mu_0\mu_1 A_1(\lambda_0 - 1)t_{\omega} > 0.$$
(4.64)

Moreover, there exists an m-parameter family of such solutions if, among the roots of the algebraic equation, there are m roots (with regard of multiplicities) whose real part has the same sign as the function  $(1 - \lambda_0)t_{\omega}$ . Furthemore, each solution of this kind admits the asymptotic representations

$$\frac{y(t)}{F(y(t))} = \delta[(\lambda_0 - 1)t_{\omega}]^n p(t) \prod_{i=1}^{n-1} \frac{1}{A_i} (1 + o(1)) \quad as \ t \to \omega,$$
(4.65)

$$\frac{y^{(k)}(t)}{y^{(k-1)}(t)} = \frac{A_k}{(\lambda_0 - 1)t_\omega} (1 + o(1)), \ k = 1, \dots, n-1, \quad \text{as } t \to \omega.$$
(4.66)

*Proof.* First we introduce the useful function

$$G(u) = \int_{B}^{u} \frac{\mathrm{d}s}{F(s)}, \quad B = \begin{cases} b & \text{for } \left| \int_{B}^{T} \frac{\mathrm{d}s}{F(s)} \right| = \infty \\ T & \text{for } \left| \int_{B}^{T} \frac{\mathrm{d}s}{F(s)} \right| < \infty. \end{cases}$$

Note its properties needed in the sequel. Since G'(u) > 0 for  $u \in \Delta_T(b)$ , we have  $G : \Delta_T(b) \to \Delta_Z(c)$ , where

$$\Delta_Z(c) = \begin{cases} [c, Z) & \text{for } \Delta_T(b) = [b, T) \\ (Z, c] & \text{for } \Delta_T(b) = (T, b], \end{cases}$$
$$c = \int_B^b \frac{\mathrm{d}s}{F(s)}, \quad Z = \begin{cases} 0 & \text{for } B = T \\ \infty & \text{for } B = b < T \\ -\infty & \text{for } B = b > T, \end{cases}$$

and moreover,

$$\lim_{u \to T, u \in \Delta_T(b)} F(u) = \frac{\mu_0}{1 - \alpha} \lim_{u \to T, u \in \Delta_T(b)} |u|^{1 - \alpha} = Z,$$
(4.67)

and there exists an inverse continuously differentiable increasing function  $G^{-1}$ :  $\Delta_Z(c) \rightarrow \Delta_T(b)$  such that

$$\lim_{z \to Z, \, z \in \Delta_Z(c)} G^{-1}(z) = T.$$
(4.68)

By virtue of properties of  $\mathcal{RV}_T$  functions and the L'Hospital rule, we have the relation

$$\lim_{u \to T, u \in \Delta_T(b)} \frac{u}{G(u)F(u)} = 1 - \alpha.$$
(4.69)

Necessity. Let

$$\lambda_0 = \mathbb{R} \setminus \left\{0, \frac{1}{2}, \frac{2}{3}, \dots, \frac{n-2}{n-1}, 1\right\}$$

and let  $y : [t_0, \omega) \to \Delta_T$ , be an arbitrary  $\mathcal{P}_{\omega}(T, \lambda_0)$  solution of equation (4.59). Then there exists  $t_1 \in [t_0, \omega)$  such that  $y(t) \in \Delta_T(b)$ , sgn  $y(t) = \mu_0$ , and sgn  $y'(t) = \mu_1$  for  $t \in [t_1, \omega)$ . In addition, by [36, Lemma 2.1], we have the asymptotic relations

$$y^{(k-1)}(t) \sim [(\lambda_0 - 1)t_{\omega}]^{n-k} \prod_{i=1}^{n-1} \frac{1}{A_i} y^{(n-1)}(t), \ k = 1, \dots, n-1, \text{ as } t \to \omega,$$

where  $A_i \neq 0$ , i = 1, ..., n - 1. They imply asymptotic representations (4.66), and, by virtue of (4.61), we have

$$y^{(n)}(t) \sim \frac{1}{\lambda_0} \left( \frac{y^{(n-1)}(t)}{y^{(n-2)}(t)} \right)^2 y^{(n-2)}(t) = \frac{1}{\lambda_0} \left( \frac{y^{(n-1)}(t)}{y^{(n-2)}(t)} \right)^2 \frac{y^{(n-2)}(t)}{y^{(n-3)}(t)} \cdots \frac{y^{\prime\prime}(t)}{y^{\prime}(t)} y^{\prime}(t)$$
$$\sim \frac{\lambda_0}{(\lambda_0 - 1)^2 t_{\omega}^2} \cdot \frac{A_{n-2}}{(\lambda_0 - 1)t_{\omega}} \cdots \frac{a_2}{(\lambda_0 - 1)t_{\omega}} y^{\prime}(t) = \frac{\prod_{i=2}^{n-1} A_i}{[(\lambda_0 - 1)t_{\omega}]^{n-1}} y^{\prime}(t)$$

as  $t \to \omega$ . Therefore by virtue of (4.59), we have (4.65). By integrating this asymptotic relation from  $t_1$  to  $t, t \in (t_1, \omega)$ , and by taking into account conditions (4.60) and (4.67), we obtain the relation

$$G(y(t)) = \delta(\lambda_0 - 1)^{n-1} \prod_{i=2}^{n-1} \frac{1}{A_i} J(t) [1 + o(1)]$$

as  $t \rightarrow \omega$ , which, together with (4.69), implies the representation

$$\delta(\lambda_0 - 1)^{n-1} \prod_{i=2}^{n-1} \frac{1}{A_i} J(t) [1 + o(1)]$$
(4.70)

as  $t \rightarrow \omega$ . It follows from (4.65) and (4.70) that

$$\frac{y'(t)}{y(t)} = \frac{t_{\omega}^{n-1}p(t)}{(1-\alpha)J(t)}(1+o(1))$$

as  $t \to \omega$ . Therefore, by virtue of (4.66) for k = 1, we have

$$\frac{A_1}{(\lambda_0 - 1)t_{\omega}} = \frac{t_{\omega}^{n-1}p(t)}{(1 - \alpha)J(t)}(1 + o(1))$$

as  $t \to \omega$ . Consequently, condition (4.63) is satisfied, and relation (4.70) can be represented in the form (4.65). In addition, since sgn  $y(t) = \mu_0$  and sgn  $y'(t) = \mu_1$  for  $t \in [t_1, \omega)$ , it follows from (4.65) and (4.66) that inequalities (4.64) hold.

Sufficiency. Let conditions (4.63) and (4.64) be satisfied for some

$$\lambda_0 = \mathbb{R} \setminus \left\{ 0, \frac{1}{2}, \frac{2}{3}, \dots, \frac{n-2}{n-1}, 1 \right\},\,$$

and let the algebraic equation (4.62) have no roots with zero real part. Let us show that, in this case, (4.59) has  $\mathcal{P}_{\omega}(T, \lambda_0)$  solutions that admit asymptotic representations (4.65) and (4.66). By applying the transformation

$$F(y(t)) = q(t)(1+v_1(\tau)), \quad \frac{y^{(k)}(t)}{y^{(k-1)}(t)} = \frac{A_k}{(\lambda_0 - 1)t_\omega}(1+v_{k+1}(\tau)), \tag{4.71}$$

k = 1, ..., n - 1, to equation (4.59), where

$$\tau = \beta \ln |t_{\omega}|, \quad \beta = \begin{cases} 1 & \text{for } \omega = \infty \\ -1 & \text{for } \omega < \infty, \end{cases} \quad q(t) = \delta(\lambda_0 - 1)^{n-1} \prod_{i=2}^{n-1} \frac{1}{A_i} J(t),$$

we obtain the system of differential equations

$$\begin{aligned} v_1' &= \beta \left( \frac{t_{\omega} J'(t)}{J(t)} (1+v_1) + \frac{A_1}{\lambda_0 - 1} \cdot \frac{Y(t, v_1)}{F(Y(t, v_1))q(t)} (1+v_2) \right), \\ v_k' &= \beta \left( 1+v_k + \frac{A_k}{\lambda_0 - 1} (1+v_k)(1+v_{k+1}) - \frac{A_{k-1}}{\lambda_0 - 1} (1+v_k)^2 \right), \\ v_n' &= \beta \left( 1+v_n - \frac{A_{n-1}}{\lambda_0 - 1} (1+v_n)^2 + \frac{p(t)t_{\omega}^n}{A_1 J(t)} \cdot \frac{q(t)F(Y(t, v_1))}{Y(t, v_1) \prod_{k=2}^{n-1} (1+v_k)} \right), \end{aligned}$$
(4.72)

k = 2, ..., n - 1, where  $t = t(\tau)$  is the inverse function of  $\tau = \beta \ln |t_{\omega}|$  and

$$Y(t, v_1) = G^{-1}(q(t)(1 + v_1)).$$

By virtue of conditions (4.63) and (4.64),  $\lim_{t\to\omega} q(t) = Z$ , and there exists a number  $t_0 \in [a, \omega)$  such that  $q(t)(1 + v_1) \in \Delta_Z(c)$  for  $t \in [t_0, \omega)$  and  $|v_1| \le 1/2$ . Therefore,  $Y(t, v_1) \in \Delta_T(b)$  for  $t \in [t_0, \omega)$  and, by virtue of (4.68),

 $\lim_{t \to \omega} Y(t, v_1) = T \text{ uniformly with respect to } v_1 \in [-1/2, 1/2].$ 

This, together with (4.69), implies that

$$\lim_{t \to \omega} \frac{Y(t, v_1)}{F(Y(t, v_1))G(Y(t, v_1))} = 1 - \alpha$$

uniformly with respect to  $v_1 \in [-1/2, 1/2]$ , i.e.,

$$\frac{Y(t,v_1)}{F(Y(t,v_1))} = [1 - \alpha + R_1(t,v_1)]G(Y(t,v_1))$$
and

$$\frac{F(Y(t,v_1))}{Y(t,v_1)} = \frac{1/(1-\alpha) + R_2(t,v_1)}{F(Y(t,v_1))},$$

where the functions  $R_i$ , i = 1, 2, satisfy the conditions

$$\lim_{t \to \omega} R_i(t, v_1) = 0, \ i = 1, 2, \tag{4.73}$$

uniformly with respect to  $v_1 \in [-1/2, 1/2]$ . Therefore, by virtue of the form of the function  $Y(t, v_1)$ , we have the representations

$$\frac{Y(t,v_1)}{F(Y(t,v_1))} = [1 - \alpha + R_1(t,v_1)]q(t)(1+v_1),$$

$$\frac{F(Y(t,v_1))}{Y(t,v_1)} = \frac{1/(1-\alpha) + R_2(t,v_1)}{q(t)(1+v_1)}.$$
(4.74)

By taking into account these representations and by setting

$$h(t) = \frac{(\lambda_0 - 1)t_{\omega}J'(t)}{A_1(1 - \alpha)J(t)},$$

we rewrite system (4.72) in the form

$$v_{1}' = \frac{\beta}{\lambda_{0} - 1} [f_{1}(\tau, v_{1}, v_{2}) + A_{1}(1 - \alpha)v_{2} + V_{1}(v_{1}, v_{2})],$$
  

$$v_{k}' = \frac{\beta}{\lambda_{0} - 1} [-A_{k-1}v_{k} + A_{0}v_{k+1} + V_{k}(v_{k}, v_{k+1})], \quad k = 2, \dots, n - 1,$$
  

$$v_{n}' = \frac{\beta}{\lambda_{0} - 1} \left( f_{2}(\tau, v_{1}, \dots, v_{n-1}) - \sum_{i=1}^{n-1} v_{i} - (\lambda_{0} + 1)v_{n} + V_{n}(\tau, v_{1}, \dots, v_{n}) \right),$$
  
(4.75)

where

$$\begin{aligned} f_1(\tau, v_1, v_2) &= A_1[R_1(t, v_1)(1 + v_2) + (1 - \alpha) - (1 - \alpha)h(t)](1 + v_1), \\ V_1(v_1, v_2) &= A_1(1 - \sigma)v_1v_2, \\ V_k(v_k, v_{k+1}) &= A_k v_k v_{k+1} - A_{k-1}v_k^2, \ k = 2, \dots, n - 1, \\ f_2(\tau, v_1, \dots, v_{n-1}) &= (1 - \alpha)h(t)R_2(t, v_1)\prod_{k=1}^{n-1}\frac{1}{1 + v_k} + (h(t) - 1)\left(1 - \sum_{k=1}^{n-1}v_k\right), \\ V_n(\tau, v_1, \dots, v_n) &= -A_{n-1}v_n^2 + h(t)\left(\prod_{k=1}^{n-1}\frac{1}{1 + v_k} - 1 + \sum_{k=1}^{n-1}v_k\right) \end{aligned}$$

Consider the resulting system on the set  $[\tau_0, \infty) \times \mathbb{R}^n_{1/2}$ , where  $\tau_0 = \beta \ln |t_\omega|$  and  $\mathbb{R}^n_{1/2} = \{(v_1, \ldots, v_n) \in \mathbb{R}^n : |v_k| \le 1/2, k = 1, \ldots, n\}$ . Since  $\tau'(t) = \beta/t_\omega > 0$  for  $t \in [t_0, \omega)$ ,  $\lim_{t\to\omega} \tau(t) = \infty$ , and  $t = t(\tau)$  is the inverse function of  $\tau : [t_0, \omega) \to [\tau_0, \infty)$ , it follows that the right-hand sides of the system are continuous on the set

 $[\tau_0, \infty) \times \mathbb{R}^n_{1/2}$ , and, by virtue of the conditions of the theorem,  $\lim_{\tau \to \infty} f_1(\tau, v_1, v_2) = 0$  uniformly with respect to  $(v_1, v_2) \in \mathbb{R}^2_{1/2}$ ,  $\lim_{\tau \to \infty} f_2(\tau, v_1, \dots, v_{n-1}) = 0$  uniformly with respect to  $(v_1, \dots, v_{n-1}) \in \mathbb{R}^{n-1}_{1/2}$ ,

$$\lim_{|v_k|+|v_{k+1}|\to 0} \frac{V_k(v_k, v_{k+1})}{|v_k|+|v_{k+1}|} = 0$$

 $k=1,\ldots,n-1,$ 

$$\lim_{|v_1|+\dots+|v_n|\to 0} \frac{V_n(\tau, v_1, \dots, v_n)}{|v_1|+\dots+|v_n|} = 0$$

uniformly with respect to  $\tau \in [\tau_0, \infty)$ . In addition, by writing out the characteristic equation det $(C - \rho E) = 0$  of the matrix *C* consisting of the coefficients  $v_1, \ldots, v_n$  in square brackets in system (4.75), that is, the matrix

$$C = \begin{pmatrix} 0 & A_1(1-\alpha) & 0 & \cdots & 0 & 0 \\ 0 & -A_1 & A_2 & \cdots & 0 & 0 \\ \vdots & \vdots & \vdots & \ddots & \vdots & \vdots \\ 0 & 0 & 0 & \cdots & -A_{n-2} & A_{n-1} \\ -1 & -1 & -1 & \cdots & -1 & -(\lambda_0+1) \end{pmatrix}$$

we obtain the algebraic equation (4.62), which, by the assumptions of the theorem, has no roots with zero real part. We have thereby shown that system (4.75) satisfies all assumptions of [35, Theorem 2.2]. By this theorem, system (4.75) has at least one solution  $(v_1, \ldots, v_n) : [\tau_1, \infty) \to \mathbb{R}^n$ ,  $\tau_1 \ge \tau_0$ , that tends to zero as  $\tau \to \infty$ . Moreover, there exists an *m*-parameter family of such solutions if, among the roots of algebraic equation (4.62), there are *m* roots whose signs coincide with that of  $\beta(1 - \lambda_0)$ . By virtue of the changes of variables (4.71) and conditions (4.69) and (4.63), to each such solution of system (4.75), there corresponds a solution of differential equation (4.59) admitting asymptotic representations (4.65) and (4.66). One can readily see that this solution is a  $\mathcal{P}_{\omega}(T, \lambda_0)$  solution of (4.59). Since  $\beta = \operatorname{sgn} t_{\omega}$ , it follows that equation (4.59) has an *m*-parameter family of such solutions if, among the roots of (4.62), there are *m* roots whose sign coincide with that of the function  $(1 - \lambda_0)t_{\omega}$  in a left neighborhood of  $\omega$ .

**Remark 4.5.** (i) Algebraic equation (4.62) necessarily has no solutions with zero real part if  $|\alpha| < 1$ .

(ii) Assume that *F* satisfies the condition

$$L_F(zg(z)) = L_F(z)[1 + o(1)] \quad \text{as } z \to T \ (z \in \Delta_T) \tag{4.76}$$

for every continuously differentiable function  $g: \Delta_T \to (0, \infty)$  with the property

$$\lim_{z\to T,\,z\in\Delta_T}\frac{zg'(z)}{g(z)}=0.$$

Compare (4.76) with (4.23), (4.95), and (4.96). If  $y : [t_0, \omega) \to \Delta_T$  is a continuously differentiable function such that

$$\lim_{t \to \omega} y(t) = T, \quad \frac{y'(t)}{y(t)} = \frac{\xi'(t)}{\xi(t)} [\vartheta + o(1)] \text{ as } t \to \omega,$$

where  $\vartheta$  is a nonzero real constant and  $\xi$  is a real function differentiable in some left neighborhood of  $\omega$  and such that  $\xi'(t) \neq 0$ , then

$$L_F(y(t)) = L_F(\mu_0 |\xi(t)|^{\vartheta}) [1 + o(1)]$$

as  $t \to \omega$ , since y(t) = z(t)g(z(t)), where  $z(t) = \mu_0 |\xi(t)|^{\vartheta}$ , and

$$\lim_{z \to T, z \in \Delta_T} \frac{zg'(z)}{g(z)} = \lim_{t \to \omega} \frac{z(t)g'(z(t))}{g(z(t))}$$
$$= \lim_{t \to \omega} \frac{z(t)(y(t)/z(t))'}{(y(t)/z(t))z'(t)}$$
$$= \lim_{t \to \omega} \left(\frac{\xi(t)y'(t)}{\vartheta\xi'(t)y(t)} - 1\right) = 0$$

By virtue of representations (4.66), a  $\mathcal{P}_{\omega}(T, \lambda_0)$  solution of (4.59) is a function regularly varying (at  $\omega$ ) of index  $\vartheta = A_1/(\lambda_0 - 1)$ . Therefore, if in Theorem 4.13 we additionally assume that *F* satisfies (4.76), then the above observations, together with (4.65), imply the asymptotic representation

$$y(t) = \mu_0 \left| \left[ (\lambda_0 - 1) t_\omega \right]^n p(t) \prod_{i=1}^{n-1} \frac{1}{A_i} L_F \left( \mu_0 | t_\omega |^{\frac{A_1}{\lambda_0 - 1}} \right) \right|^{\frac{1}{1 - \alpha}} (1 + o(1))$$

as  $t \to \omega$ . Therefore, in this case, the asymptotic formulas for  $\mathcal{P}_{\omega}(T, \lambda_0)$  solutions and their derivatives of order less than or equal to n-1 can be written out in closed form.

(iii) Note that Evtukhov et al. in their previous papers related to the topic of this chapter (some of them are mentioned above), additionally assumed that *F* is a twice continuously differentiable function on the interval  $\Delta_T$  and

$$\lim_{u\to T,\,u\in\Delta_T}\frac{uF''(u)}{F'(u)}=\alpha-1.$$

However, in the result which we describe here, the approach was improved in such a way that this condition can be omitted.

(iv) The paper [36] also deals with other types of solutions; in particular conditions guaranteeing the existence in the classes  $\mathcal{P}_{\omega}(T, \zeta)$ , where  $\zeta = \pm \infty$  or  $\zeta = 1$  or  $\zeta = (n-2)/(n-1)$  or  $\zeta = (n-i-1)/(n-i)$  with i = 2, 3, ..., n-1 or  $\zeta = 0$ , are established.

(v) In the paper [89] by Kusano and Manojlović, comments and comparisons of the results with ones in [36] are presented; note that a summary of the results

from [89] is given in the previous section and concerns equation (4.58) which is a special case of (4.59). The observations which we present next, are based on those comments and somehow adapted. Although not specifically emphasized in [36] — with some exceptions — Evtukhov and Samoilenko restricted their attention on the equation with  $\mathcal{RV}$  coefficient and focused their attention only on its  $\mathcal{RV}$  solutions. Namely, conditions imposed in [36] on the function p in (4.59) mean, due to converse half of Karamata's theorem, that p is of regular variation. That fact is neither mentioned nor used by Evtukhov and Samoilenko. Moreover, while they clearly emphasized that  $\mathcal{P}_{\infty}(T, \lambda_0)$  solutions with

$$\lambda_0 = \mathbb{R} \setminus \left\{ 0, \frac{1}{2}, \frac{2}{3}, \dots, \frac{n-2}{n-1}, 1 \right\},\,$$

and  $\lambda_0 = \infty$  are functions of regular variation (cf. (ii) of this remark), such an assertion for  $\mathcal{P}_{\infty}(T, (n-i-1)/(n-i))$  solutions with  $i \in \{1, 2, ..., n-1\}$  is missing. Also, by virtue of representations of such solutions obtained in [36] such a conclusion is almost impossible to make. But, using the fact that p is of regular variation, it becomes quite clear that each  $\mathcal{P}_{\infty}(T, \lambda_0)$  solution, with  $\lambda_0 \neq 1$ , is regularly varying, while assuming that p is of rapid variation,  $\mathcal{P}_{\infty}(T, 1)$  solutions are rapidly varying. Moreover, the converse is also true, due to the fact that  $p \in \mathcal{RV}(\sigma)$ . Indeed, denote with  $\mathcal{P}(T, \lambda_0)$  the set of all  $\mathcal{P}_{\infty}(T, \lambda_0)$  solutions of (4.58). Then, assuming that  $y \in \mathcal{RV}(\vartheta)$ , from equation (4.58) we may conclude  $y^{(n)} \in \mathcal{RV}(\sigma + \vartheta \gamma)$ , which by application of the Karamata integration theorem implies that

$$\begin{split} y^{(n-1)} &\in \mathcal{RV}(\sigma + \vartheta\gamma + 1) \implies \lim_{t \to \infty} t \frac{y^{(n)}(t)}{y^{(n-1)}(t)} = \sigma + \vartheta\gamma + 1 \\ y^{(n-2)} &\in \mathcal{RV}(\sigma + \vartheta\gamma + 2) \implies \lim_{t \to \infty} t \frac{y^{(n-1)}(t)}{y^{(n-2)}(t)} = \sigma + \vartheta\gamma + 2. \end{split}$$

Therefore, condition (4.61) in the definition of  $\mathcal{P}_{\infty}(T, \lambda_0)$  solutions becomes

$$\frac{\sigma + \vartheta \gamma + 2}{\sigma + \vartheta \gamma + 1} = \lim_{t \to \infty} \frac{t \frac{y^{(n-1)}(t)}{y^{(n-2)}(t)}}{t \frac{y^{(n)}(t)}{y^{(n-1)}(t)}} = \lim_{t \to \infty} \frac{[y^{(n-1)}(t)]^2}{y^{(n)}y^{(n-2)}(t)},$$

so that

$$y \in \mathcal{P}\left(T, \frac{\sigma + \vartheta \gamma + 2}{\sigma + \vartheta \gamma + 1}\right), \text{ i.e., } \mathcal{RS}(\vartheta) \subseteq \mathcal{P}\left(T, \frac{\sigma + \vartheta \gamma + 2}{\sigma + \vartheta \gamma + 1}\right),$$

where  $\mathcal{RS}(\vartheta) = \mathcal{RS} \cap \mathcal{RV}(\vartheta)$ ,  $\mathcal{RS}$  being the set of all  $\mathcal{RV}$  solutions of (4.58). More

specifically, it is not difficult to see that, in fact

$$\mathcal{RS}(\vartheta) = \mathcal{P}\left(T, \frac{2n-1-\vartheta}{2n-\vartheta}\right),$$
  
for  $\vartheta \in (-\infty, 2n) \setminus \{0, 1, \dots, 2n-1\}, T = \begin{cases} 0 & \text{if } \vartheta < 0 \\ \infty & \text{if } \vartheta > 0 \end{cases},$   
$$\mathcal{RS}(0) = \mathcal{P}\left(0, \frac{2n-1}{2n}\right),$$
  
$$\mathcal{RS}(k) = \mathcal{P}\left(\infty, \frac{2n+1-k}{2n-k}\right) \text{ for } k = \{1, 2, \dots, 2n-1\},$$
  
$$\mathcal{RS}(2n) = \mathcal{P}(\infty, \infty),$$

which makes the results from the previous subsection closely connected with the ones in this section and in [36].

#### **4.3.8** Strongly monotone solutions of *n*-th order nonlinear systems

This subsection is based on the paper [124] by Matucci, Řehák and on the paper [146] by Řehák. Consider the nonlinear differential system

$$\begin{cases} x'_{1} = \delta a_{1}(t)F_{1}(x_{2}), \\ x'_{2} = \delta a_{2}(t)F_{2}(x_{3}), \\ \vdots \\ x'_{n-1} = \delta a_{n-1}(t)F_{n-1}(x_{n}), \\ x'_{n} = \delta a_{n}(t)F_{n}(x_{1}), \end{cases}$$
(4.77)

 $n \in \mathbb{N}$ ,  $n \ge 2$ , where  $F_i$ , i = 1, ..., n, are continuous functions defined on  $\mathbb{R}$  with  $uF_i(u) > 0$  for  $u \ne 0$ ,  $a_i$ , i = 1, ..., n, are positive continuous functions defined on  $[T, \infty)$ ,  $T \ge 0$ , and  $\delta \in \{-1, 1\}$ .

We assume

$$a_i \in \mathcal{RV}(\sigma_i), \ \sigma_i \in \mathbb{R}, \ i = 1, \dots, n,$$

$$(4.78)$$

and

$$F_i(|\cdot|) \in \mathcal{RV}(\alpha_i), \ \alpha_i \in (0,\infty), \ i = 1, \dots, n,$$

$$(4.79)$$

when studying the asymptotics of strongly increasing solutions, or (4.78) and

$$F_i(|\cdot|) \in \mathcal{RV}_0(\alpha_i), \ \alpha_i \in (0,\infty), \ i = 1,\dots,n.$$

$$(4.80)$$

when the asymptotics of strongly decreasing solutions is considered. Further, in both cases we assume that the indices  $\alpha_1, \dots, \alpha_n$  satisfy

$$\alpha_1 \cdots \alpha_n < 1. \tag{4.81}$$

Recall that system (4.77) satisfying condition (4.81) is called *subhomogeneous* (an alternative terminology is *sub-half-linear*). The opposite (strict) inequality is called *superhomogeneity* (or *super-half-linearity*).

The aim is to study asymptotic behavior of positive solutions to system (4.77), i.e., solutions having all their components eventually positive. With  $\delta = 1$ , positive solutions are eventually increasing, while with  $\delta = -1$ , positive solutions are eventually decreasing. If one of the components in either case tends to a nonzero real constant as the independent variable tends to infinity, then the asymptotic behavior is clear from a certain point of view; asymptotic formulas can be easily derived from the integral form of system (4.77). Therefore, we are interested in the extreme cases where all the solution components tend to infinity or tend to zero. Among other, we establish conditions guaranteeing that all solutions in these classes are regularly varying and satisfy certain asymptotic formula.

Let S denote the set of all solutions of (4.77) which are defined in a neighborhood of infinity and do not eventually vanish — the so-called *proper solutions* — and whose components are eventually positive. Due to the sign conditions on the coefficients and on the nonlinearities in (4.77), it is easy to see that if one component of a solution of (4.77) is eventually of one sign, then all its components are eventually of one sign; we speak about *nonoscillatory solutions*. Further, any nonoscillatory solution of (4.77) has necessarily all components eventually monotone. Denote by  $\mathcal{DS} \subseteq \mathcal{S}$  the subset of the solutions of (4.77) which are in  $\mathcal{S}$ , and whose components are eventually decreasing; similarly, we denote by  $IS \subseteq S$  the subset of the solutions of (4.77) which are in S, and whose components are eventually increasing. We will study the set  $\mathcal{DS}$  under the condition  $\delta = -1$ , and the set IS under the condition  $\delta = 1$ ; later we explain why this setting is natural and nonrestrictive. First note that if  $\delta = -1$ , then  $\mathcal{DS} = S$ , and if  $\delta = 1$ , then IS = S. It is clear that  $\mathcal{DS}$  contains only those solutions whose components all tend to zero or at least one component tends to a positive constant and the other ones to zero (as  $t \to \infty$ ), while IS contains only those solutions whose components tend all to infinity or at least one component tends to a positive constant and the other ones to infinity (as  $t \to \infty$ ). Hence we denote

$$SDS = \{(x_1,\ldots,x_n) \in DS : \lim_{t\to\infty} x_i(t) = 0, i = 1,\ldots,n\},\$$

which are the so-called strongly decreasing solutions, and

$$SIS = \{(x_1,\ldots,x_n) \in IS : \lim_{t\to\infty} x_i(t) = \infty, i = 1,\ldots,n\},\$$

which are the so-called *strongly increasing solutions*.

The condition  $\delta = 1$  resp.  $\delta = -1$  is somehow natural when studying the classes *IS* resp. *DS*. In order to justify this assertion, we recall a standard classification of nonoscillatory solutions. At the same time, thereby we put the presented results into a broader context. Naito in [128, 129] considers the *n*-th order differential equation

$$D(\gamma_n)D(\gamma_{n-1})\cdots D(\gamma_1)x + \tilde{\delta}p(t)\Phi_\beta(x) = 0, \qquad (4.82)$$

where  $n \ge 2$ ,  $\gamma_1, \ldots, \gamma_n, \beta \in (0, \infty)$ ,  $\tilde{\delta} = 1$  or  $\tilde{\delta} = -1$ , p(t) > 0, and  $D(\gamma)x = \frac{d}{dt}(\Phi_{\gamma}(x))$ . Equation (4.82) is a special case of system (4.77). Indeed, if  $F_i = \Phi_{\alpha_i}$ ,  $i = 1, \ldots, n$ , then (4.77) can be equivalently written as

$$D_{a_{n-1}^{-\frac{1}{\alpha_{n-1}}}}\left(\frac{1}{\alpha_{n-1}}\right)\cdots D_{a_{1}^{-\frac{1}{\alpha_{1}}}}\left(\frac{1}{\alpha_{1}}\right)D_{1}(1)x_{1} = \delta^{n}a_{n}(t)\Phi_{\alpha_{n}}(x_{1}),$$
(4.83)

where  $D_f(\gamma)x = \frac{d}{dt}(f(t)\Phi_{\gamma}(x))$ . Using the substitution  $x_1(t) = \Phi_{\gamma_1}(x(t))$  and noticing that  $D_1(1)(\Phi_{\gamma_1}(x)) = D(\gamma_1)x$ , equation (4.83) reduces to (4.82) choosing  $\alpha_i = 1/\gamma_{i+1}$ , i = 1, ..., n-1,  $\alpha_n = \beta/\gamma_1$ ,  $a_i(t) \equiv 1$ , i = 1, ..., n-1,  $a_n(t) = p(t)$ , and  $\tilde{\delta} = -\delta^n$ . Naito made a basic classification of positive solutions to (4.82) extending well known results by Kiguradze and Chanturia [79] for the quasilinear equation

$$x^{(n)} + \tilde{\delta}p(t)\Phi_{\beta}(x) = 0.$$
(4.84)

Notice that (4.82) reduces to (4.84) when  $\gamma_1 = \cdots = \gamma_n = 1$ . The classification of solutions *x* to (4.82) in Naito [128] is made according to the eventual signs of  $D(\gamma_j) \cdots D(\gamma_1)x(t)$ ,  $j = 0, \dots, n-1$  (with the operator being the identity when j = 0). For solutions *x* of (4.82), the so-called *Kiguradze class of degree k* is denoted by  $\mathcal{K}_k$  and defined as

$$\begin{cases} D(\gamma_j) \cdots D(\gamma_1) x(t) > 0 & \text{for } j = 0, 1, \dots, k-1, \\ (-1)^{j-k} D(\gamma_j) \cdots D(\gamma_1) x(t) > 0 & \text{for } j = k, \dots, n-1 \end{cases}$$
(4.85)

for *t* large, where for k = 0 (resp. k = n), the first line (resp. the second line) in (4.85) is omitted. It is not difficult to see that

$$x \in \mathcal{K}_0 \iff (x_1, \dots, x_n) \in \mathcal{DS} \text{ and } x \in \mathcal{K}_n \iff (x_1, \dots, x_n) \in IS,$$
 (4.86)

where  $x_1 = \Phi_{\gamma_1}(x)$ ,  $x_i = \delta \Phi_{\gamma_i}(x'_{i-1})$ , i = 2, ..., n. If we denote the set of all eventually positive solutions of (4.82) by  $\mathcal{K}$ , then [128, Theorem 1.1] implies that

$$\begin{pmatrix}
\mathcal{K} = \mathcal{K}_1 \cup \mathcal{K}_3 \cup \cdots \cup \mathcal{K}_{n-1} & \text{for } \tilde{\delta} = 1 \text{ and } n \text{ even;} \\
\mathcal{K} = \mathcal{K}_0 \cup \mathcal{K}_2 \cup \cdots \cup \mathcal{K}_{n-1} & \text{for } \tilde{\delta} = 1 \text{ and } n \text{ odd;} \\
\mathcal{K} = \mathcal{K}_0 \cup \mathcal{K}_2 \cup \cdots \cup \mathcal{K}_{n-2} \cup \mathcal{K}_n & \text{for } \tilde{\delta} = -1 \text{ and } n \text{ even;} \\
\mathcal{K} = \mathcal{K}_1 \cup \mathcal{K}_3 \cup \cdots \cup \mathcal{K}_{n-2} \cup \mathcal{K}_n & \text{for } \tilde{\delta} = -1 \text{ and } n \text{ odd.}
\end{cases}$$
(4.87)

In view of (4.86) and the equality  $\delta^n a_n(t) = -\delta p(t)$ , the relations in (4.87) says that the condition  $\delta = -1$  and  $\delta = 1$  are quite natural and nonrestrictive when studying, respectively, positive decreasing and positive increasing solutions of (4.77).

The subscripts that indicate the components are always to be intended modulo *n* and not bigger than *n*, that is

$$u_{k} = \begin{cases} u_{k} & \text{if } 1 \le k \le n, \\ u_{k-mn} & \text{if } k > n, \end{cases}$$

$$(4.88)$$

where  $m \in \mathbb{N}$  is such that  $1 \le k - mn \le n$ . With this convention, which is used throughout the paper, system (4.77) takes the form

$$x'_{i} = \delta a_{i}(t)F_{i}(x_{i+1}), \quad i = 1, \dots, n.$$

As usual, the slowly varying component in the representation of  $f \in \mathcal{RV}(\vartheta)$ will be denoted by  $L_f$ , i.e.,  $L_f(t) = f(t)/t^\vartheta$ ; similarly for  $f \in \mathcal{RV}_0(\vartheta)$ .

For sake of simplicity, we introduce here some constants that repeatedly appear in what follows. We set

$$A_{i,j} = \prod_{k=i}^{j-1} \alpha_k, \text{ for } 1 \le i \le j \le i+n \le 2n.$$

It is easy to check that  $A_{i,n+i} = \alpha_1 \cdots \alpha_n$  and  $A_{i,i} = 1$  for all  $i = 1, \dots, n$ . We emphasize that the convention (4.88) is used only for simple subscripts and not for double ones. We indicate by  $(v_1, \dots, v_n)$  the unique solution of the linear system

$$v_i - \alpha_i v_{i+1} = \sigma_i + 1, \quad i = 1, \dots, n,$$
 (4.89)

where  $\sigma_1, \ldots, \sigma_n$  are given real numbers. Notice that the associated matrix is nonsingular thanks to the subhomogeneity condition. Further, let  $(h_1, \ldots, h_n)$  be the unique solution of

$$|v_i|h_i = h_{i+1}^{\alpha_i}, \quad i = 1, \dots, n.$$
 (4.90)

Notice that the subhomogeneity condition plays a key role in its unique solvability again. A simple calculation shows that

$$\nu_i = \frac{1}{1 - A_{i,n+i}} \sum_{k=0}^{n-1} (\sigma_{i+k} + 1) A_{i,i+k}, \quad i = 1, \dots, n,$$
(4.91)

and

$$h_i = \left(\prod_{k=0}^{n-1} |\nu_{i+k}|^{-A_{i,i+k}}\right)^{\frac{1}{1-A_{i,n+i}}}, \quad i = 1, \dots, n.$$

If we set

$$L_{i}(t) = \left(\prod_{j=0}^{n-1} \left(L_{a_{i+j}}(t)L_{F_{i+j}}(t^{\nu_{i+j+1}})\right)^{A_{i,i+j}}\right)^{\frac{1}{1-A_{i,n+i}}}, \quad i = 1, \dots, n,$$
(4.92)

then  $(L_1, ..., L_n)(t)$  is the unique solution (up to asymptotic equivalence) to the system of the relations

$$L_i(t) \sim L_{a_i}(t) L_{i+1}^{\alpha_i}(t) L_{F_i}(t^{\nu_{i+1}}) \quad \text{as } t \to \infty, \quad i = 1, \dots, n.$$
 (4.93)

Observe that if  $L_{F_1} \equiv \cdots \equiv L_{F_n} \equiv 1$ , then  $(L_1, \ldots, L_n)(t)$  reduces to the unique solution of the system

$$L_{i}(t) = L_{a_{i}}(t)L_{i+1}^{\alpha_{i}}(t), \qquad (4.94)$$

 $i = 1, \ldots, n$ .

In what follows, we assume an additional condition for the slowly varying components of the nonlinearities  $F_i$ . In particular, if  $F_i$ , i = 1, ..., n, satisfies (4.80), we assume

$$L_{F_i}(ug(u)) \sim L_{F_i}(u)$$
 as  $u \to 0+$ ,  $i = 1, ..., n$ , (4.95)

for every  $g \in SV_0$ , while if  $F_i$ , i = 1, ..., n, satisfies (4.79), we assume

$$L_{F_i}(ug(u)) \sim L_{F_i}(u) \quad \text{as } u \to \infty, \quad i = 1, \dots, n, \tag{4.96}$$

for every  $g \in SV$ . Compare conditions (4.95) and (4.96) with (4.23) and (4.76).

Now we are ready to present the main results. The first theorem gives sufficient conditions under which system (4.77) possesses a SDS or SIS solution which is regularly varying and we provide an exact asymptotic formula. We point out that  $F_i$  does not need to be monotone.

#### Theorem 4.14. Let (4.78) hold.

(i) Assume 
$$\delta = -1$$
, (4.80) and (4.95). If  $v_i < 0$ ,  $i = 1, ..., n$ , then there exists

 $(x_1,\ldots,x_n) \in SDS \cap (\mathcal{RV}(\nu_1) \times \cdots \times \mathcal{RV}(\nu_n))$ 

and

$$x_i(t) \sim h_i t^{\nu_i} L_i(t) \quad as \ t \to \infty, \quad i=1,\dots,n. \tag{4.97}$$

(*ii*) Assume  $\delta = 1$ , (4.79) and (4.96) If  $v_i > 0$ , i = 1, ..., n, then there exists

$$(x_1,\ldots,x_n) \in SIS \cap (\mathcal{RV}(v_1) \times \cdots \times \mathcal{RV}(v_n))$$

and (4.97) holds.

The proof of the above theorem will be given later. As a corollary, we get sufficient conditions for  $SDS \neq \emptyset$  and  $SIS \neq \emptyset$ .

In the second result, strengthening the assumptions on the nonlinearities  $F_i$ , we can show that

all *SDS* and *SIS* solutions are regularly varying.

**Theorem 4.15.** *Let* (4.78) *hold and*  $F_i = \Phi_{\alpha_i}$  *with*  $\alpha_i > 0$ , i = 1, ..., n.

(*i*) If  $\delta = -1$  and  $v_i < 0$ , i = 1, ..., n, then  $SDS \neq \emptyset$  and for every  $(x_1, ..., x_n) \in SDS$ , it holds  $(x_1, ..., x_n) \in \mathcal{RV}(v_1) \times \cdots \times \mathcal{RV}(v_n)$ . Further, (4.97) holds with  $L_{F_1} \equiv \cdots \equiv L_{F_n} \equiv 1$ .

(*ii*) If  $\delta = 1$  and  $v_i > 0$ , i = 1, ..., n, then  $SIS \neq \emptyset$  and for every  $(x_1, ..., x_n) \in SIS$ , *it holds*  $(x_1, ..., x_n) \in \mathcal{RV}(v_1) \times \cdots \times \mathcal{RV}(v_n)$ . Further, (4.97) holds with  $L_{F_1} \equiv \cdots \equiv L_{F_n} \equiv 1$ .

An alternative expression of sufficient conditions in Theorem 4.14 and Theorem 4.15 can be found in subsequent Lemma 4.8.

In order to prove Theorem 4.14 and Theorem 4.15, we need some technical lemmas. The first two lemmas analyze conditions (4.95), (4.96) and show how they lead to the unique solvability (up to asymptotic equivalence) in the class SV of relation (4.93).

Lemma 4.3. If (4.95) holds, then

$$L_{F_i}(t^{\nu}h(t)) \sim L_{F_i}(t^{\nu})$$
 (4.98)

as  $t \to \infty$  for any v < 0 and  $h \in SV$ . Analogously, condition (4.96) implies that (4.98) holds as  $t \to \infty$  for any v > 0 and  $h \in SV$ .

*Proof.* Taking the substitution  $u = t^{\nu}$ , relation (4.98) is transformed into

$$L_{F_i}(uh(u^{-\nu})) \sim L_{F_i}(u), \quad u \to 0+.$$
 (4.99)

Since  $g(u) := h(u^{-\nu}) \in \mathcal{RV}_0(0) = \mathcal{SV}_0$ , condition (4.95) implies that (4.99) holds. Similarly we can show that (4.96) implies (4.98) for any  $\nu > 0$ .

**Lemma 4.4.** Let  $v_i < 0$  for i = 1, ..., n. Then the system of asymptotic relations (4.93) has a unique solution (up to asymptotic equivalence) belonging to the set  $SV_0$ , and this solution is given by (4.92). Further, the system of asymptotic relations

$$\tilde{L}_{i}(t) \sim \frac{1}{|\nu_{i}|} L_{a_{i}}(t) \tilde{L}_{i+1}^{\alpha_{i}}(t) L_{F_{i}}(t^{\nu_{i+1}}) \quad as \ t \to \infty, \quad i = 1, \dots, n,$$
(4.100)

has the unique solution (up to asymptotic equivalence)

 $\tilde{L}_i(t) = h_i L_i(t),$ 

where  $h_i$  satisfies (4.90), for all i = 1, ..., n. Analogous statements hold under the assumption  $v_i > 0$ , i = 1, ..., n.

Proof. The first statement follows from the fact that system (4.93) can be written as

$$L_{i}(t) \sim L_{a_{i}}(t)L_{F_{i}}(t^{\nu_{i+1}}) \left( L_{a_{i+1}}(t)L_{F_{i+1}}(t^{\nu_{i+2}})L_{i+2}^{\alpha_{i+1}}(t) \right)^{\alpha_{i}} \sim \cdots \sim \\ \sim \prod_{j=0}^{n-1} \left( L_{a_{i+j}}(t)L_{F_{i+j}}(t^{\nu_{i+j+1}}) \right)^{A_{i,i+j}} L_{i+n}^{A_{i,i+n}}(t)$$

as  $t \to \infty$ , i = 1, ..., n, and taking into account that  $L_{i+n}(t) = L_i(t)$ . The second statement is immediate from the definition of  $h_i$ , i = 1, ..., n. Indeed it results

$$\tilde{L}_{i}(t) \sim \frac{h_{i+1}^{\alpha_{i}}}{|\nu_{i}|} L_{a_{i}}(t) L_{i+1}^{\alpha_{i}}(t) L_{F_{i}}(t^{\nu_{i+1}}) = \frac{1}{|\nu_{i}|} L_{a_{i}}(t) \tilde{L}_{i+1}^{\alpha_{i}}(t) L_{F_{i}}(t^{\nu_{i+1}})$$

The following lemma provides some properties of the constants  $v_i$ , i = 1, ..., n defined by (4.89).

**Lemma 4.5.** Let  $(p_1, \ldots, p_n)$  be the unique solution of the system

$$\begin{cases} p_1 + \dots + p_n = 1, \\ \alpha_i p_i + p_{i+2} = p_{i+1}(\alpha_{i+1} + 1), \quad i = 1, 2, \dots, n-2, n. \end{cases}$$
(4.101)

Then

$$p_i = \frac{1 + \sum_{k=i+1}^{n+i-1} A_{k,n+i}}{n + \sum_{k=1}^{n} (A_{k,k+1} + A_{k,k+2} + \dots + A_{k,k+n-1})} > 0, \quad i = 1, \dots, n,$$
(4.102)

and the following identity holds

$$v_1 + \dots + v_n = \frac{(\sigma_1 p_1 + \dots + \sigma_n p_n + 1)}{1 - \xi},$$
 (4.103)

where  $\xi$  is defined by  $p_1 + \cdots + p_{n-2} + p_{n-1}(\alpha_{n-1} + 1)$  and satisfies  $\xi < 1$ . Further, if  $L_{F_1} \equiv \cdots \equiv L_{F_n} \equiv 1$ , then

$$L_1(t)\cdots L_n(t) = \left(L_{a_1}^{p_1}(t)\cdots L_{a_n}^{p_n}(t)\right)^{\frac{1}{1-\xi}}$$
(4.104)

*Proof.* It is easy to check that  $(p_1, ..., p_n)$  given by (4.102) solves (4.101). Positivity of  $p_i$  is clear and the uniqueness can readily be seen when writing (4.101) in a matrix form. The inequality  $\xi < 1$  is equivalent to the subhomogeneity assumption. A series of routine and tedious computations (where we can expand the explicit expressions for  $v_i$  and  $p_i$ , and compare corresponding summands in the resulting formulas) shows that identities (4.103) and (4.104) hold.

Define

$$B_{i,j} = \begin{cases} A_{i,j} & \text{for } 1 \le i < j \le n, \\ \prod_{k=i}^{j-1+n} \alpha_k & \text{for } 1 \le j \le i \le j+n-1 \le 2n-1. \end{cases}$$

It is easy to verify that the following relations hold

$$B_{i,i} = \alpha_1 \cdots \alpha_n,$$
  

$$B_{i,j}B_{j,i} = \alpha_1 \cdots \alpha_n = B_{i,i},$$
  

$$B_{i,j}B_{j,\ell} = B_{i,\ell},$$

where  $i, j, \ell \in \{1, ..., n\}$ , and for the last equality we assume  $i < j < \ell$  or  $\ell < i < j$  or  $j < \ell < i$ . Now let

$$\varrho_{i,j} = \nu_i - B_{i,j}\nu_j, \quad i, j \in 1, \dots, n.$$
(4.105)

In the subsequent lemmas we derive several properties of the constants  $\varrho_{i,j}$ ,  $i, j \in 1, ..., n$ . Their proofs are rather technical and since they do not directly concern regular variation we omit them. For details see [124].

**Lemma 4.6.** Let  $v_i < 0$  for i = 1, ..., n. Then  $\varrho_{i,j}$ ,  $i, j \in 1, ..., n$ , satisfy the following relations:

- (*i*)  $\varrho_{i,i} < 0$  for i = 1, ..., n.
- (ii) If there exist  $i, j \in \{1, ..., n\}, i \neq j$ , such that  $\varrho_{i,j} \ge 0$ , then  $\varrho_{j,i} < 0$ .
- (iii) If there exist  $i, j, \ell \in \{1, ..., n\}$ , with  $i < j < \ell$ , such that  $\varrho_{i,j} < 0$  and  $\varrho_{j,\ell} \le 0$ , then  $\varrho_{i,\ell} < 0$ .
- (iv) If there exist  $i, j, \ell \in \{1, ..., n\}$ , with  $j < i < \ell$ , such that  $\varrho_{i,j} < 0$  and  $\varrho_{\ell,j} \ge 0$ , then  $\varrho_{i,\ell} < 0$ .

**Lemma 4.7.** Let  $\gamma_{i,j}$ ,  $i, j = 1, ..., n, n \ge 2$ , be any numbers which obey the rules (*i*)-(*iv*) in Lemma 4.6. Then the matrix  $(\gamma_{i,j})_{1 \le i,j \le n}$  has at least one row whose elements are negative.

The previous two lemmas play an important role in the proof of the following statement which, among other, enables us to find an alternative expression for sufficient conditions in Theorem 4.14 and Theorem 4.15.

**Lemma 4.8.** The following equivalence holds:  $v_i < 0$  (> 0) for all i = 1, ..., n if and only if  $\varrho_{i,m} < 0$  (> 0) for all i = 1, ..., n and some  $m \in \{1, ..., n\}$ .

**Lemma 4.9.** The numbers  $\varrho_{i,j}$ , i, j = 1, ..., n, defined by (4.105), satisfy the relations

$$\begin{cases} \varrho_{j+k-1,j} = \sigma_{j+k-1} + 1 + \alpha_{j+k-1}\varrho_{j+k,j}, & k = 1, \dots, n-1, \\ \varrho_{j+n-1,j} = \sigma_{j+n-1} + 1, \end{cases}$$
(4.106)

 $j=1,\ldots,n.$ 

Now we are ready to prove the main theorems.

*Proof of Theorem* 4.14. (i) First we prove that (4.77) has at least one solution  $\mathbf{x} = (x_1, \ldots, x_n) \in SDS$ . Since (4.80) is valid, for every  $i = 1, \ldots, n$  there exists  $\tilde{F}_i \in C(\mathbb{R}) \cap \mathcal{RV}_0(\alpha_i)$ , nondecreasing, such that  $\tilde{F}_i(u) \sim F_i(u)$  as  $u \to 0$ . Thus there exists  $u_0 > 0$  such that

$$\frac{1}{\sqrt[4]{2}}\tilde{F}_{i}(u) \leq F_{i}(u) \leq \sqrt[4]{2}\tilde{F}_{i}(u), \quad \forall u \in [0, u_{0}], i = 1, \dots, n.$$
(4.107)

Notice that  $\tilde{F}_i$  satisfies conditions (4.92) and (4.94) if  $F_i$  does, for i = 1, ..., n. Let  $(k_1, ..., k_n)$ , be the unique solution of the linear system  $k_i - \alpha_i k_{i+1} = 1$ . System (4.89) reduces to this if  $\sigma_i = 0$  for all i = 1, ..., n, and therefore, from (4.91), it results  $k_i > 0$  for all i. Let  $\tilde{L}_i(t) = h_i L_i(t)$ , i = 1, ..., n, see Lemma 4.4. Now, taking into account that  $v_i < 0$ , i = 1, ..., n, properties of  $\mathcal{RV}$  functions, assumptions (4.98) and (4.100) imply that  $t_0$  sufficiently large exists such that

$$2^{k_i} \tilde{L}_i(t) t^{\nu_i} \le u_0, \tag{4.108}$$

$$L_{\tilde{F}_{i}}(2^{k_{i+1}}\tilde{L}_{i+1}(t)t^{\nu_{i+1}}) \le \sqrt[4]{2}L_{\tilde{F}_{i}}(t^{\nu_{i+1}}),$$
(4.109)

$$L_{\tilde{F}_{i}}(2^{-k_{i+1}}\tilde{L}_{i+1}(t)t^{\nu_{i+1}}) \ge \frac{1}{\sqrt[4]{2}}L_{\tilde{F}_{i}}(t^{\nu_{i+1}})$$
(4.110)

$$\frac{1}{\sqrt[4]{2}}\tilde{L}_{i}(t) \leq \frac{1}{|\nu_{i}|} L_{a_{i}}(t)\tilde{L}_{i+1}^{\alpha_{i}}(t)L_{\tilde{F}_{i}}(t^{\nu_{i+1}}) \leq \sqrt[4]{2}\tilde{L}_{i}(t),$$
(4.111)

$$\frac{1}{\sqrt[4]{2}} t^{\nu_i} \tilde{L}_i(t) \le |\nu_i| \int_t^\infty s^{\nu_i - 1} \tilde{L}_i(s) \, \mathrm{d}s \le \sqrt[4]{2} t^{\nu_i} \tilde{L}_i(t), \tag{4.112}$$

for all  $t \in [t_0, \infty)$ , and i = 1, ..., n. Let  $\Omega \subset (C[t_0, \infty))^n$  be the set

$$\Omega = \{(x_1,\ldots,x_n): x_i \in C[t_0,\infty),\$$

$$2^{-k_i}\tilde{L}_i(t)t^{\nu_i} \le x_i(t) \le 2^{k_i}\tilde{L}_i(t)t^{\nu_i}, i = 1, \dots, n\}, \quad (4.113)$$

and let **T** :  $\Omega \rightarrow (C[t_0, \infty))^n$  be the operator defined by

$$\mathbf{Tx} = (T_1 x_2, T_2 x_3, \dots, T_n x_1),$$

with

$$(T_i x_{i+1})(t) = \int_t^\infty a_i(s) F_i(x_{i+1}(s)) \, \mathrm{d}s, \quad i = 1, \dots, n.$$

First of all notice that **T** is well defined in  $\Omega$ . Indeed, for all  $x \in \Omega$  we have

$$0 \le a_i(t)F_i(x_{i+1}(t)) \le \sqrt[4]{2}a_i(s)\tilde{F}_i(x_{i+1}(t)) \le \sqrt[4]{2}a_i(s)\tilde{F}_i(2^{k_{i+1}}\tilde{L}_{i+1}(t)t^{\nu_{i+1}}),$$

where we used (4.108) and (4.107). Since  $\tilde{F}_i \in \mathcal{RV}_0(\alpha_i)$  and  $a_i \in \mathcal{RV}(\sigma_i)$ , the last term in the above inequality belongs to the class  $\mathcal{RV}(\sigma_i + \alpha_i \nu_{i+1}) = \mathcal{RV}(\nu_i - 1)$ , see (4.89), with  $\nu_i < 0$ , and therefore it is integrable on  $[t_0, \infty)$ . In particular, taking into account (4.109), (4.111), and (4.112), for every  $t \ge t_0$  it holds

$$\begin{split} (T_{i}x_{i+1})(t) &\leq \sqrt[4]{2} \int_{t}^{\infty} a_{i}(s)\tilde{F}_{i}(2^{k_{i+1}}\tilde{L}_{i+1}(s)s^{\nu_{i+1}}) \,\mathrm{d}s \\ &= 2^{\alpha_{i}k_{i+1}+\frac{1}{4}} \int_{t}^{\infty} s^{\sigma_{i}}L_{a_{i}}(s)\tilde{L}_{i+1}^{\alpha_{i}}(s)s^{\alpha_{i}\nu_{i+1}}L_{\tilde{F}_{i}}(2^{k_{i+1}}\tilde{L}_{i+1}(s)s^{\nu_{i+1}}) \,\mathrm{d}s \\ &= 2^{k_{i}-\frac{3}{4}} \int_{t}^{\infty} s^{\nu_{i}-1}L_{a_{i}}(s)\tilde{L}_{i+1}^{\alpha_{i}}(s)L_{\tilde{F}_{i}}(2^{k_{i+1}}\tilde{L}_{i+1}(s)s^{\nu_{i+1}}) \,\mathrm{d}s \\ &\leq 2^{k_{i}-\frac{1}{2}} \int_{t}^{\infty} s^{\nu_{i}-1}L_{a_{i}}(s)\tilde{L}_{i+1}^{\alpha_{i}}(s)L_{\tilde{F}_{i}}(s^{\nu_{i+1}}) \,\mathrm{d}s \\ &\leq 2^{k_{i}-\frac{1}{4}} |\nu_{i}| \int_{t}^{\infty} s^{\nu_{i}-1}\tilde{L}_{i}(s) \,\mathrm{d}s \leq 2^{k_{i}}t^{\nu_{i}}\tilde{L}_{i}(t). \end{split}$$

Analogously, from (4.110), (4.111), and (4.112), for every  $t \ge t_0$  it holds

$$(T_{i}x_{i+1})(t) \geq 2^{-\frac{1}{4}} \int_{t}^{\infty} a_{i}(s)\tilde{F}_{i}(2^{-k_{i+1}}\tilde{L}_{i+1}(s)s^{\nu_{i+1}}) ds$$
  
$$= 2^{-k_{i}+\frac{3}{4}} \int_{t}^{\infty} s^{\nu_{i}-1}L_{a_{i}}(s)\tilde{L}_{i+1}^{\alpha_{i}}(s)L_{\tilde{F}_{i}}(2^{-k_{i+1}}\tilde{L}_{i+1}(s)s^{\nu_{i+1}}) ds$$
  
$$\geq 2^{-k_{i}+\frac{1}{2}} \int_{t}^{\infty} s^{\nu_{i}-1}L_{a_{i}}(s)\tilde{L}_{i+1}^{\alpha_{i}}(s)L_{\tilde{F}_{i}}(s^{\nu_{i+1}}) ds$$
  
$$\geq 2^{-k_{i}+\frac{1}{4}} |\nu_{i}| \int_{t}^{\infty} s^{\nu_{i}-1}\tilde{L}_{i}(s) ds \geq 2^{-k_{i}}t^{\nu_{i}}\tilde{L}_{i}(t).$$

Thus **T** maps  $\Omega$  into itself. In order to apply the Schauder-Tychonoff fixed point theorem, we have to show that **T** is completely continuous. Since **T**( $\Omega$ )  $\subseteq \Omega$ , functions in **T**( $\Omega$ ) are equibounded on [ $t_0$ ,  $\infty$ ); further, the inequality

$$0 \ge (T_i x_{i+1})'(t) \ge -\sqrt[4]{2} a_i(t) \tilde{F}_i(2^{k_{i+1}} \tilde{L}_{i+1}(t) t^{\nu_{i+1}})$$

which is valid for  $t \ge t_0$ , i = 1, ..., n, and for every  $(x_1, ..., x_n) \in \Omega$ , ensure that functions in  $\mathbf{T}(\Omega)$  are equicontinuous. The relative compactness of  $\mathbf{T}(\Omega)$  follows from the Ascoli-Arzelà theorem. To prove the continuity of  $\mathbf{T}$  we have to show that for any sequence  $\mathbf{x}^m = (x_1^m, ..., x_n^m)$  in  $\Omega$  which converges to  $\bar{\mathbf{x}} = (\bar{x}_1, ..., \bar{x}_n) \in \Omega$  as  $m \to \infty$  uniformly on any compact subset of  $[t_0, \infty)$ , it holds  $(\mathbf{T}\mathbf{x}^m)(t) \to (\mathbf{T}\bar{\mathbf{x}})(t)$  as  $m \to \infty$  uniformly on compact subset of  $[t_0, \infty)$ . But this is a direct consequence of the Lebesgue dominated convergence theorem. Since all the assumptions of the Schauder-Tychonoff fixed point theorem are fulfilled, we obtain the existence of at least one fixed point  $\mathbf{x} \in \Omega$  of the operator  $\mathbf{T}$ . This fixed point  $\mathbf{x} = (x_1, ..., x_n)$ is a positive solution of (4.77), and from the definition of the set  $\Omega$  it follows that  $x_i(t) \to 0$  as  $t \to \infty$ , i = 1, ..., n, i.e.,  $\mathbf{x} \in SDS$ .

Now we prove that  $x_i \in \mathcal{RV}(v_i)$ , i = 1, ..., n. Since  $\mathbf{x} \in \Omega$ , we have  $x_i(t) \asymp t^{v_i} \tilde{L}_i(t)$ as  $t \to \infty$ , i = 1, ..., n. Taking into account that  $\tilde{L}_i(\lambda t)/\tilde{L}_i(t) \to 1$  as  $t \to \infty$  for every  $\lambda > 0$ , i = 1, ..., n, we can find  $m_i, M_i \in (0, \infty)$ , i = 1, ..., n, such that

$$m_i \le \tau_i(t) \le M_i$$
, where  $\tau_i(t) := \frac{x_i(\lambda t)}{x_i(t)}$ ,  $i = 1, \dots, n$ , (4.114)

for  $t \ge t_0$ , and so

$$\liminf_{t\to\infty}\tau_i(t)=:\underline{\Lambda}_i\in(0,\infty),\quad\limsup_{t\to\infty}\tau_i(t)=:\overline{\Lambda}_i\in(0,\infty).$$

From the uniform convergence theorem for  $SV_0$  functions, we get

$$\begin{aligned} \left| \frac{L_{F_i}(x_{i+1}(\lambda t))}{L_{F_i}(x_{i+1}(t))} - 1 \right| &= \left| \frac{L_{F_i}(\tau_{i+1}(t)x_{i+1}(t))}{L_{F_i}(x_{i+1}(t))} - 1 \right| \\ &\leq \sup_{\xi \in [m_i, M_i]} \left| \frac{L_{F_i}(\xi x_{i+1}(t))}{L_{F_i}(x_{i+1}(t))} - 1 \right| = o(1) \end{aligned}$$

as  $t \to \infty$ . Thus,

$$\underline{\Lambda}_{i} \geq \liminf_{t \to \infty} \frac{\lambda x_{i}'(\lambda t)}{x_{i}'(t)} = \liminf_{t \to \infty} \frac{\lambda a_{i}(\lambda t) x_{i+1}^{\alpha_{i}}(\lambda t)}{a_{i}(t) x_{i+1}^{\alpha_{i}}(t)} \cdot \frac{L_{F_{i}}(x_{i+1}(\lambda t))}{L_{F_{i}}(x_{i+1}(t))} \\
\geq \lambda^{1+\sigma_{i}} \left(\liminf_{t \to \infty} \frac{x_{i+1}(\lambda t)}{x_{i+1}(t)}\right)^{\alpha_{i}} \geq \lambda^{1+\sigma_{i}} \left(\liminf_{t \to \infty} \frac{\lambda x_{i+1}'(\lambda t)}{x_{i+1}'(t)}\right)^{\alpha_{i}} \\
= \lambda^{1+\sigma_{i}} \left(\liminf_{t \to \infty} \frac{\lambda a_{i+1}(\lambda t) x_{i+2}^{\alpha_{i+1}}(\lambda t)}{a_{i+1}(t) x_{i+2}^{\alpha_{i+1}}(\lambda t)} \cdot \frac{L_{F_{i+1}}(x_{i+2}(\lambda t))}{L_{F_{i+1}}(x_{i+2}(t))}\right)^{\alpha_{i}} \\
\geq \lambda^{1+\sigma_{i}+\alpha_{i}(1+\sigma_{i+1})} \left(\liminf_{t \to \infty} \frac{\lambda x_{i+2}'(\lambda t)}{x_{i+2}'(t)}\right)^{\alpha_{i}\alpha_{i+1}} \\
\geq \dots \\
\geq \lambda^{1+\sigma_{i}+A_{i,i+1}(1+\sigma_{i+1})+A_{i,i+2}(1+\sigma_{i+2})+\dots+A_{i,i+n-1}(1+\sigma_{i+n-1})} \left(\liminf_{t \to \infty} \tau_{i+n}(t)\right)^{A_{i,i+n}} \\
= \lambda^{\nu_{i}(1-A_{i,i+n})} \underline{\Lambda}_{i}^{A_{i,i+n}},$$
(4.115)

where we used (4.91). Realizing now that  $A_{i,i+n} = \alpha_1 \cdots \alpha_n < 1$ , we obtain  $\underline{\Lambda}_i \ge \lambda^{\nu_i}$ . Similarly we get  $\overline{\Lambda}_i \le \lambda^{\nu_i}$ . This implies that there exists the limit  $\lim_{t\to\infty} x(\lambda t)/x(t)$  and it is equal to  $\lambda^{\nu_i}$ . Since  $\lambda$  was arbitrary, we get  $x_i \in \mathcal{RV}(\nu_i)$ , i = 1, ..., n.

Finally, we establish asymptotic formula (4.97). We have  $x_i(t) = t^{\nu_i} \bar{L}_i(t)$ , i = 1, ..., n, where  $\bar{L}_i \in SV$  has to be determined. Then, taking into account (4.89), (4.95), and (1.8), it results

$$\begin{split} t^{\nu_i} \bar{L}_i(t) &= \int_t^\infty a_i(s) F_i(x_{i+1}(s)) \, \mathrm{d}s = \int_t^\infty s^{\sigma_i} L_{a_i}(s) x_{i+1}^{\alpha_i}(s) L_{F_i}(x_{i+1}(s)) \, \mathrm{d}s \\ &= \int_t^\infty s^{\nu_i - 1} L_{a_i}(s) \bar{L}_{i+1}^{\alpha_i}(s) L_{F_i}(s^{\nu_{i+1}} \bar{L}_{i+1}(s)) \, \mathrm{d}s \\ &\sim \int_t^\infty s^{\nu_i - 1} L_{a_i}(s) \bar{L}_{i+1}^{\alpha_i}(s) L_{F_i}(s^{\nu_{i+1}}) \, \mathrm{d}s \\ &\sim \frac{1}{|\nu_i|} t^{\nu_i} L_{a_i}(t) \bar{L}_{i+1}^{\alpha_i}(t) L_{F_i}(t^{\nu_{i+1}}) \end{split}$$

as  $t \to \infty$ , i = 1, ..., n. Hence,  $\overline{L}_i(t) \sim \widetilde{L}_i(t) = h_i L_i(t)$ , i = 1, ..., n, see Lemma 4.4, and (4.97) follows.

The proof of (ii) is similar and hence omitted. Just note that relation (1.9) finds its extensive application here.  $\hfill \Box$ 

The following lemma, which follows the well known generalized AM-GM inequality, is needed to prove Theorem 4.15.

**Lemma 4.10.** Let  $u_1, ..., u_n > 0$  and  $p_1, ..., p_n \ge 0$  with  $p_1 + \cdots + p_n = 1$ . Then

$$\sum_{i=1}^{n} u_i \ge \prod_{i=1}^{n} u_i^{p_i}.$$
(4.116)

*Proof of Theorem 4.15.* (i) Take any  $(x_1, ..., x_n) \in SDS$ , which indeed exists by Theorem 4.14. From (4.77),  $x_i$  satisfies the integral equation

$$x_i(t) = \int_t^\infty a_i(s) x_{i+1}^{\alpha_i}(s) \,\mathrm{d}s, \tag{4.117}$$

i = 1, ..., n. From Lemma 4.8, there exists  $m \in \{1, 2, ..., n\}$  such that  $\varrho_{i,m} < 0$  for all i = 1, ..., n, where the constants  $\varrho_{i,j}$  are defined by (4.105) and satisfy also (4.106). Iterating (4.117), starting from i = m we get

$$\begin{aligned} x_m(t) &= \int_t^\infty a_m(s_1) x_{m+1}^{\alpha_m}(s_1) \, \mathrm{d}s_1 \\ &= \int_t^\infty a_m(s_1) \left( \int_{s_1}^\infty a_{m+1}(s_2) x_{m+2}^{\alpha_{m+1}} \, \mathrm{d}s_2 \right)^{\alpha_m} \, \mathrm{d}s_1 \\ &= \dots = \int_t^\infty a_m(s_1) \left( \int_{s_1}^\infty a_{m+1}(s_2) \times \right. \\ & \times \left( \dots \left( \int_{s_{n-1}}^\infty a_{m+n-1}(s_n) x_{m+n}^{\alpha_{m+n-1}}(s_n) \, \mathrm{d}s_n \right)^{\alpha_{m+n-2}} \dots \right)^{\alpha_{m+1}} \, \mathrm{d}s_2 \right)^{\alpha_m} \, \mathrm{d}s_1 \end{aligned}$$

Since  $x_{m+n} = x_m$  is eventually decreasing,  $x_{m+n}(s_n) \le x_m(t)$  for  $s_n \ge t$ , t being sufficiently large. Further,  $a_{m+n-1}(t) = t^{\sigma_{m+n-1}}L_{a_{m+n-1}}(t)$ , with  $\sigma_{m+n-1} = \varrho_{m+n-1,m} - 1 < -1$ , see (4.106) and Lemma 4.8. Thus we can apply (1.8) obtaining the existence of a positive constant  $k_n$  such that

$$\int_{s_{n-1}}^{\infty} a_{m+n-1}(s_n) x_{m+n}^{\alpha_{m+n-1}}(s_n) \, \mathrm{d}s_n \le k_n x_m(s_{n-1}) s_{n-1}^{\sigma_{m+n}} L_{a_{m+n-1}}(s_{n-1})$$

We can proceed in a similar way for all the iterated integrals; note that (4.106) and Lemma 4.8 assure that we can apply (1.8) at each step. We obtain that  $k \in (0, \infty)$  exists such that for *t* large

$$\begin{aligned} x_{m}(t) &\leq x_{m}^{\alpha_{m} \cdots \alpha_{m+n-1}}(t) \times \\ &\times \int_{t}^{\infty} s_{1}^{\sigma_{m}} L_{a_{m}}(s_{1}) \left( \dots \left( \int_{s_{n_{1}}}^{\infty} s_{n}^{\sigma_{m+n-1}} L_{a_{m+n-1}}(s_{n}) \, \mathrm{d}s_{n} \right)^{\alpha_{m+n-2}} \dots \right)^{\alpha_{m}} \mathrm{d}s_{1} \\ &\leq k x_{m}^{A_{m,m+n}} L_{a_{m}}(t) L_{a_{m+1}}^{A_{m,m+1}}(t) \cdots L_{a_{m+n-1}}^{A_{m,m+n-1}}(t) t^{\nu_{m}(1-A_{m,m+n})} \end{aligned}$$

where we used the equality  $\nu_m(1 - A_{m,m+n}) = \sigma_m + 1 + A_{m,m+1}(\sigma_{m+1} + 1) + \cdots + A_{m,m+n-1}(\sigma_{m+n-1} + 1)$ , see (4.91). Since  $A_{m,m+n} = \alpha_1 \cdots \alpha_n < 1$ , from (4.92) with  $L_{F_1} \equiv \cdots \equiv L_{F_n} \equiv 1$ , there exists  $d_m \in (0, \infty)$  such that  $x_m(t) \leq d_m t^{\nu_m} L_m(t)$  for large t.

Now we show that  $x_i(t) \le d_i t^{v_i} L_i(t)$  for large *t* and for all i = 1, ..., n, with  $d_i > 0$ . From the estimate for  $x_m = x_{m+n}$ , we can now easily get the estimation for  $x_{m+n-1}$ . Recall that (4.89) and (4.94) hold. From (4.117) and (1.8), in view of  $v_m < 0$ , we have

$$\begin{aligned} x_{m+n-1}(t) &= \int_{t}^{\infty} a_{m+n-1}(s) x_{m}^{\alpha_{m+n-1}}(s) \, \mathrm{d}s \\ &\leq d_{m}^{\alpha_{m+n-1}} \int_{t}^{\infty} s^{\sigma_{m+n-1}+\alpha_{m+n-1}\nu_{m}} L_{a_{m+n-1}}(s) L_{m}^{\alpha_{m+n-1}}(s) \, \mathrm{d}s \\ &\leq d_{m+n-1} t^{\nu_{m+n-1}} L_{a_{m+n-1}}(t) L_{m}^{\alpha_{m+n-1}}(t) = d_{m+n-1} t^{\nu_{m+n-1}} L_{m+n-1}(t) \end{aligned}$$

for large *t*, where  $d_{m+n-1}$  is a suitable positive constant. Repeating this process, and taking into account the modulo *n* convention, it results  $x_i(t) \le d_i t^{v_i} L_i(t)$  for large *t* and for all i = 1, ..., n.

Next we derive lower estimates for  $x_i$ 's. For brevity we sometimes omit arguments. Again take any  $(x_1, ..., x_n) \in SDS$ . Then

$$-(x_1 x_2 \cdots x_n)' = -\sum_{i=1}^n x_i' x_{i+1} \cdots x_{i+n-1} = \sum_{i=1}^n a_i x_{i+1}^{\alpha_i} x_{i+1} \cdots x_{i+n-1}$$
$$= \sum_{i=1}^n H_i, \quad \text{where } H_i = a_i x_{i+1}^{\alpha_i} \prod_{k=1; k \neq i}^n x_k.$$

Consider the (positive) numbers  $p_1, \ldots, p_n$  defined in Lemma 4.5. System (4.101) is equivalent to the system

$$\begin{cases} p_1 + \dots + p_n = 1, \\ p_2 + p_3 + \dots + p_{n-1} + p_n(\alpha_n + 1) = p_1(\alpha_1 + 1) + p_3 + \dots + p_n, \\ p_1(\alpha_1 + 1) + p_3 + \dots + p_n = p_1 + p_2(\alpha_2 + 1) + p_4 + \dots + p_n, \\ \vdots \\ p_1 + \dots + p_{n-3} + p_{n-2}(\alpha_{n-2} + 1) + p_n = p_1 + \dots + p_{n-2} + p_{n-1}(\alpha_{n-1} + 1). \end{cases}$$

$$(4.118)$$

From Lemma 4.10 we get that

$$-(x_1x_2\cdots x_n)' = \sum_{i=1}^n H_i \ge \prod_{i=1}^n H_i^{p_i}$$
  
=  $a_1^{p_1}a_2^{p_2}\cdots a_n^{p_n}x_1^{p_2+p_3+\cdots+p_{n-1}+p_n(\alpha_n+1)}x_2^{p_1(\alpha_1+1)+p_3+\cdots+p_n} \times$   
 $\times \cdots \times x_{n-1}^{p_1+p_2+\cdots+p_{n-2}(\alpha_{n-2}+1)+p_n}x_n^{p_1+p_2+\cdots+p_{n-2}+p_{n-1}(\alpha_{n-1}+1)}$ 

Observe that except of the first equality all sides of the equalities in (4.118) are mutually equal and denote any of them by  $\xi$ ; this is the same  $\xi$  as in Lemma 4.5. Then, from the last estimate, we get

$$-(x_1\cdots x_n)' \ge a_1^{p_1}\cdots a_n^{p_n}(x_1\cdots x_n)^{\xi}.$$
(4.119)

By Lemma 4.5, we have  $\xi < 1$ . Dividing (4.119) by  $(x_1 \cdots x_n)^{\xi}$  and integrating from t to  $\infty$ , we obtain

$$(x_1(t)\cdots x_n(t))^{1-\xi} \ge (1-\xi) \int_t^\infty a_1^{p_1}(s)\cdots a_n^{p_n}(s) \,\mathrm{d}s. \tag{4.120}$$

From the upper estimates for  $x_i$ 's we have

$$x_1(t)\cdots x_n(t) \le l_1 x_i(t) t^{\sum_{k=1; k\neq i}^n \nu_k} \prod_{\substack{k=1; k\neq i}}^n L_k(t)$$
(4.121)

for large *t*, where  $i \in \{1, ..., n\}$  and  $l_1$  is some positive number. Taking into account that  $\sum_{i=1}^{n} \sigma_i p_i < -1$ , see (4.103), from (1.8) there exists  $l_2 \in (0, \infty)$  such that

$$\int_{t}^{\infty} a_{1}^{p_{1}}(s) \cdots a_{n}^{p_{n}}(s) \,\mathrm{d}s \ge l_{2} t^{\sigma_{1}p_{1}+\cdots+\sigma_{n}p_{n}+1} L_{a_{1}}^{p_{1}}(t) \cdots L_{a_{n}}^{p_{n}}(t)$$
(4.122)

for large *t*. Combining (4.120), (4.121), and (4.122), we find  $c_i \in (0, \infty)$  such that

$$x_i(t) \ge c_i t^{\bar{\nu}_i} \bar{L}_i(t)$$

for large *t*, where

$$\bar{\nu}_i = \frac{1}{1-\xi} (\sigma_1 p_1 + \dots + \sigma_n p_n + 1) - \sum_{k=1; k \neq i}^n \nu_k \text{ and } \bar{L}_i = \frac{\left( L_{a_1}^{p_1} \cdots + L_{a_n}^{p_n} \right)^{\frac{1}{1-\xi}}}{\prod_{k=1; k \neq i}^n L_k}$$

Identities (4.103) and (4.104) now imply  $\bar{v}_i = v_i$  and  $\bar{L}_i(t) = L_i(t)$ , i = 1, ..., n. Thus we have proved  $x_i(t) \approx t^{v_i}L_i(t)$  as  $t \to \infty$ , i = 1, ..., n. This implies (4.114). The property  $x_i \in \mathcal{RV}(v_i)$ , i = 1, ..., n, and asymptotic formula (4.97) follow by the same arguments as those used in the second part of the proof of Theorem 4.14-(i).

(ii) The proof of this statement can be found in [146], and uses similar arguments as part (i). Note that certain additional condition is assumed in [146]. But in view of Lemma 4.8, under the assumptions of Theorem 4.15-(ii), that additional condition is automatically satisfied. □

Theorem 4.15 can be applied, for instance, to the equations

$$x^{(n)} = (-1)^n p(t) \Phi_\beta(x), \tag{4.123}$$

and

$$x^{(n)} = p(t)\Phi_{\beta}(x),$$
 (4.124)

where  $p(t) = t^{\varrho}L_p(t)$ ,  $L_p \in SV$  and  $0 < \beta < 1$ , leading to the following results.

(a) If  $\rho + n < 0$ , then (4.123) possesses a solution x such that  $\lim_{t\to\infty} x^{(i)}(t) = 0$ , i = 0, ..., n - 1, and for any such a solution it holds

$$x \in \mathcal{RV}\left(\frac{\varrho+n}{1-\beta}\right) \tag{4.125}$$

with

$$x^{1-\beta}(t) \sim t^{\varrho+n} L_p(t) \prod_{j=1}^n \frac{1-\beta}{-\varrho-n+(1-\beta)(j-1)} \text{ as } t \to \infty.$$

(b) If  $\rho + 1 + \beta(n-1) > 0$ , then equation (4.124) possesses a solution *x* such that  $\lim_{t\to\infty} x^{(i)}(t) = \infty$ , i = 1, ..., n-1, and for any such a solution (4.125) holds with

$$x^{1-\beta}(t) \sim t^{\varrho+n} L_p(t) \prod_{j=1}^n \frac{1-\beta}{\varrho+n-(1-\beta)(j-1)}$$
 as  $t \to \infty$ .

The theorems of this subsection can clearly be applied to the equation of the form (4.82), or even to the more general equation

$$D_{q_n}(\gamma_n) D_{q_{n-1}}(\gamma_{n-1}) \cdots D_{q_1}(\gamma_1) x + \delta p(t) \Phi_{\beta}(x) = 0$$
(4.126)

with  $D_f(\gamma)x = \frac{d}{dt}(f(t)\Phi_{\gamma}(x))$ . As a special case of (4.126), we get the (above discussed) equation

$$(r(t)\Phi_{\alpha}(x'))' = p(t)\Phi_{\beta}(x).$$
 (4.127)

We point out that equations of this form (but this holds also for some higher order equations or for second order systems), are often studied in separate settings depending on whether the integral  $\int_a^{\infty} r^{-1/\alpha}(s) \, ds$  diverges or not. The results of this subsection are in such a form that this distinction is not necessary and all possible combinations which may occur in second order systems or higher order equations are included in our setting.

In Subsection 5.1.2 we indicate another application of Theorem 4.14, it concerns a scalar second order equation with generalized  $\Phi$ -Laplacian.

Chapter **J**\_\_\_\_\_

# Some other nonlinear differential equations

## **5.1** Equations with a general Φ-Laplacian

As a typical prototype of the objects from the title of this section, let us consider the second order equation

$$(r(t)G(y'))' = p(t)F(y), (5.1)$$

where r > 0, p are continuous functions on  $[a, \infty)$ , *G* is an increasing odd homeomorphism defined on an open interval  $(-\rho, \rho)$ ,  $0 < \rho \le \infty$ , and  $\text{Im } G = (-\sigma, \sigma)$ ,  $0 < \sigma \le \infty$ , *F* is a real continuous function on  $\mathbb{R}$  such that uF(u) > 0 for  $u \neq 0$ .

In fact, Emden-Fowler type equations (which are studied in the previous chapter) are sometimes in so general form that include also equation (5.1). For instance, the general setting of Subsection 4.3.8 allows us to apply (after a modification) Theorem 4.14 to equation (5.1). This fact will be illustrated in Subsection 5.1.2, where we examine certain boundary value problem involving equation (5.1).

Note that typically — in considerations within the theory in the framework of regular variation — it is assumed that  $G^{-1}$  or a generalized inverse of *G* is regularly varying. Examples satisfying these assumptions are the classical *p*-Laplacian

$$G(u) = \Phi_{\alpha}(u) = |u|^{\alpha} \operatorname{sgn} u \in \mathcal{RV}(\alpha) \cap \mathcal{RV}_0(\alpha)$$

(even this function is trivially regularly varying), or

$$G(u) = \Phi_{\alpha}(u) \ln |u| \in \mathcal{RV}(\alpha) \cap \mathcal{RV}_0(\alpha),$$

or

$$G(u) = u^{\delta}(A + Bu^{\beta})^{\gamma} \in \mathcal{RV}(\delta + \beta\gamma) \cap \mathcal{RV}_{0}(\delta)$$

if  $\delta, \beta, \gamma > 0$ . Other, more special and typical, prototypes are  $G(u) = \Phi_C(u)$  or  $G(u) = \Phi_R(u)$ , where

$$\Phi_C(u) = \frac{u}{\sqrt{1+u^2}} \text{ and } \Phi_R(u) = \frac{u}{\sqrt{1-u^2}},$$

these operators arise in studying radially symmetric solutions of partial differential equations with the mean curvature operator and the relativity operator, respectively. Note that  $\Phi_C^{-1} = \Phi_R$ ,  $\Phi_R^{-1} = \Phi_C$ , and  $\Phi_C$ ,  $\Phi_R \in \mathcal{RV}_0(1)$ .

Recall that the inverse of an increasing regularly varying function of index  $\vartheta > 0$  is in  $\mathcal{RV}(1/\vartheta)$ . A similar statement can be proved for functions in  $\mathcal{RV}_0(\vartheta)$ .

In some considerations, thanks to the theorem about asymptotic inversions of  $\mathcal{RV}$  functions, the assumption on the monotonicity of *G* can be omitted; one can then work, for instance, with the generalized inverse instead of the inverse.

In working with some forms of *G*, like, for instance,  $G(x) = x \ln |x|$ , the Lambert *W* function may play an important role. The inverse of *G* is in this case  $G^{-1}(x) = e^{W(x)}$ . Note that *W* cannot be expressed in terms of elementary functions.

#### 5.1.1 A boundary value problem on a half-line

The results of this subsection are based on the paper [27] by Došlá, Marini, and Matucci. We are interested in solving the BVP on the whole half-line associated to (5.1), r(t) > 0, p(t) being defined for  $t \in [0, \infty)$ , especially when the weight p changes its sign, that is, if there exist  $t_1, t_2 \ge 0$  satisfying  $p(t_1)p(t_2) < 0$ . In addition to the hypotheses at (5.1), it is assumed that F is nondecreasing and  $\liminf_{t\to\infty} r(t) > 0$ . The boundary conditions read as

$$y(0) = c > 0, \ y(t) > 0, \ 0 < \lim_{t \to \infty} y(t) < \infty, \ \lim_{t \to \infty} y'(t) = 0.$$
(5.2)

Let  $p_+, p_-$ , denote respectively the positive and the negative part of p. Clearly,  $p(t) = p_+(t) - p_-(t)$ .

All the cases when

the inverse 
$$G^{-1}$$
 of G is  $\mathcal{RV}_0$  or  $\mathcal{RPV}_0$  or  $\mathcal{SV}_0$ 

are discussed in [27]. For illustration, we present one selected result and the shortened proof. A crucial role in all the proofs is played by the Schauder-Tychonoff fixed point theorem. Very helpful are also various facts which are consequences of simple properties of  $\mathcal{RV}$  and  $\mathcal{RPV}$  functions like, for instance,  $g(u) \leq Mu^{\alpha-\varepsilon}$ ,  $0 < \varepsilon < \alpha$ , or  $g(\lambda u) \leq M\lambda^{\alpha}g(u), \lambda \in (0,1]$ , on (0,T] for some M > 0, provided  $g \in \mathcal{RV}_0(\alpha), \alpha > 0$ , etc.

**Theorem 5.1.** Let  $G^{-1} \in \mathcal{RV}_0(\alpha)$  with  $\alpha > 0$ , and assume that

$$\lim_{u \to 0+} \frac{F(u)}{u^{1/\beta}} = 0,$$
(5.3)

and for some  $\varepsilon \in (0, \alpha - \beta)$ ,

$$\int_{0}^{\infty} \left(\frac{1}{r(s)} \int_{s}^{\infty} p_{+}(\tau) \, \mathrm{d}\tau\right)^{\alpha-\varepsilon} \mathrm{d}s < \infty,$$

$$\int_{0}^{\infty} \left(\frac{1}{r(s)} \int_{s}^{\infty} p_{-}(\tau) \, \mathrm{d}\tau\right)^{\alpha-\varepsilon} \mathrm{d}s < \infty.$$
(5.4)

Then the boundary value problem (5.1), (5.2) is solvable for any small positive c. Moreover, every solution is of bounded variation on  $[0, \infty)$ .

*Proof.* Choose  $\mu > 0$  such that

$$T_{\mu} = \mu \max\left\{ \max_{t \ge 0} \left( \frac{1}{r(t)} \int_{t}^{\infty} p_{+}(s) \, \mathrm{d}s \right), \min_{t \ge 0} \left( \frac{1}{r(t)} \int_{t}^{\infty} p_{-}(s) \, \mathrm{d}s \right) \right\} < \sigma.$$

Since  $G^{-1} \in \mathcal{RV}_0(\alpha)$ , having a fixed  $\varepsilon < \alpha - \beta$ , a positive constant M exists such that  $G^{-1}(u) \le Mu^{\alpha-\varepsilon}$  for  $0 < u \le T_{\mu}$ . Choose c > 0 sufficiently small such that  $F(2c) \le \mu$ . Let  $\Omega$  be the subset of the Fréchet space  $C[0, \infty)$  given by

$$\Omega = \left\{ u \in C[0,\infty) : \frac{c}{2} \le u(t) \le 2c \right\}$$

and define in  $\Omega$  the operator  $\mathcal{T}$  as follows

$$\mathcal{T}(u)(t) = c + \int_0^t G^{-1}\left(\frac{1}{r(s)}\left(\int_s^\infty p_-(\tau)F(u(\tau))\,\mathrm{d}\tau - \int_s^\infty p_+(\tau)F(u(\tau))\,\mathrm{d}\tau\right)\right).$$

By the previous considerations,  $\mathcal{T}$  is well defined. It can be shown that  $\mathcal{T}(\Omega) \subset \Omega$ ,  $\mathcal{T}(\Omega)$  is relative compact, and  $\mathcal{T}$  is continuous. Thus we can apply the Schauder-Tychonoff fixed point theorem, which guarantees the existence of  $y \in \Omega$  such that  $y(t) = \mathcal{T}(y)(t)$ , i.e. y is a solutions of (5.1). Clearly,

$$y'(t) = G^{-1}\left(\frac{1}{r(t)}\left(\int_t^{\infty} p_{-}(s)F(y(s))\,\mathrm{d}s - \int_t^{\infty} p_{+}(s)F(y(s))\,\mathrm{d}s\right)\right)$$

and so

$$-G^{-1}\left(\frac{F(2c)}{r(t)}\int_{t}^{\infty}p_{+}(s)\,\mathrm{d}s\right) \le y'(t) \le G^{-1}\left(\frac{F(2c)}{r(t)}\int_{t}^{\infty}p_{-}(s)\,\mathrm{d}s\right).$$

Since 1/r(t) is bounded as  $t \to \infty$ , we have  $\lim_{t\to\infty} y'(t) = 0$  and, from (5.4),  $y' \in L^1[0,\infty)$ . Thus, y is of bounded variation on  $[0,\infty)$  and the limit  $\lim_{t\to\infty} y(t)$  is finite. Since y belongs to  $\Omega$ , the assertion follows.

Note that condition (5.3) is satisfied, for instance, if  $F \in \mathcal{RV}_0(\gamma)$ , where  $\gamma > 1/\beta$ .

#### 5.1.2 Regularly varying solutions

Theorem 4.14 can be modified, for instance, for a system of the form

$$x'_i = a_i(t)G_i(b_i(t), x_{i+1}),$$

i = 1, ..., n, where  $x_{n+1}$  means  $x_1, a_i, b_i$  are regularly varying, and  $G_i$  are regularly varying with respect to both variables. Such a modification enables us to include equations like (5.1), where r, p are positive continuous functions on  $[a, \infty)$  with  $r \in \mathcal{RV}(\sigma), p \in \mathcal{RV}(\varrho)$ , and F, G are continuous functions on  $\mathbb{R}$  with uF(u) > 0, uG(u) > 0for  $u \neq 0$ ,  $|G(| \cdot |)| \in \mathcal{RV}_0(\alpha)$ ,  $|F(| \cdot |)| \in \mathcal{RV}_0(\beta)$ ,  $\alpha, \beta \in (0, \infty)$ , G being increasing in a neighborhood of zero. In fact, the theorem about asymptotic inversions of  $\mathcal{RV}$ allows us to omit the assumption on the monotonicity of G, but for simplicity we assume it. Equation (5.1) can be written as

$$\begin{cases} x'_1 = p(t)F(x_2), \\ x'_2 = G^{-1}\left(\frac{1}{r(t)}x_1\right) \end{cases}$$

where  $G^{-1}$  stand for the inverse of *G*; it holds  $G^{-1} \in \mathcal{RV}_0(1/\alpha)$ . Assume that

$$L_F(ug(u)) \sim L_F(u), \quad L_{G^{-1}}(ug(u)) \sim L_{G^{-1}}(u) \text{ as } u \to 0+$$
 (5.5)

for every  $g \in SV_0$ . The subhomogeneity condition reads as  $\alpha > \beta$ . If

$$\varrho + 1 < \min\left\{\sigma - \alpha, \frac{\beta}{\alpha}(\sigma - \alpha)\right\},$$

then (5.1) possesses an eventually positive decreasing solution x such that

$$\lim_{t \to \infty} x(t) = \lim_{t \to \infty} r(t)G(x'(t)) = 0,$$
(5.6)

 $x \in \mathcal{RV}(v)$ , and

$$x^{\alpha-\beta}(t) \sim \frac{1}{-(\varrho+1+\beta\nu)(-\nu)^{\alpha}} \cdot \frac{L^{\alpha}_{G^{-1}}(t^{\varrho+1+\beta\nu-\sigma})L_F(t^{\nu})L_p(t)}{L_r(t)} \cdot t^{\nu(\alpha-\beta)}$$
(5.7)

as  $t \to \infty$ , where  $v = (\alpha - \sigma + \rho + 1)/(\alpha - \beta)$ . Note that also x'(t) tends to zero as  $t \to \infty$ . For completeness we add that if, in addition,  $F = \Phi_{\beta}$  and  $G = \Phi_{\alpha}$ , then every solution x of (5.1) with (5.6) then satisfies  $x \in \mathcal{RV}(v)$  and (5.7). We already know that other examples of G, different from the classical p-Laplacian case, are  $G_C(u)$  and  $G_R(u)$ , the *curvature operator* and the *relativity operator*, respectively, both are in  $\mathcal{RV}_0(1)$ , and are mutually inverse. Note that condition (5.5) is clearly fulfilled for  $G_C, G_R$ .

### 5.2 Partial differential equations

#### 5.2.1 Radial $\mathcal{RV}$ solutions of partial differential systems

It is clear that many of the results in this text (that are established for ordinary differential equations) give useful information about asymptotic form of radial solutions to associated partial differential equations. For example, let us show how Theorem 4.11 can be applied to the partial differential system

in an exterior domain in  $\mathbb{R}^N$ ,  $N \ge 2$ , where  $\bar{\varphi}(t) = t^{\bar{\sigma}} L_{\bar{\varphi}}(t) \in \mathcal{RV}(\bar{\sigma})$ ,  $\bar{\psi}(t) = t^{\bar{\varrho}} L_{\bar{\psi}}(t) \in \mathcal{RV}(\bar{\varrho})$ . If we assume

$$\bar{\nu} := \Lambda(\beta(\alpha + 1 + \bar{\sigma}) + \lambda(\beta + 1 + \bar{\varrho})) < 0,$$
  
$$\bar{\omega} := \Lambda(\alpha(\beta + 1 + \bar{\varrho}) + \mu(\alpha + 1 + \bar{\sigma})) < 0,$$
  
$$N < \min\{1 - (\bar{\nu} - 1)\alpha, 1 - (\bar{\omega} - 1)\beta\},$$

then the existence of a positive (strongly) decreasing radial solution of (5.8) is guaranteed, and any such a solution (u, v) satisfies

$$\begin{split} &\lim_{||z||\to\infty}\frac{u(z)}{|z|^{\bar{v}}L_{\bar{\varphi}}^{\beta\Lambda}(||z||)L_{\bar{\psi}}^{\lambda\Lambda}(||z||)} = \left(\bar{K}_1\bar{K}_2^{\frac{\lambda}{\alpha}}\right)^{\alpha\beta\Lambda},\\ &\lim_{||z||\to\infty}\frac{v(z)}{|z|^{\bar{\omega}}L_{\bar{\varphi}}^{\mu\Lambda}(||z||)L_{\bar{\psi}}^{\alpha\Lambda}(||z||)} = \left(\bar{K}_2\bar{K}_1^{\frac{\mu}{\beta}}\right)^{\alpha\beta\Lambda}, \end{split}$$

where  $\bar{K}_1 = -\left(\bar{v}\left((1-\bar{v})\alpha - N+1\right)^{\frac{1}{\alpha}}\right)^{-1}$ ,  $\bar{K}_2 = -\left(\bar{\omega}\left((1-\bar{\omega})\beta - N+1\right)^{\frac{1}{\beta}}\right)^{-1}$ . Indeed, a radial function (u(z), v(z)) is a solution of (5.8) in  $\Sigma_a = \{z \in \mathbb{R}^N : ||z|| \ge a\}$  if and only if (x(t), y(t)), t = ||z||, given by (x(||z||), y(||z||)) = (u(z), v(z)), satisfies the ordinary differential system

$$\begin{cases} \left(t^{N-1}\Phi_{\alpha}(x')\right)' = t^{N-1}\bar{\varphi}(t)\Phi_{\lambda}(y), \\ \left(t^{N-1}\Phi_{\beta}(y')\right)' = t^{N-1}\bar{\psi}(t)\Phi_{\mu}(x), \end{cases}$$
(5.9)

 $t \ge a$ . System (5.9) has the same structure as (4.40), with  $p(t) = q(t) = t^{N-1}$ ,  $\varphi(t) = t^{N-1}\overline{\varphi}(t), \psi(t) = t^{N-1}\overline{\psi}(t)$ . Thus  $\gamma = \delta = N - 1, L_p(t) = L_q(t) = 1, \sigma = N - 1 + \overline{\sigma}$ ,  $\varrho = N - 1 + \overline{\varrho}, L_{\varphi}(t) = L_{\overline{\varphi}}(t), L_{\psi}(t) = L_{\overline{\psi}}(t)$ . Applying Theorem 4.11 to system (5.9) and going back to the original variables we get the result.

Note that if Theorem 4.11 was obtained just for system (4.40) with p(t) = q(t) = 1, then, after a suitable transformation of the independent variable, only partial systems of the form (5.8) satisfying the conditions  $\alpha = \beta$  and  $\alpha + 1 > N$  would be detectable. Arbitrariness of p, q and of the convergence or divergence of the

integrals in (4.41) enables us to omit these restrictions. Moreover, our general setting allows us to consider even more general partial differential systems where the leading coefficients are formed by elliptic matrices of certain special forms.

Similarly we can apply Theorem 4.14 and Theorem 4.15 when studying positive radial solutions to partial differential systems of the form

$$\begin{cases} \operatorname{div}(\|\nabla u_1\|^{\lambda_1 - 1} \nabla u_1) &= \varphi_1(\|z\|) G(u_2), \\ \operatorname{div}(\|\nabla u_2\|^{\lambda_2 - 1} \nabla u_2) &= \varphi_2(\|z\|) G(u_3), \\ &\vdots \\ \operatorname{div}(\|\nabla u_k\|^{\lambda_k - 1} \nabla u_k) &= \varphi_k(\|z\|) G(u_1). \end{cases}$$
(5.10)

For some other information concerning related partial differential systems see e.g. [22, 24] and the references in [24, 127, 162]. Recall that systems similar to (5.8) are in the literature sometimes called as of *Lane-Emden type*. For information about applications of such systems see e.g. [169].

#### 5.2.2 Almost radial symmetry

In this subsection we give another demonstration how  $\mathcal{RV}$  functions can appear in the qualitative theory of differential equations. This time we consider the partial differential equation

$$\Delta u = \varphi(|x|)u^{\lambda}, \quad x \in \mathbb{R}^n \text{ and } |x| \text{ large}, \tag{5.11}$$

where  $n \ge 3$  is an integer,  $\lambda \in (1, \infty)$  and, for some  $t_0 > 0$ ,  $\varphi : [t_0, \infty) \to (0, \infty)$  is a continuous function.

We present one commented result from Taliaferro [159] without a proof, and the interesting fact is to observe how the subject of regular variation is related to the conditions on  $\varphi$  in the theorem. It is worthy of note — as claimed by Taliaferro in the paper — that is was Omey who pointed him out this fact.

We are interested in whether all positive solutions of (5.11) are almost radial when |x| is large, i.e.,

$$\frac{u(x)}{\bar{u}(|x|)} \to 1 \quad \text{as } |x| \to \infty,$$

where  $\bar{u}(t)$  is the average of *u* on the sphere |x| = t. Assume that

$$\int_{t_0}^{\infty} s^{\alpha} \varphi(s) \, \mathrm{d}s < \infty \tag{5.12}$$

for some  $\alpha > 1$  and note also other (mutually exclusive) cases are discussed in [159].

**Theorem 5.2.** Suppose that u(x) is a positive solution of (5.11). Assume that

$$\frac{\varphi(t)}{\psi(t)} \to 1 \quad \text{as } t \to \infty \tag{5.13}$$

for some continuously differentiable function  $\psi : [t_0, \infty) \to (0, \infty)$  such that

$$t^{2n-2}\psi(t)$$
 is monotone on  $[t_0,\infty)$ , (5.14)

and

$$\lim_{t \to \infty} \left(\frac{1}{\sqrt{\psi(t)}}\right)' \int_t^\infty \sqrt{\psi(s)} ds = M \text{ for some } M \in \mathbb{R} \cup \{\pm\}.$$
 (5.15)

Then

$$\lim_{|x| \to \infty} \frac{u(x)}{\bar{u}(|c|)} = 1 \text{ and } \lim_{|x| \to \infty} [u(x) - \bar{u}(|x|)] = 0.$$

*Moreover,*  $1 \le M \le (\alpha + 1)/(\alpha - 1)$  *and either* 

$$\lim_{|x| \to \infty} u(x) \left( \frac{\lambda - 2}{2} \int_{|x|}^{\infty} \sqrt{\psi(s)} \, \mathrm{d}s \right)^{\frac{2}{\lambda - 1}} = \left( 1 + \frac{(n - 2)(\lambda - 1)(M - 1)}{2} \right)^{\frac{1}{\lambda - 1}},$$

or

$$u(x) = c + o(1) \quad as \ |x| \to \infty,$$

or

$$u(x) = \frac{c + o(1)}{|x|^{n-2}} \quad as \ |x| \to \infty,$$

where c is some positive constant.

**Remark 5.1.** Let  $\psi : [t_0, \infty) \to (0, \infty)$  be a continuously differentiable function. Using methods of the subject of regular variation one can easily show that  $\psi$  satisfies both  $\int_{t_0}^{\infty} \sqrt{\psi(s)} \, ds < \infty$  and (5.15) with M = 1 if and only if

$$\psi(t) = \psi(t_0) \exp\left\{-2 \int_{t_0}^t \frac{1 + h'(s)}{h(s)} \, \mathrm{d}s\right\}$$

for  $t \ge t_0$  for some continuously differentiable function  $h : [t_0, \infty) \to (0, \infty)$  such that  $h'(t) \to 0$  as  $t \to \infty$ . In this case

$$\int_{t}^{\infty} \sqrt{\psi(s)} \, \mathrm{d}s = \left( \int_{t_0}^{\infty} \sqrt{\psi(s)} \, \mathrm{d}s \right) \exp\left\{ - \int_{t_0}^{t} \frac{1}{h(s)} \, \mathrm{d}s \right)$$

for  $t \ge t_0$  and for each  $\beta \in \mathbb{R}$  there exists  $t_1 > t_0$  such that  $\psi(t)t^\beta$  is strictly decreasing on  $[t_1, \infty)$ . Therefore, since there are functions h(t) as above which tend arbitrarily fast to zero as  $t \to \infty$ , we see that Theorem 5.2 allows  $\psi(t)$  to tend arbitrarily fast to zero as  $t \to \infty$  and allows solutions of (5.11) to tend arbitrarily fast to  $\infty$  as  $|x| \to \infty$ . However conditions (5.13), (5.14), and (5.15) do require that  $\psi(t)$  does not oscillate too much as  $t \to \infty$ . **Remark 5.2.** Let  $\varphi : [t_0, \infty) \to (0, \infty)$  be a continuous function such that

$$\int_{t_0}^{\infty} \sqrt{\psi(s)} \, \mathrm{d}s < \infty.$$

(As pointed out in the previous remark, this will be the case if (5.12) holds.) Then there exists a continuously differentiable function  $\psi : [t_0, \infty) \rightarrow (0, \infty)$  satisfying both (5.13) and (5.15) for some  $M \in (1, \infty)$  if and only if  $\varphi \in \mathcal{RV}(-2\beta)$  for some  $\beta \in (1, \infty)$ . In this case  $\beta M = \beta + M$ 

# 5.3 *RV* and *RPV* solutions for a class of third order nonlinear differential equations

The results of this section are based on the paper [65] by Jaroš, Kusano, and Marić. Let us consider the third order differential equation of the form

$$x''' + 2p(t)x' + p'(t)x = F(t, x),$$
(5.16)

where *p* is continuously differentiable on  $[a, \infty)$ , *F* is continuous on  $[a, \infty) \times \mathbb{R}$  and

$$|F(t,u)| \le G(t,u)|,$$

where  $G : [a, \infty) \times \mathbb{R} \to [0, \infty)$  is a continuous function which is nondecreasing in the second variable for  $t \ge a$ . Along with (5.16) consider the self-adjoint linear equation

$$y''' + 2p(t)y' + p'(t)y = 0.$$
 (5.17)

With the help of the results for linear second order equations (Theorems 2.1, 2.2, 2.3, 2.4), we indicate the situation in which (5.17) has a fundamental set of solutions consisting of regularly and rapidly varying functions. We then establish the conditions under which (5.16) possesses a set of solutions which are asymptotic as  $t \to \infty$  to the indicated *RV* and *RPV* solutions of (5.16).

#### 5.3.1 The self-adjoint equation

**Theorem 5.3.** Let *C* be a constant such that C < 1/4 and let  $\varrho, \sigma$ , with  $\varrho < \sigma$ , denote the real roots of  $\lambda^2 - \lambda + C = 0$ . If *p* is integrable on  $[a, \infty)$  and satisfies

$$\lim_{t \to \infty} t \int_t^\infty p(s) \, \mathrm{d}s = 2C, \tag{5.18}$$

then (5.17) has a fundamental set of  $\mathcal{RV}$  (hence nonoscillatory) solutions  $y_i(t)$ , i = 1, 2, 3, of the form

$$y_1(t) = t^{2\varrho} L_1(t), \ y_2(t) = t L_2(t), \ y_3(t) = t^{2\sigma} L_3(t),$$
 (5.19)

where  $L_1 \in SV$ ,  $L_2(t) \to 1/(1-2\varrho)$ , and  $L_3(t) \sim 1/((1-2\varrho)^2 L_1(t))$  as  $t \to \infty$ .

*Proof.* By Theorems 2.2 and 2.3, condition (5.18) is necessary and sufficient for the equation

$$z'' + \frac{1}{2}p(t)z = 0 \tag{5.20}$$

to possess two linearly independent  $\mathcal{RV}$  solutions of the form  $u(t) = t^{\varrho}L_1(t)$  and  $v(t) = t^{\varrho}L_2(t)$ , where  $L_1 \in \mathcal{SV}$  and  $L_2(t) \sim 1/((1 - 2\varrho)^2 L_1(t))$  as  $t \to \infty$ . It is known that if  $\{u, v\}$  is a fundamental set of solutions of (5.20), then  $\{y_1, y_2, y_3\}$ , where  $y_1 = u^2$ ,  $y_2 = uv$ ,  $y_3 = v^2$ , is a fundamental set of solutions of (5.17). Simple application of  $\mathcal{RV}$  functions now gives the result.

The borderline case (C = 1/4) between oscillation and nonoscillation of (5.20) is treated in the next theorem.

**Theorem 5.4.** Suppose that

$$\lim_{t\to\infty}\int_t^\infty p(s)\,\mathrm{d}s=\frac{1}{2}.$$

*Suppose furthermore that the function* 

$$\phi(t) = t \int_t^\infty p(s) \, \mathrm{d}s - \frac{1}{2}$$

satisfies

$$\int^{\infty} \frac{|\phi(t)|}{t} \, \mathrm{d}t < \infty$$

and

$$\int_{t}^{\infty} \frac{|\psi(t)|}{t} \, \mathrm{d}t < \infty, \quad \text{where } \psi(t) = \int_{t}^{\infty} \frac{|\phi(s)|}{s} \, \mathrm{d}s < \infty.$$

Then equation (5.17) has a fundamental set of  $\mathcal{RV}(1)$  (hence nonoscillatory) solutions  $y_i(t)$ , i = 1, 2, 3, of the form

$$y_1(t) = tL_1(t), \ y_2(t) = t \ln tL_2(t), \ y_3(t) = t \ln^3 tL_3(t),$$

where  $L_1(t) \to k \in (0, \infty)$ ,  $L_2(t) \to 1$ ,  $L_3(t) \to 1/k^2$  as  $t \to \infty$ .

*Proof.* One proceeds exactly as in the proof of the previous theorem, using this time Theorem 2.4.

**Theorem 5.5.** Let p(t) < 0 for  $t \ge a$ . If for each  $\lambda > 1$ 

$$\lim_{t \to \infty} \left( -t \int_t^{\lambda t} p(s) \, \mathrm{d}s \right) = \infty, \tag{5.21}$$

then (5.17) has at least two RPV solutions such that the first of these solutions decreases and is of the class  $RPV(-\infty)$  whereas the second one increases and is of the class  $RPV(\infty)$ .

*Proof.* From Theorem 2.1, (5.20) has solutions  $u \in \mathcal{RPV}(-\infty)$  and  $v \in \mathcal{RPV}(\infty)$  if and only if (5.21) holds. It is clear from the definition of  $\mathcal{RPV}$  functions that  $y_1 = u^2 \in \mathcal{RPV}(-\infty)$  and  $y_3 = v^2 \in \mathcal{RPV}(\infty)$ .

Note that the third linearly independent solutions  $y_2 = uv$  need not to be  $\mathcal{RPV}$  at all. This is shown by the example:  $u(t) = e^{-t}$ ,  $v(t) = e^t$ , so that  $y_2(t) = 1$ .

#### 5.3.2 RV and RPV solutions of the perturbed equation

First we establish sufficient conditions for (5.16) to have solutions  $x_1, x_2, x_3$  with the same asymptotic behavior as the solutions  $y_1 = u^2$ ,  $y_2 = uv$ ,  $y_3 = v^2$  of (5.17), respectively. Then we apply these results to construct  $\mathcal{RV}$  and  $\mathcal{RPV}$  solutions of (5.16).

**Theorem 5.6.** *If for some*  $\alpha > 0$ 

$$\int_{a}^{\infty} v^{2}(s)G(s,\alpha u^{2}(s))\,\mathrm{d}s < \infty, \tag{5.22}$$

then there exists an eventually positive solution  $x_1$  of (5.16) such that  $x_1(t) \sim \alpha u^2(t)/2$  as  $t \to \infty$ .

*Proof.* Choose  $T \ge a$  such that

$$\int_T^\infty v^2(s)G(s,\alpha u^2(s))\,\mathrm{d} s\leq \frac{\alpha}{2},$$

which is possible by (5.22). Define the set  $X_1$  by

$$\mathcal{X}_1=\{y\in C[T,\infty): 0\leq x(t)\leq \alpha u^2(t),t\geq T\}$$

and the integral operator  $\mathcal{F}_1$  by

$$\mathcal{F}_1 x(t) = \frac{\alpha}{2} u^2(t) + u^2(t) \int_t^\infty \left( \int_t^s \frac{1}{u^2(s_2)} \int_t^{s_2} \frac{1}{u^2(s_1)} \, \mathrm{d}s_1 \, \mathrm{d}s_2 \right) u^2(s) F(s, x(s)) \, \mathrm{d}s,$$

 $t \ge T$ . It can be shown that all the hypotheses of the Schauder-Tychonoff fixed point theorem are fulfilled, and thus it ensures the existence of  $x_1 \in X_1$  such that  $x_1 = \mathcal{F}_1 x_1$ . So  $x_1$  is also a solution of (5.16). The asymptotics of  $x_1$  is an immediate consequence of the integral equation  $x_1 = \mathcal{F}_1 x_1$ .

The proofs of the next two theorems are also based on the Schauder-Tychonoff fixed point theorem; we omit details.

**Theorem 5.7.** *If for some*  $\beta > 0$ 

$$\int_{a}^{\infty} u(s)v(s)G(s,\beta u(s)v(s)) \,\mathrm{d}s < \infty, \tag{5.23}$$

then there exists an eventually positive solution  $x_2$  of (5.16) such that  $x_2(t) \sim \beta u(t)v(t)/2$ as  $t \to \infty$ .

**Theorem 5.8.** *If for some*  $\gamma > 0$ 

$$\int_{a}^{\infty} u^{2}(s) G\left(s, \gamma v^{2}(s)\right) \, \mathrm{d}s < \infty, \tag{5.24}$$

then there exists an eventually positive solution  $x_3$  of (5.16) such that  $x_3(t) \sim \gamma v^2(t)/2$  as  $t \to \infty$ .

If *p* satisfies condition (5.18), then the existence of a fundamental set of  $\mathcal{RV}$  solutions to (5.17) given by (5.19) is guaranteed. Then, of conditions (5.22), (5.23) and (5.24) hold, Theorems 5.6, 5.7, 5.8, respectively, guarantee the existence of a fundamental set of solutions  $x_1, x_2, x_3$  of (5.16) such that  $x_i(t) \sim \alpha_i y_i(t)$  as  $t \to \infty$ ,  $\alpha_i > 0$ , i = 1, 2, 3. But this also means that  $x_i$  are in  $\mathcal{RV}$ . As an illustration, take

$$F(t,x) = t^{\omega} M(t) |x|^{\delta} \operatorname{sgn} x, \ \omega \in \mathbb{R}, \ \delta \in (0,\infty), \ M \in \mathcal{SV},$$

where for a suitable choice of  $\omega$ ,  $\delta$  the conditions posed on *G* are fulfilled.

To obtain the existence and asymptotic behavior of  $\mathcal{RPV}$  solutions of the perturbed equation (5.16) with p(t) < 0, one proceeds in the same way as above in the  $\mathcal{RV}$  case.

# 5.4 Classification of convergence rates of solutions to perturbed first order ODE's

The content of this section is based on selected results from Appleby, Patterson [6]. We are interested in a classification of the rates of convergence to a limit of the solutions of the scalar differential equation

$$x'(t) = -f(x(t)) + g(t), \quad t > 0, \quad x(0) = \xi.$$
(5.25)

We assume that the unperturbed equation

$$x'(t) = -f(x(t)), \quad t > 0, \quad x(0) = \zeta$$
(5.26)

has a unique globally stable equilibrium (which we set to be at zero). This is characterized by the condition xf(x) > 0 for  $x \neq 0$ , f(0) = 0. Further we assume that  $f \in C(\mathbb{R}, \mathbb{R})$ ,  $g \in C([0, \infty), \mathbb{R})$ , f is locally Lipschitz continuous on  $\mathbb{R}$ . We suppose that f(x) does not have linear leading order behavior as  $x \to \infty$ ; moreover, we do not ask that f forces solutions of (5.26) to hit zero in finite time. We define

$$F(x) = \int_{x}^{1} \frac{1}{f(u)} du, \ x > 0,$$

and avoiding solutions of equation (5.26) to hitting zero in finite time forces  $\lim_{x\to 0^+} F(x) = \infty$ . We notice that  $F : (0, \infty) \to \mathbb{R}$  is a strictly decreasing function, so it has an inverse  $F^{-1}$ , and we have  $\lim_{t\to\infty} F^{-1}(t) = 0$ . The significance of the functions F and  $F^{-1}$  is that they enable us to determine the rate of convergence of solutions of (5.26) to zero, because  $F(y(t)) - F(\zeta) = t$  for  $t \ge 0$  or  $y(t) = F^{-1}(t+F(\zeta))$  for  $t \ge 0$ . It is then of interest to ask whether solutions of (5.25) will still converge to zero as  $t \to \infty$ , and how this convergence rate modifies according to the asymptotic behavior of g. In order to do this with reasonable generality we find it convenient and natural to assume at various points that the functions f and g are regularly varying.

The main result of [6], which characterizes the rate of convergence of solutions of (5.25) to zero, can be summarized as follows: Suppose that  $f \in \mathcal{RV}_0(\beta)$ ,  $\beta > 1$ , and that *g* is positive and regularly varying at infinity, in such a manner that

$$\lim_{t \to \infty} \frac{g(t)}{(f \circ F^{-1})(t)} =: M \in [0, \infty]$$

exists. If M = 0, the solution of (5.25) inherits the rate of decay to zero of y, in the sense that

$$\lim_{t\to\infty}\frac{F(x(t))}{t}=1.$$

If  $M \in (0, \infty)$  we can show that the rate of decay to zero is slightly slower, obeying

$$\lim_{t \to \infty} \frac{F(x(t))}{t} = \Lambda = \Lambda(L) \in (0, 1)$$

and a formula for  $\Lambda$  purely in terms of *M* and  $\beta$  can be found. Finally, in the case that  $M = \infty$  can be shown that

$$\lim_{t\to\infty}\frac{F(x(t))}{t}=0$$

If it is presumed that *g* is regularly varying at infinity with negative index, or *g* is slowly varying and is asymptotic to a decreasing function, then the exact rate of convergence can be found, namely that  $\lim_{t\to\infty} f(x(t))/g(t) = 1$ . These asymptotic results are proven by constructing appropriate upper and lower solutions to the differential equation (5.25) as in [3].

In order to simplify the analysis, we assume that g(t) > 0, t > 0;  $x(0) = \xi > 0$ . Note that the results can rapidly be extended in the case g(t) < 0 and  $\xi < 0$ .

We will not give the proofs of all the results presented here; note that some of them are rather technical. We prefer to present — for illustration — just a sketch of one selected proof. However, we give several comments.

The first result says that the global convergence of solutions of (5.25), as well as the rate of convergence of solutions to 0 is preserved provided the perturbation g decays sufficiently rapidly. In order to guarantee this, we request only that f be asymptotic to a monotone function close to zero.

**Theorem 5.9** (Appleby, Patterson [6]). Let there exist  $\phi$  such that

$$\lim_{t \to 0+} \frac{f(t)}{\phi(t)} = 1, \ \phi \ is \ increasing \ on \ (0, \delta).$$
(5.27)

If

$$\lim_{t \to \infty} \frac{g(t)}{(f \circ F^{-1})(t)} = 0,$$
(5.28)

then the unique continuous solution of (5.25) obeys

$$\lim_{t \to \infty} \frac{F(x(t))}{t} = 1.$$
 (5.29)

Immediately this theorem presents a question: Is it possible to find slower rates of decay of  $g(t) \rightarrow 0$  as  $t \rightarrow \infty$  than exhibited in (5.28), for which the solution xof (5.25) still decays at the rate of the unperturbed equation, as characterized by (5.29)? In some sense, the next theorem says that the rate of decay of g in (5.28) cannot be relaxed, at least for functions f which are regularly varying at zero with index  $\beta > 1$ , or which are rapidly varying at zero.

In the case when f is regularly varying at 0 with index 1 (and  $f(x)/x \rightarrow 0$  as  $x \rightarrow 0$ ), the condition (5.28) is not necessary in order to preserve the rate of decay embodied by (5.29). A more careful analysis is needed to characterize the asymptotic behavior of solutions of (5.25).

**Theorem 5.10** (Appleby, Patterson [6]). Let *x* be the unique continuous solution of (5.25). Suppose that there exists  $\phi$  such that (2.24) holds, and suppose further that there exists M > 0 such that

$$\lim_{t \to \infty} \frac{g(t)}{(f \circ F^{-1})(t)} = M.$$
(5.30)

Then  $x(t) \to 0$  as  $t \to \infty$ .

(*i*) If  $f \in \mathcal{RV}(\beta)$  for  $\beta > 1$ , then

$$\lim_{t \to \infty} \frac{F(x(t))}{t} = \Lambda_*(M) \in (0, 1),$$
(5.31)

where  $\Lambda_*$  is the unique solution of  $(1 - \Lambda_*)\Lambda_*^{-\beta/(\beta-1)} = M$ . (*ii*) If  $f \circ F^{-1} \in \mathcal{RV}(-1)$ , then

$$\lim_{t\to\infty}\frac{F(x(t))}{t}=\Lambda_*(M)\in(0,1),$$

where  $\Lambda_*$  is the unique solution of  $(1 - \Lambda_*)/\Lambda_* = M$ .

If *y* is the solution of (5.26), we have  $y(t)/F^{-1}(t) \to 1$  as  $t \to \infty$ . Moreover, in the case when  $\beta > 1$ , as  $F^{-1} \in \mathcal{RV}(-1/(\beta - 1))$ , we have

$$\lim_{t \to \infty} \frac{x(t)}{y(t)} = \lim_{t \to \infty} \frac{x(t)}{F^{-1}(t)} \lim_{t \to \infty} \frac{F^{-1}(\Lambda_* t)}{F^{-1}(t)} = \Lambda_*^{-\beta/(\beta-1)} > 1.$$

Therefore, the solution of (5.25) is of the same order as the solution of (5.26), but decays more slowly by a factor depending on *M*. In the second case, when  $F^{-1} \in SV$ , we have

$$\lim_{t \to \infty} \frac{x(t)}{y(t)} = \lim_{t \to \infty} \frac{x(t)}{F^{-1}(t)} \lim_{t \to \infty} \frac{F^{-1}(\Lambda_* t)}{F^{-1}(t)} = 1$$

so once again the solution of (5.25) is of the same order as the solution of (5.26)

There is a greater alignment of the hypotheses that appears at a first glance. When  $f \in \mathcal{RV}_0(\beta)$  for  $\beta > 1$ , it follows that  $F \in \mathcal{RV}_0(1 - \beta)$  and therefore that  $F^{-1} \in \mathcal{RV}(-1/(\beta-1))$  and  $f \circ F^{-1} \in \mathcal{RV}(-\beta/(\beta-1))$ . Hence we see that the hypothesis of part (ii) are in some sense the limit of those in part (i) when  $\beta \to \infty$ . This suggests that part (ii) of the theorem applies in the case when f is a rapidly varying function at 0, and the solutions of the unperturbed differential equation are slowly varying at infinity. Moreover, the solution of the perturbed differential equation should also be slowly varying in this case. If we suppose that  $f \circ F^{-1} \in \mathcal{RV}(-1)$ , then  $F^{-1} \in \mathcal{SV}$ . Therefore, we do not need to assume this second hypothesis in part (ii) of the above theorem. Further note that if  $f \in \mathcal{RPV}_0(\infty)$ , then  $F^{-1} \in \mathcal{SV}$ , see [6].

We notice that viewed as a function of M,  $\Lambda_* : (0, \infty) \to (0, 1)$  is decreasing and continuous with  $\lim_{M\to 0+} \Lambda_*(M) = 1$ ,  $\lim_{M\to\infty} \Lambda_*(M) = 0$ . The first limit demonstrates that the limit in (5.31) is a continuous extension of the limit observed in Theorem 5.9, because the hypothesis (5.28) can be viewed as (5.30) with M = 0, while the resulting limiting behavior of the solution (5.29) can be viewed as (5.31) where  $\Lambda_* = 1$ . The monotonicity of  $\Lambda_*$  in M indicates that the slower the decay rate of the perturbation is (i.e., the greater is M) the slower the rate of decay of the solution of (5.25) is. Since  $\lim_{M\to\infty} \Lambda_*(M) = 0$ , this result also suggests that

$$\lim_{t \to \infty} \frac{g(t)}{(f \circ F^{-1})(t)} = \infty$$
(5.32)

implies

$$\lim_{t \to \infty} \frac{F(x(t))}{t} = 0, \tag{5.33}$$

so that the solution of the perturbed differential equation entirely loses the decay properties of the underlying unperturbed equation when the perturbation gexceeds the critical size indicated by (5.30), and decays more slowly yet. This conjecture is borne out by virtue of the next theorem.

**Theorem 5.11.** Let x be the unique continuous solution of (5.25). Suppose that there exists  $\phi$  such that (2.24) holds, and suppose further that f and g obey (5.32). Suppose finally that  $x(t) \to 0$  as  $\to \infty$ . If  $f \in \mathcal{RV}_0(\beta)$  for  $\beta > 1$  or  $f \circ F^{-1} \in \mathcal{RV}(-1)$ , then the unique solution of (5.25) obeys (5.33).

*Proof.* We give only a sketch of the proof when  $f \circ F^{-1} \in \mathcal{RV}(-1)$  and  $F^{-1} \in \mathcal{SV}$ . The other part is proved similarly. For details see [6]. Since  $f(x)/\phi(x) \to 1$  as  $x \to 0+$  and  $F^{-1}(t) \to 0$  as  $t \to \infty$ , we have

$$\lim_{t \to \infty} \frac{\phi(F^{-1}(t))}{f(F^{-1}(t))} = 1.$$

Hence  $h = \phi \circ F^{-1} \in \mathcal{RV}(-1)$ . Let  $\varepsilon \in (0, 1/2)$ . By (5.32), we have that there exists  $T_1(\varepsilon) > 0$  such that  $h(t) < \varepsilon^2 g(t)$  for  $t \ge T_1(\varepsilon)$ . Also, as  $h \in \mathcal{RV}(-1)$ , we have that  $h(\varepsilon t)/h(t) \to 1/\varepsilon$  as  $t \to \infty$ . Hence there exists  $T_2(\varepsilon) > 0$  such that  $h(\varepsilon t) < 2h(t)/\varepsilon$  for  $t \ge T_2(\varepsilon)$ . Define  $T(\varepsilon) = 1 + \max\{T_1(\varepsilon), T_2(\varepsilon)\}$ . Now define

$$K = \max\left\{2, \frac{2x(T)}{x_1(\varepsilon)}, \frac{x(T)}{F^{-1}(\varepsilon T)}\right\},\,$$

where  $x_1(\varepsilon) > 0$  is such that  $f(x) < (1 + \varepsilon)\phi(x)$  for  $x \le x_1(\varepsilon)$ , and also define

$$x_L(t) = F^{-1}[\varepsilon(t-T) + F(x(T)/K)],$$

 $t \ge T$ . It can be shown that  $x'_L(t) < -f(x_L(t)) + g(t), t \ge T$ ;  $x_L(T) < x(T)$ . Therefore, we have that  $x_L(t) < x(t)$  for  $t \ge T$ . One can then get

$$0 \leq \liminf_{t \to \infty} \frac{F(x(t))}{t} \leq \limsup_{t \to \infty} \frac{F(x(t))}{t} \leq \varepsilon.$$

Letting  $\varepsilon \rightarrow 0+$  gives (5.33).

We observe that the hypothesis that  $x(t) \rightarrow 0$  as  $\rightarrow \infty$  has been appended to the theorem. This is because the slow rate of decay of g may now cause solutions to tend to infinity, if coupled with a hypothesis on f which forces f(x) to tend to zero as  $x \rightarrow \infty$  at a sufficiently rapid rate. We prefer to add this hypothesis, rather than sufficient conditions on f and g which would guarantee  $x(t) \rightarrow \infty$ .

Several further situations are considered in [6]. For instance, g is assumed to be regularly varying at infinity.

### 5.5 Nearly linear differential equations

The results of this section are taken from the paper [149] by Řehák. We consider the nonlinear equation

$$(G(y'))' = p(t)F(y), (5.34)$$

where *p* is a positive continuous function on  $[a, \infty)$  and *F*, *G* are continuous functions on  $\mathbb{R}$  with uF(u) > 0, uG(u) > 0 for  $u \neq 0$ . To simplify our considerations we suppose that *F* and *G* are increasing and odd. Nonlinearities *F* and *G* are further assumed to have regularly varying behavior of index 1 at zero. More precisely, we require

$$F(|\cdot|), G(|\cdot|) \in \mathcal{RV}_0(1);$$
 (5.35)

the class  $\mathcal{RV}_0$  being defined below. This condition justifies the terminology a *nearly linear equation*. Indeed, if we make a trivial choice F = G = id, then (5.34) reduces to a linear equation. It is however clear that in contrast to linear equations, the solution space of (2.1) is generally neither additive nor homogeneous. Examples of F(u) and G(u) which lead to a nonlinear equation and can be treated within this theory are:  $u |\ln |u||$ , or  $u / |\ln |u||$ , or  $u / \sqrt{1 \pm u^2}$ , and many others.

As we could see in Chapter 4, theory of regular variation has been shown very useful in studying asymptotic properties of Emden-Fowler type equations, e.g. of the form  $y'' = q(t)|y|^{\gamma} \operatorname{sgn} y$  or, more generally,

$$y^{\prime\prime} = q(t)\varphi(y), \tag{5.36}$$

where  $|\varphi(|\cdot|)| \in \mathcal{RV}(\gamma)$  or  $|\varphi(|\cdot|)| \in \mathcal{RV}_0(\gamma)$ ,  $\gamma > 0$ . Usually the sub-linearity condition resp. the super-linearity condition is assumed, i.e.,  $\gamma < 1$  resp.  $\gamma > 1$ , and such conditions play an important role in the proofs. Notice that from this point of view, equation (5.34) (which arises as a variant of (5.36) with specific nonlinearities) is neither super-linear nor sub-linear, since the indices of regular variation of *F* and *G* are the same. Therefore, asymptotic analysis of (5.34) in the framework of regular variation requires an approach which is different from the usual ones for the above mentioned Emden-Fowler equation with  $\gamma \neq 1$ . The crucial property is now the fact that the nonlinearities in (5.34) are somehow close to each other (they can differ by a slowly varying function). It turns out that a modification of some methods known from the linear theory is a useful tool. However, as we will see, some phenomena may occur for (5.34) which cannot happen in the linear case.

We are interested in asymptotic behavior of solutions y to (5.34) such that y(t)y'(t) < 0 for large t. Without loss of generality, we restrict our study to eventually positive decreasing solutions of equation (5.34); such a set is denoted as DS. As we will see, for any  $y \in DS$ ,  $\lim_{t\to\infty} y'(t) = 0$ .

We start with the simple result which gives the conditions guaranteeing slow variation of any solution in DS.

#### **Theorem 5.12.** Assume that

$$\lim_{t \to \infty} t \int_{t}^{\infty} p(s) \, \mathrm{d}s = 0, \tag{5.37}$$

$$\limsup_{u \to 0+} L_F(u) < \infty \quad and \quad \liminf_{u \to 0+} L_G(u) > 0. \tag{5.38}$$

Then

$$\emptyset \neq \mathcal{DS} \subset \mathcal{NSV}.$$

*Proof.* Rewrite equation (5.34) as an equivalent system of the form

$$y' = -G^{-1}(u), \quad u' = -p(t)F(y),$$

where  $G^{-1}$  is the inverse of *G*. Then we apply the existence result [20, Theorem 1] to obtain  $\mathcal{DS} \neq \emptyset$ .

Take  $y \in DS$ , i.e., y(t) > 0, y'(t) < 0,  $t \ge t_0$ . Then  $\lim_{t\to\infty} y'(t) = 0$ . Indeed, G(y') is negative increasing and so is y'. If  $\lim_{t\to\infty} y'(t) = -c < 0$ , then  $y(t) - y(t_0) \sim -c(t-t_0)$  as  $t \to \infty$ , which contradicts eventual positivity of y. Integration of (5.34) from t to  $\infty$  yields

$$-G(y'(t)) = \int_t^\infty p(s)F(y(s)) \,\mathrm{d}s.$$

Hence,

$$|y'(t)|L_G(|y'(t)|) = \int_t^\infty p(s)y(s)L_F(y(s)) \, \mathrm{d}s \le y(t) \int_t^\infty p(s)L_F(y(s)) \, \mathrm{d}s.$$
Thus,

$$\frac{-ty'(t)}{y(t)} \le \frac{t}{L_G(|y'(t)|)} \int_t^\infty p(s) L_F(y(s)) \,\mathrm{d}s \le \frac{tM}{N} \int_t^\infty p(s) \mathrm{d}s,\tag{5.39}$$

where *M*, *N* are some positive constants which exist thanks to (5.38). Since the expression on the right hand side of (5.39) tends to zero, it follows that  $ty'(t)/y(t) \rightarrow 0$  as  $t \rightarrow \infty$ , and  $y \in NSV$  follows.

**Remark 5.3.** Condition (5.37) is somehow necessary. Indeed, take  $y \in DS \cap SV$  and assume that  $\liminf_{u\to 0^+} L_F(u) > 0$  and  $\limsup_{u\to 0^+} L_G(u) < \infty$ . First note that because of monotonicity of y', we have  $ty'(t)/y(t) \to 0$ , and so  $y \in NSV$ . Set w = G(y')/y. Then w satisfies

$$w' = p(t)\frac{F(y)}{y} - w\frac{y'}{y}$$
(5.40)

for large *t*. There exists  $N \in (0, \infty)$  such that

$$0 < -tw(t) \le -Nt \frac{y'(t)}{y(t)} \to 0$$

as  $t \to \infty$ . Hence,

$$\int_{0}^{\infty} w(s) \frac{y'(s)}{y(s)} \, \mathrm{d}s < \infty \text{ and } \lim_{t \to \infty} t \int_{t}^{\infty} w(s) \frac{y'(s)}{y(s)} \, \mathrm{d}s = 0.$$

Integration (5.40) from *t* to  $\infty$  and multiplying by *t*, we get

$$-tw(t) = t \int_t^\infty p(s) \frac{F(y(s))}{y(s)} \,\mathrm{d}s - t \int_t^\infty w(s) \frac{y'(s)}{y(s)} \,\mathrm{d}s,$$

which implies  $\lim_{t\to\infty} t \int_t^{\infty} p(s)L_F(y(s)) ds = 0$ . Since  $M \in (0, \infty)$  exists such that  $L_F(y(t)) \ge M$  for large *t*, condition (5.37) follows.

A necessity is discussed from certain point of view also in Remark 5.6.

**Remark 5.4.** Observe that in Theorem 5.12 we are dealing with all SV solutions of (5.34). It follows from the fact that SV solutions cannot increase. Indeed, for a positive increasing solution u of (5.34), due to convexity, we have  $u'(t) \ge K_1$  for some  $K_1 > 0$ . By integrating,  $u(t) \ge K_1t + K_2$ , which contradicts the fact the  $u \in SV$ .

**Remark 5.5.** The statements of Theorem 5.12 and Remark 5.3 can be understood as a nonlinear extension of Theorem 2.1-(i).

In the next result, we derive asymptotic formulae for SV solutions provided p is regularly varying of index –2. Define

$$\hat{F}(x) = \int_{1}^{x} \frac{\mathrm{d}u}{F(u)}, \quad x > 0.$$

The function  $\hat{F}(x)$  is increasing on  $(0, \infty)$ . The constant 1 in the integral is unimportant; it can be replaced by any positive constant. Denote the inverse of  $\hat{F}$  by  $\hat{F}^{-1}$ . We have  $|\hat{F}| \in SV_0$  and in general  $\lim_{u\to 0+} |\hat{F}(u)|$  can be finite or infinite. Denote

$$H(t) = \frac{tp(t)}{L_G(1/t)}$$

and note that  $H \in \mathcal{RV}(-1)$  provided  $p \in \mathcal{RV}(-2)$ .

**Theorem 5.13.** Assume that  $p \in \mathcal{RV}(-2)$ ,  $\lim_{u\to 0^+} |\hat{F}(u)| = \infty$ , and

$$L_G(ug(u)) \sim L_G(u) \quad as \ u \to 0+, \tag{5.41}$$

for all  $g \in SV_0$ . If  $y \in DS \cap SV$ , then  $-y \in \Pi(-ty'(t))$ . Moreover: (i) If  $\int_a^{\infty} H(s) ds = \infty$ , then

$$y(t) = \hat{F}^{-1} \left\{ -\int_{a}^{t} (1 + o(1))H(s) \,\mathrm{d}s \right\}$$
(5.42)

(and  $y(t) \to 0$ ) as  $t \to \infty$ .

(*ii*) If  $\int_{a}^{\infty} H(s) \, \mathrm{d}s < \infty$ , then

$$y(t) = \hat{F}^{-1}\left\{\hat{F}(y(\infty)) + \int_{t}^{\infty} (1+o(1))H(s)\,\mathrm{d}s\right\}$$
(5.43)

(and  $y(t) \to y(\infty) \in (0, \infty)$ ) as  $t \to \infty$ .

*Proof.* Take  $y \in \mathcal{DS} \cap \mathcal{SV}$  and let  $t_0$  be such that y(t) > 0, y'(t) < 0 for  $t \ge t_0$ . Then

$$(G(y'))' = pF(y) \in \mathcal{RV}(-2+1 \cdot 0) = \mathcal{RV}(-2)$$

provided  $y(t) \to 0$  as  $t \to \infty$ . If  $y(t) \to C \in (0, \infty)$ , then we get the same conclusion since  $F(y(t)) \to F(C) \in (0, \infty)$ , and so  $pF(y) \in \mathcal{RV}(-2)$ . Thus

$$G(-y'(t)) = -G(y'(t)) = \int_t^\infty (G(y'(s)))' \mathrm{d}s \in \mathcal{RV}(-1).$$

In view of  $-y' = G^{-1}(G(-y'))$ , we get  $-y' \in \mathcal{RV}(-1)$ . Hence,

$$\frac{-y(\lambda t) + y(t)}{-ty'(t)} = \int_t^{\lambda t} \frac{-y'(u)}{-ty'(t)} \,\mathrm{d}u = \int_1^\lambda \frac{-y'(st)}{-y'(t)} \,\mathrm{d}s \to \int_1^\lambda \frac{\mathrm{d}s}{s} = \ln\lambda \tag{5.44}$$

as  $t \to \infty$ , thanks to the uniformity. This implies  $-y \in \Pi(-ty'(t))$ . Define

$$\Psi(t) = tG(y'(t)) - \int_{t_0}^t G(y'(s)) \,\mathrm{d}s.$$

Then  $\Psi'(t) = tp(t)F(y(t)) \in \mathcal{RV}(-1)$ , which implies  $\Psi \in \Pi(t\Psi'(t))$ , similarly as in (5.44). Further, we claim  $\Psi \in \Pi(-tG(y'(t)))$ . Indeed, fix  $\lambda > 0$ , and then

۰.

$$\frac{\Psi(\lambda t) - \Psi(t)}{-tG(y'(t))} = \frac{\lambda G(y'(\lambda t))}{-G(y'(t))} + 1 - \frac{\int_t^{\lambda t} G(y'(s)) \, \mathrm{d}s}{-tG(y'(t))}$$
$$= \frac{\lambda G(y'(\lambda t))}{-G(y'(t))} + 1 + \int_1^{\lambda} \frac{G(y'(tu))}{G(y'(t))} \, \mathrm{d}u \to \int_1^{\lambda} \frac{\mathrm{d}u}{u} = \ln \lambda$$

as  $t \to \infty$ , thanks to  $G(-y') \in \mathcal{RV}(-1)$  and the uniformity. From the uniqueness of the auxiliary function up to asymptotic equivalence, we obtain

$$-G(y'(t)) \sim tp(t)F(y(t))$$
 (5.45)

as  $t \to \infty$ . Condition (5.41) is equivalent to  $L_G(v(t)/t) \sim L_G(1/t)$  as  $t \to \infty$ , for all  $v \in SV$ . Hence,

$$-G(y'(t)) = -y'(t)L_G(L_{|y'|}(t)/t) \sim -y'(t)L_G(1/t)$$

as  $t \to \infty$ . Combining this relation with (5.45), we get

$$\frac{-y'(t)}{F(y(t))} \sim \frac{tp(t)}{L_G(1/t)}$$

as  $t \to \infty$ , that is

$$\frac{y'(t)}{F(y(t))} = -(1+o(1))H(t)$$
(5.46)

as  $t \to \infty$ . By integrating this relation over  $(t_0, t)$  we obtain

$$\hat{F}(y(t)) = \hat{F}(y(t_0)) - \int_{t_0}^t (1 + o(1))H(s) \,\mathrm{d}s, \tag{5.47}$$

which implies (5.42) provided  $\int_{a}^{\infty} H(s) ds = \infty$ . Clearly then  $y(t) \to 0$  as  $t \to \infty$ , otherwise we get a contradiction with the divergence of the integral in (5.47). If  $\int_{a}^{\infty} H(s) ds < \infty$  holds, then we integrate (5.46) over  $(t, \infty)$  obtaining (5.43). In this case, y(t) must tend to a positive constant as  $t \to \infty$ . Indeed, if  $y(t) \to 0$  as  $t \to \infty$ , then the left-hand side of (5.47) becomes unbounded which is a contradiction.  $\Box$ 

**Remark 5.6.** A closer examination of the proof of Theorem 5.13 shows that the condition  $\lim_{u\to 0+} |\hat{F}(u)| = \infty$  is somehow needed. Indeed, if we assume that this limit is finite and that  $\int_a^{\infty} H(s) ds = \infty$ , then in view of (5.47) we get contradiction. As a by-product we then have a non-existence of SV solutions. If  $\lim_{u\to 0+} |\hat{F}(u)| < \infty$  holds when  $\int_a^{\infty} H(s) ds < \infty$ , then no conclusion whether  $y(\infty) = 0$  or  $y(\infty) > 0$  can generally be drawn. Note that such phenomena cannot occur in the linear case.

**Remark 5.7.** There exists an alternative way how to prove (5.45). Indeed, denote  $\tilde{L}(t) = L_p(t)F(y(t))$  and observe that  $\tilde{L} \in SV$ . Therefore,

$$\int_t^\infty p(s)F(y'(s))\,\mathrm{d}s = \int_t^\infty s^{-2}\tilde{L}(s)\,\mathrm{d}s \sim \frac{1}{t}\tilde{L}(t) = tp(t)F(y(t))$$

as  $t \to \infty$  by the Karamata theorem. Since  $-G(y'(t)) = \int_t^{\infty} p(s)F(y(s)) ds$ , we obtain (5.45).

**Remark 5.8.** Observe that to prove asymptotic formulae for decreasing SV solutions of (5.34) we do not require (even one-sided) boundedness conditions on  $L_F$  and  $L_G$  such as (5.38). As for condition (5.41) from Theorem 5.13, it is not too restrictive. Many functions satisfy it, for example,  $L_G(u) \rightarrow C \in (0, \infty)$  as  $u \rightarrow 0+$ , or  $L_G(u) = |\ln |u||^{\alpha_1} |\ln |\ln |u||^{\alpha_2}$ ,  $\alpha_1, \alpha_2 \in \mathbb{R}$ . Compare also with (4.23).

The value -2 in the condition  $p \in \mathcal{RV}(-2)$  is natural and consistent within our setting. Indeed, since we work with  $\mathcal{SV}$  solutions, the expression on the left-hand side of (5.34), which is somehow close to the second derivative, is expected to be in  $\mathcal{RV}(-2)$ .

**Corollary 5.1.** Assume that  $p \in \mathcal{RV}(-2)$  and  $\lim_{t\to\infty} L_p(t) = 0$ . Let (5.38) and (5.41) hold. Then any solution  $y \in \mathcal{DS}$  belongs to  $\mathcal{NSV}$ . Moreover,  $-y \in \Pi(-ty'(t))$  and asymptotic formulae (5.42) or (5.43) hold.

*Proof.* By the Karamata theorem,

$$t \int_t^\infty p(s) \, \mathrm{d}s = t \int_t^\infty s^{-2} L_p(s) \, \mathrm{d}s \sim L_p(t) \to 0$$

as  $t \to \infty$ , and so (5.37) follows. Further, in view of  $\limsup_{u\to 0+} L_F(u) < \infty$ , exists M > 0 such that we have for x < 1,

$$\hat{F}(x) \leq -\int_{x}^{1} \frac{\mathrm{d}u}{uM} = \frac{\ln x}{M},$$

which implies  $\lim_{x\to 0^+} \hat{F}(x) = -\infty$ . The statement now follows from Theorem 5.12 and Theorem 5.13.

**Remark 5.9.** Corollary 5.1 can be seen as a nonlinear extension of Theorem 2.11.

**Example 5.1.** Consider the equation

$$(y'L_G(|y'|))' = \frac{L_p(t)y}{t^2|\ln|y||},$$
(5.48)

where  $L_G \in SV_0$  and  $L_p \in SV$ . Then  $\hat{F}(x) = \frac{-(\ln x)^2}{2}$ ,  $x \in (0, 1)$ ,  $\hat{F}(x) \to -\infty$  as  $x \to 0+$ , and  $\hat{F}^{-1}(u) = \exp\{-\sqrt{-2u}\}$ , u < 0. We restrict our considerations to positive (decreasing) solutions y of (5.48) such that y(t) < 1 for  $t \ge t_0$ ; we have this requirement because we need F(u) to be increasing at least in a certain neighborhood of zero (here it is (0, 1)). Note that a slight modification of F, namely  $F(u) = x/|\ln|x/k||$ ,  $k \in (0, \infty)$ , ensures the required monotonicity of F on the (possibly bigger) interval (0, k).

(i) Let  $G(u) = u |\ln|u||$  and  $L_p(t) = \frac{1}{\ln t + h(t)}$ , where *h* is a continuous function on  $[a, \infty)$  with  $|h(t)| = o(\ln t)$  as  $t \to \infty$ , and such that  $\ln t + h(t) > 0$  for  $t \in [a, \infty)$ . Examples of *h* are  $h(t) = \cos t$  or  $h(t) = \ln(\ln t)$ . Note that the required monotonicity of *G* is ensured in a small neighborhood of zero But this is not a problem since we have y' as the argument of *G*, and y'(t) tends to zero as  $t \to \infty$ . Nevertheless, we could modify *G* similarly as in the above mentioned modification of *F*. The function *H* reads as

$$H(t) = \frac{1}{t(\ln t + h(t))|\ln(1/t)|} = \frac{1}{t(\ln t + h(t))\ln t} \sim \frac{1}{t(\ln t)^2}$$

as  $t \to \infty$ . Thus,  $\int_{0}^{\infty} H(s) ds < \infty$  and we have  $\int_{t}^{\infty} H(s) ds \sim \frac{1}{\ln t}$  as  $t \to \infty$ . From Corollary 5.1, we get that for any eventually decreasing positive solution *y* of (5.48) (with y(t) < 1 for large *t*), -y is in  $\Pi$  (*y* is in *NSV*), *y* tends to  $y(\infty) > 0$  and satisfies the formula

$$y(t) = \exp\left\{-\sqrt{(\ln y(\infty))^2 - \frac{2(1+o(1))}{\ln t}}\right\}$$

as  $t \to \infty$ .

(ii) Let  $L_p$  be the same as in (i) and  $G(u) = \frac{u}{\sqrt{1 \pm u^2}}$ . Then

$$H(t) = \frac{\sqrt{1 \pm \frac{1}{t^2}}}{t(\ln t + h(t))} \sim \frac{1}{t \ln t}$$

as  $t \to \infty$ . Note that  $(\ln(\ln t))' = \frac{1}{t \ln t}$ , and so  $\int_{\infty}^{\infty} H(s) ds = \infty$ . From Corollary 5.1, we get that for any eventually decreasing positive solution *y* of (5.48), -y is in  $\Pi$  (*y* is in *NSV*), *y* tends to zero and satisfies the formula

$$y(t) = \exp\left\{-\sqrt{(1+o(1))\ln(\ln t)}\right\}$$

as  $t \to \infty$ .

(iii) Let  $L_p(t) = \frac{1}{(\ln t + h(t))^2}$ , where *h* is as in (i), and *G* = id. Then

$$H(t) = \frac{1}{t(\ln t + h(t))^2} \sim \frac{1}{t(\ln t)^2}$$

as  $t \to \infty$ . Applying Corollary 5.1, we get that any eventually decreasing positive solution *y* of (5.48) (with y(t) < 1 for large *t*) obeys the same asymptotic behavior as *y* in (i).

Among others, the above examples show how the convergence / divergence of the integral  $\int_{-\infty}^{\infty} H(s) ds$  can be affected by the behavior of both *p* and *G*.

Under the conditions of Theorem 5.13-(i), it does not follow that

$$y(t) \sim \hat{F}^{-1} \left\{ -\int_a^t H(s) \, \mathrm{d}s \right\}$$

as  $t \to \infty$ ; this fact was observed already in the linear case, see [45, Remark 2], see also the text after Theorem 2.11. However, we can give a lower estimate under quite mild assumptions. For technical reasons we consider positive decreasing solutions (5.34) on  $[0, \infty)$  (provided  $p \in C([0, \infty))$ ).

**Theorem 5.14.** (*i*) Let  $\liminf_{u\to 0^+} L_G(u) > 0$  and (5.37) hold. Then  $y \in \mathcal{DS} \cap SV$  satisfies the estimate

$$\liminf_{t \to \infty} \frac{y(t)}{\hat{F}^{-1}\left\{\hat{F}(y(0)) - M \int_0^t sp(s) \,\mathrm{d}s\right\}} \ge 1,\tag{5.49}$$

where M is some positive constant. The constant M can be taken such that  $M = 1/\inf_{u \in [0,|y'(0)|]} L_G(u)$ .

(*ii*) In addition to the conditions in (*i*), assume that  $\limsup_{u\to 0+} L_F(u) < \infty$  holds. Then  $y \in \mathcal{DS}$  implies  $y \in NSV$  and

$$\liminf_{t\to\infty} y(t) \exp\left\{N\int_0^t sp(s)\,\mathrm{d}s\right\} \ge y(0),$$

where N is some positive constant. The constant N can be taken such that  $N = \sup_{u \in [0, u(0)]} L_F(u) / \inf_{u \in [0, |y'(0)|]} L_G(u)$ .

*Proof.* (i) Take  $y(t) \in \mathcal{DS} \cap \mathcal{SV}, t \ge 0$ . For  $\lambda \in (0, 1)$ , we have

$$\frac{-G(y'(\lambda t)) + G(y'(t))}{F(y(\lambda t))} = \frac{1}{F(y(\lambda t))} \int_{\lambda t}^{t} p(s)F(y(s)) \, \mathrm{d}s \le \int_{\lambda t}^{t} p(s) \, \mathrm{d}s, \tag{5.50}$$

t > 0. Thanks to  $\liminf_{u \to 0^+} L_G(u) > 0$ , there exists M > 0 such that

$$\frac{-y'(\lambda t)}{MF(y(\lambda t))} + \frac{G(y'(t))}{F(y(\lambda t))} \le \frac{-y'(\lambda t)L_G(|y'(t)|)}{F(y(\lambda t))} + \frac{G(y'(t))}{F(y(\lambda t))} \le \int_{\lambda t}^t p(s) \,\mathrm{d}s, \tag{5.51}$$

t > 0, where the last estimate follows from (5.50). Integration over  $\lambda \in (0, 1)$  yields

$$\frac{-1}{Mt}[\hat{F}(y(t)) - \hat{F}(y(0))] + \frac{G(y'(t))}{t} \int_0^t \frac{\mathrm{d}s}{F(y(s))} \le \frac{1}{t} \int_0^t sp(s) \,\mathrm{d}s, \tag{5.52}$$

where we substituted  $s = \lambda t$  in  $\int_0^1 \frac{d\lambda}{F(y(\lambda t))}$  and we applied the Fubini theorem in  $\int_0^1 \int_{\lambda t}^t p(s) \, ds \, d\lambda$ . From (5.52), we get

$$y(t) \ge \hat{F}^{-1} \left\{ \hat{F}(y(0)) + MG(y'(t)) \int_0^t \frac{\mathrm{d}s}{F(y(s))} - M \int_0^t sp(s) \,\mathrm{d}s \right\}.$$
 (5.53)

Since  $F(y) \in SV$ , the Karamata theorem yields

$$0 < -G(y'(t)) \int_0^t \frac{\mathrm{d}s}{F(y(s))} \sim \frac{-tG(y'(t))}{F(y(t))}$$
$$= \frac{t}{F(y(t))} \int_t^\infty p(s)F(y(s)) \,\mathrm{d}s \le t \int_t^\infty p(s) \,\mathrm{d}s$$

where the asymptotic relation holds as  $t \rightarrow \infty$ . Hence,

$$-G(y'(t))\int_0^t \frac{\mathrm{d}s}{F(y(s))} = o(1)$$

as  $t \to \infty$ . Formula (5.49) now easily follows from (5.53).

(ii) Take  $y(t) \in \mathcal{DS}, t > 0$ . Then  $y \in \mathcal{NSV}$  follows from Theorem 5.12. Thanks to (5.38) which is in fact assumed, there exists N > 0 such that  $-NG(y'(\lambda t))/F(y(\lambda t)) \ge -y'(\lambda t)/y(\lambda t), t > 0$ . As in the proof of (i), we then get

$$\frac{-y'(\lambda t)}{y(\lambda t)} + \frac{G(y'(t))}{F(y(\lambda t))} \le \int_{\lambda t}^{t} p(s) \, \mathrm{d}s.$$

Since this estimate is a special case of (5.51), the rest of the proof is now clear.  $\Box$ 

**Remark 5.10.** It is reasonable to require the conditions  $\lim_{u\to 0^+} |\hat{F}(u)| = \infty$  and  $\int_0^{\infty} sp(s) ds = \infty$  when applying Theorem 5.14. Further notice that the proof of Theorem 5.14 does not require  $p \in \mathcal{RV}(-2)$ , in contrast to the approach known from the linear case, cf. [45, Remark 2]. From this point of view, the result is an improvement even in the linear case. Nevertheless, in order to see Theorem 5.14 as a partial refinement of information about solutions treated in Theorem 5.13-(i), it is reasonable to assume  $p \in \mathcal{RV}(-2)$ .

We now consider more general equation

$$(r(t)G(y'))' = p(t)F(y), (5.54)$$

where *r* and *p* are positive continuous functions on  $[a, \infty)$  and *F*, *G* are as before. First note that in the case when G = id and  $\int_{a}^{\infty} 1/r(s) \, ds = \infty$ , equation (5.54) can be transformed into the equation of the form (2.1) and the type of the interval (on which the equation is considered) is preserved. Indeed, denote  $R(t) = \int_{a}^{t} 1/r(s) \, ds$ and introduce new independent variable s = R(t) and new function  $z(s) = y(R^{-1}(t))$ . Then (5.54) is transformed into

$$\frac{d^2 z}{ds^2} = \tilde{p}(s)F(z), \text{ where } \tilde{p}(s) = p(R^{-1}(s))r(R^{-1}(s)),$$

 $s \in [R(a), \infty)$ . For a general *G* however such a transformation is not at disposal, and we must proceed directly. Let  $\mathcal{DS}_r$  denote the set of all eventually positive decreasing solutions of equation (5.54). An extension of Theorem 5.12 to (5.54) reads as follows.

**Theorem 5.15.** Assume that

$$\lim_{t \to \infty} \frac{t}{r(t)} \int_t^\infty p(s) \, \mathrm{d}s = 0, \tag{5.55}$$

 $\limsup_{u\to 0+} L_F(u) < \infty,$ 

$$\int_{a}^{\infty} G^{-1}\left(\frac{M}{r(s)}\right) \mathrm{d}s = \infty \tag{5.56}$$

for all  $M \in (0, \infty)$ , and  $L_G(u) \ge N$ ,  $u \in (0, \infty)$ , for some N > 0. Then  $\emptyset \neq \mathcal{DS}_r \subset \mathcal{NSV}$ .

*Proof.* We give only a concise proof. Existence of solutions in  $\mathcal{DS}_r$  again follows from [20]. Take  $y \in \mathcal{DS}_r$ . Then  $r(t)G(y'(t)) \to 0$  as  $t \to \infty$ . Otherwise we get contradiction with eventual positivity of y, because of condition (5.56). Similarly as in the proof of Theorem 5.12 we get that there exists  $K \in (0, \infty)$  such that

$$\frac{-ty'(t)}{y(t)} \le \frac{tL_F(y(t))}{L_G(|y'(t)|)r(t)} \int_t^\infty p(s) \, \mathrm{d}s \le \frac{tK}{r(t)} \int_t^\infty p(s) \, \mathrm{d}s$$

for large *t*. Hence,  $y \in NSV$ .

For  $p \in \mathcal{RV}(\beta)$  and  $r \in \mathcal{RV}(\beta + 2)$  with  $\beta < -1$ , denote

$$H_r(t) = \frac{tp(t)}{(-\beta - 1)r(t)}$$

and note that then  $H_r \in \mathcal{RV}(-1)$ . An extension of Theorem 5.13 to (5.54) reads as follows.

**Theorem 5.16.** Assume that  $p \in \mathcal{RV}(\beta)$  and  $r \in \mathcal{RV}(\beta + 2)$ , with  $\beta < -1$ . Further, let  $\lim_{u\to 0^+} |\hat{F}(u)| = \infty$  and (5.41) hold. If  $y \in \mathcal{DS}_r \cap \mathcal{SV}$ , then  $-y \in \Pi(-ty'(t))$ . Moreover: (i) If  $\int_a^{\infty} H_r(s) ds = \infty$ , then (5.42) with  $H_r$  instead of H holds and  $y(t) \to 0$  as  $t \to \infty$ . (ii) If  $\int_a^{\infty} H(s) ds < \infty$ , then (5.43) with  $H_r$  instead of H holds and  $y(t) \to y(\infty) \in (0, \infty)$  as  $t \to \infty$ .

*Proof.* We give again only a concise proof. Take  $y \in \mathcal{DS}_r \cap S\mathcal{V}$ . Then  $(rG(y'))' \in \mathcal{RV}(\beta)$ . Hence,  $-rG(y') \in \mathcal{RV}(\beta + 1)$ , as so  $-y' \in \mathcal{RV}(-1)$ , which implies  $-y \in \Pi(-ty')$  by (5.44). If  $\tilde{L} = L_p F(y)$ , then  $\tilde{L} \in S\mathcal{V}$  and we have

$$-r(t)y'(t)L_G\left(\frac{1}{t}\right) \sim -r(t)G(y'(t)) = \int_t^\infty p(s)F(y(s))\,\mathrm{d}s = \int_t^\infty s^\beta \tilde{L}(s)\,\mathrm{d}s$$
$$\sim \frac{t^{\beta+1}}{-(\beta+1)}\tilde{L}(t) = \frac{t}{-(\beta+1)}p(t)F(y(t))$$

as  $t \to \infty$ , where we applied the Karamata theorem. Asymptotic formulae then follow similarly as (5.42) and (5.43) in the proof of Theorem 5.13.

**Remark 5.11.** If  $p \in \mathcal{RV}(\beta)$ ,  $\beta < -1$ , and  $r \in \mathcal{RV}(\beta + 2)$ , then (5.55) holds provided  $L_p(t)/L_r(t) \to 0$  as  $t \to \infty$ . Indeed, by the Karamata theorem we have,

$$\frac{t}{r(t)} \int_{t}^{\infty} p(s) \, \mathrm{d}s \sim \frac{t L_{p}(t) t^{\beta+1}}{-(\beta+1) t^{\beta+2} L_{r}(t)} = \frac{L_{p}(t)}{-(\beta+1) L_{r}(t)}$$

as  $t \to \infty$ , and the claim follows.

**Remark 5.12.** Assume that  $p \in \mathcal{RV}(\beta)$  and  $r \in \mathcal{RV}(\beta + 2)$  with  $\beta > -1$ . Recall that in the previous theorem we assumed  $\beta < -1$ . Take  $y \in \mathcal{DS}_r \cap SV$ . Then we get  $\int_a^{\infty} p(s)F(y(s)) ds = \infty$  since the index of regular variation of pF(y) is bigger than -1. Integrating (5.54) from  $t_0$  to t, where  $t_0$  is such that  $y(t) > 0, y'(t) < 0, t \ge t_0$ , we obtain  $r(t)G(y'(t)) = r(t_0)G(y'(t_0)) + \int_{t_0}^t p(s)F(y(s)) ds$ . Hence, if we let t tend to  $\infty$ , then r(t)G(y'(t)) tends to  $\infty$ . Thus y' is eventually positive, which contradicts  $y \in \mathcal{DS}_r$ . In other words, this observation indicates that SV solutions should not be searched among  $\mathcal{DS}_r$  solutions in this setting. We conjecture that we should take an increasing solution in order to remain in the set SV. Of course, some logical adjustments then have to be made, like taking  $\mathcal{RV}$  instead of  $\mathcal{RV}_0$  in (5.35). As for  $\mathcal{DS}_r$ , we conjecture that this class somehow corresponds to  $\mathcal{RV}(-1)$  solutions. Note that the integral  $\int_a^{\infty} 1/r(s) ds$  (which is "close" to the integral  $\int_a^{\infty} G^{-1}(M/r(s)) ds$ ) is divergent for  $\beta < -1$  resp. convergent for  $\beta > -1$  since  $1/r \in \mathcal{RV}(-\beta - 2)$ .

We have not mentioned the remaining possibility so far, namely  $\beta = -1$ . This border case is probably the most difficult one, and surely will require a quite different approach. The direct use of the Karamata theorem is problematic in contrast to the corresponding situations in other cases. If  $p \in \mathcal{RV}(-1)$  and  $r \in \mathcal{RV}(1)$ , we cannot even say whether  $\int_a^{\infty} p(s) ds$ ,  $\int_a^{\infty} 1/r(s) ds$  are convergent or divergent. In fact, the situation is more tangled because of presence of nonlinearities *F*, *G*, where SV components  $L_F$ ,  $L_G$  are supposed to have a stronger effect than in the case  $\beta \neq -1$ .

As a conclusion of this section, we indicate some further directions for a possible future research. Asymptotic theory of nearly linear equations offers many interesting questions. This section contains some answers but there are many issues which could be followed further. There is also some space for improving the presented results. We conjecture that the results can be generalized in the sense of replacing condition (5.35) by  $F(|\cdot|)$ ,  $G(|\cdot|) \in \mathcal{RV}_0(\gamma)$ ,  $\gamma > 0$ , which would lead to a "nearly half-linear equation." It is expected that — within our setting, with taking  $\mathcal{RV}$  instead of  $\mathcal{RV}_0$  in (5.35) — increasing solutions of (5.34) are in  $\mathcal{RV}(1)$  and asymptotic formulae can be established. In contrast to the linear case, a reduction of order formula is not at disposal. A topic which would also be of interest is to obtain more precise information about  $\mathcal{SV}$  solutions of (2.1), for instance, by means of the class  $\Pi R_2$ , cf. Theorem 2.13.

# Chapter **6**\_

### Concluding remarks

It is evident that many other works can be found in which the theory of regularly varying (or of somehow related) functions is applied to study differential equations (or dynamic equations). However, the aim of this text is such that it cannot cover everything (and with details). In the last chapter we briefly mention at least few another works.

## 6.1 More on differential equations in the framework of regular variation

Appleby in [4] considers the rate of convergence to equilibrium of Volterra integrodifferential equations with infinite memory. It is shown that if the kernel of Volterra operator is regularly varying at infinity, and the initial history is regularly varying at minus infinity, then the rate of convergence to the equilibrium is regularly varying at infinity, and the exact pointwise rate of convergence can be determined in terms of the rate of decay of the kernel and the rate of growth of the initial history. The result is considered both for a linear Volterra integrodifferential equation as well as for the delay logistic equation from population biology.

The concept of subexponential functions (which are somehow related to  $\mathcal{RV}$  functions) is used, for instance, by Appleby, Györi, and Reynolds in [5]; the paper examines the asymptotic behavior of solutions of scalar linear integro-differential equations. See also Appleby, Reynolds [8, 7].

Regular variation was used by Marić, Radašin in [109] to study equations arising in boundary-layer theory, see also [105, Chapter 4]. The core of such considerations is based on the work by McLeod [118]. Van den Berg in [15] established a result on the asymptotics of some solutions to a first order nonlinear differential equation; a conspicuous feature of his consideration is its relation to the nonstandard asymptotic analysis.

Several papers exist devoted to investigation of second order equations with

deviating argument in the framework of regular variation, see Kusano, Marić [95, 96, 97, 98] for linear case and Kusano, Manojlović, Tanigawa [92] and Tanigawa [161, 163] for half-liner case. Note that the last mentioned paper utilizes the concept of generalized regular variation. Common features of these works are that a crucial role is played by the results for equations without retarded and advanced arguments which can be seen as modifications of some theorems in Chapters 2 and 3. A desired solution is then obtained by the fixed point technique.

## 6.2 Regularly varying sequences and difference equations

As mentioned in Subsection 1.3.3, the concept of regularly varying sequences was introduced already by Karamata. Much later this theory was used to study asymptotic properties of linear and half-linear difference equations in Matucci and Řehák [119, 120, 121, 122], discrete versions of some of the statements from Chapters 2 and 3 can be revealed there. It should however be emphasized that the discrete case requires to find new ways of the proofs at many points. A somewhat different approach to the use of regularly varying sequences in linear difference equations is represented by Kooman [81], see also related paper by Kooman [80]. Second order Emden-Fowler type difference equations is investigated in the framework of regular variation in [1] by Agarwal and Manojlović. Other approach in the study of Emden-Fowler type difference equations is represented by the papers [76, 77, 78] of Kharkov. Some considerations in Kharkov's papers can be seen as a discrete analogues of the results by Evtukhov et al., see e.g. Subsection 4.3.7.

#### 6.3 Regular variation on time scales and dynamic equations

The concept of regularly varying functions on time scales (or measure chains) is introduced in [141] by Řehák in order to study asymptotic behavior of dynamic equations (which unify and extend differential and difference equations), see Subsection 1.3.3. Other applications to linear and half-linear dynamic equations on time scales can be found in Řehák and Vítovec [152, 153]. The results can be viewed as a unification and extension of some of the statements from Chapters 2 and 3 and corresponding statements for difference equations. Note that an important role is played by an additional condition on the graininess  $\mu(t)$ ; it is somehow necessary to assume that  $\mu(t) = o(t)$  as  $t \to \infty$ .

#### 6.4 *q*-regular variation and *q*-difference equations

The concept of q-regularly varying functions was introduced in [151] by Řehák and Vítovec in order to study asymptotic behavior of q-difference equations, see

Subsection 1.3.3. Linear *q*-difference equations in the framework of this theory are studied in Řehák [142, 144]. Half-linear *q*-difference equations are studied in [143] Řehák and Vítovec [154]. These results can be understood as a *q*-version of some of the statements from Chapters 2 and 3. It is worthy of note that a specific approach is used for *q*-difference equations; this is due to pleasant properties of *q*-regularly varying functions. Certain generalization of *q*-regular variation was introduced in Řehák [145] and applied in the study of general linear second order *q*-difference equations. See also Řehák [148] where the classical Poincaré-Perron type result was applied to examine generalized *q*-regularly varying solutions of *n*-th order linear *q*-difference equations.

### Bibliography

- R. P. Agarwal, J. V. Manojlović, On the existence and the asymptotic behavior of nonoscillatory solutions of second order quasilinear difference equations, Funkcial. Ekvac. 56 (2013), 81–109.
- [2] S. Aljančić, R. Bojanić, M. Tomić, Sur la valeur asymptotique d'une classe des intégrales définies (French), Acad. Serbe Sci. Publ. Inst. Math. 7 (1954), 81–94.
- [3] J. A. D. Appleby, E. Buckwar, A constructive comparison technique for determining the asymptotic behaviour of linear functional differential equations with unbounded delay, Differ. Equ. Dyn. Syst. 18 (2010), 271–301.
- [4] J. A. D. Appleby, On regularly varying and history-dependent convergence rates of solutions of a Volterra equation with infinite memory, Adv. Difference Equ. 2010, Art. ID 478291, 31 pp.
- [5] J. A. D. Appleby, I. Györi, D. W. Reynolds, Subexponential solutions of scalar linear integro-differential equations with delay, Funct. Differ. Equ. 11 (2004), 11–18.
- [6] J. A. D. Appleby, D. D. Patterson, Classification of convergence rates of solutions of perturbed ordinary differential equations with regularly varying nonlinearity, arXiv:1303.3345 [math.CA].
- [7] J. A. D. Appleby, D. W. Reynolds, Subexponential solutions of linear integrodifferential equations and transient renewal equations, Proc. Roy. Soc. Edinburgh Sect. A 132 (2002), 521–543.
- [8] J. A. D. Appleby, D. W. Reynolds, Subexponential solutions of linear integrodifferential equations, Dynamic systems and applications, Vol. 4, Dynamic, Atlanta, GA, 2004, 488–494.
- [9] V. G. Avakumović, Sur l'équation différentielle de Thomas-Fermi (French), Acad. Serbe Sci. Publ. Inst. Math. 1 (1947), 101–113.

- [10] V. G. Avakumović, Sur l'équation différentielle de Thomas-Fermi, II, (French. Serbian summary), Acad. Serbe Sci. Publ. Inst. Math. 2 (1948), 223–235.
- [11] A. A. Balkema, J. L. Geluk, L. de Haan, An extension of Karamata's Tauberian theorem and its connection with complementary convex functions, Quart. J. Math. Oxford Ser. 30 (1979), 385–416.
- [12] R. Bellman, Stability theory in differential equations, Dover Publications, Inc., New York 1969.
- [13] M. A. Bilozerowa, V. M. Evtukhov, Asymptotic representations of solutions of the differential equation  $y^{(n)} = \alpha_0 p(t) \prod_{i=0}^{n-1} \phi_i(y^{(i)})$ , Miskolc Math. Notes 13 (2012), 249–270.
- [14] N. H. Bingham, C. M. Goldie, J. L. Teugels, Regular Variation, Encyclopedia of Mathematics and its Applications, Vol. 27, Cambridge University Press, 1987.
- [15] I. P. van den Berg, On solutions of polynomial growth of ordinary differential equations, J. Differential Equations 81 (1989), 368–402.
- [16] M. Bohner, A. C. Peterson. Dynamic Equations on Time Scales: An Introduction with Applications, Birkhäuser, Boston, 2001.
- [17] R. Bojanić, E. Seneta, A unified theory of regularly varying sequences, Math. Z. 134 (1973), 91–106.
- [18] N. Bourbaki, "Fonctions d'une Variable Réele," Chapter V (Étude Locale des Fonctions), 2nd ed., Hermann, Paris, 1961.
- [19] M. Cecchi, Z. Došlá, M. Marini, On nonoscillatory solutions of differential equations with *p*-Laplacian, Adv. Math. Sci. Appl. 11 (2001), 419–436.
- [20] T. A. Chanturiya, Monotone solutions of a system of nonlinear differential equations (Russian), Ann. Polon. Math. 37 (1980), 59–70.
- [21] J. Chover, P. Ney, S. Wainger, Functions of probability measures, J. Analyse Math. 26 (1973), 255–302.
- [22] P. Clément, R. Manásevich, E. Mitidieri, Positive solutions for a quasilinear systems via blow up, Comm. Partial Differential Equations 18 (1993), 2071– 2106.
- [23] W. A. Coppel, Stability and asymptotic behavior of differential equations, D. C. Heath and Co., Boston, Mass. 1965.
- [24] R. Dalmasso, Existence of positive solutions to Dirichlet boundary-value problems for some recursive differential systems, Differential Integral Equations 24 (2011), 829–944.

- [25] D. Djurčić, Lj. D. R. Kočinac, M. R. Žižović, Some properties of rapidly varying sequences, J. Math. Anal. Appl. 327 (2007), 1297–1306.
- [26] D. Djurčić, Lj. D. R. Kočinac, M. R. Žižović, Classes of sequences of real numbers, games and selection properties, Topology and its Applications 156 (2008), 46–55.
- [27] Z. Došlá, M. Marini, S. Matucci, A boundary value problem on a half-line for differential equations with indefinite weight, Commun. Appl. Anal. 15 (2011), 341–352.
- [28] O. Došlý, P. Řehák, Half-linear Differential Equations, Elsevier, North Holland, 2005.
- [29] O. Došlý, M. Unal, Conditionally oscillatory half-linear differential equations, Acta Math. Hungar. 120 (2008), 147–163.
- [30] Å. Elbert, A half-linear second order differential equation, Colloq. Math. Soc. János Bolyai 30 (1979), 158–180.
- [31] R. Emden, Anwendungen der mechanischen Warmen-theorie auf Kosmologie und meteorologische Probleme, Leipzig, Teubner, 1907.
- [32] V. M. Evtukhov, M. D. Abu Él'-Shaur, Asymptotic representations of solutions of a class of second-order ordinary differential equations (Russian), Nelinijni Koliv. 14 (2011), 203–213; translation in Nonlinear Oscil. (N. Y.) 14 (2011), 211–221.
- [33] V. M. Evtukhov, M. A. Belozerova, Asymptotic representations of solutions of second-order essentially nonlinear nonautonomous differential equations (Russian), Ukrain. Mat. Zh. 60 (2008), 310–331; translation in Ukrainian Math. J. 60 (2008), 357–383.
- [34] V. M. Evtukhov, V. M. Kharkov, Asymptotic representations of solutions of second-order essentially nonlinear differential equations (Russian), Differ. Uravn. 43 (2007), 1311–1323, 1437.
- [35] V. M. Evtukhov, A. M. Samoilenko, Conditions for the existence of solutions of real nonautonomous systems of quasilinear differential equations vanishing at a singular point, Ukrainian Math. J. 62 (2010), 56–86.
- [36] V. M. Evtukhov, A. M. Samoilenko, Asymptotic representations of solutions of nonautonomous ordinary differential equations with regularly varying nonlinearities (Russian), Differ. Uravn. 47 (2011), 628–650; translation in Differ. Equ. 47 (2011), 627–649.
- [37] V. M. Evtukhov, A. A. Stekhun, Asymptotic representations of solutions of a class of third-order nonlinear nonautonomous differential equations

(Russian), Ukrain. Mat. Zh. 59 (2007), 1363–1375; translation in Ukrainian Math. J. 59 (2007), 1528–1543.

- [38] V. M. Evtukhov, E. S. Vladova, Asymptotic representations of solutions of essentially nonlinear two-dimensional systems of ordinary differential equations, Ukrainian Math. J. 61 (2009), 1877–1892.
- [39] V. M. Evtukhov, E. S. Vladova, On the asymptotics of solutions of nonlinear cyclic systems of ordinary differential equations, Mem. Differential Equations Math. Phys. 54 (2011), 1–25.
- [40] E. Fermi, Un metodo statistico per la determinazione di alcune proprietá dell atomo, Rend. Accad. Naz. Lincei 6 (1927), 602–607.
- [41] R. H. Fowler, The form near infinity of real, continuous solutions of a certain differential equation of the second order, Quart. J. Math. (Oxford) 45 (1914), 289–350.
- [42] R. H. Fowler, Further studies of Emden's and similar differential equations, Quart. J. Math. (Oxford) 2 (1931), 259–288.
- [43] J. Galambos, E. Seneta, Regularly varying sequences, Proc. Amer. Math. Soc. 41 (1973), 110–116.
- [44] J. L. Geluk, Π-Regular Variation, Proc. Amer. Math. Soc. 82 (1981), 565–570
- [45] J. L. Geluk, On slowly varying solutions of the linear second order differential equation. Proceedings of the Third Annual Meeting of the International Workshop in Analysis and its Applications. Publ. Inst. Math. (Beograd) (N.S.) 48 (1990), 52–60.
- [46] J. L. Geluk, Note on a theorem of Avakumović, Proc. Amer. Math. Soc. 112 (1991), 429–431.
- [47] J. L. Geluk, L. de Haan, Regular Variation, Extensions and Tauberian Theorems, CWI Tract 40, Amsterdam, 1987.
- [48] J. L. Geluk, V. Marić, M. Tomic, On regularly varying solutions of second order linear differential equations, Differential Integral Equations 6 (1993), 329–336.
- [49] L. J. Grimm, L. M. Hall, Self adjoint differential equations and Karamata functions. Proceedings of the Third International Colloquium on Differential Equations (Plovdiv, 1992), 77–87, VSP, Utrecht, 1993.
- [50] L. de Haan, On Regular Variation and its Applications to the Weak Convergence of Sample Extremes, Mathematisch Centrum Amsterdam, 1970.

- [51] M. Hačik, E. Omey, On the zeros of oscillatory solutions of linear secondorder differential equations, Publ. Inst. Math. (Beograd) (N.S.) 49 (1991), 189–200.
- [52] G. H. Hardy, Orders of Infinity, Cambridge Univ. Press, 1954.
- [53] P. Hartman, Ordinary differential equations, Classics in Applied Mathematics 38, Society for Industrial and Applied Mathematics (SIAM), Philadelphia, PA, 2002.
- [54] P. Hartman, A. Wintner, Asymptotic integrations of linear differential equations, Amer. J. Math. 77 (1955), 45–86.
- [55] S. Hilger, Ein Maßkettenkalkül mit Anwendung auf Zentrumsmannigfaltigkeiten. Ph.D. dissertation, Universität Würzburg, 1988.
- [56] H. C. Howard, V. Marić, Regularity and nonoscillation of solutions of second order linear differential equations, Bull. Cl. Sci. Math. Nat. Sci. Math. 22 (1997), 85–98.
- [57] H. Howard, V. Marić, Z. Radašin, Asymptotics of nonoscillatory solutions of a second order linear differential equation. Zb. Rad. Prirod. – Mat. Fak. Ser. Mat. 20 (1990), 107–116.
- [58] J. Jaroš, T. Kusano, Remarks on the existence of regularly varying solutions for second order linear differential equations; Publ. Inst. Math. (Beograd) (N.S.) 72 (2002), 113–118.
- [59] J. Jaroš, T. Kusano, T. Tanigawa, Nonoscillation theory for second order halflinear differential equations in the framework of regular variation, Results Math. 43 (2003), 129–149.
- [60] J. Jaroš, T. Kusano, Self-adjoint differential equations and generalized Karamata functions, Bull. Cl. Sci. Math. Nat. Sci. Math. 29 (2004), 25–60.
- [61] J. Jaroš, T. Kusano, Existence and precise asymptotic behavior of strongly monotone solutions of systems of nonlinear differential equations, Differ. Equ. Appl 5 (2013), 185–204.
- [62] J. Jaroš, T. Kusano, Asymptotic behavior of positive solutions of a class of systems of second order nonlinear differential equations, Electron. J. Qual. Theory Differ. Equ. 2013, No. 23, 23 pp.
- [63] J. Jaroš, T. Kusano, Slowly varying solutions of a class of first order systems of nonlinear differential equations, Acta Math. Univ. Comen. 82 (2013), 265– 284.

- [64] J. Jaroš, T. Kusano, J. Manojlović, Asymptotic analysis of positive solutions of generalized Emden-Fowler differential equations in the framework of regular variation, Cent. Eur. J. Math. 11 (2013), 2215–2233.
- [65] J. Jaroš, T. Kusano, V. Marić, Existence of regularly and rapidly varying solutions for a class of third order nonlinear ordinary differential equations, Publ. Inst. Math. (Beograd) (N.S.) 79 (2006), 51–64.
- [66] J. Jaroš, T. Kusano, T. Tanigawa, Nonoscillatory half-linear differential equations and generalized Karamata functions, Nonlinear Anal. 64 (2006), 762– 787.
- [67] J. Jaroš, T. Kusano, T. Tanigawa, Asymptotic analysis of positive solutions of a class of third order nonlinear differential equations in the framework of regular variation, Math. Nachr. 286 (2013), 205–223.
- [68] J. Jaroš, T. Kusano, T. Tanigawa, Existence and precise asymptotics of positive solutions for a class of nonlinear differential equations of the third order, Georgian Math. J. 20 (2013), 493–531.
- [69] J. Jaroš, T. Kusano, T. Tanigawa, Asymptotic analysis of positive solutions of a class of third order nonlinear differential equations in the framework of regular variation, Math. Nachr. 286 (2013), 205–223.
- [70] K. Kamo, H. Usami, Asymptotic forms of positive solutions of second-order quasilinear ordinary differential equations, Adv. Math. Sci. Appl. 10 (2000), 673–688.
- [71] K. Kamo, H. Usami, Asymptotic forms of positive solutions of secondorder quasilinear ordinary differential equations with sub-homogeneity, Hiroshima Math. J. 31 (2001), 35–49.
- [72] K. Kamo, H. Usami, Characterization of slowly decaying positive solutions of second-order quasilinear ordinary differential equations with subhomogeneity, Bull. Lond. Math. Soc. 42 (2010), 420–428.
- [73] J. Karamata, Sur certain "Tauberian theorems" de M. M. Hardy et Littlewood, Mathematica Cluj 3 (1930), 33–48.
- [74] Karamata, J. Sur un mode de croissance réguliere des functions (French), Mathematica 4 (1930), 38–53.
- [75] J. Karamata, Sur un mode de croissance régulière. Théorèmes fondamentaux, Bull. Soc. Math. France 61 (1933), 55–62.
- [76] V. Kharkov, Asymptotic behavior of a class of solutions of second-order Emden-Fowler difference equations, Difference equations and applications, Ugur-Bahcesehir Univ. Publ. Co., Istanbul, 2009, 219–226.

- [77] V. Kharkov, Asymptotic representations of a class of solutions of a secondorder difference equation with a power nonlinearity (Russian), Ukrain. Mat. Zh. 61 (2009), 839–854; translation in Ukrainian Math. J. 61 (2009), 994–1012
- [78] V. Kharkov, Positive solutions of the Emden-Fowler difference equation, J. Difference Equ. Appl. 19 (2013), 234–260.
- [79] I. T. Kiguradze, T. A. Chanturiya, Asymptotic properties of solutions of nonautonomous ordinary differential equations, Mathematics and its Applications 89, Kluwer Acad. Pub., Dordrecht, 1993.
- [80] R. J. Kooman, Asymptotic behaviour of solutions of linear recurrences and sequences of Möbius-transformations, J. Approx. Theory 93 (1998), 1–58.
- [81] R. J. Kooman, An asymptotic formula for solutions of linear second-order difference equations with regularly behaving coefficients, J. Difference Equ. Appl. 13 (2007), 1037–1049.
- [82] A. A. Kozma, Asymptotic behavior of a class of solutions of second-order nonlinear nonautonomous differential equations (Russian), Neliniini Koliv. 14 (2011), 468–481.
- [83] T. Kusano, J. Manojlović, V. Marić, Increasing solutions of Thomas-Fermi type differential equations—the sublinear case, Bull. Cl. Sci. Math. Nat. Sci. Math. 36 (2011), 21–36.
- [84] T. Kusano, J. Manojlović, Precise asymptotic behavior of solutions of the sublinear Emden-Fowler differential equation, Appl. Math. Comput. 217 (2011), 4382–4396.
- [85] T. Kusano, J. Manojlović, Asymptotic analysis of Emden-Fowler differential equations in the framework of regular variation, Ann. Mat. Pura Appl. 190 (2011), 619–644.
- [86] T. Kusano, J. Manojlović, Asymptotic behavior of positive solutions of sublinear differential equations of Emden-Fowler type, Comput. Math. Appl. 62 (2011), 551–565.
- [87] T. Kusano, J. Manojlović, Positive solutions of fourth order Emden-Fowler type differential equations in the framework of regular variation, Appl. Math. Comput. 218 (2012), 6684–6701.
- [88] T. Kusano, J. Manojlović, Positive solutions of fourth order Thomas-Fermi type differential equations in the framework of regular variation, Acta Appl. Math. 121 (2012), 81–103.
- [89] T. Kusano, J. Manojlović, Asymptotic behavior of positive solutions of odd order Emden-Fowler type differential equations in the framework of regular variation, Electron. J. Qual. Theory Differ. Equ. 2012, No. 45, 23 pp.

- [90] T. Kusano, J. Manojlović, Complete asymptotic analysis of positive solutions of odd-order nonlinear differential equation, Lith. Math. J. 53 (2013), 40–62.
- [91] T. Kusano, J. Manojlović, J. Milošević, Intermediate solutions of second order quasilinear ordinary differential equations in the framework of regular variation, Appl. Math. Comput. 219 (2013), 8178–8191.
- [92] T. Kusano, J. Manojlović, T. Tanigawa, Existence of regularly varying solutions with nonzero indices of half-linear differential equations with retarded arguments, Comput. Math. Appl. 59 (2010), 411–425.
- [93] T. Kusano, J. Manojlović, T. Tanigawa, Existence and asymptotic behavior of positive solutions of fourth order quasilinear differential equations, Taiwanese J. Math. 17 (2013), 999–1030.
- [94] T. Kusano; V. Marić, Nonoscillatory linear and half-linear differential equations having regularly varying solutions, Adv. Math. Sci. Appl. 14 (2004), 351–357.
- [95] T. Kusano; V. Marić, On a class of functional differential equations having slowly varying solutions, Publ. Inst. Math. (Beograd) (N.S.) 80 (2006), 207– 217.
- [96] T. Kusano, V. Marić, Slowly varying solutions of functional differential equations with retarded and advanced arguments, Georgian Math. J. 14 (2007), 301–314.
- [97] T. Kusano, V. Marić, T. Tanigawa, The asymptotic behaviour of solutions of a class of nonlinear second-order differential equations, Bull. Lond. Math. Soc. 39 (2007), 413–418.
- [98] T. Kusano, V. Marić, On the existence and the asymptotic behaviour of regularly (slowly) varying solutions of some classes of second order functional differential equations with deviating argument, Int. J. Pure Appl. Math. 46 (2008), 231–236.
- [99] T. Kusano, V. Marić, T. Tanigawa, Asymptotics of some classes of nonoscillatory solutions of second-order half-linear differential equations, Bull. Cl. Sci. Math. Nat. Sci. Math. 28 (2003), 61–74.
- [100] T. Kusano, V. Marić, T. Tanigawa, An asymptotic analysis of positive solutions of generalized Thomas-Fermi differential equations—the sub-halflinear case, Nonlinear Anal. 75 (2012), 2474–2485.
- [101] T. Kusano, A. Ogata, H. Usami, On the oscillation of solutions of second order quasilinear ordinary differential equations. Hiroshima Math. J. 23 (1993), 645–667.

- [102] J. H. Lane, On the theoretical temperature of the sun under the hypothesis of a gaseous mass maintaining its volume by its internal heat and depending on the laws of gases known to terrestrial experiment, The American Journal of Science and Arts, 2nd series 50 (1870), 57–74.
- [103] A. R. Liddle, L. H. Lyth, Cosmological Inflation and Large-Scale Structure, Cambridge Univ. Press, Cambridge, 2000.
- [104] J. Manojlović, V. Marić, An asymptotic analysis of positive solutions of Thomas-Fermi type sublinear differential equations. Mem. Differential Equations Math. Phys. 57 (2012), 75–94.
- [105] V. Marić, Regular Variation and Differential Equations, Lecture Notes in Mathematics 1726, Springer-Verlag, Berlin-Heidelberg-New York, 2000.
- [106] V. Marić, Z. Radašin, Asymptotic behavior of solutions of the equation  $y'' = f(x)\phi(\psi(y))$ , Glas. Mat. Ser. III 23 (1988), 27–34.
- [107] V. Marić, Z. Radašin, Asymptotics of solutions of the equation  $y'' = f(g(x))\phi(\psi(y))$ , Differential equations (Xanthi, 1987), 459—466, Lecture Notes in Pure and Appl. Math., 118, Dekker, New York, 1989.
- [108] V. Marić, Z. Radašin, On a method in asymptotics of solutions of a class of second order nonlinear differential equations, Bull. Acad. Serbe Sci. Arts Cl. Sci. Math. Natur. 20 (1995), 1–11.
- [109] V. Marić, Z. Radašin, Regularity and asymptotics of a similarity solution of an equation arising in boundary layer theory, World Congress of Nonlinear Analysts '92, Vol. I–IV (Tampa, FL, 1992), 1469–1473, de Gruyter, Berlin, 1996.
- [110] V. Marić, M. Tomić, Asymptotic properties of solutions of the equation  $y'' = f(x)\phi(y)$ , Math. Z. 149 (1976), 261—266.
- [111] V. Marić, M. Tomić, Regular variation and asymptotic properties of solutions of nonlinear differential equations, Publ. Inst. Math. (Beograd) (N.S.) 21 (1977), 119–129.
- [112] V. Marić, M. Tomić, Asymptotics of solutions of a generalized Thomas-Fermi equation, J. Differential Equations 35 (1980), 36–44.
- [113] V. Marić, M. Tomić, A trichotomy of solutions of second order linear differential equations, Univ. u Novom Sadu Zb. Rad. Prirod.-Mat. Fak. Ser. Mat. 14 (1984), 1–11.
- [114] V. Marić, M. Tomić, A classification of solutions of second order linear differential equations by means of regularly varying functions. Proceedings of the Third Annual Meeting of the International Workshop in Analysis and its Applications, Publ. Inst. Math. (Beograd) (N.S.) 48 (1990), 199–207.

- [115] V. Marić, M. Tomić, Slowly varying solutions of second order linear differential equations, Slobodan Aljančić memorial volume, Publ. Inst. Math. (Beograd) (N.S.) 58 (1995), 129–136.
- [116] V. Marić, M. Tomić, A note on some results of Grimm and Hall, PRIM '96 (Budva), 169–174, Univ. Novi Sad, Novi Sad, 1997.
- [117] J. Mařík, M. Ráb, Asymptotische Eigenschaften von Lösungen der Differentialgleichung y'' = A(x)y im nichtoszillatorischen Fall. (German), Czechoslovak Math. J. 10 (85) 1960, 501–522.
- [118] J. B. McLeod, The existence and uniqueness of a similarity solution arising from separation at a free stream line, Quart. J. Math. Oxford Ser. 23 (1972), 63–77.
- [119] S. Matucci, P. Řehák; Regularly varying sequences and second order difference equations, J. Difference Equ. Appl. 14 (2008), 17–30.
- [120] S. Matucci, P. Řehák, Second order linear difference equations and Karamata sequences, Int. J. Difference Equ. 3 (2008), 277–288.
- [121] S. Matucci, P. Rehák, Rapidly varying decreasing solutions of half-linear difference equations, Math. Comput. Modelling 49 (2009), 1692–1699.
- [122] S. Matucci, P. Řehák, Regularly varying solutions of second order difference equations with arbitrary sign coefficient, Advances in Difference Equations, vol. 2010, Article ID 673761, 16 pages, 2010.
- [123] S. Matucci, P. Řehák, Asymptotics of decreasing solutions of coupled *p*-Laplacian systems in the framework of regular variation, to appear in Ann. Mat. Pura Appl.
- [124] S. Matucci, P. Rehák, Extreme solutions of a system of *n* nonlinear differential equations and regularly varying functions, submitted.
- [125] Ž. Mijajlović, N. Pejović, S. Šegan, G. Damljanović, On asymptotic solutions of Friedmann equations, Appl. Math. Comp. 219 (2012), 1273–1286.
- [126] J. D. Mirzov, On some analogs of Sturm's and Kneser's theorems for nonlinear systems, J. Math. Anal. Appl. 53 (1976), 418–425.
- [127] B. Muatjetjeja, C. M. Khalique, F. M. Mahomed, Group Classification of a Generalized Lane-Emden System, J. Appl. Math., Volume 2013, Article ID 305032, 12 pages.
- [128] M. Naito, Existence and asymptotic behavior of positive solutions of higherorder quasilinear ordinary differential equations, Math. Nachr. 279 (2006), 198–216.

- [129] M. Naito, Existence of positive solutions of higher-order quasilinear ordinary differential equations, Ann. Mat. Pura Appl. 186 (2007), 59–84.
- [130] E. Omey, Regular variation and its applications to second order linear differential equations, Bull. Soc. Math. Belg. Sér. B 33 (1981), 207–229.
- [131] E. Omey, Rapidly varying behaviour of the solutions to a second order linear differential equation, Proc. of the 7th int. coll. on differential equations, Plovdiv, August 18-23, 1996. Utrecht: VSP (1997), 295–303.
- [132] E. Omey, A note on the solutions of  $r''(x)r(x) = g(x)r'^2(x)$ , Proc. of the 9th int. coll. on differential equations, Plovdiv, August 18-23, 1998. Utrecht: VSP (1999), 293–298.
- [133] E. Omey, On the class gamma and related classes of functions, retrieved April 19, 2013, from <www.edwardomey.com/nonsave/ OmeyBeurlingJuli2011Final.pdf>.
- [134] E. Omey, E. Willekens, Π-variation with remainder, J. London Math. Soc. 37 (1988), 105–118.
- [135] Z. Pátíková, Asymptotic formulas for non-oscillatory solutions of perturbed Euler equation, Nonlinear Anal. 69 (2008), 3281–3290.
- [136] Z. Pátíková, Asymptotic formulas for solutions of half-linear Euler-Weber equation, E. J. Qualitative Theory of Diff. Equ., Proc. 8th Coll. Qualitative Theory of Diff. Equ., 15 (2008), 1–11.
- [137] Z. Pátíková, Asymptotic formulas for nonoscillatory solutions of conditionally oscillatory half-linear equations, Math. Slovaca 60 (2010), 223–236.
- [138] O. Perron, Uber summengleichungen und Poincarésche differenzengleichungen (German), Math. Ann. 84 (1921), 1–15.
- [139] J. A. Peacock, Cosmological Physics, Cambridge Univ. Press, Cambridge, 1999.
- [140] H. Poincaré, Sur les equations lineaires aux differentielles ordinaires et aux differences finies (French), Amer. J. Math. 7 (1885), 203–258.
- [141] P. Řehák, Regular variation on time scales and dynamic equations, Aust. J. Math. Anal. Appl. 5 (2008), 1–10.
- [142] P. Rehák, Second order linear *q*-difference equations: nonoscillation and asymptotics, Czechoslovak Mathematical Journal 61 (2011), 1107–1134.
- [143] P. Řehák, Asymptotic behavior of solutions to half-linear *q*-difference equations, Abstr. Appl. Anal. , vol. 2011, Article ID 986343, 12 pp., 2011.

- [144] P. Řehák, A note on asymptotics and nonoscillation of linear *q*-difference equations, E. J. Qualitative Theory of Diff. Equ., Proc. 9'th Coll. Qualitative Theory of Diff. Equ., No. 12 (2011), 1–12.
- [145] P. Řehák, On certain asymptotic class of solutions to second order linear q-difference equations, J. Phys. A: Math. Theor. 45 (2012) 055202.
- [146] P. Řehák, Asymptotic behavior of increasing solutions to a system of n nonlinear differential equations, Nonlinear Analysis Series A: Theory, Methods & Applications 77 (2013), 45–58.
- [147] P. Řehák, De Haan type increasing solutions of half-linear differential equations, J. Math. Anal. Appl. 412 (2014), 236–243.
- [148] P. Řehák, A few remarks on Poincaré-Perron solutions and regularly varying solutions, submitted.
- [149] P. Řehák, On decreasing solutions of second order nearly linear differential equations, to appear in Bound. Value Probl.
- [150] P. Rehák, V. Taddei, Solutions of half-linear differential equations in the classes Gamma and Pi, in preparation.
- [151] P. Řehák, J. Vítovec, *q*-regular variation and *q*-difference equations, J. Phys. A: Math. Theor. 41 (2008) 495203, 1–10.
- [152] P. Rehák, J. Vítovec, Regularly varying decreasing solutions of half-linear dynamic equations, in: Discrete Dynamics and Difference Equations, Proceedings of the Twelfth International Conference on Difference Equations and Applications (Lisbon, 2007), World Scientific Publishing Co., London, 2010, 375–382.
- [153] P. Řehák, J. Vítovec, Regular Variation on Measure Chains, Nonlinear Anal. 72 (2010), 439–448.
- [154] P. Řehák, J. Vítovec, *q*-Karamata functions and second order *q*-difference equations, E. J. Qualitative Theory of Diff. Equ. 24 (2011), pp. 1–20.
- [155] M. Rosenlicht, Hardy fields, J. Math. Anal. Appl. 93 (1983), 297–311.
- [156] E. Seneta, Regularly Varying Functions, Lecture Notes in Mathematics 508, Springer-Verlag, Berlin-Heidelberg-New York, 1976.
- [157] Swanson, C. A. Comparison and oscillation theory of linear differential equations. Mathematics in Science and Engineering, Vol. 48. Academic Press, New York-London, 1968.
- [158] S. D. Taliaferro, Asymptotic behavior of solutions of  $y'' = \phi(t)f(y)$ , SIAM J. Math. Anal. 12 (1981), 853–865.

- [159] S. D. Taliaferro, Radial symmetry of positive solutions of nonlinear elliptic equations, J. Differential Equations 156 (1999), 219–253.
- [160] S. D. Taliaferro, Asymptotic behavior of positive decreasing solutions of y'' = F(t, y, y'), Geometric analysis and nonlinear partial differential equations (Denton, TX, 1990), 105—127, Lecture Notes in Pure and Appl. Math., 144, Dekker, New York, 1993.
- [161] T. Tanigawa, Regularly varying solutions of half-linear functional differential equations with retarded arguments, Acta Math. Hungar. 120 (2008), 53–78.
- [162] T. Tanigawa, Positive decreasing solutions of systems of second order quasilinear differential equations, Funkcial. Ekvac. 43 (2000), 361–380.
- [163] T. Tanigawa, Generalized regularly varying solutions of second order nonlinear differential equations with deviating arguments, Mem. Differential Equations Math. Phys. 57 (2012), 123–162.
- [164] L. H. Thomas, The calculation of atomic fields, Proc. Cambridge Phil. Soc. 23 (1927), 542–548.
- [165] E. C. Titchmarsh, Eigenfunction expansions associated with second-order differential equations, Part I, Second Edition, Clarendon Press, Oxford 1962.
- [166] D. Willett, Classification of second order linear differential equations with respect to oscillation, Advances in Math. 3 (1969) 594–623.
- [167] A. Wiman, A. Über eine Stabilitätsfrage in der Theorie der Linearen Differentialgleichungen (German), Acta Math. 66 (1936), 121–145.
- [168] P. Wong, Existence and asymptotic behavior of proper solutions of a class of second-order nonlinear differential equations, Pacific J. Math. 13 (1963), 737–760.
- [169] H. Zou, A priori estimates for a semilinear elliptic system without variational structure and their applications, Mathematische Annalen 323 (2002), 713– 735.